Science Advice for Policy by European Academies

TOWARDS SUSTAINABLE FOOD CONSUMPTION

UNPUBLISHED DRAFT

This document is not in its final form. Its contents are subject to change. It must not be distributed in any form except with the express permission of SAPEA. For more information, contact celine.tschirhart@sapea.info.



This Evidence Review Report has been produced under the auspices of the SAPEA consortium, which is funded by the European Union.

The text of this work is licensed under the terms of the Creative Commons Attribution licence which permits unrestricted use, provided the original author and source are credited. The licence is available at http://creativecommons.org/licenses/by/4.0. Images reproduced from other publications are not covered by this licence and remain the property of their respective owners, whose licence terms may be different. Every effort has been made to secure permission for reproduction of copyright material. The usage of images reproduced from other publications has not been reviewed by the copyright owners prior to release, and therefore those owners are not responsible for any errors, omissions or inaccuracies, or for any consequences arising from the use or misuse of this document.

The consortium identified all relevant interests of the experts requested to contribute to this Evidence Review Report, assessed whether an interest constituted a conflict of interests, and took — where relevant — measures to exclude that an interest could compromise or be reasonably perceived as compromising the impartiality or objectivity of the report. Further information about SAPEA's quality guidelines and working processes is available at <u>www.sapea.info</u>.

The information, facts and opinions set out in this report are those of the authors. They do not necessarily reflect the opinion of the European Union or the European Commission, which are not responsible for the use which may be made of the information contained in this report by anyone.

- ISBN 978-3-9823562-2-8
- DOI 10.26356/foodconsumption
- Downloadable from https://sapea.info/topic/food-consumption/

Publisher

SAPEA c/o acatech Pariser Platz 4a 10117 Berlin, Germany

Contact

SAPEA communications office Rue d'Egmont 13 1000 Brussels, Belgium <u>contact@sapea.info</u>

SAPEA, Science Advice for Policy by European Academies. (2023). *Towards sustainable food consumption*. Berlin: SAPEA. <u>https://doi.org/10.26356/foodconsumption</u>

Version history

Version	Date	Summary of changes	
~0.3	22 May 2023	NON-FINAL DRAFT embargoed version	



Towards sustainable food consumption

Informs the Scientific Opinion of the European Commission Group of Chief Scientific Advisors

UNPUBLISHED DRAFT

This document is not in its final form. Its contents are subject to change. It must not be distributed in any form except with the express permission of SAPEA. For more information, contact celine.tschirhart@sapea.info.

Table of contents

Foreword

7

13

Members of the working group 8

Executive summary	9
Food system impacts	9
Food environment and consumer behaviour	9
Policies and strategies	10

Introduction

The current policy mix	13
The Farm-to-Fork Strategy	16
Approach in this report	17

Chapter 1. Food systems, food environments and their drivers 19

		-
1.1.	The food system concept: drivers,	
	structures and dynamics	20
1.2.	Inside the food system	26
1.3.	3. Situating the food environment within the	
	food system	35
1.4.	The digital food environment	40
1.5.	Food consumption behaviour	46
1.6.	Key messages	50

Chapter 2. Nutritional and environmental food system outcomes

2.1.	Dietary patterns	52
2.2.	Healthy diets	54
2.3.	Environmentally sustainable diets	56
2.4.	Integrating environmental sustainability and health	63
2.5.	Food waste	64
2.6.	Key messages	66

Chapter 3. Consumer behaviour: Barriers to sustainable and healthy food consumption 68

3.1.	Theoretical framework to identify barriers in sustainable and healthy food	
	consumption	69
3.2.	Barriers to healthy and sustainable food	
	consumption	70
3.3.	Entry points for policy interventions	80
3.4.	Key messages	81

Chapter 4. Shaping the food environment for healthy and sustainable diets 83

4.1.	Economic environment and fiscal food	
	policies	85
4.2.	Physical availability	93
4.3.	Food composition	99
4.4.	Information environment	100
4.5.	Social environment	110
4.6.	Combinations of tools	111
4.7.	Key messages	114

Chapter 5. Selected examples
of policies to promote
healthy and sustainable food
consumption1165.1. Sugar reduction policies1185.2. Meat reduction policies123

5.2.	Medicies	123
5.3.	Promoting organic food consumption	127
5.4.	Key messages	135

Chapter 6. Evidence-based policy elements 136

- 6.1. The systemic nature of food systems and environments in which consumption is embedded 137
- 6.2. Direction of change: high-impact behaviours reducing negative

52

environmental and health outcomes related to food systems	138	Annex 3. Evidence review	490
6.3. Overcoming barriers to more sustain and healthy food consumption	able 139	process	182
References	144	Annex 4. Policy landscape summary	188
Annex 1. Responsibilities and working structure within the Scientific Advice Mechanism		Annex 5. Literature search strategies	189
	180	Annex 6. Acknowledgement	ts 192
Annex 2. Selection of experts	s 181		

About SAPEA

SAPEA brings together outstanding expertise from natural, applied, and social sciences and humanities, from over 120 academies, young academies and learned societies in more than 40 countries across Europe.

SAPEA is part of the European Commission's Scientific Advice Mechanism. Together with the Group of Chief Scientific Advisors, we provide independent scientific advice to European Commissioners to support their decision-making.

We also work to strengthen connections between Europe's academies and Academy Networks, and to stimulate debate in Europe about the role of evidence in policymaking.

Europe's academies draw on the best scientific expertise to provide independent, balanced and authoritative scientific advice. This approach makes SAPEA a critical source of evidence for policymakers and the wider public.

Our Academy Networks collectively represent over a hundred academies, young academies and learned societies across Europe. SAPEA works to strengthen these academies and provides a means for close collaboration in a unique and interdisciplinary way.

For further information about SAPEA, visit www.sapea.info.



With the approval of the EU Green Deal in 2020, the European Union embarked on a mission to respond to the climate and environmental emergencies of our time, while leaving no one behind. To achieve these objectives requires extraordinary coordination among a wide range of policy sectors.

There is now general recognition that making our food system more sustainable, healthy and fair is a key lever in this overall transition, which is why the Farm-to-Fork Strategy lies at the heart of the Green Deal. To inform the development of the Strategy, the Scientific Advice Mechanism already delivered scientific advice on *Towards a sustainable food system* in 2020.

Building on this previous advice, and to inform the 2023 revision of the Farm-to-Fork Strategy, the Commissioners for Health and Food Safety and for Innovation, Research, Culture, Education and Youth asked the Group of Chief Scientific Advisors to the European Commission to deliver science policy advice on a more specific area of the food system: food consumption and the food environment. They asked:

What tools could be used at EU level, in addition to those mentioned in the 2020 Farm-to-Fork Strategy, to overcome the barriers preventing consumers [from adopting] sustainable and healthy diets, fostering the necessary change towards sustainability in the food environment? The Group's advice should be based on an analysis that identifies the elements refraining consumers from making healthy and sustainable choices.

To address this question, SAPEA assembled an outstanding, independent, international and interdisciplinary working group of experts in the field, nominated by academies of science and academy networks. Between October 2022 and April 2023, the working group reviewed and compiled the latest evidence on the subject to create this 12th SAPEA Evidence Review Report. This report informs the accompanying Scientific Opinion of the Group of Chief Scientific Advisors, which contains the requested policy recommendations. This project was coordinated by ALLEA, the European Federation of Academies of Sciences and Humanities, acting as the lead network on behalf of SAPEA.

We warmly thank all working group members for their voluntary contributions and dedication, and especially the chair of the SAPEA working group, Professor Erik Mathijs. We would also like to express our sincere gratitude to everyone involved in pulling this report together, as well as to the science academies across Europe.

Professor Antonio Loprieno Chair of the SAPEA board

Members of the working group

This report was written by an independent working group consisting of the following members:

- Erik Mathijs, KU Leuven, Belgium (Chair)
- Janis Baird, University of Southampton, United Kingdom
- Rune Blomhoff, University of Oslo, Norway
- Andrea Büttner,
 Fraunhofer Institute for
 Process Engineering and
 Packaging IVV, Germany
- Carsten Daugbjerg,
 University of Copenhagen,
 Denmark
- Francesca Galli, University of Pisa, Italy

- Wencke Gwozdz, Justus-Liebig-University, Giessen, Germany
- Meike Janssen,
 Copenhagen Business
 School, Denmark
- Petr Jehlička, Czech Academy of Sciences, Czechia
- Linus Mattauch, Technical University Berlin, Germany
 - **Jutta Roosen**, Technical University of Munich, Germany
- Elin Röös, Swedish
 University of Agricultural
 Sciences, Sweden

- Tanja Schneider, University of St Gallen, Switzerland
- Antonia Trichopoulou, Academy of Athens Centre for Public Health Research and Education, Greece
- Mónica Truninger, Unversidade de Lisboa, Portugal
- Jenny van Doorn,
 University of Groningen,
 Netherlands
- Stefanie Vandevijvere, Sciensano, Belgium

The above experts were identified with the support of:

- The Academy of Athens
- The Czech Academy of Sciences
- Die Junge Akademie
- The National Academy of Science and Engineering in Germany (acatech)
- The Norwegian Academy of Science and Letters
- The Royal Netherlands Academy of Arts and Sciences
- The Swiss Academies of Arts and Sciences

Executive summary

Healthy and sustainable food consumption is a key component of the EU's transition to a sustainable food system. This report unpacks the evidence to identify levers and actions that will shape food consumption for the mutual benefit of environmental sustainability and public health, including health-related inequalities. We look at current evidence and potential policies that can provide impact in this area, alongside tools to overcome the barriers that hinder sustainable and healthy consumption.

Food system impacts

The current food system is a major driver of environmental impacts, especially biodiversity loss, eutrophication, water stress, land degradation and climate change. Poor dietary quality is also linked to obesity and risks of noncommunicable diseases. To address these environmental and health issues, there is widespread agreement that diets need to shift towards more plant-based ingredients, rich in vegetables, fruits, wholegrains, pulses and fish and seafood sourced from sustainably managed stocks, with moderate amounts of low-fat dairy products and limited in red meat, processed meat, salt, added sugar and high-fat animal products.

Changing food consumption is a key lever to achieve the objectives of the EU's Green Deal and Farm-to-Fork Strategy. The food system is a dynamic and complex area, where a mix is needed of coherent evidence-based policies that consider multiple areas of impact to promote sustainable and healthy food consumption. While reducing sugar and animal product consumption are key to increasing health, reducing consumption of animal products is the key mitigation option to reduce environmental impacts. Lowinput production systems like organic production can reduce chemical use and support a shift to sustainable food systems when combined with reduced animal products and decreased food waste. Prevention of and reduction in food waste can further reduce resource use and emissions. Local foods can contribute to important social aspects of the EU food system, but their climate benefits are limited.

Food environment and consumer behaviour

The food environment is the consumers' primary interface with the food system and is the focus of our attention in this report. It is the context in which food is accessed and eaten, and entails both an external domain (physical availability, the infrastructural environment, the price of food, the information environment and labelling, the social environment)

Executive summary

and an individual domain (affordability, accessibility, convenience and desirability) which relates to the conditions for individual daily routines and practices.

There is a huge diversity of dietary patterns in the EU, but also in the ways consumers source food, from formal markets to informal exchanges, food redistribution and more, with increasing importance of the digital environment. There are also some commonalities, with high caloric shares of processed food products in the diets of many EU countries and added sugar and salt intakes exceeding dietary recommendations. Creating a food environment that drives improvements in socioeconomic, health and environmental outcomes will have to turn trade-offs and conflicts between differing food system outcomes into a situation in which healthy and sustainable diets are the easy and the preferred choices for consumers.

The barriers to consumption changes are situated both at the individual and the contextual level. Barriers at individual level include a perceived lack of consumer motivation and capabilities, while at the contextual level there is a lack of physical, financial and social opportunities to acquire more healthy and sustainable foods. Trade-offs are exacerbated for those from disadvantaged backgrounds who may lack agency to make healthy or sustainable food choices. The obesogenic effects of the food environment will be greater for disadvantaged groups, indicating the need for disruptive measures that alter the food environment without the need for agency from consumers.

Guiding consumer behaviours or practices requires the need to understand three key socio-psychological processes. The development of effective strategies depends on the acknowledgement that consumer behaviour is driven by:

- habits, routines, and semi-automatic processes
- cognitive processes, i.e. deliberate thinking and information processing
- affective processes, i.e. emotions and impulsive reactions

Most current public policy interventions aim to increase motivation and personal capabilities by providing information or education. However, given that habits, routines, semi-autonomous and affective processes are strong determinants of food choice, policy measures need to shift food environments affecting individual behaviour significantly.

Policies and strategies

To create shifts in behaviour, disruptive measures such as taxes, bans and product reformulations should be considered, as well as softer measures that can influence and reshape social norms. These approaches help to create food environments that allow consumers to prioritise healthy and sustainable foods without the need for high agency and they also help to reduce dietary inequalities.

Five areas are considered:

- The economic environment and fiscal food policies. People are most likely to respond to direct incentives that make less healthy or sustainable diets more expensive. Sugar taxes in Mexico and the UK have shown their effectiveness in reducing purchases of sugary drinks and stimulating reformulation of beverages towards healthier versions. Making animal products reflect the true cost of their associated impacts is economically efficient, and may require levying carbon pricing on greenhouse gas emissions from agriculture. Consumption taxes on animal products are also effective in reducing the environmental impacts from animal products. It is important to consider the equity effects of these policies, which can be neutralised by returning the tax proceeds to citizens appropriately.
- Physical availability. Prominent placement of healthy and sustainable food options in the food environment can help to influence consumer behaviour, while the removal of unhealthy options from prominent places also has a positive effect. This applies to both retail and food service environments such as schools and canteens.
- Food composition. At a product development level, we can shape the healthiness of the food environment by reformulating food products to reduce fat, salt or sugar content. The introduction of more healthy plant-based alternatives can also widen consumer choices to replace meat products. There is a caveat here, by way of previous policies that show limited effect of reformulation when based on voluntary agreements. Comprehensive and mandatory inclusion of the food industry with clear target agreements are needed to make this policy instrument more effective.
- The information environment. Labelling foods for health impacts only has low to moderate impact, because consumers need to be interested and motivated to use them and because in the European context front-of-pack labelling to date remains voluntary rather than mandatory. However, some impact has been shown on food reformulation which then indirectly impacts diets and health. Warning labels, as introduced in Chile, show a comparatively better effect, while for sustainability labels the evidence is mixed.

The effectiveness of any labelling initiative relies on developing trust. There are many new sustainability labels appearing and a coherent approach to these schemes is needed to avoid confusion.

The social environment. There are many ways that choices are shaped. Peer influence has been shown to be successful in improving fruit and vegetable intake and limiting fast food consumption. There is also evidence that social influence can lead to reduced meat consumption. The digital food environment offers further possibilities, such as personalised feedback on food choices by enabling healthier and more sustainable options at check-out, but also huge risks of stimulating unhealthy and unsustainable consumption through industry marketing strategies.

Executive summary

Governments will need to introduce expanded use of policy instruments to create shifts in consumption towards healthier and more sustainable diets. A policy mix of hard and soft measures will help to overcome the barriers preventing consumers from adopting sustainable and healthy diets. The current policy focus on providing information and education is not effective enough, and must be mixed with many other policy elements. A comprehensive policy package is needed that considers addressing lobbying attempts by industry to influence policies that they might consider harmful to their own interests.

Within the policy space, there are positive examples to look towards. A meat tax framed as "animal welfare levy" is being considered in Germany, and New Zealand is implementing a price for greenhouse gas emissions from agriculture. Local initiatives across European cities indicate that public procurement of more plant-based and fewer animal products is viable and effective, and could be rolled out at the national or EU level. Experiences from Denmark and Sweden demonstrate governments can stimulate demand for targeted food categories, for example organic food, through public procurement and campaigns directed at households.

Ultimately, a legislative framework is needed for sustainable food systems that specifies the guiding principles for effective and efficient actions at multiple policy levels, in a coherent and systemic way. When approached with a dynamic, joined-up approach, these policy areas offer the prospect of positive influence over the food environment and in the end, the improved health and sustainability of our food system.

Introduction

Food consumption is an important lever to move food systems towards more healthy and sustainable outcomes. Governments — including the EU — tend to focus on informing consumers as the main policy intervention. However, the SAPEA report on food systems (SAPEA, 2020) posited that information provision by public authorities is insufficient to change consumer behaviour, due to the complexity of the food environment¹ influencing consumer behaviour on the one hand, and the habit-based nature of consumer behaviour on the one hand, and the habit-based nature of Chief Scientific Advisors to the European Commission recommended to ensure a combination of "regulatory, financial, behavioural, information, communication, and education measures" to reshape the food environment. In this context, "information-based initiatives should be a part of the policy mix despite the fact that on their own they would be insufficient to change behaviour" (GCSA, 2020, pp. 41–42).

Building on this previous advice, the Commissioner for Innovation, Research, Culture, Education and Youth Mariya Gabriel and the Commissioner for Health and Food Safety Stella Kyriakides have asked the Advisors to deliver advice in the form of a scientific opinion on the topic of sustainable food consumption by the second quarter of 2023, as it will feed into the revision of the Farm-to-Fork strategy from the middle of 2023 onwards.

This SAPEA evidence review report gathers the relevant scientific evidence to inform the Advisors' Scientific Opinion. It addresses issues described in a scoping paper² which sets out the formal request for advice from the European College of Commissioners to the Advisors. The aim of this report is to analyse:

What concrete actions could be taken at EU level, in addition to those announced in the 2020 Farm-to-Fork Strategy, to overcome the barriers preventing consumers [from adopting] sustainable and healthy diets, fostering the necessary change towards sustainability in the food environment?

The current policy mix

The current policy mix related to sustainable and healthy food consumption is highly fragmented, rooted in the General Food Law established in 2002 on the one hand, and in many pieces of legislation that influence the sustainability of the food system, either directly or indirectly, on the other.

¹ As it is central in this report, the concept of the food environment is introduced and defined at length in chapter 1.

^{2 &}lt;u>https://research-and-innovation.ec.europa.eu/system/files/2022-07/scoping-paper_sustainable-food-consumption.pdf</u>

Introduction

The 2002 General Food Law³ is rooted in EU food safety policy that falls under the EU Treaty's articles on public health (article 168) and consumer protection (article 169). The Law lays down the general principles governing food safety, establishing the European Food Safety Authority (EFSA) as the main governing body. Issues addressed include the regulation of food additives (vitamins, minerals, food supplements); health and nutrition claims made on foods; food intended for infants, young children and special medical purposes; and the impact of foods on food allergies.⁴

In its White Paper on *A strategy for Europe on nutrition, overweight and obesity-related health issues,* the European Commission stresses that:

any public action, including those possibly undertaken at Community level, in this field should take into account three factors. Firstly, the individual is ultimately responsible for his lifestyle, and that of his children, while recognising the importance and the influence of the environment on his behaviour. Secondly, only a well-informed consumer is able to make rational decisions. Finally, an optimal response in this field will be achieved by promoting both the complementarity and integration of the different relevant policy areas (horizontal approach), and of the different levels of action (vertical approach).

(European Commission, 2007)

Hence, actions should result in better-informed consumers, making the healthy option available, and encouraging physical activity, with focus on priority groups and settings — particularly children living in low socioeconomic conditions.

As a result, legislation starts from the premise that a well-informed consumer is able to make rational decisions and tends to focus mainly on food labelling. Most recently, food labelling is regulated under Regulation (EU) No 1169/2011 on providing food information to consumers, merging earlier regulations on food labelling and nutritional labelling. The regulation prescribes the mandatory information that needs to be communicated to consumers, including the country of origin or place of provenance for some products.

Sustainability dimensions of the food system are regulated by various pieces of legislation, a comprehensive overview of which is difficult to realise given the complexity of food systems. EU policies aim "to protect the environment and biodiversity, minimise risks to human health, and promote the transition to a circular economy".⁵ The food system relates to several environmental issues, including clean air, chemicals, industrial emissions, waste and recycling, water, soil and land, and nature and biodiversity. Particularly worth mentioning in relation to sustainable food consumption are the EU initiatives to make sustainable products the norm in the EU and substantiate green claims

³ Regulation (EC) No 178/2002 of the European Parliament and of the Council of 28 January 2002 laying down the general principles and requirements of food law, establishing the European Food Safety Authority and laying down procedures in matters of food safety.

⁴ https://www.efsa.europa.eu/en/applications/nutrition/regulationsandguidance

^{5 &}lt;u>https://environment.ec.europa.eu/index_en</u>

on products. These initiatives are intended to replace current Regulation (EC) No 66/2010 of the European Parliament and of the Council of 25 November 2009 on the EU Ecolabel.

The regulation of climate change mitigation differentiates between energy-intensive sectors of the economy that are covered by the EU's Emissions Trading System (ETS1) and other sectors, including domestic transport (excluding aviation), buildings, agriculture, small industry and waste. For the latter, the Effort-Sharing Regulation has been in place since 2018, stipulating that these sectors must reduce emissions by 30% by 2030 compared to 2005. In 2023, the Regulation was amended, increasing the reduction target to 40%.⁶

The European Commission also plans to revise the Regulation on land use, land use change and forestry "to provide more powerful incentives for member states to grow and improve their natural carbon sinks in line with the European Climate Law, and to reduce the complexity of the current rules".⁶ Particularly promising is the CSR Directive that entered into force in January 2023, requiring all large companies to list information on risks and opportunities arising from social and environmental issues.⁷

Another important instrument is public procurement. Directive 2014/24/EU on public procurement currently prescribes that public contracts must be based on the most economically advantageous tender. This method of assessment "allows for considering price, or cost, and other criteria that relate, among other factors, to quality, social, environmental and innovative aspects as well as delivery conditions such as delivery date, delivery process and delivery period. In practice, this method rewards the bids which are compliant with specific criteria".⁸

Last but not least, the Common Agricultural Policy (CAP) significantly impacts the sustainability of agricultural production. The CAP remains the major EU policy directed at the food sector. As an agricultural policy, farm interests have been at the centre of policymaking. However, demands from outside the farming community that the CAP should respond to sustainability issues have resulted in policy evolution in which provision of public goods, in the form of compliance with specified environmental measures, has been added to the direct farm income support payments as an eligibility criterion. Sustainability in the CAP domain has been defined entirely from an environmental perspective; it has done very little to address public health (De Schutter et al., 2020). While most CAP support is not coupled to production, a considerable

^{6 &}lt;u>https://climate.ec.europa.eu/eu-action/effort-sharing-member-states-emission-targets/effort-sharing-2021-2030-targets-and-flexibilities_en</u>

⁷ Directive (EU) 2022/2464 of the European Parliament and of the Council of 14 December 2022 amending Regulation (EU) No 537/2014, Directive 2004/109/EC, Directive 2006/43/EC and Directive 2013/34/EU, as regards corporate sustainability reporting

^{8 &}lt;u>https://op.europa.eu/en/publication-detail/-/publication/b1b7d65b-5334-11e8-be1d-01aa75ed71a1/language-en</u>

Introduction

amount of subsidies coupled to livestock production still exist. The most recent reform has added the instrument of eco-schemes to the existing set of instruments including the conditionality of obtaining income support on a set of food safety, environmental and animal welfare standards in the first pillar of the CAP and the rural development measures of the second pillar of the CAP addressing issues related to nature, the environment and climate change. Important to mention are also the Regulation and the Resolution regarding the aid scheme for the supply of fruit and vegetable and the EU's quality policy. The EU protects and promotes certain foods that are linked to their geographical origin and to traditional know-how. Geographical indications include Protected Designation of Origin, Protected Geographical Indication, and Geographical Indications for spirits.

The Farm-to-Fork Strategy

To address major environmental and climate change challenges across all policy fields, the Commission published the European Green Deal in December 2019 and the Farmto-Fork Strategy in May 2020.⁹ The strategy aims to accelerate the European Union's transition to a sustainable food system, addressing the food environment in two ways.

First, to facilitate the shift towards healthy and sustainable diets, the strategy proposed the following actions:

- proposal for a harmonised mandatory front-of-pack nutrition labelling to enable consumers to make health-conscious food choices
- proposal to require origin indication for certain products
- determine the best modalities for setting minimum mandatory criteria for sustainable food procurement to promote healthy and sustainable diets, including organic products, in schools and public institutions
- proposal for a sustainable food labelling framework to empower consumers to make sustainable food choices
- review of the EU promotion programme for agricultural and food products with a view to enhancing its contribution to sustainable production and consumption
- review of the EU school scheme legal framework with a view to refocus the scheme on healthy and sustainable food

^{9 &}lt;u>https://food.ec.europa.eu/horizontal-topics/farm-fork-strategy_en</u>

Second, the strategy also proposes a set of actions to stimulate food processors, retailers, hospitality and food services companies to become more sustainable:

- initiative to improve the corporate governance framework, including a requirement for the food industry to integrate sustainability into corporate strategies
- develop an EU code and monitoring framework for responsible business and marketing conduct in the food supply chain
- launch initiatives to stimulate reformulation of processed food, including the setting of maximum levels for certain nutrients
- set nutrient profiles to restrict promotion of food high in salt, sugars or fat
- proposal for a revision of EU legislation on food contact materials to improve food safety, ensure citizens' health and reduce the environmental footprint of the sector
- proposal for a revision of EU marketing standards for agricultural, fishery and aquaculture products to ensure the uptake and supply of sustainable products
- enhance coordination to enforce single market rules and tackle food fraud, including by considering a reinforced use of OLAF's investigative capacities¹⁰

Most (but not all) of the proposed actions focus on information and marketing, and are thus in line with the aforementioned legislator's main focus on well-informed and rational consumers as the starting point of policies supporting the shift towards healthy and sustainable diets.

Approach in this report

Given the complexity and breadth of the food system, choices have been made concerning the evidence that the working group looked into to compile this report. To frame the report, we would like to make explicit three important considerations that widen the scope of this review to non-conventional aspects of the food system on the one hand, but narrow down its scope on the other.

First, it needs to be recognised that hegemonic approaches to alternatives to the conventional food system and its transformation along sustainability lines are derived from research in Western European contexts and are based on ethical consumerism, food commodification, certification and marketisation. These alternative approaches are largely reminiscent of the "ABC policy model in which attitudes (A) drive the kinds of behaviour (the B) that individuals choose (the C) to adopt" (Shove, 2010, p. 1274). The dominance of the ABC model in these narratives of transformation can be criticised on the grounds that these approaches are "incapable of conceptualising transformation in

¹⁰ OLAF is the European Anti-Fraud Office: <u>https://anti-fraud.ec.europa.eu/index_en</u>

Introduction

the fabric of daily life on the scale and at the rate required" (Shove, 2010, p. 1238). Clearly, there is a strong need to widen the scope of what is considered a sustainable food system. In this respect, the already existing and largely sustainable — environmentally and socially — informal food practices make a compelling case.

Second, the food system is a highly complex system with a range of interacting social, cultural, technological, economic and ecological components. Its complexity is key, but also threatens its resilience: it guarantees diversified resources and supply chains, but compromises sustainability if not guided appropriately on a global scale. Food consumption not only impacts human and planetary health, but also a range of socioeconomic factors, including working conditions for farmers and farm workers, animal welfare, power imbalances among firms, food affordability, etc. (HLPE, 2017).

However, the focus of this report is on the nutritional, health and environmental outcomes of food systems. When designing future sustainable food systems and policies to steer towards such systems, these wider food system aspects should also be considered. However, there are limitations to what can and should be controlled by food consumption interventions, and some aspects that are associated to food consumption and indirectly affected by food choice might still be more effective to address with policies directed at that issue specifically. For example, it is more effective to regulate working conditions through labour legislation than by encouraging consumers to buy food products that are produced under such conditions.

Third, the report focuses on how the food environment shapes food consumption, which means that several aspects have received less attention, such as the role of physical activity and alcohol consumption. Given the focus of most of the literature on the consumption of meat, fruit, vegetables and foods high in sugar content, certain foods also received less attention, including fish and marine products and nuts.

To respond to the scoping question, we reviewed published scientific evidence, including input from social sciences, and took a systemic approach which considers the complex architecture of the food environment. Chapter 1 conceptualises the structural dimensions of food systems and particularly the food environment that shapes food consumption. Chapter 2 defines what healthy and sustainable diets and food consumption entails. Chapter 3 identifies the various barriers to changing food consumption. Chapter 4 summarises the evidence on the impact of interventions in the various dimensions of the food environment on food consumption. Chapter 5 provides examples of successful policy mixes in selected areas that go beyond information provision. Chapter 6 concludes with a set of policy elements that could constitute a policy mix fostering healthy and sustainable diets.

Chapter 1. Food systems, food environments and their drivers

Food systems are a focal point of interaction between humans and their environment. The earlier SAPEA report *A sustainable food system for the European Union* (SAPEA, 2020) adopted a social science perspective to identify viable pathways towards a more socially just and sustainable food system. It found that promoting a transition to a more sustainable food system requires addressing several interrelated challenges, including malnutrition; population growth and urbanisation; biodiversity; globalisation; territorial imbalances and geopolitical uncertainties; and the social and environmental consequences of intensive agriculture and industrialised food production.

This new report continues to support a systemic view of actors, activities and relations, recognising the interactions between food systems at different territorial scales, allowing for both common EU and region-specific approaches.

Food and nutrition security is central in food systems thinking, as hunger and malnutrition found at a consumer level are determined by income, price, access to food, and more. An important external factor in consumption patterns is the food environment, which is the physical and social environment that affects what people eat. Therefore, this report adopts a food systems perspective to address food consumption — currently considered one of the most important drivers of both global environmental change and the state of human health — and the barriers within food environments that prevent consumers from adopting more sustainable and healthier diets.

This chapter introduces key food system concepts, the drivers of food system change, food system dynamics, and food system outcomes. Referring to the production and distribution mechanisms within the food system, we provide an overview of the different ways people access food, namely through food value chains, alternative networks, informal exchanges, and donations. We then define the food environment in terms of its key elements, the context in which food is accessed and consumed, and we address the importance of digital technology in shaping the food environment. The mix of policy instruments found later in this report (Chapters 4 and 5) target the key elements of the food environment which ultimately influence food consumer behaviour. Figure 1 summarises the contents of this chapter in the context of the following chapters.



Figure 1. A food system perspective on food environment and food consumption: overview of report's chapters (1-5)

1.1. The food system concept: drivers, structures and dynamics

The food system has become an important concept as researchers and policymakers are confronted with complex (also called "wicked") food-related problems for which linear or reductionist thinking cannot provide sufficient answers (Fanzo, Haddad, et al., 2021). The food systems approach stems from systems thinking applied to agricultural and food science and is considered comprehensive and integrative enough to address the grand challenges of sustainability (Eakin et al., 2017).

Based on Ostrom (2009), food systems have been interpreted as complex socioecological systems that integrate social, economic, technological, and environmental processes. The food system includes the biophysical resources for food production, the resource use requirements of food processors and retailers, and consumer behaviour, including food preferences, preparation, and distribution patterns within households (Ericksen et al., 2010). Particularly during recent years of severe economic crisis, there has been renewed attention to food and nutrition security in relation to food system vulnerabilities (Zurek et al., 2018). By capturing the interconnected food relationships at multiple scales in spatial, temporal, and legal terms, food systems thinking allows for an explicit focus on key tensions such as food and nutrition insecurity, environmental degradation, and chronic poverty (Godfray et al., 2010). Food systems represent one of the most critical entry points for achieving the multiple objectives of the Sustainable Development Goals, as over half of them relate to global food security and nutrition, with SDG2 (Zero Hunger) being the most critical (Fanzo, 2019).

There are a set of critical aspects in food systems thinking, starting with the description of the components and the relationships between them. The key components of food systems are structures that comprise actors, resources and activities (i.e., production, aggregation, processing, distribution, consumption and disposal, including loss or waste, of food products), that interact based on coordination mechanisms and governance (van Bers et al., 2019), to deliver outcomes such as food and nutrition security and other socioeconomic and environmental effects (Ericksen, 2008; HLPE, 2017; Ingram, 2011). As food systems are very diverse and location-specific, yet interconnected, they can be represented in different ways. Conceptualising food systems entails defining systems boundaries, building-blocks and linkages between them, while simultaneously being connected to other systems such as health, education, environment, economy and governance, and science and innovation systems.

The food system concept critically puts the emphasis on the outcomes of the processes taking place within it (p. 22). As stated in the earlier SAPEA report (2020), a sustainable food system is "one that contributes to food security and nutrition for all people in a way that sustains the economic, social, cultural, and environmental resources to provide food security and nutrition for future generations". Food system outcomes are identified and assessed in terms of food and nutrition security, socioeconomic impacts (income, employment), and the environment (biodiversity, climate). In most recent frameworks (HLPE, 2017, 2020), food and nutrition security entails six dimensions. Beyond availability, access, utilisation and temporal stability, there are two further overarching dimensions:

- Agency refers to the ability of individuals or groups to make their own decisions about what food they eat, what food they produce, and how that food is produced, processed, and distributed within food systems, including their ability to participate in processes that shape food systems policy and governance.
- Sustainability refers to the long-term ability of food systems to provide food security and nutrition in ways that do not compromise the economic, social and environmental foundations that create food security and nutrition for future generations (FAO, 2022a).

Distinguished authors emphasise the growing awareness of food system inequities and the connections between food and ecological systems that highlight the importance of

agency and sustainability (Clapp et al., 2022). Both agency and sustainability are important factors influencing both drivers and outcomes of food systems and the food environment.

Explicitly linking food system outcomes to the structures and activities in the food system is important for research and policy, because food security results from a complex set of interactions in multiple domains that are often not highlighted in conventional sector or food chain analyses which focus on food yields and flows (Borman et al., 2022; Ericksen et al., 2012).

Below is a list of food system outcomes (adapted from HLPE, 2020; van Berkum et al., 2018):

- Socioeconomic outcomes:
 - **i**ncome
 - livelihoods
 - employment
 - wealth
 - social and political capital
 - human capital
- Food and nutrition security (incorporating stability, agency and sustainability):
 - food utilisation (nutritional value, social value, food safety)
 - food access (affordability, allocation, preference)
 - food availability (production, distribution, exchange)
- Environmental outcomes:
 - land
 -) soil
 - fossil fuel
 - minerals
 - biodiversity
 - water
 - l climate

Systems thinking views the behaviour of a system as a dynamic interaction between variables and across sub- or supra-systems, which goes beyond a simple chain of causeand-effect relationships and aims to capture the root causes behind unsustainable food system outcomes. Food systems dynamics are expressions of specific humanenvironment interactions, characterised by uncertainty, error, learning, and adaptation (Rivera-Ferre et al., 2013). Both positive/reinforcing and negative/mitigating feedback effects explain the dynamic effects of interactions on the initial variables. Feedback loops are a distinguishing aspect because they occur between parts of the food chain and from the socioeconomic and environmental outcomes of food production and consumption, back to that production and consumption. Mapping system dynamics using causal loop diagrams allows the identification of entry points and opportunities for change in the system (Brzezina et al., 2016; Galli et al., 2019; Kopainsky et al., 2018). Equilibrium and disequilibrium states, co-evolution of system components, self-organising properties, nonlinear dynamics, and multivariable structures are all consequences of the dynamic and uncertain behaviour of food systems (SAPEA, 2020).

Ultimately, systems thinking aims to identify the drivers behind systems dynamics, to address the root causes of the lack of sustainability and food and nutrition security. At the same time, food systems themselves are drivers of global change (Ericksen et al., 2012) as they have socioeconomic and environmental impacts. The relation between food systems and the environment is indeed complex because environmental changes are both a driver and an outcome of food systems (Fanzo, Bellows, et al., 2021). Such bidirectional relationships (between climate change and food and nutrition security) require adaptation and mitigation interventions that are both climate-smart and nutrition-sensitive (Fanzo et al., 2018).

Food systems drivers can be classified as internal or external, depending on how the boundaries of the system are defined. Drivers can be identified in relation to the durability of their effects on the food system: shocks and stressors that alter the food system equilibrium may not necessarily be drivers of change. They can be considered drivers to the extent that their recurrence or increase in frequency and intensity (e.g. of climate events) will lead people, individually or collectively, to adapt by changing their behaviours, eventually altering the system in a durable way. Three general categories of drivers can be identified (Béné et al., 2020):¹¹

- drivers relevant to demand and consumers, such as population demographic transition, rise in consumer income, urbanisation and lifestyle change, and growing attention to diet
- drivers related to production and supply, such as technological innovation, intensification of the agricultural sector, improved access to infrastructure, general degradation in agro-ecological conditions, and climate change
- drivers related to trade and distribution, such as policies facilitating and mitigating trade, internationalisation of private investments, and growing concerns for food safety

The pathways to increased sustainability are open to debate, whereas drivers are often correlated and overlapping and show more or less significant correlations with food system sustainability (Béné et al., 2020; Béné et al., 2019). A recent report by FAO (2022a)

¹¹ Recent analysis (Béné et al., 2020) concludes that only a few drivers display both significant and positive associations with food system sustainability. Changes in population growth, agricultural area, female employment in services and urban population are negatively correlated with country food system sustainability scores.

Food systems, food environments and their drivers

analyses eighteen interconnected socioeconomic and environmental drivers, and the related trends that could shape the future of agrifood systems:

- Systemic (overarching) drivers:
 - population dynamics and urbanisation
 - economic growth, structural transformation and the macroeconomic outlook
 - cross-country interdependencies
 - big data generation, control, use and ownership
 - geopolitical instability and increasing conflicts
 - uncertainties (e.g. pandemic)
- Drivers directly affecting food access and livelihoods:
 - rural and urban poverty
 - Inequalities are widespread and deep-rooted
 - food prices
- Drivers directly affecting food and agricultural production and distribution processes:
 - innovation and science
 - public investment in agrifood systems
 - capital and information intensity of production
 - input and output market concentration
- Drivers regarding environmental systems:
 - consumption and nutrition patterns
 - scarcity and degradation of natural resources
 - epidemics and degradation of ecosystems
 - climate change
 - the 'sustainable ocean economies'

Interrelations among drivers occurring simultaneously, or through cause-effect relationships, contribute to determining agrifood systems outcomes, at different scales. The FAO (2022a) report provides a detailed understanding of the role of each driver in determining possible future patterns of agrifood systems.

Among the multiple food system drivers assessed in the FAO (2022a) report, changes in consumption and nutrition patterns over time play a key role, and occur as a consequence of a number of factors that interact in a complex manner. These include income, prices, demographic changes, urbanisation, trade, individual preferences and beliefs, cultural traditions and social norms, and modifications of lifestyle. Changes in dietary patterns and consumer behaviour are extensively addressed in Chapter 2 and Chapter 3 of this report. For now, it is sufficient to say that changes in the past decades have not been limited to what people eat, but also to how people consume foods in relation to lifestyle changes (including the digital environment, which is further addressed in this chapter). Other key drivers, beyond the scope of the present report, are the consequence of sudden crises such as the COVID-19 pandemic, whose impact on eating behaviours is yet to be clearly understood (Alamri, 2021; González-Monroy et al., 2021; Hunter et al., 2023), or the outbreak of war, such as recently in Ukraine. Different forms of conflict and violence, including ecological violence, are at once driven by food systems and drivers of food insecurity, as stated by a recent report by the UN special rapporteur on the right to food (Fakhri, 2022), who emphasises "systemic violence and structural inequality in food systems" favoured by relationships of dependence (among individuals, countries, international financial institutions and corporations) and extractive practices that undermine human and environmental health.

Multiple perspectives and interests in food system activities and outcomes, along with differences in power across levels and scales, also means that it is very difficult to agree on solutions to food system problems. This is at the heart of trade-offs among food and nutrition security, socioeconomic and environmental outcomes (Ericksen et al., 2012; FAO, 2022b; van Berkum et al., 2018). Different perspectives, power relations and interests in different contexts determine how trade-offs are evaluated and hence how policy and other decisions are made. On the other hand, synergies arise if, for instance, socioeconomic outcomes such as growth in income increase food availability, and better food use (e.g. reducing waste) leading to positive environmental impacts. Trade-offs could instead arise if an increase in food production puts pressure on the environmental outcomes and choices concerning food access, leading to unequal socioeconomic outcomes. Dependence on foreign food production and imports versus the expansion of domestic cropland and productivity for self-provision is a key tension (MacDonald, 2013). Other examples of food system trade-offs can be identified in the tensions between food and biofuel energy production (see Gasparatos et al., 2011, for a systematic critical review), between organic farming, the use of genetically modified crops and productivity (Azadi & Ho, 2010), or between sustainable products and accessibility for low-income groups.

The following sections explore the structures and activities of the food system and briefly discuss food supply chains (p. 26), the food environment, and consumer behaviour. As mentioned earlier, it is important to keep in mind that these food system constituents are influenced by the various drivers mentioned above and help to shape dietary habits and determine the ultimate nutritional, health, economic, environmental and social outcomes of food systems. After addressing the core activities of the food system, we review the concept of the food environment, including its digital dimensions (p. 40). We then move from the general context to the individual level, discussing food consumption behaviour (p. 46). This leads to Chapter 2, p. 52, which focuses on dietary habits and takes a closer look at the impact of the food system on health and the environment.

1.2. Inside the food system

This section looks inside the food system by focusing on the main processes and relations that enable the exchange and consumption of food, leading to diverse food and nutrition security outcomes (Ingram, 2011).

Food systems encompass food value chains insofar as they include "a chain of activities from production ('the field') to consumption ('the table'), with particular emphasis on processing and marketing and the multiple transformations of food that these entail" (Ericksen, 2008). Key processes include (HLPE, 2017):

- **production:** all activities involved in the production of the raw food material
- **storage and distribution:** the range of intermediaries who link producers, processors, packaging to the final market
- **processing and packaging:** the various transformations that the raw food material undergoes before it is sent to the retail market for sale
- distribution and markets: the many actors involved in transport, delivery, logistics, warehousing, trading and marketing.

This section further distinguishes the domains through which people access food: either via the formal food system, in which exchanges are regulated mainly by contractual and market arrangements, or through other informal means, encompassing all other ways to obtain and exchange food beyond purchasing it on the market.

The economic relations within which food is produced, distributed and consumed extend beyond market-based interactions and include alternative market and non-market economic relations (including donations, for example). This is leading to a growing recognition of "more-than-capitalist configurations" (capitalist, alternative capitalist, noncapitalist) in agrifood systems (Koretskaya & Feola, 2020). Moreover, "the variations in and between agrifood systems" can be understood as a form of "diverse economy" (Gibson-Graham, 2006) where none of the market, non-market or alternative food systems should be conceptualised as monolithic or hegemonic, as they all overlap and interact with each other.

The inadequacy of reducing the economic system — and by extension, the food system — to its market-based variant is powerfully captured in the allegory of an iceberg (Gibson-Graham, 2002). The visible tip of the iceberg represents the formal, market-based part of the economy, while the multiplicity of alternative market and non-market relations are submerged.

Within the overall food system, we find the formal food system, alternative food networks (alternative to market food systems) and the informal food system including informal food production, gathering and sharing. Conceptualising the food system as more-

than-capitalist and hence comprising formal and informal elements also relates to the question of food system resilience in two ways. First, we can conceptualise resilience as the capacity to recover from a shock or disturbance: "The underlying idea is that the greater diversity of entitlements available to a person, household or community, the more likely they are to avoid becoming vulnerable and to be food-secure and consequently resilient" (Fendrychová & Jehlička, 2018). Alternatively, we can conceptualise resilience as future-oriented and change-enabling (DeVerteuil & Golubchikov, 2016). Individuals, households and communities that are able to combine diverse entitlements have greater opportunity to experiment and create ways of obtaining and consuming food that are novel and more sustainable (both environmentally and socially).

Many food consumers in Europe participate simultaneously in several variants of the food system (formal, market-based; alternative market; and non-market) and in their different configurations. The extent of their engagement in the latter varies significantly in different societies.

Formal and market-based value chains

Agrifood systems in the Global North are often understood and governed through the concept of value chains, which links the different functions of production, processing, trade and retail (Gereffi et al., 2005). Modern agrifood supply chains are governed by public regulations but also, importantly, by private standards concerning safety, quality and price, which structure the relationships between the actors via enforceable contracts and strict supply conditions (Fulponi, 2007).

The analysis of food value chains requires an understanding of nutrition, agriculture, food technology, economics, marketing, and more. Food value chain actors identify innovative ways to improve the availability, affordability and acceptability of nutritious foods, in the contexts of both undernutrition and overnutrition, and there is currently a push for conducting food value chain analyses in an integrated manner with multiple and diverse stakeholders (Fanzo et al., 2017). However, the potentially negative implications of the food industry, and how retail concentration can impact food system outcomes, are important concerns related to these actors (see e.g. Clapp, 2023; Wood et al., 2021).

Companies in the food value chain have an incentive to enlarge due to two mechanisms. Firstly, the relative decrease in agricultural prices over time has led to a higher concentration in industries purchasing raw inputs (Gaigné & Le Mener, 2014), while technology has fuelled economies of scale in logistics, purchasing and production. Secondly, companies have an incentive to escape competitive pressure by product differentiation, requiring economies of scale in marketing, which in turn is fuelled by technological innovation, free trade and globalisation. As a result, the concentration in multinational food companies and retailers has increased tremendously. A recent study based on Euromonitor 2018 data found C4 concentration ratios (the sum of the%age of market share of the four largest firms) as high as 60% across the EU in packed food subsectors such as breakfast cereals, carbonated drinks, confectionery, energy drinks, ice cream, soups and sports drinks (Van Dam et al., 2022). The authors found concentration ratios to be significantly higher in subsectors offering ultra-processed packed food products. Therefore, the question arises of whether corporate concentration and globalisation have a negative impact on diets.

Evidence shows that technology, globalisation and corporate concentration may correlate with each other, but the exact mechanisms of cause and effect are more complex, leading to mixed evidence (see Zimmermann & Rapsomanikis, 2021 for a recent overview). Evidence supports both positive and negative effects of globalisation, free trade and communication on healthy and sustainable diets. For example, trade increases dietary quality and reduces malnutrition (Cuevas García-Dorado et al., 2019), but also increases access to less healthy and less sustainable foods and drinks, which results in an association between trade and obesity. In the same way, free communication may spread messages about the less healthy 'Westernisation' of diets, but at the same time also provide wider access to knowledge about healthy diets (Knutson & de Soysa, 2019).

An often-used argument (found in Fox et al., 2019), the Dependency/Worlds System Theory, is that economic and cultural globalisation has paved the way for the diffusion of a lifestyle that is detrimental to human and planetary health, which has been accelerated by free trade, profit-driven multinational corporations and technological changes which support more effective distribution and communication systems. An alternative argument (the Modernisation Theory) is that domestic processes of modernisation resulting from economic development and thus higher incomes, have both positive and negative consequences for diets. Urbanisation and democratisation leads to a nutrition transition and decreasing rates of physical activity, while women's empowerment results in higher participation in the workforce and thus higher demand for convenience. Using longitudinal data from 190 countries from 1980 to 2008, Fox et al. (2019) conclude that rising obesity is robustly predicted by processes of economic development, rather than globalisation.¹²

Evidence also shows both positive and negative effects of trade on environmental outcomes. The positive effect of trade includes a better use of natural resources globally (Roux et al., 2021). However, increased demand for imports can elevate environmental impacts in exporting countries beyond acceptable thresholds.

¹² This statement seems to contradict observations that particularly the poor suffer from obesity (for example, Yach et al., 2005). However, those observations have not been confirmed by empirical studies. For instance, Goryakin and Suhrcke (2014) used a large set of demographic and health survey finding a positive and concave relationship between national per capita income and obesity.

A growing number of studies are raising concerns about large multinational corporations as driving forces in the increased consumption of sugar-sweetened beverages and processed food rich in salt, sugar and fat (Stuckler et al., 2012). Increasingly, studies point to the political activities of these companies (Baker et al., 2020; Sievert, Lawrence, Parker, & Baker, 2022), giving rise to a new literature on the commercial determinants of health (Kickbusch et al., 2016; Maani et al., 2022) and the role of power in particular (Lacy-Nichols & Marten, 2021). Recently, Gilmore et al. (2023) defined the commercial determinants of health as "the systems, practices, and pathways through which commercial actors drive health and equity", and proposed a comprehensive framework to analyse these determinants. In addition to political practices, such as lobbying, Gilmore et al. emphasise that companies are not only operationalising their power though political practices such as lobbying, but also by using practices related to finance, marketing, science, supply chain and waste, labour and employment and reputational management.

Alternative food chains, informal food provisioning and donation

This section provides a more holistic or 'real world' understanding of the way people obtain and exchange food to fulfil their food-related needs. As mentioned in the previous section, the concentration, specialisation and financialisation of largely global and market-based food systems and value chains has been criticised because of the externalities in terms of social, economic and environmental impacts (Ericksen et al., 2010). Here, we explore 'alternative market' concepts with respect to the conventional market, as well as the 'non-formalised' (non-market) exchange networks via key aspects linked to sustainable food consumption, and finally the role of food donation and thirdsector networks.

Alternative food networks and short food supply chains

Various terms and concepts are used interchangeably to identify farmers' markets, community-supported agriculture, box schemes and other similar initiatives. There are dichotomies between conventional vs alternative, short vs long, localised vs global and traditional food chains that have been widely discussed in relation to how sustainable they are.

The term 'short food supply chain' refers to the physical and social distance between the producer (farmer) and the consumer. Local and global food chains are defined using several criteria that ultimately refer to the spatial distribution of the actors involved, with no clear distinction between them. Analyses show a highly dynamic local-global continuum exemplified by multinational companies adopting localisation strategies and local producers attempting strategies to integrate into the global market (Brunori & Galli, 2016).

Food systems, food environments and their drivers

Various movements and initiatives are committed to alternative models in the pursuit of sustainability goals, such as alternative food networks (AFNs) (Forssell & Lankoski, 2015; Michel-Villarreal et al., 2019). Such networks are 'alternative' because they allow consumers to make connections to the place of production, the production methods used, and/or the people who produce food (Ilbery et al., 2005; Marsden et al., 2002; Renting et al., 2003).

Bowen and Mutersbaugh (2014) identify two different approaches in alternative food research that focus on food (re-)localisation. The first approach, prevalent in Mediterranean Europe, emphasises 'terroir', 'slow food', and geographical origins to differentiate products by linking product quality to the specific conditions of production in a particular area. European Protected Designation of Origins and Protected Geographical Indications are emblematic in this sense and include a wide variety of different schemes used to create a link between a place and a product (Barham, 2003; Libery & Kneafsey, 1998). Because of their focus on food quality rather than social and economic reconfiguration, Watts et al. (2005) described these types of alternative food networks as "weak" alternatives.

The second approach, originated mainly in the United States and United Kingdom, focuses more on local food system initiatives that aim to reconnect producers and consumers (Albrecht & Smithers, 2018), to "resocialise" and "respatialise" food (Hinrichs, 2000; Milestad et al., 2010). Due to the emphasis placed on social embeddedness, Watts et al. (2005) referred to these food alternatives as "stronger" alternative food networks. Community-led initiatives are seen as an opportunity to foster networks founded on practical common interests that could help to enhance resilience and bring about broader food system transformation (Blay-Palmer et al., 2016).

It is important to emphasise that short food supply chains are not isolated models; they coexist, cooperate or competitively interact with other food chains, and they are always evolving (Thomé et al., 2021). 'Hybrid' or 'mid-tier' chains are also terms used to identify examples between local and global. A number of authors contest the simplistic opposition between conventional and alternative models (Holloway et al., 2007; Sonnino & Marsden, 2005) highlighting the risk of "defensive localism" (Winter, 2003), or missing the opportunity to take into account the diversity of local food systems (Mount, 2012). Furthermore, several short-chain initiatives involve more than one intermediary, without prejudice to those intermediaries who draw upon both local and global resources. Such examples combine conventional food system infrastructure with the more alternative goal of building local food systems (Bloom & Hinrichs, 2011).

The sustainability performance of conventional, alternative or hybrid food supply chains has received significant attention from researchers, finding there are no best-performing supply chains on all sustainability attributes. Evidence shows that local chains perform better than their global counterparts for some sustainability attributes and worse for others. Results dependent strongly on the context, the actual behaviour of supply chain actors, the assessment methodologies, and the perception of external observers (Brunori et al., 2016).

While there seems to be a general agreement about the use of alternative food networks as a response to environmental and social consequences of the conventional agrifood sector, there is less of a consensus "when it comes to [their] ability to ameliorate or mitigate these issues" (Fendrychová & Jehlička, 2018). For instance, Sonnino and Griggs-Trevarthen (2013) note that even in affluent regions such as south-east England, with its history of strong civic activism, the persistence of alternative networks is dependent on dedicated individuals or small activist groups and external sources of funding. One key aspect to consider in relation to food access is that sustainability performance centres on the predominance of middle classes within alternative food networks (Goodman et al., 2013). This debate emphasises the inequitable nature of such alternative networks as a path to sustainability: higher prices relate to buying power and ultimately access to food (Smith et al., 2015).

Informal food provisioning

When considering informal and non-market food provisioning, we refer to Sen's (1981, 1984) concepts of endowment as elaborated by Osmani (1993). 'Endowment' denotes the resources owned by a person, both tangible (land, equipment) and intangible (skills, knowledge, membership in a community; and entitlement, all combinations of goods and services a person can obtain using their resources (Jehlička et al., 2019). These resources can be used to obtain food in three ways:

- **exchange**, where people use their labour to earn money to purchase food
- **transfer**, as members of their community people obtain food as a gift
- **production**, where people produce their food using their land and skills

In the Global South, many food markets operate informally. In industrialised countries, food trade outside of formal value chains is highly unusual, and often remains under the radar of research and policy (Pinto-Correia et al., 2021). Research efforts have helped to shed light on the unseen circuits of food self-provisioning across Europe, in which small farms play a central role. 'Unseen' means food exchange or consumption that happens under the radar of traditional economic assessment. This unseen food does not enter formal marketing routes, and is either consumed by the household, or exchanged or gifted amongst neighbours and family (Fendrychová & Jehlička, 2018; Schupp & Sharp, 2012; Teitelbaum & Beckley, 2006).

Food systems, food environments and their drivers

Non-market and informal food exchanges are captured particularly by literature from eastern Europe.¹³ There is significant difference in the extent of engagement in food self-provisioning between eastern European EU countries and the rest of Europe. The%age of the population who grew food in their own households is:

- 51% of Czechs (in 2022 unpublished source)
- 36% of Hungarians in 2013 (Balázs et al., 2016)
- 50% of Croats (Jehlička et al., 2021)
- 54% of Poles in 2011 (Smith & Jehlička, 2013)

Along with home-grown food received as a gift within the omnipresent food-sharing networks (for instance, 61% of Czechs receive food as a gift), this informal food production accounts for significant volumes of food. In the case of Czech food-growing households, this informal economy is responsible for about 40% of their consumption of temperate zone fruit and vegetables (Jehlička & Daněk, 2017).

The food produced and then consumed by households and shared within interhousehold networks holds value that is "expressed in terms other than price" (McMichael, 2015). The dominant motivations for food self-provisioning are healthy and fresh food, while financial savings ranking only third or fourth, with much lower significance than the first two motivations. While informal food production and sharing are predominantly motivated by the desire for fresh and healthy food, on a few occasions these motivations extend to sustainability or environmental protection. These informal practices represent the shortest possible food supply chain, relying on largely organic production methods and contributing to strong social bonds. Despite the weak environmental motivation behind these practices, they bring significant and tangible sustainability benefits, and for this reason they have been referred to as "sustainability by outcome rather than intention" (Jehlička et al., 2020).

Food donations and redistribution

Household food insecurity has been defined as "the inability of people to acquire or consume an adequate or sufficient amount of food in a socially acceptable manner, or the uncertainty that one will be able to do so" (Dowler & O'Connor, 2012). In this sense, food poverty is primarily conceptualised as the lack of access to a safe and healthy diet, explained by insufficient income and poverty (O'Connor et al., 2016). It also refers to psychological wellbeing, which is affected by feelings of uncertainty and consequently fear, as well as possible exclusion from social life (Riches & Silvasti, 2014).

¹³ For instance, data show that between a fifth and a quarter of Polish and Czech populations are involved in the informal production of organic food and vegetables, avoiding use of industriallymade fertilisers and pesticides and accounting for a large proportion of their consumption of these foodstuffs (Smith et al., 2015). Hence, the extent of organic food production and consumption is far greater than the level suggested by analyses based only on organic food purchases.

Food aid initiatives take different forms, with several variations existing in practice to complement or innovate with more traditional initiatives (Hebinck et al., 2018), leading to an extremely diverse landscape. Lambie-Mumford and Silvasti (2020) provide a broad typology, which includes:

- food parcels provisioning, such as parcels of food that people can take away, prepare and eat, assuming that recipients have the facilities to do so
- prepared food, such as soup kitchens that provide cooked food that people can eat or take away
- new forms of food aid such as subsidised food shopping, reliant on EU-subsidised foods and supplied through additional donations, for example, social supermarkets with some form of membership card
- subsidised prepared food, such as in social cafés and canteens

Additionally, food drives and food recovery initiatives are often organised at local or national level to respond to growing demand, involve citizens and raise awareness of food surplus and redistribution to meet social needs.¹⁴

Decisions at the EU level impact on the operation of social welfare in European countries, but there is no common social policy at the EU level, in contrast with the Common Agricultural Policy. Responsibility for social and poverty policies lies primarily with national governments. Historically, there have been major differences in the governance of welfare and the political and cultural histories across Europe. In the European Union, Programme Européen d'Aide alimentaire aux plus Démunis (PEAD), a programme of surplus food redistribution rooted in the Common Agricultural Policy, existed for almost 30 years to make agricultural surplus from the CAP available to European food charities for redistribution to disadvantaged people (Caraher and Cavicchi, 2014).

In 2014, PEAD was replaced by FEAD (Fund for the European Aid for the most Deprived), which is part of social policy, compulsory and co-funded by all EU member states. FEAD integrates actions carried out by EU countries, to provide food aid and material assistance to the most deprived, such as essential goods for personal use, beyond food. The aim is to provide short-term assistance and immediate, emergency support on the most basic needs. This helps provide the necessary (but not sufficient) conditions to find a job and step out of poverty.

Charitable organisations were, and still are, appointed as frontline distributors of FEAD resources. The provision of food by charities has a long tradition in Europe and in some countries, such as Italy and Spain, religious actors have played a major role in the history

¹⁴ The European Federation of Food Banks was founded in 1986 and has members from 24 European countries.

of welfare state development and the operation of charitable food assistance (Galli et al., 2018; Lambie-Mumford & Silvasti, 2020).

In 2019, a mid-term evaluation of FEAD was carried out for the period 2014–2017. The Commission Staff Working Document¹⁵ presents the main findings that confirm the fund's effectiveness (helping beneficiaries make ends meet), coherence, and relevance. There was some criticism relating to the efficiency and cost of the policy. After the covid-19 pandemic outbreak, the European Commission made an amendment to the FEAD regulations, most recently in 2021, to increase available funds for the following years.

Over the years, particularly in the US and Europe, research on charitable food aid has increased. This research highlights a number of criticisms of current food aid policies and practices. For example, there is conflicting evidence on the effectiveness of food banks and other food assistance practices in alleviating household food insecurity (Loopstra, 2018) with reference to the U.S. context). The contradiction between food poverty affecting a large part of the global population and the everyday wastage of food, particularly in high-income countries, has pointed academic and public attention to the practice of using food surplus (sometimes referred to as 'food waste') to fulfil the needs of the poor. This gives rise to possible tensions between food poverty alleviation and food waste reduction. On the one hand, actors in the food chain should prevent the generation of food waste, by collecting and redistributing edible food according to a hierarchy of destinations.¹⁶ On the other hand, food surplus is a key resource for food poverty mitigation, so the dependence of food assistance organisations on food surplus for poverty mitigation makes them vulnerable and exposes them to several drivers of change (Galli et al., 2019).

There is a fundamental tension between charity and the right to food. Several authors in the field of social policy criticise charitable initiatives as insufficient measures that prevent governments from responsibly protecting the right to food. The retreat of welfare states reinforces the role of charitable food assistance, which by definition is not universal (Lambie-Mumford & Silvasti, 2020; Riches, 2018). There are also major concerns about the nutritional quality of food assistance, but new practices in charitable food assistance appear to address this issue. While food banks play an important role in providing emergency solutions to severe food deficiencies, they are limited in their ability to improve overall food security because they provide insufficient quantities of nutrient-dense foods, especially dairy products, vegetables and fruits (Bazerghi et al., 2016). This research suggests that food banks have the potential to improve food security

^{15 &}lt;u>https://op.europa.eu/en/publication-detail/-/publication/0e03aa7b-025f-11e8-b8f5-01aa75ed71a1</u>

¹⁶ See, for example, <u>https://www.eca.europa.eu/Lists/ECADocuments/SR16_34/SR_FOOD_WASTE_EN.pdf</u>, Figure 1, p. 10.

if operational resources are adequate, perishable food groups are available, and client needs are identified and addressed.

1.3. Situating the food environment within the food system

The food environment is a pivotal concept in this report, describing the context in which food is acquired and eaten. It is a key constituent of the food system and comprises the multidimensional physical, economic, political and sociocultural situations in which consumers carry out the practices which relate to the planning, acquisition, transport, storage, preparation, eating and disposal of food.

In concrete terms, this includes a huge variety of settings, not only retail but also hospitality services, food banks, homes, workplaces, public facilities, restaurants, transport, other public and private venues, social media platforms, and so forth. One recent and comprehensive definition of food environments, intended to be applicable to low-, middle-, and high-income countries, is:

the consumer interface with the food system that encompasses the availability, affordability, convenience, promotion and quality, and sustainability of foods and beverages in wild, cultivated, and built spaces that are influenced by the sociocultural and political environment and ecosystems within which they are embedded

(Downs et al., 2020)

The food environment is the 'interface' which mediates people's 'interaction', and these two facets need to be emphasised in order to explicitly link it to people's daily lives, activities and diets. The food environment is an interface in the sense that it mediates the acquisition of foods by people within the wider food system and lies between an external and an individual domain (Turner et al., 2018). The external domain of the food environment includes the physical space where food is made accessible, in which infrastructure such as the built environment is set, and prices, information and promotion are provided. The individual domain of the food environment relates to the conditions for individual consumers' food access, which relate to daily routines and practices which can be affected by multiple factors, such as physical distance from points of purchase, time availability, individual mobility (for instance, working close to home or commuting; the ability to move autonomously or dependence on other people's support), purchasing power, convenience, the possibility of self-producing, harvesting and preparing one's own foods, cultural conventions and acceptability, and proximity to knowledge and skills. The external domain of the food environment has been studied more extensively than the personal domain (Penney et al., 2014; Turner et al., 2018).

Food systems, food environments and their drivers

In absolute terms, the food environment is unique for each individual (and community) and it is clear that such interaction is shaped by multiple interlinked factors. In order to understand the drivers of change, and the key underlying mechanisms, the food environment should be contextualised within the food system of reference. The food environment evolves alongside the ways in which people access, prepare and consume food. These methods have shifted progressively, over time and space, from locally produced and purchased foods to globally supplied, long-distance transported products and purchases (Herforth & Ahmed, 2015; Mozaffarian, 2016).

The food environment can also be approached at different levels, like the wider food system, ranging from the individual, to the household/community, to the higher national, regional and global levels (Caspi et al., 2012). Each level brings significant implications in terms of the monitoring and assessment of food environments (Downs et al., 2020). In addition, the conceptualisation of the food environment can be more or less narrow, or expansive at each level, depending on which aspects need to be considered (and ultimately, which problem needs to be addressed). The following broad categories are important elements of the food environment that influence consumer food activities and diets: physical and economic access to food; food promotion; advertising and information; and food quality and safety (Hawkes et al., 2015; Swinburn et al., 2015).

The underlying assumption, or hypothesis, is that healthy food environments enable consumers to engage in nutritionally balanced food practices, with the potential to improve diets and reduce the burden of malnutrition. Conversely, unhealthy food environments promote less healthy dietary practices by consumers. Conceptualising 'accessibility' to healthy food was fundamental to the development of food environment re-definitions, with food access dimensions including: availability, as the presence of the supply; accessibility, in terms of how easy to get to the location by time and distance; affordability, meaning price, or worth relative to cost; acceptability in terms of attitude and satisfaction; and accommodation, how well local food sources adapt to local residents' needs (Penchansky & Thomas, 1981).

The literature on healthy food environments has evolved considerably with respect to 'food deserts', and barriers to healthy food access (Walker et al., 2010). There is growing evidence that a healthy food environment plays an important role in healthy food choices, although it is important to remember that choices are not influenced solely by the food environment. Healthy conditions in the food environment are not necessarily a guarantee that healthy choices will be made, but it becomes very likely that they are (Caspi et al., 2012).

Conceptualising food purchasing and consumption patterns in terms of the number of grocery stores or the availability of food risks viewing people as passive consumers who are simply confronted with a physical space that provides a set of market-based
consumption incentives. Consumers also play an active role in shaping the food environment by demanding the availability of certain foods, such as the increasing availability of organic foods, fair trade foods, or plant-based foods and meat substitutes in supermarkets (see Fuentes & Fuentes, 2022). On the other hand, buycotts and boycotts are additional means by which consumers can express their concerns, priorities, values, and interests about food.

Each dimension of the food environment — physical (availability, quality, and promotion), economic (cost), policy ('rules') and sociocultural (norms and beliefs) — has a substantial impact on consumers' food activities and diets (see Figure 2 below, enlarged from Figure 1, p. 20). Unhealthy food environments that are dominated by energy-dense, nutrient-poor, ultra-processed food products, relatively inexpensive and heavily promoted, create a supply-side 'push' effect on less healthy diets and energy overconsumption, which is a driver of population-level food habits and negative health outcomes (Brouwer et al., 2021).



Figure 2. Focus on food environment dimensions. Authors' elaboration, based on Turner et al. (2018) and Downs et al. (2020). This is a section of Figure 1, p. 20..

The dimensions of the food environment that are addressed later in this report (Chapter 4, p. 83), with regard to the impact of public and private policy instruments on healthy and sustainable food consumption, are briefly summarised here:

Price. The economic environment plays a fundamental role in sustainable and healthy diets, as prices drive people to adopt behaviours that may prove more or less sustainable. Changing the incentive structure, whether through taxes or subsidies, affects prices and in turn people's choices, particularly in relation to their budget constraint (affordability).

Food systems, food environments and their drivers

- Physical availability. This relates to the physical product and the extent to which food availability affects sustainable and healthy food consumption in different settings, such as supermarkets, shops, restaurants, canteens, schools and neighbourhoods.
- Product placement. Prominent placement of food products (e.g. at eye-level or on displays) is a promotion tool frequently used by supermarkets and food companies to increase sales with target products. These measures can also be used to promote healthy and sustainable foods.
- Out-of-home consumption opportunities. In addition to the effect of physical availability in the (grocery) purchase environment, placement also matters in other contexts of food consumption. These include, among others, food consumption in schools or restaurants and canteens.
- Neighbourhoods. The neighbourhood food environment refers to the broader context in which individuals find themselves. Its impact on diet quality and health outcomes can be understood in terms of the diversity, density, and proximity of food stores.
- Food quality and composition. In recent years, attention has shifted to the composition of individual foods (for example, ingredients, formulations, and types of processing), because the level of processing has a major impact on dietary health. Reformulation of products can help to improve nutritional quality.
- Marketing, information campaigns and advertising. This refers to the information environment created for consumers by food businesses competing in the market along the food value chain.
- Labelling. Food labelling is a broad area of research that spans nutrition and health as well as environment and ethical qualities of food products and processing. Food labels are effective to the extent to which the buyer reads and understands the message that the labelling aims to convey.
- Social environment. People make reference to others within their social environment and this exerts a powerful influence on behaviours, including food consumption. Social influence and norms can be part of both the problem and the solution to sustainable and healthy food systems.

Among the multiple factors that contribute to shaping food environments, food systems institutions (the system of rules embodied into collective behaviour; the 'rules of the game') are crucial to the modes of interaction that are set and implemented in food systems. These include cognitive rules (i.e., cultural assumptions), normative rules (i.e., social obligations) and regulative rules (i.e., legislation and institutional rule systems) (Scott, 1995).

Food supply is mostly provided by the private food industry, which determines to a large extent its availability, quality and price. It also promotes the consumption of ultra-processed foods and fast food, while shaping social norms and beliefs about food.

Governments' policies, laws and regulations provide the rules by which the private sector must operate (Swinburn et al., 2013). Policy instruments shape the food environment by changing production and processing (supply), by affecting consumption (demand), and by directly changing features of the local food environment (Galli et al., 2020). Governments can also influence sociocultural norms through health promotion and social marketing.

Based on cultural norms for food and cuisines, and through traditional, cultural and religious practices, individuals interact with the food environment and shape their diets. Personal habits, preferences, education and income all come into play. In addition, there are interactions between the food industry, governments and society, not only at the food environment interface but also on many other levels, such as through policymaking, science funding, lobbying and agenda-setting. Particular concern has been raised recently about the increasingly high level of influence that the private sector has on governments, with its enormous market concentration and lobbying power, especially when regulations and fiscal policies are proposed. Industry bodies are often given a seat at the 'policy-development table' (for example, government advisory committees, task forces) even when deep conflicts of interest exist between commercial benefits and public health benefits (Maani et al., 2022; Swinburn et al., 2013).

In order to reduce obesity and diet-related noncommunicable diseases, there is a call for research to better understand interventions on the external and structural food environment dimensions which can support individual behaviours (Swinburn et al., 2011). Policy should also support and enable 'healthy food environments' which shift population diets, especially those of socially disadvantaged populations, towards diets that meet dietary guidelines.

There is a need for careful and comprehensive monitoring of food environments, including the evaluation of the impact of public and private sector policies (Hawkes et al., 2015; Swinburn et al., 2013). The INFORMAS framework includes several modules that monitor the impact of the nutrient composition of available foods, food labelling, the extent and nature of food marketing, the provision of foods, the availability of foods in communities, prices and the affordability of foods, and the risks to food environments within trade and investment agreements (see among others, Ni Mhurchu et al., 2013; Swinburn et al., 2015; Vandevijvere & Swinburn, 2014). The effectiveness of food environment policies in improving population diets is addressed by a recent review (Hansen et al., 2022) and demonstrates significant potential for the EU to strengthen its policies, priority actions and infrastructure support in order to improve food environments (Djojosoeparto et al., 2022).

1.4. The digital food environment

For more than a decade, digitalisation has propelled changes within the food environment. Increasingly, digital infrastructures and technologies mediate how people seek, share and interpret food-related and eating-related information and practices (Schneider & Eli, 2021). There are many practices that have become common for European consumers, including online grocery shopping, ordering and requesting the delivery of takeaway food through digital platforms, dietary self-tracking, and searching for recipes online. Although unequal access to the internet and digital devices limits this concept, some researchers propose that consumers are living in a "digital food environment" (Granheim, 2019; Granheim et al., 2022). Even if "no evidence-based conceptual framework or definition of the term exists", Granheim (2019) proposes that "three factors [...] could influence diet-related outcomes: digital actors (such as governments, the food industry, the media and individuals), digital settings where such actors operate, and digital activities performed by such actors in the digital sphere".

Granheim et al. (2022) conducted a systematic scoping review of 357 research articles that covered two main search terms, "digital technology" and the "food environment", aiming to map existing research on the topic.¹⁷ They identified research trends relating to digital food environments and explored how the external and individual domains of food environments (see p. 35 of this report) change in a digital society. In short, they found that:

all food environment dimensions are subject to digital transformation. Food environments are increasingly experienced through technology, and also shaped by it in many ways *(Granheim et al., 2022)*

They also identified three emerging issues that go beyond, or do not easily fit into the dimensions of, Turner et al.'s (2018) conceptualisation of the food environment: digital settings, digital food culture and the interconnectedness of physical and digital food environments. Thus, digitalisation adds further complexity to an already intricate food environment that shapes food and eating practices.

We review Granheim et al.'s findings in more detail in the following sections, and supplement their scoping review with findings from additional relevant studies that have examined the digitalisation of food and eating.¹⁸ As the interdisciplinary field of digital food studies is a rapidly-growing field (Granheim et al., 2022) (see also Leer & Krogager, 2021; Lewis, 2020; Lupton & Feldman, 2020; Schneider et al., 2018), it is important to draw

¹⁷ The team acknowledges that the reviewed studies were "heavily skewed towards high-income countries, where 88% (n=314) of studies were conducted" (Granheim et al., 2022, p. 6).

¹⁸ Granheim et al. did not include and consider these papers for one of two reasons: either the studies did not engage with the concept of the food environment, or they had been published online since December 2019, as Granheim et al.'s review paper only reviewed papers published before this date.

on more recent publications as well as on those that use different terminology, such as "digital foodscapes" (Goodman & Jaworska, 2020; Schneider & Eli, 2021), which refer to the same underlying issue.

Granheim et al. (2022) emphasise that the external domain of the digital food environment has been studied more extensively than the personal domain. In particular, the dimensions of desirability, vendor and product properties, marketing and regulation have been studied most.

Digitalisation enables (Granheim et al., 2022):

- new forms of selling and buying food, for example, online grocery shopping and food delivery services (see also Fuentes & Samsioe, 2021; Samsioe & Fuentes, 2022)
- the potential to increase the availability of foods
- digital information and services (for example, online recipe portrayals) are the product or service that is consumed
- digital food marketing and related personal data harvesting have expanded
- the use of different computer interfaces, product placement on websites, and the kind of food images presented, which have been found to influence consumer choice

The digital food environment can best be understood as an expansion of the physical food environment and as an augmented experience of it, mediated by digital technologies (Granheim et al., 2022). Granheim et al. also call for more attention to "seemingly immaterial transformations" in digital settings that remediate experiences and relationships with food and eating. A number of researchers from sociology, human geography, media studies and related fields have started to investigate these transformations, with a focus on different digital settings including digital platforms such as Instagram (e.g. Contois & Kish, 2022) or restaurant reviewing platforms (Kobez, 2018; Onorati & Giardullo, 2020), and mobile phone apps for purposes such as self-tracking and food delivery (e.g. Lupton, 2021).

Part of what is referred to as the digital food environment is the so-called 'sharing economy'. In the area of food, this includes both profit-making commercial and noncommercial forms of food deliveries mediated via digital platforms. The profit-making platform-based companies connect providers of food (a meal prepared in a food outlet) with customers via an online platform or mobile app. The providers often claim to offer cheaper meals in a more convenient way (in terms of making choices and the setting of consumption) to the customer as a result of 'disrupting' the restaurant business. However, the environmental benefits of this model of the profit-making sharing economy are questionable.

The original idea behind the digital platform-based, non-profit-making sharing economy was to exploit unused or underused resources through efficient and trust-

Food systems, food environments and their drivers

assuring communication and, as a result, to reduce the environmental impacts of food consumption. With the transformation into the profit-making sharing economy, these digital-platform based companies no longer seek to make use of the otherwise unused or idle resources and instead offer new, on-demand services with attendant environmental consequences that may not have arisen without the availability of efficient digital communication. These commercial transactions facilitated by digital platforms can be contrasted with non-commercial forms of food-sharing that can be either facilitated by digital platforms or by personal communication. This non-commercial, food-sharing economy tends to be invisible to policymakers and researchers but is surprisingly widespread in a number of affluent countries of the Global North as diverse as Japan (Kamiyama et al., 2016) and Czechia (Jehlička & Daněk, 2017). Whether digitally based or not, these forms of the sharing economy are likely to bring more environmental and social benefits than the commercial ones.

Another scoping review has systematically explored the link between digital food communication and analogue food behaviour (Bartelmeß & Godemann, 2022). The authors suggest that "traditional approaches to food behavior no longer capture the complexity of food actions in digitalized societies and require an update with the emergence and increasing usage of social media". The authors of the review build on Contento and Koch's (2021) socioecological model of food behaviour that considers food behaviour as "determined by a conglomerate of different variables and processes operating in three distinct, interrelated domains: the food-related, the person-related, and the socioecological domains" (Contento & Koch, 2021, as cited in Bartelmeß & Godemann, 2022). They emphasise that this model considers communication as part of the information environment that forms part of the socioecological domain.¹⁹

Despite the potential advantages of drawing on the socioecological model of food behaviour, Bartelmeß & Godemann stress that the model's assumption that considers media and social media (as a form of communication) as part of the information environment needs to be critically reflected on, as it advances a teleological understanding of communication. Ultimately, they propose that behavioural models that conceptualise communication as having a linear influence on behaviour are outdated in a media environment that is defined by fast-paced social media communication in which food and eating are very prominent topics (see also Feldman, 2021; Goodman & Jaworska, 2020; Kent, 2021).

¹⁹ Some other models, such as the theory of planned behaviour or health belief model, fail to address communication, or view it as part of the individual behaviour variable and not in relation to the social communication context.

To illustrate the linkages between digital food communication and three areas of food behaviour (food choice, dietary intake and eating behaviour), we quote Bartelmeß & Godemann (2022):

Food choice (preferences, preparation and intentions):

- "Variables of all domains of the socioecological food behavior model are linked to digital food consumption in food choice" (p. 8)
- "Social media users are now socialized in their food behavior in multiple ways, leading to the emergence of new, socially informed preferences and food choice patterns" (p. 8)
- "In the experience-related dimension, personal physiological conditioning becomes less important, as other communicators now share their experiences with certain foods, dishes, or restaurants online, evaluate them, and thus, act as deputy experiencers" (p. 8)
- "Variables such as knowledge or social and cultural norms are elevated from a formerly intrapersonal level to an interpersonal level through food communication on social media, blurring the boundaries of categories even within domains" (p. 9)
- Online communities serve as culinary support by providing guidance, motivation, and inspiration for using and recombining, contributing to a reservoir of practical knowledge within online communities that guides member users in analog food preparation and planning" (p. 9)
- Engaging in social media food communication is understood as a reciprocal process in which different variables of the food behavior model interact and influence each other and affect certain outcomes of food choice behavior" (p. 9)

Dietary intake (healthy or unhealthy, food components, etc.):

- The area of dietary intake is the only one for which there is empirical evidence of a direct link between communication and behavior, possibly due to the fact that outcomes in this area can be measured" (p. 9)
- This field assumes a very abbreviated, linear understanding of the linkages of food communication on social media platforms and the analog behaviors performed; comparably, dietary intake also addressed the fewest behavioral variables, with links made only to determinants in the socioecological environment and the person-related domain" (p. 9)
- "Studies are typically behavioral science studies that are experimental in design and measure the effects of exposure to food communication on behavioral outcomes in laboratory settings" (p. 10)
- Analogue eating behaviour (habits, occasions, portions, etc.):
 - "Review found that in the area of eating behaviour, habits, occasions, dieting, and disordered eating may be linked to social media food communication" (p. 6)

Food systems, food environments and their drivers

- In the experience-related dimension, the studies reviewed indicate that social media platforms function as a social affective context that provides a digital platform for the social conditioning of analog eating behaviors" (p. 7)
- The linkages of social media food communication and analog food behavior are described as performative, in that the use of social media to communicate about and during eating directly influences eating behavior outcomes" (p. 7)
- Within the person-related domain, intra- as well as interpersonal determinants are affected by social media communication and are inferred to subsequently inform analog eating behaviors; it is argued that eating habits or dieting practices are governed through digitally conveyed food meanings and virtual social relationship networks that serve as collective identity pools that construct and provide symbolic structures for the orientation of analog eating behavior practices" (p. 7)

In conclusion, Bartelmeß and Godemann (2022) propose that, for the most part, variables of the person-related domain and the social-ecological environment have an effect on the linkages between digital communication and analogue behaviour. Yet, their review did not find a demonstrable overarching impact on the three areas of food behaviour, or the influence of social media food communication on analogue behavioural outcomes.

The impact of digital innovation on the food environment

Digitalisation is a driver of change in the food environment that has an impact on both external and personal domains. External domains include food availability, prices, vendor and product properties, marketing and regulation, while personal domains involve individual level dimensions, including food accessibility, affordability, convenience and desirability. Although research on the topic is increasing, the majority of studies do not assess the effect of digitalisation on all aspects of the food environment simultaneously. Moreover, only limited research considers the digitally-enabled interactions between external and personal domains. This means that current evidence is predominantly available about certain dimensions such as convenience in relation to specific case studies (for example, food delivery apps).

In this section, we selectively review existing research on digital innovation with a focus on digital food retail services (following Fernandez & Raine, 2021) and dietary tracking, seeking to emphasise how digitalisation reshapes the food environment and the impact this can have on healthy and sustainable food consumption. Food ordering and delivery apps and food tracking apps are the most downloaded food-related apps (together with games; Lupton, 2021). In Europe, the number of users of the online food delivery market including grocery and meal delivery have been steadily increasing between 2017 and 2021 (Statista, 2023).²⁰ In 2022, the estimated number of European users of online meal delivery platforms amounted to 211.1 million users and of online grocery delivery up to 164 million users (Statista, 2023). Nutrition apps, which includes dietary tracking apps, have also grown in popularity. Overall, the number of users of nutrition apps has been growing between 2017 and 2020 in Europe (Statista, 2023a).²¹ Beyond commercial platforms, it is important to consider the role of non-commercial platforms and their impact (Bos & Owen, 2016; Eli et al., 2015; Hoelscher & Chatzidakis, 2021), while acknowledging potential digital labour and data extractivism of digital platforms that are free of charge (Schneider & Eli, 2022).

In their review paper on digital food retail, Fernandez and Raine (2021) foreground how the digitalisation of the food environment (see Granheim et al., 2022) provides opportunities for food retailers that have led to the emergence of various novel digital food retail services that enable consumers to purchase groceries and meals through websites and apps. They state that three categories of digital food retail services are particularly prominent given their reach and market penetration, whose impact is as follows:

- Online groceries. There is great potential for online groceries to support healthy eating, as online grocery shopping reduces the amount consumers spend on food and the quantity of less healthy foods they purchase. "Generally, consumers are more likely to spend more money on items when benefits are immediate and tangible (for example, in-store purchases) and are more likely to purchase items with short-term benefits (for example, tasty foods high in sugar, salt, fat) rather than long-term benefits (for example, healthy foods)" (Fernandez & Raine, 2021).
- Food delivery. Food delivery apps tend to stimulate out-of-home food consumption and therefore the intake of nutrient-poor and energy-dense foods (Fernandez & Raine, 2021). This also poses a risk factor for the increase of noncommunicable diseases (see also WHO Regional Office for Europe, 2021).

Meal kits. The evidence on the impact of meal kit services suggests they can support healthy eating, although they have limited reach today. "Meal kits can remove the food preparation burden (planning and shopping), making it easier to actually cook meals at home and prepare foods that fit with dietary preferences, restrictions, or health concerns" (Fernandez & Raine, 2021; see also Hertz & Halkier, 2017). Other studies explore the potential of digital food provisioning platforms for more sustainable shopping (Fuentes & Samsioe, 2021; Heidenstrøm & Hebrok, 2022; Samsioe & Fuentes, 2022). Meal box providers advertise a subscription to a meal box as a way to reduce food waste because no food will be left over if each meal is

²⁰ https://www.statista.com/forecasts/1297721/users-online-food-delivery-europe

^{21 &}lt;u>https://www.statista.com/outlook/dmo/digital-health/digital-fitness-wellbeing/digital-fitness-wellbeing-apps/nutrition-apps/europe</u>

prepared and consumed according to the instructions that accompany the meal box ingredients (Heidenstrøm & Hebrok, 2022). In addition, meal kits containing mainly local and vegan foods are presented as a sustainable means of food consumption. Potential limits on the uptake of meal boxes are the cost of the service, which leads to households with higher incomes subscribing to a service (Fernandez & Raine, 2021), and the accompanying re-organising of household practices that meal kit subscription instigate (Fuentes & Samsioe, 2021).

Digitalisation can also enable the bypassing of the food retail sector. New digitallyenabled initiatives have emerged that seek to foster direct producer consumer relations. For instance, the company Crowdfarming allows farmers to sell directly to consumers across the EU. The company has built a digital platform which provides logistics and customer service so that farmers can sell their produce directly to end consumers. Consumers can also adopt produce which gives farmers the chance and certainty to grow a crop knowing that someone will buy and consume it. This reduces potential food waste and guarantees an income to farmers. In this case, digitalisation enables the establishment of an alternative supply chain, which could support the development of a sustainable food system.

However, we currently lack systematic reviews and evidence on the impact of digital platforms on supporting the development of a more resilient and sustainable food system. Studies that have focused on the digitalisation of agricultural production (not consumption) problematise overly optimistic outlooks related to social sustainability by pointing out that new digital technologies and platforms impact labour and rural communities and attest a need to develop equitable, non-exploitive agricultural technology (Carolan, 2020; Rotz et al., 2019; for a review see Klerkx et al., 2019).

1.5. Food consumption behaviour

One of the key constituent elements of food systems is consumer behaviour. Consumer behaviour is a rich but disparate and divided field of research (Warde, 2022). There is no single and uniform way of defining consumption and consumers, given the varied disciplinary traditions in this area. According to Warde (2022), and tracing back to the etymological root of the verb 'to consume', there are two general meanings of consumption. One has a clear affinity with the disciplinary tradition of orthodox economics and psychology, focusing on exchange and commodification. In this case, the 'consumer' is an 'individual' that shops or buys products and services in the 'market' according to their individual preferences and choices. The other meaning points towards appropriation and utilisation encompassing the scientific field of sociology, cultural geography, anthropology, among other social sciences. In this second meaning, what is emphasised

is social embeddedness, i.e. the context that frames the sociomaterial use of goods and services, and everyday social activity (Warde, 2022). In this vein, consumption is framed as a collective and sociocultural phenomenon where issues of identity, material culture and practice are highlighted.

There has been an effort to bring both meanings together, by focusing on consumption as encompassing all the activities of planning, acquiring, using and maintaining as well as disposing of goods, services, activities and experiences over time. One consequence of such a multidisciplinary approach is that consumers can be portrayed in a variety of ways, as choosers, buyers, citizen-consumers, practitioners, users, caregivers, carereceivers, and identity-seekers, among others. It is important to note that such a plurality of framings for consumption and consumers is conducive to different forms of understanding, which depart from different research questions asked, leading to diverse conclusions which serve distinctive purposes. Thus, all approaches have their merits and limitations. Each one highlights aspects of consumption, consumers and their behaviour, while neglecting the contribution of other factors.

In this report we take an interdisciplinary approach, departing from a definition of consumption that has the advantage of offering versatility, to encompass manifold problem framings and agendas:

consumption is a process whereby agents engage in appropriation and appreciation, whether for utilitarian, expressive or contemplative purposes, of goods, services, performances, information or ambience, whether purchased or not, over which the agent has some discretion

(Warde, 2005, p. 137)

This definition stresses that consumers have some (and not whole) discretion in consumption processes. This means that they are often not in control over what to choose, being dependent on the configuration of the food environment, and the food system at large, for shifting their practices. It highlights that food consumer behaviour involves a set of moments, namely acquisition, appropriation and appreciation of food (Warde, 2010).

Acquisition refers to the dynamics, arrangements and conditions of economic and social exchange in consumption whereby goods and services are procured. Appropriation covers the variations of how consumers use goods and services and what is being done with goods and services in which processes. Appreciation concerns the meaning-making made in relation to consumption activities.

Halkier et al. (2017, pp. 1–2)

However, given the recent response that these moments are centred at the 'front end' of consumption, it is important to consider the counterpart to each of these three moments at the other end, namely disposal, divestment and devaluation (Evans, 2019).

Food systems, food environments and their drivers

Disposal can be considered the counterpart to acquisition, as goods, services, and experiences are acquired they can also be disposed of in a variety of forms. Some can go to the waste stream, while others can be reused or transformed (for example, leftovers for future meals). Devaluation can be considered the counterpart to appreciation, as people cease to attribute value (either economic or symbolic) and meaning to eating particular foods. For example, in the transition to an increasingly plant-based diet, a process of devaluing the importance of eating meat on a regular basis takes place. Finally, divestment is considered the counterpart of appropriation: just as goods, services and experiences around food can be used, "personalized and domesticated, so too can these attachments be undone" (Evans, 2019, p. 506). For example, people may divest from cooking with familiar kitchen tools (for example, pots and pans) and replace them with less domesticated cooking robots or digitally mediated technologies that demand to be appropriately handled by learning new skills to navigate competently, with new cooking equipment and technologies. In this case, they divest time and skills in particular cooking practices and reinvest new skills and time learning how to cook in digitally mediated food environments.

By including disposal, devaluing and divestment in the definition of food consumer behaviour we address all the modes and ways in which consumers "get rid of things, empty them of meaning, throw out, re-use and re-craft them" (Halkier et al., 2017, p. 2).

One advantage of this definition is the compatibility of the various consumption moments that exist before — acquisition, appropriation, appreciation, devaluation, divestment and disposal — with the more established 'food journey' concept. The latter encompasses all consumer-related activities with regards to food and nutrition across the lifespan of food, including planning, purchase, transport, storage, preparation, eating and disposal. In each of these phases of the food journey, consumers engage in instances of:

- acquisition when planning and shopping in market-based value chains, either faceto-face or online, but also when acquiring food through informal food provisioning chains and donations
- appropriation when transporting, storing and preparing the food with recourse to technologies, kitchen equipment and tools
- appreciation when eating alone or having a meal with others while judging the quality and taste of the food
- disposal, including food waste

They can also be engaged in divestment and devaluation, instances that are visible in social change processes when one food practice is substituted by another. For example, consumers divest from preparing meals with high environmental impact ingredients and empty them of meaning (devaluing) and attach new meanings (and motivations) to eating more climate-friendly meals (revaluing).

Another advantage of this definition of food consumer behaviour is that it enables the inclusion of both deliberative and semi-automatic processes, allowing recognition of the importance of the unreflective, emotive and habitual processes in food consumer behaviour. This adds nuance to the overestimated conscious deliberation and decision-making processes associated with the sovereign consumer that is often portrayed as "meaningfully *choosing*" what to eat (Warde, 2016, emphasis in original). This opens the opportunity to work with systematic and integrative approaches to behaviour change that are interdisciplinary and integrate evidence from distinct but complementary disciplines that address food consumer behaviour and change, such as psychology, sociology, history, economics, marketing, political science, agricultural and food science (Graça et al., 2019).

One potential integrative framework that was developed within the field of psychology but achieved out of disciplinary compromises is the COM-B model,²² further detailed in Chapter 3, p. 68. The COM-B model has shown promise across contexts and domains, and we draw on this comprehensive model of behaviour change to clearly frame the barriers to healthy and sustainable food consumption. The model assumes that, to trigger behaviour change (Graça et al., 2019, p. 20):

- consumers need to be capable of performing the behaviour, which includes psychological and physical/dexterity features
- the context must facilitate and support the behaviour, providing **opportunity**, which includes both social and physical features, and can be considered the external domain of the food environment
- consumers have to be **motivated** to perform the behaviour, which includes both reflective and automatic psychological features

Thus, in this model, behaviour change is always the outcome of the relations and links between these three elements: capability (skills, competences, body and mental abilities), opportunity (material and social contexts), and motivation (for example, automatic and reflective conduct).

²² It is important to note that this model still positions individual behaviour as the unit of analysis and the target of policy intervention, which is incompatible with alternative sociological frameworks that offer a critique to methodological individualism, such as practice theory. Another incompatibility is in the component of 'motivation', which in the COM-B model resides in the mind (or brain) of individuals. The brain, according to cognitive science, is composed of two systems that generate behaviour: system 1 that is more automatic and emotion-driven, and system 2 more reflective and deliberative (Warde, 2016). In a sociological approach informed by practice theory, the individual and the mind (cognitive processes) are decentred and it is practice that is the unit of analysis and of policy intervention. In this vein, it is through consumers' engagement in daily practices that motivation emerges and is configured. In other words, the pleasure and motivation of doing something derives from the very act of doing that something. Similar to learning by doing (or by practising), people are also motivated by and because of engaging in social practices. Still, the COM-B model has the merits for bringing to the fore the importance of capability, opportunity and the automatic and non-reflective features of behaviour, which marks a significant advance compared to previous orthodox consumer behaviour models that overestimated deliberate and voluntary decision-making processes.

1.6. Key messages

- The food systems approach focuses on food system functions (food and nutrition security) and identifies actors and activities that contribute to fulfilling them. By focusing on dynamics and feedbacks, it aims to understand the impacts that external drivers of change have on food systems and in turn, the impacts that the food system has in terms of socioeconomic and environmental outcomes.
- The multiple perspectives on food system activities and outcomes, along with differences in power across levels and scales, also means that it is very difficult to agree on solutions to food system problems. This is at the heart of the trade-offs among food and nutrition security, social, economic and environmental objectives and values.
- It is important to acknowledge the diversity of food systems in Europe and to account for both formal and informal food provisioning systems, not only market but crucially non-market based relations of exchange (for example, gift economy, donations and food redistribution). They not only continue to have importance in Eastern and Southern European countries as legacies of their recent rural histories, but are also gaining traction everywhere in Europe as places of social innovation and experimentation to mitigate the effects of economic, climate, energy, food and sanitary crises. Many Europeans in their daily lives navigate and move seamlessly across various combinations of these systems.
- The food environment is the context in which food is accessed and eaten and entails an external domain (i.e., physical availability, the infrastructural environment, the price of food, the information environment and labelling, the social environment) and the individual domain (i.e., affordability, accessibility, convenience and desirability) which relates to the conditions for individual daily routines and practices.
- Public and private policies contribute to directly shaping the food environment's external dimensions, indirectly impacting on the personal ones in different ways in diverse contexts.
- Digitalisation has become a key driver of change of food environments. Increasingly, digital infrastructures and technologies mediate how people seek, share and interpret food and eating-related information and practices. The effects of the digital food environment on (un)sustainable and (un)healthy food consumption are mixed and interrelated, while there is a risk of "technological determinism" in the debate.

- There is increasing evidence that consumer behaviour in everyday life is less deliberative and reflective and more automatic, emotion-driven and the outcome of habituation. Consumers tend to eat in a partially-distracted way (or semiautomatic way) and these features should be increasingly taken into account in sustainable food consumption policy.
- Consumer behaviour models should draw on systemic, integrative and interdisciplinary knowledge and include the relations between social, material and digital contexts; cultural conventions, social norms and values; meanings, beliefs and motivations; mental and physical/body features; emotions and feelings; knowhow, skills and competences.

Chapter 2. Nutritional and environmental food system outcomes

The research fields of food and health, food systems, and sustainability are largely multidisciplinary areas that bridge natural sciences, social sciences and several other disciplines. Data sources between these fields are not always identical and cannot always be harmonised, which brings challenges in interoperability and traceability, as well as scientific concepts being used differently across these research fields.

In this chapter, we will define core concepts that relate to the nutritional and environmental outcomes of food systems. These definitions will serve as a background for the subsequent chapters, on topics including dietary patterns, food consumption behaviours and sustainability dimensions. We will also introduce, define and set out boundaries for the main challenges and conflicts between health and sustainability.

2.1. Dietary patterns

Definition

A dietary pattern comprises the combination of foods that constitute the usual dietary intake over time, i.e., what is actually eaten. A dietary pattern is defined as the quantities, proportions, variety, or combination of different foods, and the frequency with which they are habitually consumed.²³ Instead of isolated nutrients, food components, foods or food groups, the diet is considered as a whole. The nutritional quality of a dietary pattern can be assessed by comparing its nutrient contents with dietary reference values. However, the focus of nutritional sciences has gradually shifted from the health effects of single nutrients to foods and dietary patterns.

There are several reasons for this shift. Since each food contains numerous compounds with complex interactions, it is not feasible to isolate and examine their separate effects on disease in a traditional reductionistic manner. Food components may have additive, synergistic or antagonistic effects. The food matrix, the physical micromolecular or macromolecular structure and domain, may also affect the bioavailability and bioactivity of nutrients and other food components. Thus, nutrients may allow us to make potential predictions about health effects in a few cases. However, establishing direct relationships

²³ https://www.dietaryguidelines.gov/

between single nutrients to specific outcomes is challenging, especially in view of individual and divergent physiological conditions.

Many health effects of foods cannot be traced back to the health effects of specific nutrients and their mode of administration. Thus, nutritional sciences do not only focus on the role of food as a source of nutrients eliminating deficiencies, but also on the role of food in the prevention of chronic diseases. Furthermore, predicting the role of individual foods is challenging since an increase in the consumption of one type of food will most often lead to a decrease in the consumption of others. The increased unravelling of the complexity and interactions of dietary intake with health outcomes has contributed to the recent recognition that studies on dietary patterns as a holistic approach are necessary alongside those on individual foods and individual nutrients.²⁴

Dietary pattern diversity in Europe

There are considerable differences in dietary patterns and food consumption among European countries (Ranta, 2014; Slimani et al., 2002; Woolhead et al., 2015). It should be noted that generalisations of dietary patterns in regions and countries should be done with care. However, generally speaking, the highest fruit, vegetable and pulses consumption is in Mediterranean countries, followed by northern central European countries, and lowest in eastern European countries. Mediterranean countries also generally have a lower consumption of processed meats than countries in central and northern Europe (Linseisen et al., 2002). In general, the Mediterranean countries share a dietary pattern characterised by more plant foods (apart from potatoes) and lower consumption of processed foods. The consumption of beverages (alcoholic and nonalcoholic) is generally higher among central and northern Europe are more diverse than the eastern European countries.

A north/south gradient in European dietary patterns can also be observed. It reflects climatic, agricultural, and economic disparities between the respective populations. In most instances, the variation in eating practices is a response to varying climatic circumstances. Many traditional foods have been consumed regionally or locally for a significant period of time, in some cases dating back centuries, and are fundamental to nutritional diversity, demonstrating key elements of the differences in dietary patterns. People evolved agricultural, processing, and preservation techniques in order to be self-sufficient. As time progressed and communities evolved, nutritional preferences, especially traditional foods, became a part of their collective identity, as revealed by

²⁴ In nutritional sciences, dietary patterns can in principle be defined in two ways: (1) an investigatordriven manner where the dietary patterns is developed based on *a priori* knowledge (such as Mediterranean, Western, prudent, Nordic, and plant-based dietary patterns) (Suri et al., 2020; Schulz et al., 2021), or (2) a data-driven manner which is developed by statistical methods such as principal component analysis or cluster analysis (Zhao et al., 2021).

Béhar (1976). Traditional foods represent cultural inheritance and have left their mark on different dietary patterns, even though current lifestyles discourage their preservation in our everyday lives and traditions (Trichopoulou et al., 2007).

The Comprehensive Food Consumption Database by EFSA is a source of information on food consumption in many, but not all, European countries.²⁵ EU member states, however, use different methods to collect food consumption data, which makes it difficult to carry out EU-wide analyses or country-to-country comparisons. EFSA's EU Menu project aims to provide standardised information on what people eat in all countries and regions across the EU.²⁶ This new project may improve the comparability and usability of the Comprehensive Food Consumption Database.

Dietary patterns and food consumption do not only vary considerably from country to country and region to region in Europe, but also over time. For example, the traditional Mediterranean diet, which is the dietary pattern prevailing among the people of the olive tree-growing areas of the Mediterranean basin before the mid-1960s, is quite different from the diet in present-day Mediterranean countries (Trichopoulou et al., 2014). There may also be a more acute change during a national or international crisis such as the supply shock provoked by the blockade of Ukrainian exports, coupled with record price levels for energy and basic commodities, which led to unpredictability in the global food supply.

2.2. Healthy diets

Updated national food-based dietary guidelines (FBDGs) are probably the best source of information for a healthy dietary pattern. Dietary patterns defined in an investigator-driven manner, where the dietary pattern is developed based on *a priori* knowledge, are given priority in most recent national FBDGs. Thirty-three countries in Europe have established national FBDGs, and in some cases regional guidelines such as the Nordic Nutrition Recommendations (2012; see also Herforth et al., 2019) guide their populations towards a healthier diet.

There is a wide consensus in the recommended dietary patterns across countries to predominantly eat a plant-based diet, rich in vegetables, fruits, whole grains, pulses and fish,²⁷ with moderate amounts of low-fat dairy products, and limited amounts of red

²⁵ https://www.efsa.europa.eu/en/data-report/food-consumption-data

²⁶ https://www.efsa.europa.eu/en/data-report/food-consumption-data

²⁷ The Norwegian Scientific Committee for Food and Environment has come to the general conclusion that the benefits of fish and seafood consumption by far outweigh food safety risks. Seafood is generally regarded as nutritious and the benefits to human health are related to many physiological functions. Seafood, with a few exceptions, compares favourably to other protein sources. In seafoods, lean proteins are combined with healthy long chain polyunsaturated fatty acids of the

and processed meat, salt, added sugar and high fat animal products. Some guidelines also include guidance on increased consumption of nuts. The Dietary Guidelines for Americans²⁸ and the World Health Organization²⁹ have also reached a similar conclusion on healthy dietary patterns. With small variations, these dietary guidelines are similar to the traditional Mediterranean dietary pattern with its focus on olives and olive oil (Obeid et al., 2022; Trichopoulou et al., 2014), the Nordic dietary pattern's focus on berries (Akesson et al., 2013) and the Dietary Approaches to Stop Hypertension diet.³⁰ In addition to a holistic approach to the role of diet for chronic diseases, these dietary patterns also take into account the role of single nutrients and foods for deficiencies and chronic disease, such as the role of sugar-sweetened beverages for obesity or added sodium for hypertension.

Dietary patterns defined in a data-driven manner (developed by statistical methods such as principal component analysis or cluster analysis) by the Global Burden of Disease collaboration³¹ have recently been used as an additional tool to inform the formulation of national FBDGs.³² This collaboration uses data from 195 countries and territories, as well as subnational data from many countries, to model diet and other lifestyle exposures with health outcome associations. The dietary risk factors studied include low intake of fruits, vegetables, legumes, whole grains, nuts and seeds, milk, seafood, n-3 fatty acids, n-6 polyunsaturated fatty acids, calcium and fibre, as well as high intake of red meat, processed meat, sugar-sweetened beverages, trans fatty acids, and sodium. While the collaboration does not consider all dietary exposures, the majority of its findings strongly support most recent national FBDGs.^{33,34,35}

A main challenge in Europe and elsewhere is the rise in overweight and obesity over the last few decades, and the increased risk caused by obesity for many chronic diseases. The WHO European Regional Obesity Report 2022 estimates that overweight and obesity affect almost 60% of adults and about 30% of children in European countries (WHO Regional Office for Europe, 2022d). Overweight and obesity is now the fourth most common risk factor for chronic diseases in the region, after high blood pressure, dietary risks and tobacco. It is also the leading risk factor for disability.

31 https://www.healthdata.org/gbd/2019

omega-3 class (marine n-3 LCPUFAs) and highly bioavailable micronutrients. Seafood is a vital part of a balanced diet and contributes important nutrients (VKM, 2022).

²⁸ https://health.gov/our-work/nutrition-physical-activity/dietary-guidelines

²⁹ https://www.euro.who.int/__data/assets/pdf_file/0003/294474/European-Food-Nutrition-Action-Plan-20152020-en.pdf

³⁰ https://www.nhlbi.nih.gov/education/dash-eating-plan

³² https://www.helsedirektoratet.no/english/nordic-nutrition-recommendations-2022

³³ https://health.gov/our-work/nutrition-physical-activity/dietary-guidelines

^{34 &}lt;u>https://www.euro.who.int/__data/assets/pdf_file/0003/294474/European-Food-Nutrition-Action-</u> <u>Plan-20152020-en.pdf</u>

³⁵ https://www.helsedirektoratet.no/english/nordic-nutrition-recommendations-2022

A recent study documented the rise in dietary supplements (vitamins, food supplements) in Europe, a rise that was said to be accelerated by the COVID pandemic.³⁶ According to Ronis et al. (2018), there is little evidence of any health benefits for well-nourished adults who use these supplements.

2.3. Environmentally sustainable diets

Current food systems are major drivers of environmental impacts (Willett et al., 2019). In terms of climate change, the global food system contributes approximately one third of global anthropogenic greenhouse gas emissions. In industrialised regions like the EU, approximately half of emissions associated with the food system come from agriculture and associated land use, including land use change (i.e., the conversion of other land, such as forest, to agricultural land). The other half come from energy-related activities downstream, including transports, processing and packing (Crippa et al., 2021). The ratio of food-related emissions to total greenhouse gas emissions is smaller in industrialised countries due to overall higher total energy use. In the EU, approximately 14% of emissions associated with final consumption come from food, beverages and tobacco (Beylot et al., 2019). The EU food sector also uses considerable amounts of energy (17% of EU's gross energy consumption in 2013) of which agriculture uses one-third and industrial processing uses 28% (European Commission et al., 2015). Fossil fuels dominate as the energy source in the food chain, and continued investments are needed in increasing energy efficiency and the use of renewable energy in food processing and transport.

Greenhouse gases associated with EU diets are emitted domestically and abroad, depending on where the food is produced. Approximately half of food-related emissions in member states are emitted in the country where the food is consumed, and half are emitted outside the country, predominantly in Asia, followed by other EU countries (Sandström et al., 2018).

A substantial part of trade-related emissions come from tropical deforestation. Approximately 15% of the total climate impact from food consumption in EU member states can be attributed to deforestation (Pendrill et al., 2019). Tropical deforestation driven by globally-traded agricultural commodities (for example, beef, soy and palm oil) also causes detrimental effects on biodiversity (Chaudhary & Brooks, 2019). Food has also been shown to be a major contributor to acidification, eutrophication (the addition of nutrients to water in lakes and rivers, which encourages plant growth that can take oxygen from the water and kill fish and other animals), land and water use in the EU. Food, beverages and tobacco contribute between them approximately 30–60% of the total such impacts from consumption in the EU (Beylot et al., 2019).

³⁶ https://www.ipsos.com/en/nutrition-pill-europeans-attitudes-towards-food-supplements

While the energy and transport sectors are the main contributors to climate change, agriculture is the main driver of several other environmental challenges including biodiversity loss, eutrophication and water stress (Gordon et al., 2017). Therefore, when considering the environmental impact of the food system, much attention is put on the agricultural phase in the supply chain of foods.

Food production has diverse and substantial impacts on local ecosystems and global earth system processes. Mitigation of environmental impacts caused by the food system is needed on both the supply side and the demand side. Supply-side mitigation options include abolishing fossil energy use along the whole supply chain; careful and efficient manure and nitrogen management; agricultural production systems with well-planned crop rotations that minimise the need for external inputs such as synthetic fertilisers and pesticides; and catering for biodiversity-enhancing landscapes. Incentivising such measures requires policy measures on the production side, such as regulations on the use of fertilisers and pesticides or payments to farmers for ecosystem services within agricultural policies. To some extent, production-side improvements can also be incentivised by food choice, such as through demand for organic products or those produced under other environmental certification schemes, but the impact from such measures is limited and does not reduce the need for production-side policy.

Impacts are highly context-specific and variable. However, based on the current evidence, it is possible to conclude that high-impact food behaviours include:

- wasteful food consumption behaviours, both food waste and metabolic waste (overconsumption)
- dietary patterns that are high in:
 - animal products
 - fish and seafood sourced from unsustainably managed stocks
 - products from tropical areas such as palm or coconut oil, tropical fruits, coffee, tea, cocoa
 - meat from tropical areas
 - water-demanding crops such as nuts and fruits, and crops from water-scarce areas
 - foods from monoculture cropping systems with high pesticide use

While animal products dominate the climate impacts of the food system, a nonnegligible and increasing share of impacts is associated with the consumption of foods low in nutritional value such as coffee, alcoholic beverages, sugar-sweetened beverages and confectionery. A study from Sweden showed that these products contributed approximately 20% of the climate impact of the average Swedish diet (Moberg et al., 2021). Coffee, tea, and cocoa also contributed substantially to the biodiversity impact in this study, together with tropical fruits and vegetable oils, due to these being grown in areas high in biodiversity.

Animal-sourced versus plant-based foods

When comparing different food products, animal-sourced foods have substantially higher environmental impacts compared to plant-based foods, especially in terms of climate change, both in total and per kilogram of food. Globally, production of animal-sourced food causes 57% of greenhouse gas emissions from food production (Xu et al., 2021), while supplying only 17% of global calories and 33% of global protein (FAO, 2022b). In the EU, livestock production occupies 28% of the land surface corresponding to 65% of agricultural land. Further, the livestock sector contributes 78% to terrestrial biodiversity loss, 80% of soil acidification and air pollution, 81% of climate impact, and 73% of water pollution out of agricultural environmental impacts in the EU (Leip et al., 2015).

In terms of emissions from the EU food supply, meat and egg consumption represents an average of 56% (ranging from 49%–64% across member states), while consumption of dairy products represents an average of 27% (ranging from 16%–36%). Consumption of grains accounts for 4% of EU food supply-related greenhouse gas emissions (Sandström et al., 2018). Protein from beef, pork, fish, chicken and egg respectively cause approximately 20 times higher, 4 times higher, 3 times higher, 3 times, and 2 times higher greenhouse gas emissions per kilogram than plant-based protein sources like cereals and legumes (Poore & Nemecek, 2018).

The reasons why animal products cause substantially higher greenhouse gas emissions than plant-based foods can be summarised in three main points. First, ruminant animals (for example, cattle, sheep, goats) emit methane in the digestion of feed and methane is a potent greenhouse gas. Second, the storage, handling, and spreading of manure causes emissions of methane and nitrous oxide. Third, animals eat substantial amounts of feed.

Globally, producing 1 kg of protein from animal products requires 80 kg of feed (Mottet et al., 2017). Even though a substantial part of this feed is non-digestible for humans, most importantly grass from grassland, on a global level, 3.1 kg of human edible crops is used to produce 1 kg of meat (Mottet et al., 2017). Cultivation of crops causes greenhouse gas emissions from:

- land use, such as nitrous oxide from fertilisers and carbon dioxide from carbon-rich soils
- energy, such as diesel in field machinery and the production of mineral fertilisers and other inputs
- in some cases, **land-use change**, such as deforestation

Higher inputs of feed per kilogram of meat produced leads to higher emissions per kilogram of food product. Because of high feed use and methane emissions from feed digestion, meat from ruminant animals causes substantially higher emissions than meat from pork and poultry (Poore & Nemecek, 2018). However, ruminant animals are not dependent on human-edible crops like cereals and grain legumes, as monogastric animals like pigs and poultry are. Therefore, ruminants do not cause feed/food competition if raised on grassland and other roughages. Ruminants might also be favoured over monogastric animals to minimise the cropland needed to feed a growing population (van Selm et al., 2022) or to decrease imports of soybeans for feed into the EU (Karlsson et al., 2021). In addition, there might be further trade-offs between other environmental impacts. For example, Nordborg et al. (2017), found higher ecotoxicity impacts per kilogram of meat for pork and poultry than for beef, due to high use of soy in the pork and poultry systems studied. Further, in some areas, grazing animals are important for biodiversity conservation in traditional grasslands (Rodríguez-Ortega et al., 2014). In addition, as animal welfare differs significantly across species and production systems there may also be trade-offs between reduced environmental impacts and impacts on animal welfare. For example, poultry meat causes considerably lower emissions than ruminant meat, but typical poultry systems are associated with considerable animal welfare challenges (De Jong & Guémené, 2011; Hartcher & Lum, 2020).

Despite the positive role that livestock can play for ecosystems in some landscapes, there is a broad consensus that limiting the consumption of meat and dairy, especially in affluent settings where consumption is high, is a crucial strategy to mitigate climate change, together with the reduction of food waste and supply side improvements (Bryngelsson et al., 2016; IPCC, 2022; UNEP, 2022; Willett et al., 2019). It also combats further biodiversity loss by reducing the demand for land (Henry et al., 2019; IPBES, 2019). Emissions of greenhouse gases and land use can be reduced by 50–70% from reductions of animal products in diets, the size of reduction depending mainly on the extent to which animal products are removed from diets (Hallström et al., 2015; Röös et al., 2017; Röös et al., 2022). If the land made available by decreasing livestock production is used for vegetation regrowth (such as afforestation), there is a large potential to sequester carbon from the atmosphere and mitigate climate change (Röös et al., 2017; Röös et al., 2022; Smith et al., 2013).³⁷

³⁷ Within the food group, variability of environmental impacts is also large, especially for beef, reflecting different production systems, production intensities and regional climate and soil conditions. For example, the greenhouse gas emissions from global bone-free dairy beef vary from 40 kg–210 kg CO₂e per kilogram (10th–90th%ile), and for beef from pure beef herds the corresponding values are 18 kg–51 kg CO₂e. For crops, variability in impacts are also considerable: for wheat, maize, and rice, 90th-percentile impacts are more than three times greater than 10th-percentile impacts for greenhouse gas emissions, land use, eutrophication, acidification and freshwater withdrawals (Poore & Nemecek, 2018). Within food groups, variability in a specific region with the same climate and soil conditions indicate a potential to reduce some impacts through increased efficiency and best available technology and management. However, reduction of certain

Nutritional and environmental food system outcomes

Emissions of greenhouse gas from blue foods are also highly variable, with farmed seaweeds and bivalves causing the lowest emissions, while flatfish and crustacean fisheries produce the highest (Gephart et al., 2021). Emissions from aquaculture arise mainly from the production of feed, while fuel use in fishing vessels dominates emissions in capture fisheries. Other environmental stressors from aquaculture and capture fisheries are genetic pollution, the introduction of invasive species, antibiotic use and the spread of disease (Gephart et al., 2021). In addition, overfishing and physical disturbance, as is caused by bottom-trawling in capture fisheries, are major drivers of marine biodiversity loss (IPBES, 2019). Capture fisheries also commonly change the size structure and abundance of captured species (Svedäng & Hornborg, 2014) and cause bycatch that negatively impact non-targeted species (Pérez Roda et al., 2019).

Organic food

The EU Farm-to-Fork Strategy has an explicit goal to increase the agricultural land under organic farming to 25% by 2030 (European Commission, 2020).

When organic foods are compared to conventional foods (per kilogram) using the most commonly applied environmental indicators, they are shown to cause a similar climate impact to conventional systems, but require more land and cause more eutrophication (Clark & Tilman, 2017). However, the benefits of organic farming are not well captured by these indicators (Meier et al., 2015; van der Werf et al., 2020), because a key benefit of organic farming is the avoidance of the use of synthetic pesticides. The use of pesticides is a major driver of damage to biodiversity (IPBES, 2019; Sánchez-Bayo & Wyckhuys, 2019). Despite ambitious policies for the reduction of contamination of surface and ground water in many EU member states, pesticides and veterinary drugs are still commonly found in EU waterways (Casado et al., 2019).

In addition, the comparison of organic and conventional production systems (per kilogram of food) using life-cycle assessment does not capture indirect effects, which are very difficult to assess (van der Werf et al., 2020). For example, the assumption that the additional land used by organic farming could otherwise be used for natural revegetation or the prevention of further deforestation has burdened organic products with the emissions caused by the missed 'carbon opportunity', making organic products considerably worse from a climate perspective (Searchinger et al., 2018; Smith et al., 2019). While in theory land 'spared' by more intensive forms of farming could be used for reforestation and hence carbon sequestration and climate mitigation, there is no such direct causal link. Land use dynamics are difficult to capture, and there are examples of

impacts (such as greenhouse gas emissions) per kilogram of produce through e.g. yield increases can involve trade-offs with other environmental impacts and other sustainability dimensions like animal welfare (Röös et al., 2018).

intensification leading to higher profits and therefore further land expansion rather than land sparing (Barretto et al., 2013; van der Werf et al., 2020).

Organic production has some undisputable benefits, most importantly the avoidance of synthetic pesticides but also, in some cases, superior soil fertility and social impacts (Reganold & Wachter, 2016; Seufert & Ramankutty, 2017). However, to supply the same amounts of food, more land is needed compared to more intensive forms of agriculture. If total demand for food and feed can be managed through dietary change (fewer animal products) and reductions in food waste, there are possibilities to reap the benefits of organic farming while avoiding further agricultural land expansion.

Local foods

There is a growing consumer interest in local foods, which are perceived by consumers as more environmentally sustainable and healthier (Hasselbach & Roosen, 2015; Hempel & Hamm, 2016; Wägeli et al., 2016). However, it has been shown repeatedly that savings of greenhouse gas emissions from choosing local foods are minor, due to the relatively small contribution from transport to overall emissions from the food system. For example, from EU diets, transport caused only 6% of overall emissions (Sandström et al., 2018). For substantial reductions in greenhouse gas emissions, reduced intake of animal products is vastly more effective.

However, as rapid emission reductions are urgently needed to mitigate climate change, reduction of emissions in all supply chain stages are likely needed. Global emissions from refrigeration supply chains are expected to rise (Crippa et al., 2021) which points to a growing importance of considering transport and storage emissions. In addition, when including additional indicators, such as indicators capturing social aspects, into the assessment, the benefits of local versus more global products are less clear. For example, a study performing a multicriteria ranking exercise between products of the same kind (cheese, ham, bread and wine respectively) along a gradient of localness found that products from more globally oriented supply chains generally were ranked last (Schmitt et al., 2017). More local products were favoured over more global products for aspects such as nutrition, biodiversity, information and communication, creation and distribution of added value, territoriality, resilience, animal welfare and governance.

For these reasons, comparing sustainability outcomes of local versus global foods is heavily influenced by the indicators used and how 'local' and 'global' are defined. Brunori et al. (2016) caution about comparing 'local' and 'global' supply chains, as the way that different chains are classified is arbitrary. In addition, many sustainability issues relevant to this comparison are difficult to quantify.

BOX 1. DATA SOURCES FOR FOOD AVAILABILITY, FOOD CONSUMPTION AND ENVIRONMENTAL IMPACTS OF FOOD

In general, country-specific environmental footprints of foods, food groups and food systems are calculated by adding up the environmental impact of all resources and processes used for country-specific food availability or food consumption. The environmental impact of resources and processes is then calculated using a standard IPCC Tier 1 (i.e. bottom-up or life-cycle assessment) methodology. As with all models that analyse aggregate level data and impacts, the estimates provide an indication of impacts and a relative comparison between countries, foods and food groups. They do not calculate the impacts of food production at the granular, detailed level within countries, such as specific amounts of the numerous foods produced using a number of specific agricultural techniques.

All models for country-specific environmental footprint of foods, food groups and food systems depend on the specific data sources and handling used. In environmental analyses, datasets from the Food and Agriculture Organization (FAO) on food availability and agricultural production are often used alone, or in combination with other datasets such as surveys and meta-analyses, to develop country-specific environmental footprints. Food availability and agricultural production methods may vary considerably between countries. Every year, authorities in over 245 countries and territories submit national food and agriculture statistics, as well as micro-datasets collected through farm and household surveys, to the FAO. In the FAOSTAT database, national food supply is estimated as the average per capita food available for consumption, based on domestic production, adjusted for exports, imports, and non-food uses. National authorities, who annually supply data to FAOSTAT, are not required to use standardised methodologies when collecting, categorising and grouping data. Thus, while FAOSTAT data are important for FAO's strategic work to reduce hunger and malnutrition, the data should be used with an understanding of such limitations in country-specific environmental modelling studies.

To analyse the environmental impact of food availability, data from FAOSTAT are often paired with comprehensive country-specific databases of environmental footprints or life-cycle assessment data. To various degrees, these databases include primary production, including feed requirement, processing factors, transport, packaging, and share of processed commodities, and may also consider food loss and waste. These models build on numerous assumptions and theoretical considerations. Increasingly often, coding and data inputs used to produce such complex models are made available either via the publisher or via contact with the authorial team, but this is not standard practice. Despite their commonly held limitations with all types of complex modelling endeavours, these somewhat 'black box' models of environmental footprints are innovative and represent the state of the art.

Country-specific food availability concerns the availability of sufficient qualities and quantities of foods supplied through domestic production or imports. Food consumption or dietary pattern is defined as the foods that are actually ingested by people. Thus, the amounts of foods consumed by a national population are often much lower than the foods that are available for the country. Since composite food group categories (such as cereals) often are not identical in food availability and food consumption datasets, comparisons across datasets should be done with care. The gold standard for assessing food consumption is surveys using food records or 24-hour diet recall methodologies. When interpreting data, it is important to realise that dietary patterns and food consumption assessed by food records or 24-hour diet recall methodologies.

2.4. Integrating environmental sustainability and health

While most national food-based dietary guidelines (FBDGs) are based on the assessment of health outcomes, a small number of countries have initiated processes that integrate sustainability into national FBDGs. Recently, the FAO and WHO have developed guiding principles for developing sustainable FBDGs (FAO & WHO, 2019). They define a set of eighht principles for assessing health outcomes, five principles for assessing environmental impacts, and three principles for assessing sociocultural aspects (see Box 2). The main report, the principles and the background papers are an important step towards harmonising the process of developing such guidelines. Wood et al. (2023) also present a framework with concrete steps for developing sustainable FBDGs that integrates environmental sustainability on equal footing as health.

In a recent modelling study, Springmann et al. (2020) analysed the healthiness and sustainability of national FBDGs in 85 countries. The results suggest that adoption of national FBDGs would, on average, be associated with 15% reduction in premature mortality, as well as mixed changes in environmental resource demand, including a reduction in greenhouse gas emissions of 13% (regional range –34%–35%). They concluded that most FBDGs are incompatible with the Paris Climate Agreement, plus land use, freshwater, and nitrogen targets. This analysis suggests that national guidelines could be both healthier and more sustainable.

BOX 2. FAO/WHO GUIDING PRINCIPLES FOR SUSTAINABLE HEALTHY DIETS

Principles regarding the health aspect:

- start early in life with early initiation of breastfeeding, exclusive breastfeeding until six months of age, and continued breastfeeding until two years and beyond, combined with appropriate complementary feeding
- are based on a great variety of unprocessed or minimally processed foods, balanced across food groups, while restricting highly processed food and drink products
- include wholegrains, legumes, nuts and an abundance and variety of fruits and vegetables
- can include moderate amounts of eggs, dairy, poultry and fish, and small amounts of red meat
- include safe and clean drinking water as the fluid of choice
- are adequate (i.e. reaching but not exceeding needs) in energy and nutrients for growth and development, and to meet the needs for an active and healthy life across the lifecycle
- are consistent with WHO guidelines to reduce the risk of diet-related NCDs and ensure health and wellbeing for the general population.
- contain minimal levels, or none, if possible, of pathogens, toxins and other agents that can cause foodborne disease

Nutritional and environmental food system outcomes

Principles regarding environmental impact:

- maintain greenhouse gas emissions, water and land use, nitrogen and phosphorus application and chemical pollution within set targets.
- preserve biodiversity, including that of crops, livestock, forest-derived foods and aquatic genetic resources, and avoid overfishing and overhunting
- minimize the use of antibiotics and hormones in food production
- minimize the use of plastics and derivatives in food packaging
- reduce food loss and waste

Principles regarding sociocultural aspects:

- are built on and respect local culture, culinary practices, knowledge and consumption patterns, and values on the way food is sourced, produced and consumed
- are accessible and desirable
- avoid adverse gender-related impacts, especially with regard to time allocation (e.g. for buying and preparing food, water and fuel acquisition)

(Source: FAO & WHO, 2019)

There is a strong link between health and environmental sustainability of dietary choices (Tilman & Clark, 2014). For example, on a nutrient level, replacing animal-based foods with plant-based foods decreases premature mortality up to 12%, as well as some environmental impacts, such as greenhouse gas emissions, by up to 84%. On the other hand, it increases other environmental impacts such as freshwater use by up to 16% (Springmann, Wiebe, et al., 2018). For other dietary patterns, flexitarian, pescetarian, vegetarian or vegan diets reduce premature mortality by about 19%–22%, greenhouse gas emissions by 54%–87%, nitrogen application by 23%–25%, cropland use by 8–11% and freshwater use by 2–11% (Springmann, Wiebe, et al., 2018).

2.5. Food waste

Reducing food waste is another key mitigation option for tackling environmental impacts.³⁸ Research suggests that a 50% reduction in food waste would result in reductions in environmental impacts associated with food production in the range of 5%–10% (Bryngelsson et al., 2016; Read et al., 2020; Röös et al., 2017).³⁹ In addition,

³⁸ There is a Sustainable Development Goal target (12.3) with the aim to "halve per capita global food waste at the retail and consumer levels and reduce food losses along production and supply chains, including post-harvest losses" by 2030.

³⁹ Another type of food waste is metabolic food waste, for example, the amount of food leading to excess body fat. In addition to health problems associated with overconsumption of food, excessive intake of food can also be seen as a type of food waste causing additional emissions of greenhouse gases and other environmental impacts (Edwards & Roberts, 2009). Toti et al. (2019) assessed environmental impacts from metabolic food waste for different world regions. They found that the EU is responsible for the greatest amounts of metabolic food waste and associated environmental impacts, approximately 14 times more than sub-Saharan Africa. A study from Sweden showed that

there is a growing discrepancy between both stakeholders' and researchers' emphasis on food waste prevention (i.e. stopping food from becoming waste), in the context of the efforts to halve the volume of food waste by 2030. This is a policy top priority, including EU Directive 2008/98/EC on Waste and the waste management practices in which "food waste prevention has been subsumed into measures aimed at diverting or averting waste" (Messner et al., 2020). Some scholars criticise the reliance on consumers and households for food waste reduction as disproportionate, and as having an inconclusive effect on the effectiveness of resource use in upper segments of the food chain (Chaboud & Daviron, 2017). In the context of the search for insights regarding the capacity to prevent food from becoming waste, it is nevertheless important to study the situations where households generate significantly lower volumes of food waste than the EU average.

There are segments of European societies that already produce significantly less food waste than the EU average and it is not known why. For instance, analysing bins with municipal mixed waste from a small number of village households, Sosna et al. (2019) found that Czech rural dwellers may produce as little as 7.9 kg of edible food waste per person per year. Drawing on large-scale research in an urban environment, Kubíčková et al. (2021) discovered that Czech urban households produced 37.4 kg of edible food waste per person per year (the figure does not include food waste flushed down the toilet and food waste that ends up in organic waste bins). This is a highly reliable finding, as food waste was repeatedly and anonymously collected from 900 households in Brno, the second-largest Czech city (population 380 000), across three years. A different research project aimed at food waste was conducted in the small town of Boskovice (population 12 000, with food waste produced by 4000 inhabitants analysed) confirmed low volumes of food waste produced by Czech households. The amount of edible food waste per person per year ranged from less than 2 kg to 31 kg, depending on the type of residential area (Kormaňáková et al., 2021). For comparison, the average amount of food waste (both edible and inedible) from an EU household is estimated to be $92 \text{ kg} \pm 9 \text{ kg}$ per person per year (FUSIONS project, 2016).

Therefore, in addition to researching how to raise consumers' awareness of the meaning of 'use by' and 'best before' dates (the current preferred approach to the issue of food waste), or promoting digital solutions such as the uptake of food waste reduction mobile phone apps, a better and more comprehensive understanding is also needed of how some people in Europe already produce little or almost no food waste. In other words, in addition to understanding the barriers to wasting less food, we need to understand the enabling factors that facilitate frugality and unwasteful behaviour. The complexity of the issue of food waste requires consideration and understanding of a range of factors, including "household self-supply, food preparation at home, the financial situation of the household, sociodemographic factors, access to shops and perceiving the symbolic value of nutritives in the context of their disposal" (Kormaňáková et al., 2021).

the amount of surplus food intake was 66% larger than the total amount of avoidable household food waste per year (Sundin et al., 2021).

2.6. Key messages

- A dietary pattern is defined as the quantities, proportions, variety, or combination of different foods, and the frequency with which they are habitually consumed. Dietary patterns and food consumption vary considerably from region to region in Europe, but also over time.
- There is a broad consensus in the recommended dietary pattern across countries: to predominantly eat a plant-based diet, rich in vegetables, fruits, whole grains, pulses and fish, with moderate amounts of low-fat dairy products, and limited amounts of red and processed meat, salt, added sugar and high-fat animal products.
- National nutrient recommendations and food-based dietary guidelines are not only the main foundation of national food policies, but also affect individual food choices.
- Current food systems are major drivers of environmental impacts, especially biodiversity loss, eutrophication, water stress, land degradation and climate change.
- Dietary choices are one of the biggest drivers behind rising obesity and the prevalence of noncommunicable diseases, so shifting consumer choices towards healthier foods can help to mitigate these threats.
- Animal-sourced foods in general have substantially higher environmental impacts compared to plant-based foods, especially in terms of climate change, both per kilogram of food and in total.
- There is a broad consensus that limiting the consumption of meat and dairy, especially in affluent settings where consumption is high, is a crucial strategy to mitigate climate change, to stop biodiversity loss, halt obesity and fight chronic non-transmissible diseases.
- Ruminant animals (e.g. cattle and sheep) cause higher greenhouse gas emissions per kilogram of meat than meat from pigs and poultry. However, ruminants are not dependent on human-edible crops like cereals, grain and legumes, as pigs and poultry are.
- Organic production has benefits such as the avoidance of synthetic pesticides, but has similar climate impacts per kilogram of produce as conventional production. Organic production without demand side-changes (reduced consumption of animal products and reduced waste) is not a climate change mitigation strategy.

- Savings of greenhouse gas emissions from choosing local foods are minor due to the relatively small contribution from transport to overall emissions from the food system, but local food systems can have other (social) benefits depending on the indicators used to compare local and global supply chains.
- Reducing food waste has the potential to reduce greenhouse gas emissions by 5%-10%, but the food waste generated is highly variable.
- Food waste strategies should focus on prevention stopping food from becoming waste — rather than on diverting surplus food and food waste to redistribution channels.

Chapter 3. Consumer behaviour: Barriers to sustainable and healthy food consumption

How to encourage behaviour change among consumers is one of the central questions in the discourse about sustainable and healthy food consumption. Given the magnitude of the challenge — that is, the disparity between current consumption patterns and the target of sustainable and healthy food consumption, as discussed in Chapter 2 the answer is not simple. There is a growing body of literature on the effectiveness of interventions and tools, which we discuss later in Chapter 4, p. 83. However, before we discuss the different tools, this chapter provides a structured overview from a behavioural perspective of the different types of barriers that consumers face, in order to identify which tools are effective for different behaviours and consumer segments.

The shift to sustainable and healthy food choices is a complex challenge, especially for the individual consumer, since it requires far-reaching changes in all aspects of food consumption:

- a shift in dietary patterns: consumers need to change the composition of foods and diet types they are eating, e.g. less meat, more legumes, more fruits and vegetables, more nuts and seeds, fewer unhealthy snacks or ultra-processed foods, and so on
- a shift towards (more) sustainable production systems: e.g. from 'normal' apples to apples from organic production or other types of eco-friendly and biodiversityfriendly production systems (for a detailed discussion on the benefits and impacts of organic production, see p. 60)
- shifts in practices to reduce food waste at household level: changing food preparation, planning and shopping practices

The challenges and the barriers differ across the various impact behaviours outlined here. While the shift in dietary patterns requires consumers to give up certain foods, some of which provide large hedonic pleasure, and eat other foods instead, the shift towards food from more sustainable production systems is a matter of substitution, with the largest barriers being higher costs, lack of availability or accessibility, and lack of trust in ecolabels. Meanwhile, avoiding food waste requires changing everyday habits and practices.

The problem with the transition to sustainable and healthy food consumption is that it requires consumers to make trade-offs (Grunert, 2011) between sustainability and health on the one hand, and perceived negative outcomes on the other, such as inferior taste,

higher prices and inconvenience, contributing to the gap between consumer values, attitudes and intentions on the one hand, and their behaviour on the other hand (Vermeir et al., 2020). While more and more consumers nowadays are worried about both their personal health as well as the effect of their consumption practices on the environment, they tend not to choose foods in a consistent and systematic way, or spend their money according to these worries, a situation often referred to as a 'value-action gap', an 'attitude-behaviour gap', or an 'intention-behaviour gap'.

Moreover, it is uncontested that consumers have some agency, but not complete agency, over consumption processes when it comes to sustainable and healthy food consumption (Warde, 2005). This means that consumers are not always in control of what to choose, but are dependent on contextual factors such as the configuration of the food environment (Chapter 4, p. 83), and the food system at large (Chapter 1, p. 19). The different models for understanding consumer behaviour and identifying barriers to sustainable consumption vary in the degree to which they incorporate individual-level factors as well as contextual factors (SAPEA, 2020). Both of these factors need to be included when identifying barriers, as well as the interplay between the two.

This report uses the so-called COM-B framework for providing a comprehensive and structured overview of barriers to sustainable and healthy food consumption (introduced in the following section), and complements this approach with insights from practice theory.

3.1. Theoretical framework to identify barriers in sustainable and healthy food consumption

We draw on the COM-B framework to provide a structured overview of barriers that consumers face when it comes to sustainable and healthy food consumption, while simultaneously embracing the manifold approaches to understanding consumer behaviour (see Cane et al., 2012; Carey et al., 2018; Michie et al., 2011).

Here we explain the COM-B framework in more detail. According to this framework, behaviour (B) is generated by the interaction of motivation (M), capability (C) and opportunity (O). While personal motivation is an important driver to perform a behaviour, it is often not sufficient. Consumers additionally need the capabilities and resources to perform a behaviour, as well as the opportunities that make a behaviour possible or prompt it.

Motivation is defined as "all those brain processes that energise and direct behaviour" (Michie et al., 2011). Motivation is important when the behaviour in question has a personal relevance, as it determines the evaluation of the expected outcome of a behaviour, independently of whether the decision process is automatic or reflective. When behaviour is based on reflective decision-making processes, motivational factors such as attitudes, preferences, beliefs about capabilities or consequences, personal norms or goal-setting play a role. But most daily food-related decisions are made automatically, driven by motivational factors such as emotions or impulses. Some motivational factors. such as identity, values, likes or food-related attributes (for example, taste, texture and appearance) could direct both reflective and automatic decision-making, and anything in between.

- Capability is defined as the individual psychological and physical resources required to perform a behaviour. This includes resources such as physical capacities, cognitive capacities, nutrition-related knowledge, skills or habits as well as resources such as time, money or self-regulation capabilities.
- Opportunity is defined by the physical and social environments surrounding consumers that can either support or inhibit sustainable food consumption. For example, consumers can only acquire sustainable foods that are available and accessible in the physical environment. Important drivers in the decision-making process are, for example, convenience and salience of sustainable alternatives. As the health and sustainability consequences of behaviours are often distant and opaque, a strategy used by consumers is to refer to others (Sparkman et al., 2021). Hence, the social environment exerts a powerful influence via social and cultural norms, and those norms are often rather less healthy and sustainable.

It is important to highlight the interrelations between motivation, capability, opportunity and behaviour. For example, motivation and the underlying drivers of food preferences, beliefs, and attitudes are highly influenced by capability, opportunity, and behaviour itself (i.e., past experience). A motivation to prepare a sustainable and healthy meal and perform the behaviour can be promoted by existing knowledge about, and skills for preparing, sustainable and healthy meals (capability), but also by experienced social support or availability and easy accessibility of the required food products (opportunity). In this context, it is worth making the link to what economists refer to as 'endogenous preferences' (see 'Combinations of tools', p. 111). A motivation to consume healthy and sustainable foods can, in turn, lead consumers to self-select supportive physical and social environments. The experience of performing the desired behaviour also exerts some influence on motivation, capability and opportunity.

3.2. Barriers to healthy and sustainable food consumption

The literature on barriers to sustainable and healthy food consumption stems from different academic disciplines, such as food sciences, public health, consumer

psychology, marketing, economics and food policy, while drawing upon different theoretical frameworks, including social-cognitive theories, most prominently the theory of planned behaviour, and behaviour change frameworks such as COM-B and stages of change.

The barriers preventing consumers from engaging in sustainable and healthy food consumption are summarised in Boxes 3 to 6 (p. 72), broken down into four key impact areas:

- reducing consumption of meat and animal foods, and increasing consumption of plant-based foods
- reducing the consumption of unhealthy foods, i.e. those that are high in fat, salt and sugar, or ultra-processed
- increasing the consumption of organic foods
- reducing food waste at household level

These four key impact behaviours were chosen because of their impact on sustainable and healthy diets, and the (relatively) high agency consumers have in these areas. Other relevant types of food choices with high impacts on climate, biodiversity, land use, water, etc. are summarised in Box 7, p. 77.

The barriers are organised along the main categories of the COM-B framework. The COM-B framework serves as a means to structure the numerous and interrelated barriers into thematic groups; we are not proposing that this framework is the best model for predicting behaviour. Rather, an overview of existing major barriers towards healthier and sustainable food consumption is given without doing justice to the complexity and the interplay of those barriers within and across categories.

The following premises are important for interpreting the schematic boxes:

- Reciprocity of consumer internal and external factors: Internal drivers of consumer behaviour ('motivation' and 'capability') are strongly interconnected with the social and physical environments ('opportunity') in a reciprocal way.
- Food consumption is essentially a set of practices strongly embedded in social and cultural norms, not only fulfilling functional needs but also providing symbolic meaning. The shift to sustainable food consumption requires the individual consumer to adopt (partly) new practices and abandon existing ones.

The boxes provide an overview of common barriers in European countries. There are slight variations across the countries in terms of which barriers are most prominent, but largely speaking there are more commonalities than differences across the countries, which is why we do not go into detail about country-specific exceptions. Rather, it is important to emphasise that there are large variations across consumer segments within each country regarding the barriers that hinder the individual consumer from behaving more sustainably, and the extent to which consumers are already consuming sustainable and healthy food.

The barriers outlined here can be translated into enablers: removing a barrier can facilitate the transition to sustainable and healthy food consumption. The barriers and enablers represent potential entry points for interventions and policies aimed at fostering sustainable and healthy food consumption.

BOX 3. BARRIERS TO REDUCING THE CONSUMPTION OF MEAT AND ANIMAL FOODS AND INCREASING PLANT-BASED FOODS CONSUMPTION

Motivation:

- Liking and preferences: strong liking of meat and animal products (taste, texture, ...), disliking of plant-based foods (influence of social and cultural norms)
- Attitudes: positive attitudes towards meat and animal products, negative attitudes towards plant-based foods (influence of social and cultural norms and conventions)
- Perceived benefits & risks: for example, belief that (large amounts of) meat and animal foods are necessary for a healthy diet
- Perceived high prices of meat substitutes and vegetables (compared to animal products)
- Scepticism and prejudice towards the terms/labels 'vegetarian' and 'vegan'
- Lack of environmental or climate concern
- Lack of perceived responsibility or obligation
- Lack of perceived behavioural control
- Emotions and affect: positive emotions linked to meat and animal products, negative emotions linked to plant-based foods
- Negative expectations about outcomes (for example, costly, little self-efficacy, no environmental effect, unhealthy)

Capability:

- Food familiarity:
 - Lack of familiarity with novel foods
 - Prior experiences/aversion/liking
 - Food neophobia
- Lack of skills (planning, acquisition, storing, preparing, cooking e.g. meat-free recipes)
- Lack of nutrition-related knowledge (planning, acquisition, storing, preparing, cooking)
- Physiological factors such as hunger, satiety or appetite linked to meat and animal products
- Lack of knowledge about environmental, climate and health impacts of meat consumption
- Self-regulation (lack of self-control) personality trait: lack of conscientiousness leads to impulsive eating and loss of self-control
- Difficulty or inability to change habits and routines
- Lack of time (throughout all consumption phases, for example, to try new meat-free recipes)
Opportunity:

Physical environment:

- Perceived low availability and low salience/visibility of plant-based alternatives in supermarkets, canteens, restaurants etc.
- Perceived inconvenience of plant-based alternatives
- Prominent positioning and omnipresence of meat and animal products in supermarkets, canteens, restaurants etc.
- Prices (cheap meat products possible through subsidies and industrial factory farming?)
- Visual cues, smells or other sensory stimuli (affect perception of portion size, intake or satiation)
- Large portion sizes of meat and animal products by default
- Difficult accessibility of plant-based foods (e.g., unattractive labelling and naming/framing, e.g. as 'vegan' or 'vegetarian')
- Obesogenic/unsustainable environments (e.g., neighbourhoods with higher density of fast-food outlets and lower density of supermarkets and grocery stores - so-called food-deserts)

Social environment:

- Lack of support for eating plant-based eating foods from close others (family, friends, colleagues)
- Social norms ("meat = default")
- Food culture and tradition: Meat is traditionally part of the food culture in European countries
- Social prejudice
- Social identity and lifestyles determined by food consumption
- Symbolic meaning of meat (e.g. red meat symbolises human power)

Sources: Appleton et al., 2016; Bauer & Reisch, 2019; Benelam, 2009; Bucher et al., 2016; Chen & Antonelli, 2020; Evers et al., 2018; Graça et al., 2019; Munt et al., 2017; Nguyen et al., 2022; Stautz et al., 2018; Stoll-Kleemann & Schmidt, 2017.

Box 4. Barriers to reducing the consumption of unhealthy foods (e.g. high in fat, sugar and salt or ultra-high processed foods)

Motivation:

- Liking and preferences: strong liking of sugar-sweetened beverages, sweet and savoury snacks etc. (taste, texture, etc.)
- Attitudes: positive attitudes towards foods high in fat, sugar and salt
- Negative expectations about outcomes (e.g., little self-efficacy associated with amount of sugar consumed)
- Emotions and affect (e.g., comfort eating or stress-induced eating; negative emotions associated with increased eating in restrained eaters; positive emotions associated with increased eating)
- Mindless eating (e.g., eating while watching TV associated with consumption of foods high in fat, sugar and salt)
- Impulsive food choices in favour of foods high in fat, sugar and salt
- Hot-cold empathy gap (in a hot state more rewarding food is consumed)

Consumer behaviour: Barriers

Habits or routines in favour of unhealthy food consumption, e.g., sugar-sweetened beverages

Capability:

- Nutrition-related skills (planning, acquisition, storing, preparing, cooking)
- Nutrition-related knowledge (planning, acquisition, storing, preparing, cooking)
- Low levels of health literacy are associated with high sugar intake
- Satiety from sugars or fats is lower compared to satiety from proteins or complex carbohydrates
- Physiological factors such as hunger, satiety or appetite enhance likelihood to consume unhealthy foods (and rely heavier on automated processes)
- Lack of time (throughout all consumption phases, e.g., processed food instead of cooking)
- Misjudgement of long-term costs and short-term benefits
- Skills to solve goal conflicts (e.g., eat healthy vs. indulging in food) -> goal activation
- Low level of self-regulation
- Difficulty / inability to change habits and routines

Opportunity:

- Physical environment:
 - Portion sizes (larger portion sizes lead to more consumed food)
 - Larger packages or plates lead to higher quantities consumed (Delboeuf illusion, e.g. portions are judged in relation to plate size)
 - Parallel activities increase consumption (e.g. TV viewing)
 - Misinformation or information overload
 - Availability and salience of unhealthy products: visual cues, smells or other sensory stimuli (affect perception of portion size, intake or satiation); product placement; neighbourhood (e.g., density of fast food outlets and convenience stores)

Social environment:

- Social influence or reference points, e.g. eating with others increases unhealthy food consumption
- Social norms affect food choice and quantity eaten (e.g., portion sizes, frequency of sugar consumption)
- Social and cultural norms around hospitality and receiving guests (e.g. celebratory occasions, birthday parties, Christmas, hosting dinner parties are occasions where dishes high in fat, sugar and salt are perceived as 'proper' food)

Sources: Bauer & Reisch, 2019; Benelam, 2009; Bucher et al., 2016; Buja et al., 2021; Calabro et al., 2023; Chen & Antonelli, 2020; Evers et al., 2018; Gupta et al., 2018; Mazarello Paes et al., 2015; Munt et al., 2017; Schneider et al., 2021; Stautz et al., 2018; Warde et al., 2020.

Box 5. Barriers to increasing the consumption of organic food consumption

Motivation:

- Unwillingness to pay higher prices for organic food
- Liking and preferences:
 - Other attributes are more important than organic (e.g. taste, convenience, local origin)
 - Perception that organic = less tasty, shorter shelf-life, less appealing
- Lack of trust in organic certification
- Confusion and scepticism about the benefits of organic food and organic production (e.g. health, environment)
- Lack of environmental concern
- Lack of perceived responsibility to act
- Lack of perceived behavioural control
- Positive attitudes towards non-organic products: consumers are loyal customers of non-organic products and brands, and are satisfied with these products
- Habits: purchasing non-organic food is a habit many consumers have never questioned

Capability:

- Lack of knowledge about organic production methods, standards and the control system, including lack of knowledge about the benefits of organic food (e.g. health) and organic production (e.g. environment)
- Difficulties identifying certified organic food
- Lower income or perceived lack of financial resources
- Lower level of formal education
- Lack of time: perception that buying organic food involves extra time
- Difficulty or inability to change habits and routines

Opportunity:

- Physical environment:
 - Higher prices of organic food
 - Perceived low availability and low salience or visibility of organic food in supermarkets, canteens, restaurants etc.
 - Perceived limited variety or product assortment of organic products
- Social environment:
 - Scepticism about the environmental benefits of organic production among close others (family, friends, colleagues)
 - Lack of support for paying the organic price premium from close others

Sources: Aertsens et al., 2009; Hansmann et al., 2020; Hughner et al., 2007; Kushwah et al., 2019.

BOX 6. BARRIERS TO REDUCING FOOD WASTE AT HOUSEHOLD LEVEL

Motivation:

Competing (good) motives at the moment of purchase despite increasing awareness of negative environmental consequences (e.g., the move to healthier eating patterns has been associated with a higher propensity to waste food because consumers purchase more fruit, vegetables and other categories that have a higher risk of getting wasted)

Capability:

- Food-related skills (planning, acquisition, storing, preparing, cooking, eating, managing leftovers)
- Food-related knowledge, e.g., regarding best before and use by dates, or adequate portion sizes
- Food safety concerns when it comes to the consumption of leftovers or interpretation of date labels
- Lack of planning (meal planning, shopping lists)
- Lack of time to prepare the food that has been acquired
- Lack of time can lead consumers to substitute already purchased food that demands preparation with quicker alternatives (convenience food, take-out food, restaurant meals)
- Difficulty or inability to change habits and routines (planning and purchasing routines, but also shopping frequency and cooking routines)

Opportunity:

- Physical environment:
 - Large packages or packages that are difficult to empty lead to leftovers and waste
 - Promotions and discounts, such as 'buy 1 get 2', lead to more food purchased than can be consumed
 - Other sensory stimuli, such as visual cues, smells, product placement can induce consumers to acquire more food than needed
 - Low price of food makes avoiding waste not a priority. Wasting food may be cheaper than another trip to the grocery store for a forgotten food item
 - Ubiquity of alternatives (convenience food, take-out food, restaurant meals) makes it easy to change food-related plans, with the risk of food that has been acquired, but is not prepared, going to waste
- Social environment:
 - Social norm of providing more than enough food for family members and guests
 - Social identity of being a good provider and caterer to all family member's wishes

Sources: Aschemann-Witzel et al., 2015; Brunner et al., 2010; Evans, 2011, 2014; Graham-Rowe et al., 2014; Quested et al., 2013; Schanes et al., 2018; Welch et al., 2021.

BOX 7. OTHER HIGH-IMPACT BEHAVIOURS

The four key impact behaviours highlighted here are not the only relevant behaviour changes needed to move to sustainable and healthy food consumption (see Chapter 1, p. 19). Food consumption and production have to change even further to reduce the negative impacts on climate, environment (biodiversity, land use, soil, water, etc.), and social issues.

When it comes to consumers' food choices and practices, consumers often have little to no possibility to retrieve information about the impact of their behaviour on climate, environment, and social issues — beyond the four key areas highlighted above. For climate impacts of food choices, for instance, consumers can access information about the greenhouse gas emissions of different food categories (for example, published by NGOs), neglecting variations *within* food categories. For biodiversity impacts of food choices, consumers do not even have the possibility to retrieve information, neither at product category level nor at product level. The organic label is the only 'proxy' for biodiversity impacts consumers can currently look for, neglecting variations between farmers and products. The same holds true for impacts on water, soil, and land use. The organic label is also problematic as meat can still be labelled although the environmental impact of meat is considerably higher than plant-based foods, which can be confusing. However, considering the higher price of organic meat, recommending the consumption of organic meat might still lead to reductions in impact through decreased overall consumption.

Consumers have limited agency to consume in a sustainable manner due to a lack of transparency and information about the actual impacts of individual food choices (and products) on climate, environment and social issues.

The analysis of barriers across the four high-impact behaviours reveals that barriers exist at all levels: at the individual consumer level in the form of lack of motivation and personal capabilities, and at the contextual level in the form of lack of physical and social opportunity. These barriers represent potential entry points for policies and interventions aimed at fostering sustainable and healthy food consumption.

These high-impact behaviours share a commonality, in that large barriers exist at all levels of the COM-B framework, with slight variations regarding the significance of specific barriers (see the above boxes for details). The following summary should be understood through the lens of the premises outlined in the preceding section:

Lack of motivation: Consumers lack the motivation to change their food-related behaviours. Some consumers lack concern for sustainability and personal health altogether; others do not feel responsible to act; while others would like to change their behaviour but find the sustainable and healthy option insufficiently attractive (for instance, in terms of taste, convenience or price) and they are satisfied with their current food choices. Variations between the impact behaviours exist as to what consumers dislike about the sustainable options: taste preferences are a major barrier for meat reduction and healthy eating, while high prices are a major barrier for organic food consumption. While most consumers dislike wasting food, other motives such as making sure there is enough food available seem to be dominant at

the moment when food is acquired. Apart from these cognitive and attitudinal factors, another important motivational barrier is that consumers are locked in unsustainable habits and automatic routines, as further elaborated below.

- Lack of capability: Consumers lack the capability to make sustainable and healthy food choices. The lack of cognitive capabilities includes the lack of cooking and food preparation skills, the lack of knowledge about sustainable and healthy food, expiration dates and food literacy in general, and difficulties identifying sustainably produced food. Consumers further lack mental capabilities, e.g. self-control to resist temptations, or willpower to change their habits. In terms of resources, consumers mostly lack financial resources for paying the higher price of sustainable food, and time resources to plan and prepare food.
- Lack of opportunity: Consumers lack opportunities for sustainable and healthy food consumption. The physical environment does not adequately support sustainable and healthy food consumption in terms of food availability, accessibility, affordability and visibility. Unhealthy and unsustainable options are omnipresent in supermarkets, restaurants, canteens and so on, preventing consumers from making sustainable and healthy choices. Package sizes are often too large, and discounts and promotions can steer consumers towards unhealthy food or overprovisioning.

The social environment is another type of barrier. Sustainable and healthy food consumption is not always supported by close others (family, friends, colleagues, etc.), and social norms often work against sustainable food consumption, especially in the case of reducing the consumption of meat and animal products. With regards to other high-impact behaviours (see Box 7, p. 77), consumers have limited agency to consume in a sustainable manner due to a lack of transparency and information about the actual impacts of individual and collective food choices on climate, environment, and social issues.

These barriers to sustainable and healthy food consumption further illustrate how, at the individual consumer level, the following three types of psychological processes are involved to varying degrees when consumers make food choices:

- habits, routines, and semi-automatic processes
- cognitive processes, i.e. deliberate thinking and information processing
- affective processes, i.e. emotions and impulsive reactions

Large barriers exist at all three levels, in that habits and routines, cognitive processes, and affective processes work against sustainable and healthy food consumption. This is the case for all types of high-impact behaviours investigated here.

Habits, routines and semi-automatic processes

Food consumption is largely characterised by repeated and semi-automatic processes, and for many consumers these habits are not in line with sustainable and healthy food consumption. Food habits are notoriously difficult to change (Vermeir et al., 2020). We do not yet fully understand how food habits evolve, hence breaking and establishing new habits is difficult and an under-researched area (Rees et al., 2018). It becomes even more difficult when consumers are not motivated to change their behaviour (Vermeir et al., 2020). To add to the complication, taste preferences are also linked to habits, since taste preferences are largely developed over time (Diószegi et al., 2019) and embedded in food cultures (Wright et al., 2001). This is especially challenging since a preference for the taste of meat and unhealthy snacks, and a dislike of the taste of legumes, vegetables and wholegrain foods, is one of the strongest barriers to sustainable and healthy food consumption. Overcoming the barrier of taste dislike is an under-researched area.

Cognitive barriers

A large number of cognitive processes influence food consumption (Chen & Antonelli, 2020) and prevent consumers from making sustainable and healthy choices (Biasini et al., 2021). Consumers compare and evaluate different food options and prioritise other characteristics over sustainability and health. Their decisions and evaluations might suffer from: a lack of cognitive capabilities, including the lack of knowledge about sustainable and healthy food and food literacy in general; difficulties identifying sustainably-produced food; and a lack of cooking and food preparation skills. Some consumers lack concern for sustainability and personal health altogether, while others do not feel responsible to act.

Affective barriers

Consumers experience pleasure and positive emotions when consuming certain foods, and this is especially the case for foods that are not in line with sustainable and healthy food consumption (Evers et al., 2018). Furthermore, many food choices result from impulsive reactions triggered by situational cues from the physical and social environment. Impulsive food choices also often clash with sustainable and healthy food consumption (e.g. Wouters et al., 2018).

3.3. Entry points for policy interventions

The large magnitude and variety of barriers to sustainable and healthy food consumption calls for a broad set of policy interventions and tools to be implemented. The policy mix needs to take into account the following points, simultaneously if possible:

- **Type of barrier.** What type of barrier does the intervention primarily aim to reduce? Lack of motivation, lack of capability, or lack of opportunity? Which other barriers are intertwined with the targeted one?
- Nature of the target behaviour. Which type of psychological process dominates the target behaviour at hand? Habits and routines, cognitive processes, or affective processes? Depending on which type of psychological process dominates the food-related behaviour at hand, contextual factors and the food environments play slightly different roles in affecting consumer behaviour, and different types of interventions and tools are most promising for overcoming the barriers to sustainable and healthy food consumption.
- **Target group.** Which consumer segment is the intervention trying to reach? There are large variations across consumer segments within and across countries regarding the barriers that hinder the individual consumer from behaving more sustainably and healthy. Personalised feedback tools offer promising possibilities to effectively induce behaviour change among specific target groups (see the information on personalised feedback tools, p. 109).
- Unintended side effects. Policy interventions can have unintended side effects, steering consumer demand in the opposite direction from that which is desired (rebound effects, reactance, licensing). As an example, it is important to consider that some people do not feel responsible for acting in a sustainable way, but still perform their everyday practices with low-resource intensity and low ecological footprints, referred to as "inadvertent environmentalism" (Hitchings et al., 2013) or "quiet sustainability" (Smith & Jehlička, 2013). It is crucial to preserve these beneficial but unintended sustainable behaviours, and not demotivate these consumers through the introduction of policy measures.

Most public policy interventions use cognitive barriers and lack of personal capabilities as the entry point for behaviour change and try to build capacity through information provision and — to a lesser extent — aim to provide financial (dis)incentives. Target group-specific public policy interventions and the use of personalised feedback tools are rare. The current approach neglects that food-related behaviour is often dominated by (semi-) automatic decision-making and affective processes. Disruptive measures promoting healthier and more sustainable physical and social environments, such as taxes, bans and mandatory reformulations are needed to change consumer behaviour towards more sustainable and healthy food consumption.

Food environments are key levers for creating opportunities, strengthening consumer capabilities, and increasing consumer motivation (see Chapter 4). They offer platforms for implementing disruptive measures. Ideally, the policy mix combines broad measures with target-group specific measures.

3.4. Key messages

- From the consumer perspective, a transition to sustainable and healthy food consumption patterns involves a trade-off between sustainability and health on the one hand, with taste, price, social norms, symbolic meaning, and convenience on the other hand. It requires consumers to adopt (partly) new practices and abandon existing ones.
- Barriers preventing the transition to healthier and more sustainable food consumption patterns are situated at the individual consumer level in the form of a lack of motivation and personal capabilities, and at the contextual level in the form of a lack of physical and social opportunities, with reciprocal relationships between these levels.
- Consumers may lack individual agency to make healthy or sustainable food choices, particularly if they are from disadvantaged backgrounds and lack the resources to make such choices. The food environment becomes more influential in these circumstances and much evidence demonstrates the obesogenic nature of the current food environment, favouring less healthy and sustainable choices.
- Three types of psychological processes are involved when consumers make food choices: (a) habits, routines, and semi-automatic processes; (b) cognitive processes, i.e. deliberate thinking and/or information processing; and (c) affective processes, i.e. emotions and impulsive reactions.
- Most public policy interventions aim at increasing motivation and personal capabilities, and focus on altering the cognitive processes of food choice.
- However, given that habits, routines and semi-autonomous processes, as well as affective processes, are important determinants of food choice, policy measures need to address these as well. In this regard, interdisciplinary approaches within SSH are important to take into account, namely from sociology, anthropology, human geography and other interpretive social sciences.
- Disruptive measures that alter the context of food-related behaviour, in particular the physical and social environment (for example, taxes, bans, mandatory

reformulations), can alter routines and semi-autonomous processes. Such approaches are important in creating a food environment that favours healthy and sustainable choices without the need for high agency at consumer level, and a powerful way of reducing dietary inequalities.

Chapter 4. Shaping the food environment for healthy and sustainable diets

This chapter deals with interventions and instruments that make changes to the close food environment⁴⁰ and the implications for dietary quality in terms of health or sustainability. More general policy interventions addressing the food system as a whole and rebalancing power relationships will be addressed for specific policy packages in Chapter 5, p. 116.

Considering the mandate of the working group that wrote this report, we mostly deal with the effects of public and private policy instruments. A multitude of frameworks already exists that aim to classify policy instruments, such as the NOURISHING framework from the World Cancer Research Fund International that distinguishes between interventions in the food environment, the food system and communication tools.^{41,42} An alternative taxonomy applies the policy categories of the behavioural change wheel that address the different dimensions of the COM-B framework, as outlined in Chapter 3; it distinguishes between fiscal measures, guidelines, environmental and social planning, communication and marketing, legislation, service provision and regulation.

In this chapter, we will assess various interventions in the close food environment of consumers, according to the change in the food environment that is addressed, as listed in the external food environment in Figure 1, p. 20. We will not distinguish specifically between economic and behavioural-based approaches, instead focusing on the components of the food environment where they intervene. We consider interventions based on the concept of nudges as important, and treat them at the level which they address, for example, as nudges in the arrangement of food in the environment or informational nudges.

⁴⁰ As defined in 1.3, p. 35, the food environment describes the context in which food is accessed and eaten. Research has shown that it is the (geographically) close environment that determines food availability and accessibility. We stress this here by emphasising the proximity of the food environment.

⁴¹ See https://www.wcrf.org/policy/policy-databases/nourishing-framework/

⁴² NOURISHING stands for: Nutrition label standards and regulations on the use of claims on food; Offer healthy foods and set standards in public institutions and other specific settings; Use economic tools to address food affordability and purchase incentives, Restrict food advertising and other forms of commercial promotion; Improve the quality of the food supply; Set incentives and rules to create a healthy retail environment; Harness supply chain and actions across sectors to ensure coherence with health; Inform people about food and nutrition through public awareness; Nutrition advice and counselling in health care settings; Give nutrition education and skills. In this classification, 'NOURIS' belong to the domain of the food environment, 'H' to the domain of the food system, and 'ING' to the domain of behaviour change communication.

Interventions can come out of either private or public policies. It is important to keep in mind that they alter the incentives for various actors in the food system and trigger additional changes, for example in marketing practices or consumer behaviour. This review will summarise the existing evidence on the effectiveness, efficiency and distributional impact of these instruments.

The evidence reported in this chapter is based on different types of studies:

- Observed market outcomes are found in some studies, after an intervention has been introduced. Given the complexity of the food environment described in Chapter 2, direct causal effects are sometimes hard to identify. For example, causality may run in two ways when considering the correlation between obesity rates and the number of fast-food restaurants in a location. To disentangle simple correlations from causal effects, experimental methods are needed. Typically, either field or laboratory experiments are conducted.
- Field experiments are convincing because of their external validity, they produce data commensurate with actual consumer behaviour in typical purchase or consumption environments. In laboratory experiments, the internal validity can be better controlled, allowing for the clear identification of the causal effect (Falk & Heckman, 2009).
- Laboratory experiments may suffer from simplification of the complexity of the natural environment, where consumers typically have various behavioural options available. For example, experimenting with food prices in a vending machine may have a large impact on behaviour at the vending machine, but at the same time yield shifts in purchase behaviour to other food outlets which are not measured in the specific environment.
- When actual observations or experimental evidence are not available, **simulation studies** are often implemented, for example, when using elasticities (percentage changes in one variable of interest in response to the%age change in its determinant) to predict behavioural responses, such as a reaction to fiscal interventions like food taxes or subsidies. Simulation studies can also be used to generalise the results of consumer experiments about the effectiveness of new labels on market outcomes, including demand shifts and price changes. Simulation studies are also important in situations when various groups in the value chain have agency: consumers may not be the only ones to respond to fiscal policies; manufacturers may respond by modifying the recipes and composition of their products and retailers may alter their assortment when deciding which products to list. Consequently, the close food environment may itself respond to interventions aimed at altering consumer behaviour.

Considering the range of existing food policies in various countries, the forms of evidence on the effectiveness of changes to the food environment is quite diverse. While various

food labels on health and, increasingly, on sustainability do exist, their impact has been mostly evaluated in experimental and survey studies. Observational studies are increasingly used to understand the effects of sugar taxes that have been implemented on specific products such as sugar-sweetened beverages, including how much of the tax is passed through to consumer prices. Sustainability aspects, such as carbon taxes, have so far mostly been studied using simulation approaches.

The chapter will review the various dimensions of the food environment which the consumer interacts with: the economic environment and fiscal food policies (below); physical availability (p. 93); food composition (p. 99); the information environment (p. 100); and the social environment (p. 110). The chapter concludes with a summary of the state of knowledge on combining interventions from these different dimensions, including the relationship between pricing and changes in preferences (p. 111).

4.1. Economic environment and fiscal food policies

Introduction

The most important lesson we can take from the study of the economic environment's role on sustainable and healthy diets is that people change behaviour according to incentives, and not primarily because of information provision. These incentives need not be monetary, but monetary incentives are generally the most studied. This section examines how relative prices will need to be changed to deliver healthy and sustainable food consumption — meaning either how much cheaper healthy and sustainable diets will need to be, or how much more expensive unsustainable and unhealthy diets will need to be. There are several levers to modify prices, which can be brought about explicitly through taxes, subsidies or emissions trading.

The effectiveness of price interventions in changing dietary behaviour depends on how strongly consumers react to changes in food prices. To put price interventions in context, in a meta-review of 160 studies, Andreyeva et al. (2010) found price elasticities from 0.27 to 0.81 (absolute values) as a measure of how consumers react to changes in food prices. Food consumed out of home, soft drinks, juices, and meat are the most responsive to price changes (0.7–0.8). Focusing on healthy diets, Powell et al. (2013) provide a systematic review that found price elasticities of demand for sugar-sweetened beverages, fast food, fruits and vegetables to be –1.21, –0.52, –0.49 and –0.48 respectively, demonstrating how consumers are particularly responsive to price increases in sugarsweetened beverages compared with other food items. In a meta-analysis, Andreyeva, Marple, Moore, et al. (2022) found the effect of a subsidy on fruits and vegetables sales in the order of –0.59, but did not identify a significant change in consumption.

Shaping the food environment

There is a body of literature, highly relevant to this report's definition of healthy and sustainable diets, that models the effect of price interventions on key outcomes of interest, such as a healthy diets that follow WHO guidelines, the reduction of meat consumption and sugar, and reductions of obesity rates (Griffith & O'Connell, 2010; Hawkes et al., 2015; Latka et al., 2021; Powell et al., 2013; Roosen et al., 2022; Springmann et al., 2016; Willett et al., 2019). Studies looking at access to food for low-income households and making price interventions progressive are also considered (Allcott et al., 2019; Dubois et al., 2020; Klenert et al., 2022; Penne & Goedemé, 2021). In economics, the most developed body of literature is on pricing sugary products, which are a key lever for achieving healthier diets and reducing obesity.

This section of the report covers explicit price interventions. Implicit price changes resulting from the effects of bans, labelling, or quality standards are not covered, as the effects of these interventions, which can lead to implicit price changes, are covered in later sections.

The research assessed covers almost exclusively public interventions, i.e. fiscal policy. Private price interventions are scarce but do exist, for example as discounts on food products soon to expire. Hansen et al. (2021) find that price discounts on soon-to-expire products increase the likelihood of consumers' choosing them, and together with re-ordering by expiration date can reduce food waste in a representative store of a European grocery retailer. Nguyen et al. (2022) showed that personalised price promotions in the form of coupons are effective in leading to a healthier selection of menus in a university canteen setting.

Recent research on public price interventions in applied economics characterises which prices should be levied on key components of our diets. We highlight four key results from recent research:

- Latka et al. (2021) quantify necessary tax levels to achieve dietary recommendations. They find that specific taxes on food groups, such as animal products or food high in salt or sugar, are effective in reaching health and environmental sustainability targets, but that considerably high tax levels are required. For environmental sustainability, pricing animal products is key.
- Pieper et al. (2020) show that, in order to correctly account for climate change impacts, conventional and organic animal-based products should be priced by 146% and 71% surcharge on producer price levels, conventional dairy products by a 91% surcharge, while organic plant-based products only by a 6% surcharge.
- Funke et al. (2022) summarise the evidence for climate change impacts and nutrient pollution from meat consumption and find that the price of beef would be US \$5.75–US\$9.17 per kilogram higher if it reflected the true environmental costs to society. Accounting for diet-related health impacts, including the statistical value of life years

lost from premature mortality, further increases these costs, by up to an order of magnitude for processed meat.

Allcott et al. (2019) calculate optimal sugar taxes for the United States: accounting for concerns about the regressivity of the tax, they show optimal sugar taxes could be higher rather than lower when accounting for distribution.

As there is a large body of literature on pricing food, this section limits itself to addressing three key fields of policy interest:

- price interventions for making less healthy and sustainable foods more expensive: pricing (red) meat (below)
- price interventions for making less healthy foods more expensive: sugar taxes
 (p. 90)
- price interventions for making more healthy foods more accessible to low-income households: subsidies for healthy food (p. 92)

From these studies, it can be concluded that less healthy and sustainable foods need to be more expensive to meet societal objectives on healthy and sustainable food consumption. Importantly, pricing (red and processed) meat in accordance with its negative impacts on the environment and health is a key intervention point for delivering on both healthy and sustainable diets.

Price interventions for reducing less sustainable and healthy food: animalbased products

Animal products, and especially ruminant meat, are severely underpriced with regards to their costs to society from environmental damages that are not accounted for (Funke et al., 2022; Pieper et al., 2020). Reducing greenhouse gas emissions from the agricultural sector implies bringing down consumption of animal products and globally moving towards plant-rich diets (Clark et al., 2020). In terms of price interventions to achieve sustainable diets, we discuss the evidence around greenhouse gas-based taxes on food alongside consumption taxes on meat or dairy together.

Processed meat and red meat also contribute to the burden of a number of diseases (Godfray et al., 2018; Springmann et al., 2016, see also Chapter 2, p. 52). Despite recent discussions in Germany and the Netherlands about the introduction of an 'animal welfare levy' or 'meat tax', so far animal-based foods have not been explicitly priced in any country in the world, with the exception of New Zealand which is planning to price greenhouse gas agricultural emissions by 2025 (see Chapter 5, p. 116). However, through economic models and price elasticities, there is substantial evidence about the effectiveness, efficiency and equity effects of taxing animal-based products, and especially red meat, to deliver on healthy and sustainable diets. All animal products are

resource-intensive, with dairy and other ruminant products playing an especially large role in land-use-related environmental problems, more so than poultry (see Chapter 2), but the research available is almost exclusively on general meat taxation, including based on greenhouse gas emissions.

The effectiveness of meat pricing in reducing meat consumption depends on demand reaction and price pass-through. A meta-analysis by Gallet (2010) covering 419 earlier studies and reporting 4121 estimates finds a median own-price elasticity of -0.77 (percentage change of reduction in meat demand in response to a price increase), with considerable variation. For reducing environmental impact, Wirsenius et al. (2011) estimate that a 60€ per ton greenhouse gas emissions weighted tax on animal food products in the 27 member states of the EU would reduce agricultural emissions by 32 million tonnes, and that most of this effect results from reduced ruminant meat products. Moberg et al. (2021) studied various options for environmental taxation and VAT increases on animal products for Sweden. They find that basing taxation on weighting of several environmental impacts resulted in a reduction in the impacts for all environmental categories, for example a 0.98 Mt (-5.2%) reduction in greenhouse gas (Moberg et al., 2021). Roosen et al. (2022) have estimated elasticities for meat consumption in Germany and found the unconditional own-price elasticities is around -0.9, but with substantial income effects (i.e. changed consumption patterns due to change in income). Demand reactions are highest for beef and lowest for mixed-meat products and the tax incidence falls more on older than younger citizens. Increasing VAT to 19% would reduce German meat consumption by 11% for all meat types (Roosen et al., 2022).

Modelling studies are based on a wide range of elasticity estimates for food, meaning there is a quantifiable degree of uncertainty about the effectiveness of meat taxes. One cannot assume that taxes are fully passed through to consumers, given imperfect market structures such as market power for slaughtering and 'price psychology' (see section 4.6, p. 111). Once a jurisdiction levies a form of meat pricing, analysis of such policy could clarify these additional effects. Furthermore, the effectiveness of meat taxation for achieving environmental and health goals depends on avoiding the substitution of other environmentally harmful or unhealthy products, or cheaper, lower quality meat products (Bonnet et al., 2018; Funke et al., 2022).

Springmann, Mason-D'Croz, et al. (2018) studied the effectiveness of meat taxes on health outcomes globally. To account for the health-related costs to society attributable to red and processed meat consumption, especially their potential to cause cancer, they found that processed meat should be taxed at a 100% price increase in high-income countries, compared to only 4% for unprocessed red meat. They found "consumption of processed meat decreased by 16% on average [...] whilst red meat consumption remained stable as substitution for processed meat compensated price-related reductions" globally. Also

by 9% [...] and attributable health costs decreased by 14% [...], in each case with greatest reductions in high and middle-income countries" (Springmann, Mason-D'Croz, et al., 2018, p. 2).

Finally, the Danish tax on saturated fats in 2011 (repealed at the end of 2012) led to a significant reduction in demand for selected meat products (Jensen et al., 2016). However, Mytton et al. (2007) suggest that health-motivated meat taxes need to be carefully designed to avoid encouraging consumers to substitute with other unhealthy products. Furthermore, taxing the highest-emission food groups such as animal products to subsidise the lowest-emission food groups could have adverse health effects (Briggs et al., 2013; Caillavet et al., 2019). Subsidising meat replacement products (especially 'cultured meat') to bring down their cost could also be a valuable policy instrument if combined with pricing the environmental damages in agriculture (Treich, 2021; see p. 123 for examples on recent R&D government funding for 'cultured meat'), but there is no evidence yet of the effectiveness of such subsidies.

Taxing meat is an economically efficient policy to reduce meat consumption, as it delivers reductions of meat consumption with the least cost to society. However, for any specific environmental protection objective, pricing greenhouse gas emissions and nitrogen pollution directly at the source is more efficient and has a larger steering effect, as it provides producers with incentives for localised reduction of environmental harms, beyond the demand-side effect from reduced meat consumption (Funke et al., 2022). For reducing greenhouse gas emissions, emissions trading could lead to efficiency gains, especially including agricultural emissions into the existing and future EU emission trading systems (Grosjean et al., 2016). Schmutzler and Goulder (1997) show that, when costs to monitor compliance are high, consumption taxes can be more economically efficient than pricing emissions at the source, if production-side abatement options are limited and the taxed goods can be easily substituted. However, in practice, a significant proportion of agricultural emissions and mitigation potential could be covered by an emission trading design that targets large farms and few emission sources, thereby reducing transaction costs (Grosjean et al., 2016).

For greenhouse gas emissions, there is evidence that the potential for decarbonising traditional livestock farming in rearing and production is limited (Wirsenius et al., 2011). There is, however, some evidence from Sweden that globally desirable reduction in beef consumption conflicts with maintaining biodiversity-rich, semi-natural pastures (Moberg et al., 2021). Moreover, in an EU context, for optimal environmental pricing (levied where environmental damages occur), some form of border adjustment would be needed to ensure environmental gains are not undermined by increasing imports from overseas, and for protecting local farming industries from competitive disadvantage. Consumption taxes on meat can easily circumvent this concern, as imports can be taxed and exports exempted. Finally, in comparison to levying consumption taxes on meat in line with the

different meat type's environmental and health impacts or in comparison to greenhouse gas-based pricing on meat, simply altering the value-added tax rate is rather inefficient because it does not reflect how different types of meat can be more or less harmful to society.

When it comes to the distributional effects of meat taxation, taxes on food levied according to the greenhouse gas emissions of the product are slightly regressive with respect to income when the use of the tax proceeds is not considered (Caillavet et al., 2016; García-Muros et al., 2017; Roosen et al., 2022; Säll, 2018). In the United Kingdom, it was shown that such taxes have a disproportionate effect on households in the lowest socioeconomic class (Kehlbacher et al., 2016). The regressive effect can be mitigated by exempting certain subsistence goods such as cereals (García-Muros et al., 2017). Klenert et al. (2022) found for 24 EU countries that the distributional burden of meat taxes is mild and can be turned progressive by redistributing meat tax revenues via lump-sum transfers. Using meat tax revenues to lower value-added taxes on fruit and vegetables dampens, but does not fully offset, the mild regressive effect. As richer households in the EU consume more expensive and more greenhouse gas intensive meat products on average, differentiating tax rates based on average greenhouse gas content, or increasing VAT on meat products, have milder regressive effects than uniform taxes (Klenert et al., 2022). Pitt and Bendavid (2017) investigated the effect of changing meat prices on the burden of obesity by race and gender for the United States and found that males and black females are expected to realise greater reduction in obesity prevalence by 2030 from higher meat prices than white females, while black males benefit less in terms of longer life expectancy.

Price interventions for reducing less healthy food: sugar

Obesity is a very complex phenomenon (Hawkes et al., 2015). While overall calorie intake in diets is the key determinant, food preference learning, labelling and economic instruments have to work together as tools to deliver on obesity reduction (Alston et al., 2016; Griffith, 2022; Hawkes et al., 2015) with an important role for well-directed taxes on calories, sugar, or fat (Alston et al., 2016). The recent policy practice and evidence assessment in economics is on taxes on sugar, and sugar-sweetened beverage (SSB) taxes especially, as these are assessed as having little nutritional value when compared to most other foods. A general price rise on food to reduce its consumption would be ineffective and unjust (Griffith, 2022); however, the relative price between unhealthy and healthy foods meaningfully influences consumption choices. Specifically with high sugar consumption, taxes should be assessed jointly together with advertising restrictions and food availability. However, those who consume very high levels of sugar tend to consume a disproportionate share of soft drinks and confectionery, which explains the emphasis in present research on evidence for economic instruments on sugar taxes (Griffith et al., 2020).

For improved public health, the taxation of sugar and SSBs especially are an effective measure. Griffith et al. (2020) and Griffith (2022) discuss 27 studies of sugar-related taxes in eleven jurisdictions in the UK, and concluded that all studies find taxes lead to increased prices and reduced consumption. In settings of national economies like the UK, taxes are fully passed through to prices, while pass-through of the taxes is lower in smaller jurisdictions. Most studies found that taxes led to substantial reductions in purchases of SSBs. For instance, Powell et al. (2013) found that imposing a tax which raises prices of SSBs by 20% can lead to a reduction in consumption of around 20%. An earlier assessment by Cabrera Escobar et al. (2013) already showed that higher prices would be associated with lower demand for SSBs and lower obesity rates. In a recent meta-analysis based on 62 studies, Andreyeva, Marple, Marinello, et al. (2022) found that tax pass-through is 82%, suggesting tax undershifting. Given the high elasticity of demand (-1.5), a mean reduction in sales of SSBs by 15% is found. Recent research focuses on ex-post evidence of implemented sugar taxes in major economies outside the EU (see Chapter 5, p. 116 for existing sugar taxes and their design). In a systematic review, Teng et al. (2019) found that the equivalent of a 10% increase in SSB tax led to a 10% decline in beverage purchase and dietary intake.

We focus our discussion on the two most-studied cases, Mexico and the United Kingdom:

Mexico: Colchero et al. (2017) find that, within the first two years of the Mexican sugar tax implementation, it showed an average reduction of 7.6% in the purchase of taxed SSBs. Additionally, that research found that for households with the least income, purchases were reduced by as much as 11.7%. Conversely, a 2.1% increase was identified in purchases of untaxed beverages, especially purchased bottled water. Álvarez-Sánchez et al. (2018) found that Mexican individuals aware of the sugar tax further decreased their sugar consumption more than those not aware, pointing out the interaction between pricing and awareness campaigns (see 'Combinations of tools', p. 111).

United Kingdom: Bandy et al. (2020) find that sugar consumption from sugar sold in soft drinks was significantly reduced in reaction to the UK 'soft drink industry levy' and the consequent actions by industry. It confirms the modelling by Briggs et al. (2017), who quantified the reduction of obesity, type-2 diabetes and dental caries as a result of potential effects of the levy. Griffith et al. (2021) further simulate the outcome of a tax on all added sugar of £3 per kilogram and added salt of £6 per kilogram. They find reductions in sugar will be in the order of 10g per person per day and salt of 0.5g per person per day, depending on product reformulation and consumer responsiveness to the price increases (Griffith et al., 2021).

Sugar taxes are not efficient for obesity reduction, but might be optimal given actual policy constraints. An efficient economic policy for reducing obesity would be to increase

Shaping the food environment

the price of calories for those at risk of obesity, but in practice such an instrument is unfeasible, while a general price rise on calories to reduce their total consumption would be ineffective and unfair (Griffith, 2022; Kalamov & Runkel, 2022). With regards to changing relative prices and making less healthy foods more expensive, taxing SSBs and confectionery are most promising (as well as saturated fat taxes), as these are of little nutritional value compared to other ways of eating excess calories and demand is reactive to price increases (Powell et al., 2013). Hence, sugar taxes are an important tool for reducing obesity, along with labelling, food quality and learning food preferences (Griffith, 2022; Hawkes et al., 2015).

Dubois et al. (2020) studied who is impacted by the sugar taxes locally in the United Kingdom and the United States, finding that, for out-of-home consumption, the taxes are effective at targeting the sugar intake of the young, and are less successful at targeting the intake of those with high total dietary sugar. They are in general more effective at targeting the young than older members of the population. They further find that the welfare loss, ignoring potential health benefits, is 20% higher for those in the bottom half of the distribution of total annual grocery expenditure (as a proxy for income). Allcott et al. (2019) calculate optimal sugar taxes for the United States: accounting for concerns about the regressivity of the tax, they show optimal sugar taxes could be higher rather than lower when accounting for distribution, if poorer households make more mistakes in translating their consumption preference to purchasing behaviour.

Price interventions for healthy food

For subsidies for improving health, Pancrazi et al. (2022) explore how price distortions (when prices do not reflect demand and supply) can be part of the reason why there is a gap between the recommended and the actual intake of fruits and vegetables. They focus on the share of fixed cost in a product, which is significantly larger for healthy than less healthy products. They found that fiscal intervention could correct these distortions, so that all consumers can be made better off. Broeks et al. (2020) study the effects of introducing a meat tax of 15% and a subsidy on fruits and vegetables of 10% simultaneously in the Netherlands. They find that the health benefits of the fruits and vegetable subsidy alone could be between \leq 1.8 billion and \leq 3.3 billion, although the benefits from the meat tax increase would be twice as high.

Food subsidies are implemented as a way of making healthy food more affordable for low-income households (Hawkes et al., 2015). However, there is some field experimental evidence from the Netherlands that shows subsidising healthy foods could lead shoppers in a supermarket to purchase more calories overall (Waterlander et al., 2012). In the EU, policies that aim to address food insecurity and deliver healthier diets include food aid, nutritional education and financial incentives, but these policies are not targeted at the problem of insufficient income as an underlying cause of less healthy diets (Penne & Goedemé, 2021). In contexts in which healthy foods are more expensive to produce, there are equity reasons to provide targeted subsidies to low-income households to reduce their costs (Griffith, 2022). Indeed, a meta-review by An (2013) finds that targeted food subsidies for a healthy diets in high-income countries shift behaviour, while Black et al. (2012) caution about a lack of limited high-quality evidence. Hirvonen et al. (2020) estimate a lower bound for the affordability of a healthy diet across the world. In high-income countries (which includes all EU countries except Bulgaria), they find that less than 1% of the population cannot in principle pay for a healthy diet. Nevertheless, accounting in more detail for the cost of a healthy diet in 24 EU countries, Penne and Goedemé (2021) find that in 16 of those countries, particularly Bulgaria, Romania and Greece, at least 10% of the population in urban and suburban areas risk income-related food subsidy programmes alone, arguing that food policies need to be embedded in economic and social policies that address poverty more generally.

The nutritional quality of food donations is understudied. Mousa and Freeland-Graves (2019) assess the effect of food donations on the dietary quality of food pantry recipients.⁴³ They found that supplemental food provisioning can be an important resource for improving the nutrient intake of low-income populations. However, the study was based on 112 subjects only. A UK study on the parcels delivered by 11 food banks (Fallaize et al., 2020) found that donations were in excess of energy requirements and provided disproportionately high sugar and carbohydrate, and inadequate vitamin A and vitamin D, compared to the UK guidelines. Better cooperation with food donors and attention to dietary guidelines will be important in the use of food donations for healthy and sustainable diets.

4.2. Physical availability

This section summarises the evidence on how the physical environment influences sustainable and healthy food consumption, and more specifically how the availability and prominent placement of sustainable and healthy foods (or unsustainable and unhealthy foods) can affect consumers' food choices in supermarkets, shops, restaurants, canteens, schools and neighbourhoods. Interventions in the physical food environment are often designed and intended to correct existing market outcomes. Consumer preferences and competition among other market participants can result in a product assortment and built environment that imposes barriers on consumers' choices for healthy and sustainable products.

⁴³ Food pantries are organisations that distribute food, e.g. from a food bank.

Product placement and availability within supermarkets and other shops

Greater availability and more prominent placement of target products (for example, at eye-level or on special displays) is a promotion tool frequently used by supermarkets and food companies to increase sales with target products. These measures can also be used to promote healthy and sustainable foods — in this context they are referred to as 'choice architecture nudges'.

Healthy foods

Greater availability and more prominent placement of healthy food products in supermarkets and other food shops is associated with healthier patterns of purchasing and diet. Evidence comes from a systematic review of 17 observational and 22 interventional studies published up to 2019 (Shaw et al., 2020). The majority of studies showed that greater availability and more prominent placement of healthy foods, or reduced availability of less healthy foods, was associated with healthier diet-related behaviours, as determined by measures of diet, body mass index or sales and purchasing. Not all studies demonstrated a positive effect, and the quality of evidence varied, but the balance of evidence indicated favourable effects on healthier purchasing and diet outcomes. Similar results were found in a meta-analysis by Cadario and Chandon (2020) which found mostly modest effects. Lindstrom et al. (2023) carried out a systematic review on nudges and choice architecture. They found that nudge applications, in particular priming,⁴⁴ in food purchasing settings are an effective tool to promote healthier food choices; but a recent study based on a randomised controlled trial from a real-life online supermarket showed no overall significant effect of in-store nudges on healthy food choices (Stuber et al., 2022). More research is needed on optimal combinations of intervention strategies.

A more recent study in a UK discount supermarket chain to increase the availability of healthy foods by placing them in more prominent locations and removing unhealthy products from prominent locations has also been shown to have positive effects on customer purchasing patterns and on dietary patterns (Vogel et al., 2021). The intervention involved positioning new fruit and vegetable displays at the front of stores to create increased availability and more prominent placement, as part of the supermarket chain's phased store refurbishment scheme. Evaluation in a matched controlled cluster design showed increased store sales and customer purchases of fruit and vegetables maintained up to six months post-intervention. In the same study, removal of confectionery from checkouts and end of aisles opposite led to decreased confectionery sales, though customer purchasing was not affected. The study was relatively small-scale

^{44 &#}x27;Priming' refers to a technique where a cue is placed to activate specific associations before a target cue is placed. For example, an advertisement for fruit and vegetables viewed before a choice task leads makes the healthier choices more likely.

but is now being evaluated at scale as the supermarket chain's refurbishment scheme is rolled out across the UK.

The accumulating evidence on the influence of placement on consumer purchasing has led to the introduction of legislation in the UK to restrict the placement of foods high in fat, salt and sugar in prominent locations — end of aisles, store entrances and checkouts — in supermarkets and grocery stores of a certain size.⁴⁵ There is some evidence, however, that retailers might exploit loopholes in the legislation, creating, for example, new displays of crisps, confectionery and sugary drinks in the middle of aisles (Muir et al., 2022). The legislation was introduced on 1 October 2022 and will be evaluated in a study funded by the UK National Institute of Health Research.

Sustainable foods

Only few studies have analysed the effects of choice architecture nudges (greater availability and more prominent placement) on consumer choice of foods from sustainable production. Evidence from field experiments in supermarkets is surprisingly limited in this regard. We expect that the above-mentioned positive effects of greater availability and prominent placement observed for healthy foods are likely to be transferable to sustainable foods. The analysis of market data over time supports this assumption. For example, market data on organic food sales (from countries with larger market shares like Denmark and Germany) suggests that increased availability and higher salience of organic food in supermarkets had a large effect on consumers' uptake. As soon as 'regular' supermarket chains started selling organic food, sales of organic food, as well as the group of consumers buying organic food, grew substantially. Evidence from Denmark based on household scanner data exemplifies how consumers extend organic food purchases from one category to another over time (Juhl et al., 2017).

Meat consumption

The evidence on how consumers can be nudged towards healthier foods through increased availability and product placement of healthy options raises the hope that these measures are also successful in nudging consumers away from (red) meat and towards plant-based protein alternatives. First evidence from field experiments with supermarkets suggests that product placement of plant-based alternatives side-by-side with their meat counterpart is an effective measure for increasing sales of plant-based alternatives (Coucke et al., 2022; Vandenbroele et al., 2021). A supermarket field experiment tested the effect of an increased display of poultry products (increased size of the display area and quantity of displayed products) and a simultaneous decreased display for less sustainable meat products. The measures resulted in a significant

^{45 &}lt;u>www.gov.uk/government/consultations/restricting-promotions-of-food-and-drink-that-is-highin-fat-sugar-and-salt/outcome/restricting-promotions-of-products-high-in-fat-sugar-and-salt-bylocation-and-by-price-government-response-to-public-consultation).</u>

Shaping the food environment

increase in the sales of poultry during the intervention period, while the sales of less sustainable meat products remained unchanged (Coucke et al., 2019). Overall, we conclude there is clearly a need for more evidence from field experiments on how to nudge consumers away from meat and towards plant-based proteins in supermarkets, to untangle the effect of decreases in meat assortment size and prominence, and increases in salience and attractiveness of plant-based protein alternatives, including not only meat substitutes ('mock meat') but especially legumes and low processed plant-based alternatives.

Online stores and ordering platforms

Interestingly, there is evidence that the transition to online grocery shopping can actually support consumers in making healthier choices. Literature has found online shopping baskets to contain healthier products than their counterparts in brick and mortar stores, because the products are presented less vividly, and are therefore less gratifying and tempting (Huyghe et al., 2017).

Out-of-home consumption

Out-of-home consumption, which is an increasingly important part of consumers' diets, includes food consumption in any premises other than the household, including restaurants, canteens, schools, street food and so on (though definitions across Europe are not harmonised, sometimes including school and workplace restaurants and sometimes not; WHO Regional Office for Europe, 2022a).

Based on data from 1997–2000 in 10 European countries, Orfanos et al. (2007) estimated that out-of-home consumption contributed between 12% and 28% of calories per individual. In the UK, the share is estimated at 25% of calories and more than 50% of expenditures (WHO Regional Office for Europe, 2022a).

Schools

Placement matters in other contexts of food consumption, as well as in purchasing environments. Studies have focused on aspects of the school food environment and its impact on dietary habit and health outcomes. Neto and Gama Caldas (2018) studied the use of green criteria in the public procurement of food products and catering services. They noted a wide variety of procedures in place at national, regional and local levels, or for specific organisations (i.e., a school). Evaluating 23 green public procurement schemes, they found that the main food products covered by the criteria are fruits and vegetables, dairy, fish and seafood, and meat. Most green public procurement schemes set criteria at the early stages of the supply chain (for example, organic production, seasonality). Some criteria also focus on ethical issues, such as animal welfare, fair trade and food safety.

- Micha et al. (2018) looked at the effectiveness of school food environment policies on children's dietary behaviours and found that, in 91 interventions, direct provisioning increased fruit intake (0.27 servings per day) and vegetable intake (0.04 servings per day). Furthermore, competitive standards reduced intake of sugar-sweetened beverages and unhealthy snacks. School meal standards were also found to be important, but no effect on health (such as adiposity or metabolic factors) could be identified.
- Verdonschot et al. (2022) studied the role of fruit and vegetable provision, gaming and the school curriculum, measuring changes in both fruit and vegetable intake and nutritional knowledge. They found that provision of fruit and vegetables had the largest impact on fruit intake, while gaming/computer delivery had the largest impact on vegetable intake and the curriculum had the largest impact on on nutrition knowledge.
- Hendrie et al. (2017) studied strategies to increase children's vegetable intake in home and community settings (outside of school) and identified 22 studies between 2004 and 2014. Planning for social support, vegetable exposure and provision of staff training were commonly used techniques for effective interventions. 12 of the 22 studies showed short-term effects, while 6 showed long-term effects.
- Nathan et al. (2019) identified a weak impact of lunchbox interventions in improving children's dietary intake.
- Ismail et al. (2021) studied the effectiveness of fruit and vegetable distribution interventions (alone or as a component), and found a positive effect of distributing fruit and vegetables as a snack in schools, enhancing access and exposure.
- In an umbrella review on fruit and vegetable interventions, Wolfenden et al. (2021) found that choice architecture, in combination with strategies to increase accessibility and making food more tasty or palatable, are promising avenues to increase intake and reduce food waste.

Restaurants and canteens

Limited evidence from field experiments relating to more sustainable and healthy food choices in restaurants and canteens suggest potential for behavioural interventions combined with information-based interventions to increase healthier choices and to reduce unhealthy choices, to increase the consumption of meat-free meals, and to reduce food waste. Evidence from field experiments relating to food from sustainable production (for example, organic) is scarce.

Decreasing the convenience of unhealthy options by making them less accessible in cafeterias has been found to have some effect. In the systematic review by Cadario and Chandon (2020), summarising the effect of cognitively, affectively or behaviourally oriented nudge interventions in a range of settings and population groups, interventions

had greater effect on decreasing unhealthy food choices than on increasing healthy choices. In addition, those in cafeteria settings were more effective at decreasing unhealthy food choices than those in grocery stores, though overall effects were small. While there are inevitable biases in such experiments, such as selection bias and lack of control over the intervention, these findings do indicate the potential for reducing less healthy food choices in out-of-home settings.

The influence of menu labelling on customer choices in relation to calories has suggested modest effects in terms of reduction in calories chosen and on those consumed. Sinclair et al. (2014) conducted a systematic review and meta-analysis of 17 studies published up to 2013. Ten of the studies were experimental. Overall, calorie labelling of menus had only small effects on calories chosen or consumed. Effects were greater, however, if contextual information was added to the menu, indicating a healthier choice. Other systematic reviews published in 2015 (Long, Tobias, et al., 2015; Nikolaou et al., 2015) similarly indicate only small effects of menu calorie labelling on calories purchased or consumed.

Neighbourhoods

In the wider food environment, studies identified effects of the neighbourhood food environment on diet quality and health outcomes. Based on a review of 38 papers, Caspi et al. (2012) found moderate evidence of the effect of neighbourhood food environments on dietary health. In a systematic review for China, An, He, et al. (2020) found that variety, density and proximity of food outlets were positively associated with dietary diversity, portion size and daily caloric intake. Density and proximity of fast-food outlets and convenience stores were positively associated with adiposity in some studies, but not all. Similar results are found in Cobb et al. (2015) without a regional constraint. They also find many null associations, which may in part be related to siting of supermarkets and fast-food restaurants (Currie et al., 2010). Atanasova et al. (2022) looked at causal studies, where work and school environment studies were excluded. They found that in-kind or financial incentives, healthy food salience and health priming, but not calorie menu labelling, significantly improved dietary quality of children and adults, while effects on body mass index were null. The distance and number of fast-food or healthy food shops had the expected effects for children but only for selected groups of adults. Black et al. (2014) showed that inequalities in community and consumer nutrition environments are mostly observed in the US and less in other developed countries. Increasingly, studies have focused on the neighbourhood food environment comparing urban to rural space (for example, Love et al., 2019). In Europe, there is less evidence on spatial patterns in food environments and its resulting impact on dietary quality (Pinho et al., 2019; Titis et al., 2022).

By supporting the informal food system, interventions can equally enable healthy and sustainable food self-provisioning. Urban gardens are such an alternative system. Garcia et al. (2018) and Hume et al. (2022) provide a systematic review of the impact of urban gardens or community gardens on adequate and healthy diets, including 12 and 53 studies respectively. Impacts were identified of greater fruit and vegetable consumption, better access to health foods, greater valuing of cooking and enhanced importance of organic food. However, the authors of both reviews highlighted the need for more methodological rigour in this field of research.

4.3. Food composition

In addition to the physical availability of healthy and sustainable food, attention has shifted in recent years towards the composition of individual food products considering the large impact of (ultra-) processed food on the healthiness of the diet.

The reformulation of product recipes can help improve dietary quality. Research has shown that an assessment of the effectiveness of reformulation policies has to consider potential substitution effects with existing and new products in consumers' shopping baskets. These substitution effects limit the extent of possible improvements of the diet. Griffith et al. (2017) showed that reformulation towards lower sodium content improved consumers' shopping baskets more than an information campaign did in the UK. Spiteri and Soler (2018) showed similarly for France that reformulation can have an impact that is partly offset by new product launches and consumer substitution. Jensen and Sommer (2017) showed in a Danish study that silent reformulation had the effect of lowering the overall calorie content of consumption baskets also after accounting for substitution effects. Consumer reactions to various firm-specific or industry-wide reformulation scenarios have been studied, for example, by (Staudigel & Anders, 2020).

Experience with reformulation policies has been mixed. Often reformulation targets are built into voluntary agreements, limiting the effectiveness of the policy approach, in particular if the food industry is composed of many players and if agreements cover only a part of the product category (Leroy et al., 2016). Results in the UK indicate that the soft drink levy is more effective than the voluntary reformulation program (Public Health England, 2020).

Making healthier options more convenient or easier to consume has been found to be a comparatively effective nudge in steering consumers towards healthier choices. The same holds for interventions aiming to regulate portion sizes, such as providing larger plates for healthier and smaller plates of unhealthy options (Cadario & Chandon, 2020). Recent research has also studied the acceptance of processed food products replacing meat. Considerable effort has been undertaken for consumer acceptance. It has been shown that featuring vegetarian options as 'dish of the day' can help to improve the selection of vegetarian alternatives (Saulais et al., 2019). However, nudges that trigger greater vegetable intake must be carefully assessed to ensure that more vegetable intake is not accompanied by more food waste (Qi et al., 2022).

4.4. Information environment

To a large extent, private and public policies target the information environment. Public marketing campaigns (for example, the five-a-day campaign), labels and scores, and regulation of advertising, all belong to this set of instruments (Hobbs & Roosen, 2022). A prime example is the use of various labels signalling healthiness or sustainability of a product.

In addition to these 'classical' instruments, personalised feedback tools have emerged over recent years with the growth in digitalisation of food selection and purchase. Many of these interventions in the information environment, as in the physical environment, are based on behavioural principles and have been recently reviewed (Reisch et al., 2021).

Information campaigns

Studies have investigated the effect of the 'five-a-day' campaign (for example, Capacci & Mazzocchi, 2011). Other studies have compared the effectiveness of public campaigns in the light of ongoing reformulation efforts (Griffith et al., 2017). In this comparative assessment, it turns out that the relative contribution from reformulation is larger compared to the effect of the information campaign, and this is particularly true for households of low socioeconomic status.

Labelling

The effect of food labelling has been extensively studied, both in the field of nutrition and health and in the field of sustainability (referring to environmental, climate, social and animal welfare impacts). Labels are attractive information tools, as in theory they allow for the expression of heterogeneous consumer preferences. The effect of labels will depend on consumers' motivation to understand and use labels in their product choices (Grunert & Wills, 2007).

In the field of nutrition and health, studies in general distinguish between nutrition facts tables and front-of-pack labels. In this context, labels can be mandatory or voluntary. Because consumers may shy away from products that have negative attributes

communicated on a label, it may be necessary to implement mandatory labelling to make negative attributes also appear in the market (see 'Mandatory vs voluntary labelling schemes', p. 105).

Front-of-pack labels aim to simplify information to make it more actionable. These come in various forms, from simple check-marks that a production standard is fulfilled (for instance, the organic label) to warning labels such as 'high in salt' or 'high in sugar', to more complex scoring systems which assess various dimensions at a time (for instance, the Nutri-Score label assesses nutrients). Scoring systems usually combine a product rating that summarises various dimensions with a colour scheme. Extensive evaluations of the Nutri-Score in France and other countries have shown that such scoring systems are helpful for judging the healthiness of products, and enable ease and speed of interpretation (Hercberg et al., 2022). Experimental studies have shown that the nutritional value of food baskets improves, both in lab experiments (Crosetto et al., 2019) and in field experiments (Dubois et al., 2021).⁴⁶

Given the complexity of the existing labelling landscape, policy changes to address new issues face the question of whether to augment existing labels or create new ones. For example, deforestation is not considered when labelling food as organic, but it is very important for biodiversity protection. Whether to add a label on deforestation rules, such as the one of the Rainforest Alliance, or augment the specification of the organic label, is a difficult question. Research looking at the effect of multiple sustainability labels shows that conflicting messages may lead consumers to trust labels that they are already familiar with (De Bauw et al., 2022).

Nutritional labels and information

Several meta-analyses shed light on, and generalise empirical insights about, the effect of nutritional labels (and health-related information in general) on consumer attitudes and purchase behaviour.

Sugar:

- Based on 23 studies, An, Liu, et al. (2020) established that sugar-sweetened beverage warning labels reduced the likelihood of choosing sugar-sweetened beverages. A graphic with health effect labels had the largest impact.
- Grummon and Hall (2020) conducted a meta-analysis of 23 experimental studies on sugary drink warnings, and also found that sugary drink warnings not only cause negative emotional reactions, but also reduce consumption and purchase (and purchase intentions).

⁴⁶ Bastounis et al. (2021) provide a systematic review and meta-analysis for consumers' willingness to pay for foods with environmental sustainability labels.

Based on a meta-analysis again of 23 studies, Scapin et al. (2021) concluded that health warning messages, graphics showing sugar content in teaspoons, and warning signs all steer consumers towards products with lower sugar content.

Front-of-pack labels:

- Based on a meta-analysis of only 6 studies, Agarwal et al. (2021) found that energy and fat content labelling had the intended effect of reducing the consumption of energy and fat.
- In a systematic literature review of 15 studies, An et al. (2021) found that front-of-pack labels such as traffic lights, health star ratings, daily intake guides, health warnings and high sugar symbol labels prove effective in around half of the studies, but ineffective in the other half.
- In a meta-analysis of 14 studies on front-of-pack labels on prepackaged food, Croker et al. (2020) concluded that front-of-pack labels encourage healthy food purchasing.
- Ikonen et al. (2020) generalised the findings of 114 articles in a meta-analysis and found that front-of-pack labels have only a limited effect on steering consumers towards healthier choices. Warning labels have the strongest impact on consumers' choice of healthier food, while front-of-pack labels have little impact on food consumption. Importantly, front-of-pack labels may have the negative side effect of improving health perceptions of so-called 'vice products'. (those that provide immediate pleasurable experience, but contribute to negative long-term health outcomes).
- Based on a meta-analysis of 13 studies, Song et al. (2021) concluded that traffic light labels, nutrient information and health warnings steered consumers towards healthy products, while Nutri-Score and warning labels were effective in making consumers avoid unhealthier options. Colour-coded labels were better able to steer consumers towards healthier choices, while warning labels were more likely to discourage unhealthy choices.
- Based on a meta-analysis of 14 randomised controlled trials (three alcohol, eleven food), Clarke et al. (2021) established that health warning labels (either text-only or pictorial) make participants 26% less likely to choose the product.
- In a meta-analysis of 15 randomised controlled studies, Daley et al. (2020) concluded that physical activity calorie equivalent labels displaying the physical activity calorie equivalent of food reduced the number of calories selected and consumed compared with other types of food labelling and the absence of labels.
- Based on 60 studies, Shangguan et al. (2019) concluded that food labelling decreased consumer intake of energy by 6.6% and total fat by 10.6%. Furthermore, vegetable intake increased by 13.5%. Interestingly, they also identified industry

responses to such labelling schemes, in that products contained 8.9% less sodium and 64.3% less artificial trans-fat.

Nutrition labels in out-of-home consumption:

- Based on a meta-analysis of 22 studies among college students, Christoph and An (2018) found that nutrition labels had a moderate, yet positive effect on college students' diets.
- Based on a meta-analysis of 28 studies, Crockett et al. (2018) concluded that nutritional labelling on menus in an out-of-home consumption setting may reduce the number of calories purchased. However, they rated the quality of the evidence on which their meta-analysis was based as low, did do not provide any estimates for vending machines and supermarkets because they found the evidence on these to be too low-quality. Based on a separate meta-analysis of three randomised controlled trials, they concluded that calories per meal were reduced by 47 kcal (7.8%) when energy labelling on menus in restaurants was applied. Interestingly, they do not find any evidence for the often-cited 'dark side' of nutritional labels, that they increase instead of reducing calories purchased or consumed (see 'Negative side-effects of labels', p. 105).
- Sinclair et al. (2014) meta-analysed 17 studies and found that menu labelling with calories alone did not decrease calories selected or consumed, but the addition of contextual or interpretive nutritional information did have this effect.

Based on these reviews and meta-analyses, we can conclude that there are some effects of nutritional labelling and health warning messages on the food consumers purchase, but effects seem to be moderate and/or limited. Health warning labels seem to be comparatively effective, but also trigger negative reactions. Furthermore, in many studies, warning labels about a product's sugar content may prove effective because of the availability of sugar-free alternatives in many categories: switching from regular Coke to Coke Zero demands less changes from a consumer than adapting one's diet.

We also note several limitations in this stream of research. Firstly, the current body of literature only includes a limited number of real-world studies. In the meta-analysis of Song et al. (2021), for instance, 95% of the studies were laboratory studies. This is likely to overestimate the impact of nutritional labels and health messages on consumer choice, given that customers pay attention to many more stimuli in a real-world environment, besides the nutritional content of food. Secondly, publication bias may play a role here, as studies that show significant results may be more likely to be published. Some, but not all, meta-analyses try to account for this.

Sustainability labels and information

Sustainability labels such as organic, animal welfare and fair trade labels have proven to be an effective tool for reaching consumers who are motivated and interested in

Shaping the food environment

food that supports these types of production standards. Sustainability labels enable consumers to recognise and identify respective foods which were produced according to a defined set of sustainability standards. However, it should be noted that 'sustainability' is a so-called 'credence attribute', meaning that consumers cannot verify whether the respective production standards were met.

Potter et al. (2021) provide a review of the effect of sustainability labels. Overall, they are found to be associated with the selection and purchase of sustainable food products. Sustainability labels and claims are only successful if consumers find them trustworthy (Nagy et al., 2022), in particular regarding two aspects (Janssen & Hamm, 2012). First, consumers need to have a positive view of the underlying standards; if they are not perceived as offering substantial improvements compared to the status quo, the label risks being accused of greenwashing by consumers (as has happened with the Marine Stewardship Council label for wild fish, for instance; Ponte, 2012; Wijen & Chiroleu-Assouline, 2019). Second, consumers need to trust that the product fulfils the promised standards. Third-party certification and labelling schemes are a mechanism to overcome the information asymmetry between producers and consumers and raise consumer trust (Jahn et al., 2005).

Interestingly, sustainability labels seem to influence consumer perceptions also beyond the domain of the label itself. For instance, organic foods are viewed as healthier, and the same 'health halo' effect has been found for fair trade food (Nadricka et al., 2020; Tobi et al., 2019).

Regarding the impact of environmental sustainability labels on consumer behaviour, a recent meta-analysis (Bastounis et al., 2021) establishes that consumers are willing to pay a price premium of 3.79 purchasing power parity dollars per kilogram (95%CI 2.7, 4.89; based on 35 studies) for eco-labelled food. There was some variation in this effect, in particular for meat and dairy products, where the effect was stronger than other product categories. Consumers were also willing to pay more for food carrying an organic label than other environmental sustainability labels (for example, CO₂ emissions, water, land, pesticide use, and biodiversity loss). The authors note sizable heterogeneity between consumer segments. In particular, women and younger consumers were willing to pay more for food with an environmental sustainability label. Counterintuitively, this also holds for consumers with lower education levels, although this may be an artefact of many samples being biased towards respondents with a higher level of education. According to another study (Wägeli et al., 2016), consumers are willing to pay more for products from animal-friendly husbandry systems, where outdoor access, stocking density and floor type are viewed as important determinants of animal welfare.

Potter et al. (2021) show in their systematic review that ecolabels have a positive effect on the selection, purchase and consumption of more environmentally sustainable food and drink products. Heterogeneous responses show that labels were more effective among women and higher income or education consumers. While this systematic review highlights the risk of bias present in many studies, it also shows that the studies with a high risk of bias do not favour interventions compared to studies with a low risk of bias.

However, while there are comparatively many meta-analyses examining reactions to health-related and nutrition labels, only a few review studies cover reactions to sustainability and environmental labels, such as willingness to buy or willingness to pay. More efforts are needed to systemise this literature. Moreover, studies based on hypothetical experiments and surveys show a positive consumer response but are potentially biased, overestimating the effects of sustainability labels on actual behaviour.

The limited evidence on the effectiveness of sustainability labels from field experiments suggests that sustainability labels have a significant but small effect (e.g., Elofsson et al., 2016). Market data, for example, for organic food, suggest that sustainability labels (combined with certification, where appropriate) are an important prerequisite for successful market penetration, but not sufficient for reaching consumers who are not already highly motivated to buy sustainably produced foods, especially if sustainable foods are perceived as involving extra effort such as higher prices, inconvenience, limited selection.

Mandatory vs voluntary labelling schemes

The question of mandatory versus voluntary labelling schemes is an important topic in relation to sustainable and healthy food consumption. While nutritional facts panels have been mandatory in the EU for decades, front-of-pack nutrition labelling is voluntary, and there is no mandatory labelling relating to sustainability (climate impact, environmental impact, animal welfare).

While mandatory disclosure of information has proven to be an effective tool in the area of energy efficiency in the EU, steering consumers towards home appliances with higher energy efficiency and incentivising producers to change their products,⁴⁷ scholars argue that there is sufficient scientific evidence to recommend the implementation of mandatory front-of-pack nutrition labelling to improve population health (Roberto et al., 2021; Song et al., 2021), nor mandatory sustainability labels (for animal welfare, see Wägeli et al., 2016).

Negative side-effects of labels

There are also 'dark sides' to labels that can act as a barrier in steering consumers to a more sustainable and healthier eating pattern. Healthy foods may suffer from consumer

⁴⁷ https://commission.europa.eu/energy-climate-change-environment/standards-tools-and-labels/ products-labelling-rules-and-requirements/energy-label-and-ecodesign/about_en

inferences that they are less tasty and more expensive. Sustainable food labels may trigger inferences that the food is inferior in quality and taste. In some product categories, a sustainability label can make consumers infer that the product is not strong enough, or that quality was sacrificed for sustainability in the product development phase. Such negative inferences likely explain findings in literature that a sustainable label harms rather than boosts sales of new food products (Luchs et al., 2010; Newman et al., 2014; van Doorn et al., 2021; van Doorn & Verhoef, 2011). It is therefore important to design healthy and sustainable food that overcomes the 'unhealthy=tasty' intuition, and to build taste preferences for healthy options (Briers et al., 2020).

Labels and their effects on socioeconomic groups

Information campaigns and food labels are regulatory tools that can lead to efficient outcomes when information is costly and consumers have perfect recall and understanding. To lead to fully efficient outcomes, consumers would also need to fully internalise externalities in their actions, which is not likely. Therefore, it is unlikely that the sustainability of food choices can be fully achieved based on information alone. Also, when consumers do not pay attention, information tools such as labels have to be supported by other instruments such as taxes and subsidies (Roosen & Marette, 2011).

Overall, the effect of labels and other information will have varying impact on different socioeconomic groups. They are often considered to be most effective among the educated. Kenkel (1991) noted that better schooling may lead to better health knowledge, and could therefore explain the tendency for better health behaviour among the better educated. However, even when accounting for differences in knowledge, it was found that differences in health behaviour by education level remain; more educated people may also have a higher preference for healthy behaviour, or beliefs that motivate a greater response to labels (Nayga, 2000).

The differential impact of environmental interventions on consumption and resulting health outcomes triggers the concern that information interventions mostly favour middle-income and high-income parts of society, leading to greater health inequities. Evidence suggests that many interventions in the food environment have reached disadvantaged consumer groups to a lesser extent (von Philipsborn et al., 2019). However, information on the equity impact of many interventions is limited, not only because the necessary information on diversity dimensions⁴⁸ is not collected, but also because disadvantaged groups are less likely to participate in studies (Waters et al., 2011).

The equity effects for sustainability labels may be different. In this context too, more affluent and better-educated people respond to labels more than people from lower

⁴⁸ Disadvantage can be identified in terms of place of residence, race or ethnicity, occupation, gender, religion, education, socioeconomic position and social capital, known as the PROGRESS acronym.

socioeconomic strata of society. However, as discussed in Barkemeyer et al. (2023) they also tend to consume more unsustainably by consuming more. Eco-labelling schemes are more prominent in societies that consume more at a general level, and the propensity to use products with eco-labels is higher among consumers with a larger resource footprint. Sustainability labels may hence lead to a rebound effect.

Advertising restrictions

Another tool to address the information environment is the use of advertising restrictions, which often tale the form of voluntary industry pledges. Such an approach was evaluated by Landwehr and Hartmann (2020), where major food companies reduced their advertisements to children across the EU. Studying the change in advertising content on children's channels in Germany, the study showed that the voluntary pledge was limited in effect in reducing exposure to advertisements, partly because of its focus on children's programmes and partly due to lenient nutritional criteria.

A few studies go further to assess the effect of advertising regulation on consumption behaviour. Silva et al. (2015) looked at regulation in the UK. While television advertising regulation led to a reduction in expenditures on foods high in fat, sugar or salt, the study also observed a reallocation of advertising budgets of firms from television to other media. Dubois et al. (2017) furthermore show that an advertising ban on junk food may be countered by companies with increased price competition, limiting the effectiveness of the policy.

More recent evidence from the UK is encouraging. In early 2019, advertising restrictions on food and drinks high in fat, sugar and salt were introduced in the UK across the Transport for London network. Recent evaluations at population level, examining household purchasing patterns up to 10 months after the restrictions were introduced, demonstrated relative reductions in purchasing of these products in the intervention area compared with an area in the north of England where there were no restrictions (Yau et al., 2022). An accompanying economic modelling study of the impact of the changes in purchasing patterns on health at the individual level demonstrated a reduction of 4.8% in the proportion of individuals with obesity (equivalent to approximately 95 000 cases), with reduced incidence of diabetes and cardiovascular disease three years post-intervention (Thomas et al., 2022). The model suggested that there would be greater benefits over time for more disadvantaged groups in terms of quality of life.

Written and oral injunctions steering consumers towards healthier choices (for example, "have a tossed salad for lunch") have been found to have some effect, as have approaches which position the healthy option as hedonically more attractive, e.g. "dynamite beets" (Cadario & Chandon, 2020). The Veganuary campaign, a social media campaign for a vegan January, may have increased demand for plant-based alternatives but did not lead to a reduction in meat demand (Trewern et al., 2022).

Boyland et al. (2022) provide a systemic review of the effect of policies to restrict marketing of foods and non-alcoholic beverages to children. Outcome measures were exposure to marketing messages and their persuasive power, as well as dietary intake, choice preference, and purchasing. They found marketing policies may result in reduced purchases of unhealthy food, but stressed that results of individual studies are mixed, notably also in the way in which outcomes are measured. Therefore, it seems important that advertisement regulations state clear goals. For example, if a reduction in exposure of advertising to children is an aim, is this based on placement in children's programmes, or on advertisements generally designed to appeal to children? The authors favoured a mandatory implementation of the policy aligned with WHO recommendations.

Oke and Tan (2022) provided a review of the effectiveness of healthy food advertising in school settings to increase fruit and vegetable consumption. Only 8 studies could be included in the review. While advertisement was shown to be effective, the authors criticised the weak evidence because of non-cohesive study designs.

Food apps

The emergence of the digital food environment (see p. 40) has consequences for food choice, dietary intake and eating behaviour. Various apps exist that aim to support healthy and sustainable food purchasing and consumption. Targeted at individual consumers, these app-based solutions are typically either dietary tracking, monitoring and interventions apps to facilitate nutritiously balanced food consumption (for example, Yuka, see Soutjis, 2020), or ethical consumption apps that seek to inform consumers about the eco-score ranking of their food choice (for example, Weber, 2021), how they can 'save food' (for example, TooGoodToGo, see Fragapane & Mortara, 2022), or how to buycott or boycott certain foods (for example, Buycott, see Eli et al., 2016; Hawkins & Horst, 2020).

To date, several systematic reviews have reviewed the effectiveness of dietary tracking and intervention apps to support healthy food purchasing and consumption (DiFilippo et al., 2015; Lim et al., 2021), their impact on increasing daily fruit and vegetable consumption (Mandracchia et al., 2019), and their impact on improving diet quality (Kankanhalli et al., 2019; Rodríguez-González et al., 2023; Scarry et al., 2022). Lim et al. (2021) found "modest evidence for the efficacy of app interventions to improve healthy food purchasing and food consumption". They highlighted the advantages of apps compared to other media such as text messaging, websites, paper journals, email, face-to-face counselling and group sessions. They state that apps are effective tools for eaters seeking convenient, cost and resource saving interventions.

Mandracchia et al. (2019) found that interventions described in six of the eight studies they reviewed were effective in increasing fruit or vegetable intake. Scarry et al. (2022) found that
60% of the studies reviewed showed improved diet quality due to the use of mobile apps.⁴⁹

Evidence from field experiments shows that personalised feedback is effective in changing behaviour into the desired direction.⁵⁰ A study with restaurants in the United States tested the effect of personalised recommendations to switch from unhealthy to healthier items displayed on the receipts. This measure shifted the mix of items purchased toward the healthier alternatives (Bedard & Kuhn, 2015). A study by Armitage and Conner (2001) found that personalised feedback on fat consumption significantly reduced fat intake. Mohr et al. (2019) and VanEpps et al. (2021) evaluated the effect of real-time aggregation of the calorie content of meals to provide dynamic feedback using traffic light symbols, and found that dynamic feedback reduces calories in food orders. Similarly, Livingstone et al. (2021) confirmed that personalised nutrition advice reduced intake of discretionary foods and beverages.

Importantly, there is some evidence that dietary tracking apps might be linked to disordered eating (Hahn et al., 2022; Kent, 2020; Messer et al., 2021). Food delivery apps may also play a role in disordered eating urges (Portingale et al., 2023). As no systematic review of studies on this topic exists at the moment, the evidence is limited to individual studies that at times also focus on selected groups of users (for instance, emerging adults), but should not be ignored. As Messer et al. (2021) show, prior app users reported higher levels of thinness-oriented and muscularity-oriented disordered eating than non-users in the studies they reviewed. Users who employed the app to lose weight, or control weight or their body shape, were more likely to report that the app use contributed to disordered eating. Relatedly, social media such as TikTok play a potential role by predominantly espousing weight-normative messaging (Minadeo & Pope, 2022). In practice, it is very difficult to disentangle the different digital media stimuli that influence eaters' practices and may lead to disordered eating.

Personalised feedback tools have proven to be effective in inducing healthier food choices (Bedard & Kuhn, 2015), but there is a lack of research on the effectiveness of

⁴⁹ It is important to state that the systematic reviews referred to above mention important limitations. Lim et al. (2021) highlight that their review was limited to English language-based studies and is likely to have a publication bias. They also point out that interventions mentioned in studies older than 2013 differ from more recent ones: "Technology apps with older interventions may not be as effective as newer apps with advanced technology such as food photo recognition and personalised real-time feedback." (Lim et al. 2021, p.17). Further limitations include the different methods used to calculate nutritionally healthy and balanced diets in the reviewed studies, the inclusion of studies with non-significant findings, the amount of studies included in the reviews, the length of the intervention studies, and the targeted intervention group (often younger aged persons and women). Händel et al. (2019) also criticise the lack of rigour in systematic reviews and meta-analysis when selecting studies.

⁵⁰ From the area of residential home energy consumption, it is known that peer feedback reports (informing customers how much energy they use and how much energy their neighbours consume) have the potential to decrease energy consumption (Jachimowicz et al. 2018; Ayres et al., 2013).

personalised feedback on environmentally sustainable food choices. Retailers have started to implement personalised feedback tools to induce environmentally sustainable food choices; for instance, the Norwegian online supermarket Oda⁵¹ provides customers with an overview of their past purchases and the approximate CO₂ emissions, with the goal to steer customers towards food categories with lower greenhouse gas emissions. As indicated in section 1.4, p. 40, the appearance of food delivery apps may replace home cooking with delivered food, which is often more calorie dense. Alternatively, apps can also help consumers in food purchase and meal planning, thereby reducing food waste.

4.5. Social environment

When the consequences of our behaviours are distant or opaque, one decision-making strategy is to refer to others in our social environment (Sparkman et al., 2021). Social environments exert powerful influence on consumers' behaviour, and unhealthy and unsustainable dietary choices are often the norm. However, social norms can also be part of the solution (Nyborg et al., 2016). Many studies have suggested that descriptive norms, which communicate a desired behaviour for others, such as posters of people eating vegetables with lunch in the workplace, can be effective (Taufik et al., 2019). When a certain behaviour is not yet widely adopted, or when consumers already perform better than the descriptive norm, then injunctive norms (i.e., what individuals perceive as acceptable behaviour, what ought to be) are more effective at motivating behaviour change (Davis et al., 2018). Even minorities can change social norms on a societal level, and some studies pinpoint the existence of certain tipping points to overturn societal norms (for example, Centola et al., 2018). In addition to social norms, studies have shown the influence social support exerts on dietary-related behaviours, in general and in a situation (for example, Brug, 2008; Perino & Schwirplies, 2022).

While 'social support' in general is defined as the availability of resources, information, emotion or motivation by family or peers, for example, 'social support in the situation' is defined as the supporting or thwarting nature of individuals accompanying an individual at the point of decision (see for example, Inauen et al., 2017; McSpadden et al., 2016).

In a systematic review of reviews, Wright and Bragge (2018) found that associating a person's identity with a social group that is perceived to eat healthily, i.e., a positive positioning of social identity, is related to how much food is consumed when dining out. Chung et al. (2021) analysed peer influence on healthy eating among adolescents via social media and found significant influence on healthy eating behaviours (fruit and vegetable intake) and unhealthy ones (fast food consumption). Instagram and Facebook

⁵¹ https://sustainability.oda.com/klimaavtrykk-for-matvarer

were among the most popular social media platforms. In a review of 160 experimental interventions, Byerly et al. (2018) found some evidence that social influence and social norms can encourage reduced meat consumption. In a meta-review, Grundy et al. (2022) found that emphasising social norms is promising to reduce meat consumption. Yee et al. (2017) investigated parental influence on child food consumption and found that parental active guidance and education, psychosocial mediators and parental styles all influence child food consumption.

4.6. Combinations of tools

Increasingly, intervention studies for healthy diets focus on a combination of tools. The review of interventions in a school setting has shown that interventions that combine placement, information and education tools are particularly promising. A review by Mah et al. (2019) on retail food environment interventions similarly shows retail interventions can combine elements of geographic access with classical marketing tools in the area of price, promotion, place and product (the 'four Ps'). Programmes in schools also often combine different elements, such as providing fruit and vegetables with nutrition education elements. Furthermore, Reisch et al. (2021) show that a single intervention is used in 14% of the studies analysing climate change mitigation interventions at the consumer level, while 43% use two interventions and 34% use three simultaneously. The authors of the review conclude that a comparable share of studies evaluating multi-component interventions as compared to single component interventions had mixed or null effects on measures of health outcomes.

Regulations intervening in the close food environment often face issues that firms' action and market outcomes can strengthen or weaken policy outcomes (Hobbs & Roosen, 2022). Economic studies have turned to these dynamic interactions between policy levers. This includes reactions to labelling policies and strategic use of alternative labelling options (Villas-Boas et al., 2020), opposing pricing strategies (Dubois et al., 2017), and manipulation of ingredients or nutrients to enable front-of-pack claims. Therefore, it is important that regulatory interventions keep an eye on possible unintended consequences. For example, Kleis et al. (2020) show a ban on trans-fatty acids in Denmark has led to a replacement of these by saturated fatty acids, hence limiting effectiveness of the policy in improving dietary quality.

In addition, the release of new dietary guidelines may change consumer demand and influence the offering of and demand for specific product types. For example, Mancino et al. (2008) analysed the effect of the 2005 US Dietary Guidelines on the consumption of whole grain products. The 2005 Dietary Guidelines made its previous recommendation that encouraged the consumption of whole grains more specific, in that at least half of a

person's daily grain intake should come from whole grains. The guidelines have spurred competition among food manufacturers to introduce new products to expand the availability of whole grain foods, to certify their products as whole grain, and to use this certification in marketing communication on the food package.

In a review, Vandevijvere and Vanderlee (2019) find evidence that reformulation of food products also interacts with labelling and taxation policies, and that the effect of these policies is not limited to its effect on consumer behaviour but also creates incentives to firms to reformulate. In a more recent study, Allais et al. (2020) analyse the effect of sugar taxes on food reformulation in a simulation study. They estimate consumer preferences and, considering the cost of reformulation, the impact of a sugar tax is estimated. Hence, sugar taxes do not only influence demand and consumption directly, but can also further reduce sugar consumption by reformulating existing product ranges. Furthermore, Fernandez and Raine (2019) argue that sugar taxes should be combined with interventions that increase access to healthier beverages, to increase access to education about healthier drinks and to monitor the intended and unintended effects if sugar taxes are meant to be a viable anti-obesity policy.

Given the strategic response of food companies to various instruments that influence the proximate food environment, good monitoring tools are essential. This does not only require the regular implementation of dietary intake studies, but also the assessment of the nutrient content of food over time. In addition, given the limited effect of individual interventions and countervailing action by supply chain members, a combination of instruments is necessary to assure a substantial shift towards healthy and sustainable diets (Hawkes et al., 2015).

BOX 8. PREFERENCE FOR FOOD CONSUMPTION CHANGE WITH POLICY CHOICES

In the language of economics, it is not only the context in the moment of a decision that influences consumers. Instead, preferences, which economics assumes to be the basis for decisions, are shaped by cultural background and mental models of the individuals and hence are malleable by societal choices (Hoff & Stiglitz, 2016).

Across the social sciences studying diets, it is widely accepted that healthy preferences for food consumption can be learned and guided by society and policy (Birch, 1999; Hawkes et al., 2015). One important implication in assessing the evidence for policies is that price interventions can change preferences and lead to unintended outcomes: there is the risk that those price changes which are needed to deliver the required change in diets (see p. 85) might decrease intrinsic motivation to save the environment (a "preference change": see Bowles, 2016). Therefore, a price signal leading towards healthier and more sustainable choices should also be accompanied by behavioural interventions and informing citizens. That can lead to increased intrinsic motivation reinforcing the regulatory objective (Bowles, 2016; Gravert & Shreedhar, 2022). As an example, Lanz et al. (2018) find that in a field experiment with British supermarket shoppers, a carbon price of $£19/tCO_2$ on food decreases intrinsic motivation. Compensating for this effect requires the carbon price to rise by as much as £48/ tCO₂. However, in certain circumstances, relative price changes can increase intrinsic motivation to eat more sustainably. In particular, the "meat paradox" — people hate to harm animals, but still eat meat from factory farming — is often explained by cognitive dissonance (Buttlar & Walther, 2018; Gradidge et al., 2021) and can be understood as a case of endogenous beliefs (Hestermann et al., 2020). A tax on sugar-sweetened beverages may crowd-in intrinsic motivation for healthy drinking, at least if individuals are aware that such a tax exists (Álvarez-Sánchez et al., 2018), although whether such effects exist may depend on relevant geographies (Ahn & Lusk, 2021).

Such findings indicate that price signals will need to be guided by behavioural and informational instruments and by careful explanation of the policy to the public (see Chapter 5, p. 116) as part of policy packages (Mattauch et al., 2022).

4.7. Key messages

- People are most likely to respond to incentives, monetary or otherwise. For the EU to meet objectives on healthy and sustainable food consumption, unhealthy and unsustainable diets need to be more expensive, especially animal products and products high in sugar. Evidence from sugar taxes, in particular from Mexico and the UK, show that they are effective in reducing sugar intake and can help reduce obesity. Making the price of animal products reflect the social cost of the associated environmental damages is economically efficient, including levying carbon pricing on greenhouse-gas emissions from agriculture. In view of high monitoring costs, substitution effects and food imports, consumption taxes on animal products are also effective in reducing the environmental impacts from these products. Importantly, increasing prices on red and processed meat correctly is a key lever for delivering on both healthy and sustainable diets.
- Equity effects of pricing animal products and sugary products are mild and can be made progressive by returning the tax proceeds to citizens appropriately, for example via lump-sum or targeted transfers. Increasing the prices on products high in sugar leads to comparatively higher health benefits for lower-income households and younger generations. Targeted food subsidies for low-income suburban households in poorer EU countries can help overcome poverty as an obstacle to unhealthy diets, but will need to be complemented by social policy addressing poverty in general.
- The prominent placing of healthy and sustainable food options in the food environment can help to improve the consumers' food choices. The removal of unhealthy options from prominent places also has a positive effect. This applies to grocery shop settings as well as other food environments such as schools and canteens.
- The wider food environment has been studied in terms of accessibility of healthy food stores and fast-food restaurants. Results show the expected effects, but only for some subgroups of the populations. Literature shows that inequality in consumers' food environment is more pronounced in the US compared to other higher-income and middle-income countries. Recently, studies have begun to compare the food environment in urban vs rural settings, but overall more study of regional effects is needed.
- Improving the healthiness of the overall food environment and enhancing opportunities for healthy product choice can also be achieved through reformulating food products (reducing fat, salt or sugar). The introduction of plantbased alternatives helps to widen the choice faced by consumers to replace meat

products. Silent reformulation can help to overcome barriers to adoption, but policy experiences in the past have shown that reformulation based on voluntary agreements have a limited effect. The comprehensive or mandatory inclusion of the food industry, and clear targets, make this policy instrument more effective.

- To a large extent, policy has focused on information provision to consumers. A large variety of public and private labels and symbols exist, aimed at steering consumers towards healthier or more sustainable food choices. In general, the literature shows the effectiveness of health labels, but the impact is low to moderate. The literature also shows that only some consumers, not all, attend to labels. Warning labels, as introduced in Chile, show a comparatively better effect.
- For sustainability labels, the evidence is mixed. Consumers need to be interested and motivated to use these labels. Trust in label standards plays a prominent role in determining whether consumers adopt labels for guiding their purchase decision. Given the blossoming of new sustainability labels, a coherent approach to developing and establishing label schemes is key to building trust and avoiding confusion.
- The digital food environment offers new possibilities for personalised and dynamic feedback, enabling food 'swaps' where choices are replaced healthier or more sustainable options at checkout. Initial studies show promising results. However, it must be observed how sellers (retailers and restaurants) use these tools, and whether they help consumers to make equally cost-efficient choices.
- The social environment exerts a powerful influence on consumer choices. Past policy examples, such as the regulation of the tobacco industry, show that social norms can be shifted by using a multitude of policy instruments, from taxes and advertisement restrictions to smoking bans in public spaces. In the food environment, the effect of peer influence has been shown to be successful in improving fruit and vegetable intake and limiting fast food consumption. There is evidence that influence and social norms lead to reduced meat consumption.
- A key message of this review is that interventions can have dynamic effects. They do not lead only to consumer responses, but also to reactions by other actors in the food environment, such as processors and retailers. Food labels, advertisement bans and taxes may lead to reformulations or competitive pricing strategies that can enhance or limit the desired outcomes in consumer reactions. Therefore, the development of any intervention package must consider possible reactions.

Chapter 5. Selected examples of policies to promote healthy and sustainable food consumption

Policies aimed at improving healthy and sustainable food consumption apply one or a set of policy instruments intended to change the behaviour of individual consumers, or collective consumers such as public institutions. Essentially, policies aim to create healthy and sustainable diets, by providing information or economic incentives for sustainable and healthy food consumption, or creating or removing barriers. Hence, public policies relate to the M (motivation) and O (opportunity) components of the COM-B framework (see 'Theoretical framework to identify barriers in sustainable and healthy food consumption', p. 69). For instance, by changing the price relations between sustainable and unsustainable food products, public policies can motivate consumers to choose the sustainable options by creating economic incentives and disincentives.

The selection and combination of policy instruments play an important role in relation to whether or not policy will succeed. According to May (2003), public policies "set forth courses of action for addressing problems or for providing goods and services to segments of society". Policies can come in different forms, but they "typically contain a set of intentions or goals, a mix of instruments or means for accomplishing the intentions, a designation of governmental or nongovernmental entities charged with carrying out the intentions, and an allocation of resources for the requisite tasks" (May, 2003). While goals can independently have an influence on the behaviour of the intended target groups if clearly formulated and considered legitimate within those groups, they are often problematic as steering tools. Though ideally policy goals should be clearly defined and set a clear direction, this is often not how the world of politics works. To enable the formation of a majority coalition to adopt policies, goals can be vaguely formulated to paper over disagreements on policy focus, have mainly symbolic meaning, suffer from inconsistency, or blur hidden agendas.

Hence, to assess policy's impact on behaviour, the analytical focus is usually directed at policy instruments. Policy instruments can be defined as "the set of techniques by which governmental authorities wield their power in attempting to ensure support and effect or prevent social changes" (Vedung, 1998, p. 21). While this definition is fairly broad, it does include the essentials of policy instruments and can encompass policy pursued by governments as well as private bodies.

Hood's classic policy instrument typology is useful to consider the instruments that governing bodies can use to affect the behaviour of more or less well-defined target groups. His typology distinguishes between (Hood, 1983):

- instruments based upon information (informative instruments)
- instruments that use authority (regulatory or administrative instruments)
- instruments that are associated with treasure (economic instruments)
- organisation

The distinction between different instruments is based on the motivational mechanisms that each instrument uses to induce individuals or organisations to change behaviour in the desired direction or maintain an already existing and valued behaviour which could potentially change if people are not motivated to maintain it. The four basic types of instruments use different motivational mechanisms. Informative instruments apply learning or persuasion to motivate people to change or maintain behaviour. Regulatory instruments use rules to bring about compliance, backed by authority to apply force if necessary. Economic instruments use monetary incentives to motivate people to behave in a particular way by rewarding desired behaviour or increasing the cost of undesired behaviour. Finally, organisation uses 'architecture', i.e., building or shaping organisations in a way that induces people to behave in particular ways (Hood, 1983; Vedung, 1998). Organisation is often associated with the use of one or more of the other instrument types. For instance, implementing an organic certification and labelling scheme requires an organisation to certify and monitor compliance with the standards. This policy instrument typology has proved impressively robust over forty years and continues to be well used (Hood & Margetts, 2007; Howlett et al., 2020). Although a more fine-grained categorisation of policy instruments can often be useful in empirical research, it is exactly the parsimoniousness and simplicity of the typology which has made it attractive for many policy analysts over time.

Policies can rely on a single instrument or a combination of instruments. While there is a tendency in the academic literature to focus on single instruments, in practice many policies use a combination of policy instruments. The instrument mix can be deliberately designed when policymakers have doubts that a single instrument is sufficiently effective at bringing about a desired impact. However, often instrument mixes are the result of an evolutionary process in which instruments have been added to existing instruments to address new concerns. In such situations, policy tends to evolve without having a unifying overall logic or design principle (Howlett & Rayner, 2007; Kern & Howlett, 2009).

This chapter reviews policies adopted (or under consideration) by national governments to address three of the four key impact areas highlighted in section 3.2, p. 70. These are:

reducing consumption of unhealthy foods, focusing on sugar (below)

Selected examples of policies

- reducing consumption of meat and animal products (p. 123)
- increasing the consumption of organic food (p. 127)

The policy reviews in the following sections address the barriers to policy intervention, which instruments have been or can be selected for the policy mix, how the instruments have been used and with what effect, as well as the conditions under which they were implemented, where this is relevant for understanding implementation of the policy mixes.

5.1. Sugar reduction policies

The World Health Organization (WHO) has formulated recommendations to reduce free sugars intake throughout the life course. In both adults and children, WHO recommends reducing intake of free sugars to less than 10% of total energy intake, and less than 5% as a conditional recommendation.⁵² A high intake of free sugars is of concern because of its association with poor dietary quality, obesity and risk of noncommunicable diseases (WHO, 1990, 2003).

The majority of national food consumption surveys conducted among European populations provide information on intake of added sugars, rather than free sugars (Rippin et al., 2017). Of the European countries that reported adult daily sugar intake, all reported exceeding 5% of total energy intake, although only Estonian and Finnish women were over the 10% recommendation (Rippin et al., 2017). Regarding children and adolescents, in European countries where data on sugar intake were available, most children, particularly those aged 10 years or over, consumed more than 10% of their total energy intake from added sugars (Rippin et al., 2019). Beverages and sweet products, including confectionery, chocolates, cakes and biscuits make the most significant contribution to sugar intake across these age groups (Azaïs-Braesco et al., 2017; Graffe et al., 2020).

The WHO best-buys action plan lists the most cost-effective policies that can contribute to reducing free sugars intake and prevent diet-related noncommunicable diseases. These include economic instruments such as effective taxation on sugar-sweetened beverages (SSBs), informative instruments such as food labelling, and regulatory instruments such as marketing restrictions, all aimed at reducing intake of nutrients of concern, including sugars (WHO, 2017).

⁵² https://www.who.int/news/item/04-03-2015-who-calls-on-countries-to-reduce-sugars-intakeamong-adults-and-children

A recent literature review presents a framework, illustrating various types of policies that have been used to reduce sugar and in particular SSB consumption. It categorises policies into four categories (Krieger et al., 2021):

- Financial policies increase the price of SSBs relative to healthier beverages and include taxes, restrictions on price and volume promotions (for example, buy-one-get-one-free offers), and incentives for the purchase of unsweetened beverages.
- Information policies aim to reduce the public's exposure, especially that of children, to marketing of SSBs or to increase awareness of the health risks that they pose. These policies include front-of-pack and advertising warning labels and marketing restrictions (for example, no advertising during children's television programming).
- Default policies make the choice of a healthy beverage easier, such as requiring a healthy drink in children's meals in restaurants.
- Availability policies decrease access to SSBs, or reduce portion sizes or sugar content in products. They include beverage procurement (for example, purchase or placement of beverages within various settings) and healthy checkout aisle policies (for example, lanes in grocery shops that display healthier options) (Krieger et al., 2021).

Below, we focus on both economic instruments and informative and regulatory instruments to reduce sugar intake for which most evidence is available to date. While food reformulation has been shown to be successful for salt and trans-fatty acids reduction (meaning that people usually accept, buy and consume reformulated products, resulting in an overall improvement in the nutritional composition of food purchases), similar evidence is not yet available for energy and sugar. Note that mandatory standards are generally more effective than voluntary actions, and that multi-component strategies that include food reformulation are more promising for improving population diets than reformulation alone (Gressier et al., 2021).

Economic instruments

Most fiscal policies to date related to sugar reduction have focused on SSBs. To date, over 45 countries and other jurisdictions (i.e. cities) have introduced SSB taxes, ranging from small taxes (3%–5%) to impactful taxes (50%–75%). There are generally two types of SSB taxes to consider:

- a **sales tax** is levied on consumers, based on the price of an SSB
- an excise tax is a tax levied further up the supply chain, meaning that manufacturers, distributors, or retailers are the taxed entities, instead of consumers. The impact of this on the consumer price will depend on how much of the tax is passed through to the consumer

SSB taxes can also take different forms, as:

- a volumetric tax, which is levied on each litre of sugary beverages sold, regardless of sugar content
- a sugar content tax, which is levied on each gram of sugar over a minimum threshold
- a tiered tax, which sorts sugary drinks into different tiers according to their sugar concentrations and taxes products with high sugar concentrations at a higher rate

SSB taxes may vary in additional characteristics such as the definition of an SSB, the tax rate, and the allocation of revenues.

In their paper, Popkin and Ng (2022) show that taxes focused on sugar content have a stronger impact on the sugar content available in beverages, because they encourage reformulation. This has been seen in the UK's tiered soft drink levy based on sugar content per 100 millilitres of beverage, and in South Africa with the health promotion levy (Bandy et al., 2020; Essman et al., 2021; Scarborough et al., 2020; Stacey et al., 2021). Volume-based taxes can provide greater tax revenues, but are less likely to encourage reformulation.

In general, lower-income households and individuals are more likely to reduce their purchases in response to a tax, and therefore stand to gain more long-term health and monetary benefits (Barrientos-Gutierrez et al., 2017; Basto-Abreu et al., 2018; Sánchez-Romero et al., 2016; Torres-Álvarez et al., 2020). This was the case in evaluations of the Mexican nonessential food and SSB taxes (Batis et al., 2016; Colchero et al., 2016; Hernández-F et al., 2019; Ng et al., 2019; Sánchez-Romero et al., 2016) as well as South Africa's sugary drink levy (Essman et al., 2021). Assessing the health impact of SSB taxes is more difficult due to the long lag between tax implementation and potential health effects, and the multiple factors that contribute to health conditions associated with SSBs. However, microsimulation models predict significant reductions in obesity and cardiovascular diseases (Long, Gortmaker, et al., 2015; Peñalvo et al., 2017; Sánchez-Romero et al., 2016). A recent study suggests that the UK's soft drinks levy was associated with decreased prevalence of obesity in girls aged 10–11, 19 months after implementation, with the greatest differences in those living in deprived areas (Rogers et al., 2023). For further evidence from assessments in applied economics on taxes on SSB, see section 4.1, p. 85.

In the European context, the taxes in France, Hungary, Ireland, Latvia, Norway, Portugal and the United Kingdom were explicitly designed to reduce consumption of SSBs and/or sugar. Furthermore, taxes were designed to achieve an explicit objective of reformulation, through differential tax rates with thresholds based on sugar content, in Hungary, Latvia, the United Kingdom, Finland and France (WHO Regional Office for Europe, 2022c). Several WHO documents are available to support member states in implementing fiscal policies (WHO, 2022; WHO Regional Office for Europe, 2022b). The most common barrier to the implementation of well-designed health taxes to reduce sugar intake in Europe is the opposition of the food industry (Campbell et al., 2020; Lauber et al., 2022; Tselengidis & Östergren, 2019).

Informative and regulatory instruments

Popkin and Ng (2022) point to clear evidence that that industry self-regulation of marketing to children has not been effective (Chambers et al., 2015; Galbraith-Emami & Lobstein, 2013). They present the Chilean example, where the government introduced a marketing ban on children-focused marketing of warning labelled foods during children's television shows, and subsequently on all marketing of those foods between the times of 6:00 and 22:00. Evaluations of these bans showed a decrease of ads for foods high in energy, saturated fat, sugar, or sodium, from 41.9% of all ads before the regulations to 14.8% after implementation. During the first year of the law, this resulted in a 44.0% decrease in exposure to 'high-in' food advertisements for children, and a 58.0% decrease for adolescents (Correa et al., 2020).

In London, since February 2019, all advertising for high fat, sugar and salt products has been banned across the London transport network. A recent evaluation found an association between the implementation of restrictions on outdoor advertising of such products and relative reductions in energy, sugar, and fat from these products. Relative reductions in purchases of sugar (80.7 g, 95% CI 41.4 to 120.1) from these products were also observed (Yau et al., 2022). A process evaluation of the design and implementation of this local policy highlighted some practical and policy challenges, such as translating the concept of 'junk food' into operational policy, reported uneven impacts across industry stakeholders, balancing health and financial impacts and the perceived influence of political motivations (Meiksin et al., 2022; see section 4.4, p. 100, for detailed discussion of the evidence).

In general, policies addressing unhealthy food marketing need to be strengthened in many ways in order to be more effective (Sing & Backholer, 2023):

- increasing the definition of a child to 18 years
- broadening the focus from 'child-directed' marketing to all marketing that children are exposed to
- including multiple settings, media and techniques within regulations
- using evidence-based food classification systems
- strengthening monitoring and enforcement systems

The following discussion focuses on SSB reduction policies, and is largely based on Krieger et al. (2021).

Selected examples of policies

SSB warning labels on outdoor advertising or on beverage cans can incentivise the industry to reformulate products and provide easy-to-understand information to consumers. There are two main types of warning labels: nutrient warnings (that indicate a high amount of sugar) and health warnings (that describe health harms of SSBs). One study that simulated a mandatory SSB health warning policy in the US concluded that such a policy would reduce the average SSB intake by 25.3 calories per day and the total energy intake by 31.2 calories per day, thus reducing obesity prevalence by 3.1% age points over 5 years. This study also found that this would particularly benefit racial and ethnic minority adults and lower-income adults (Grummon et al., 2019).

A real-world evaluation of a warning label policy became possible in 2012 in Chile, when the country adopted a food labelling and marketing law which mandated warnings for products high in sugar, saturated fats, sodium or energy, based on nutrient threshold values. This was the first national policy to regulate front-of-pack warning labels, to restrict child-directed marketing, and to ban sales of foods and beverages high in added sugars, sodium, or saturated fats in schools. The law states that prepackaged foods with at least one warning label cannot be promoted to children under 14 years, cannot be sold in schools and nurseries, and cannot be provided in school and nursery food programmes. The warning label was developed based on guantitative and gualitative studies conducted with different groups in the population. The warning label performed the best among a set of labels in terms of visibility, understanding and intention to purchase (Reyes et al., 2019). One year after the implementation of this law, purchases with such labels fell by nearly 24%. Importantly, households where the head of household was lower-educated heads showed similar absolute reductions in sugary drinks with warning labels to households where the head of household was more highly educated, demonstrating that these warning labels were understood and did not widen disparities (Taillie et al., 2020). In addition, manufacturers reformulated their products substantially in response to the implementation of the policy (Reyes et al., 2020).

A recent study evaluated changes in the calorie and sugar content of food and beverage purchases after the first phase of implementation of this law (Taillie et al., 2021). Compared with the counterfactual scenario, overall calories purchased declined by 3.5%, or 16.4 kcal per person per day (95% CI -27.3 to -5.6; p=0.0031). Overall sugar declined by 10.2%, or 11.5 kcal per person per day (-14.6 to -8.4; p<0.0001). Among purchases of products with a warning label, relative to the counterfactual scenario, there were substantial declines of 23.8% or -49.4 kcal purchased per person per day (95% CI -55.1 to -43.7; p<0.0001) and 26.7% or -20.7 kcal purchased per person per day (-23.4 to -18.1; p<0.0001). Greater changes might reasonably be anticipated after the implementation of the second and third phases of the law (Taillie et al., 2021).

Generally, in countries that have adopted front-of-pack nutrition labels to date, the main implementation challenges have included identification of covered products, trade

considerations, and the time needed by industry to change packaging and reformulate products (Corvalán et al., 2019). The involvement of civil society organisations was found vital to provide support when higher limits or restrictions on the label thresholds were proposed, as well as when opposing lower limits or less strict requirements (FAO & PAHO, 2017; WCRF, 2019).

Some evidence is now also emerging on industries' attempts to influence policy areas that they perceive as detrimental to their interests. This includes misrepresenting scientific evidence; developing public relations campaigns; mounting front organisations; lobbying; shifting the blame away from their products; and highlighting positive corporate actions (Collin & Hill, 2019; Lauber et al., 2021). However, the number of sugar reduction policies has increased and success factors have started to emerge. Successful campaigns have:

increased public awareness of the health and equity issues associated with SSBs, built strong multisector coalitions, fielded effective grassroots initiatives, launched effective communications strategies to control the framing of the policy debate early on, articulated clearly the purpose of the policy and who will benefit, secured the support of elected officials, and arranged adequate funding

(Krieger et al., 2021)

In a review, von Philipsborn et al. (2019) study the effect of environmental interventions excluding taxation aimed at reducing the consumption of SSBs. As successful, they identify easily understood labels such as traffic lights; limiting availability of SSBs in schools; children's menus including healthier beverages by default; promoting healthier beverages in supermarkets; community campaigns focused on SSBs; and measures that improve the availability of low-calorie beverages at home.

No single policy will reduce sugar and sugary drinks consumption to recommended levels. Therefore, multiple policies at different levels, such as Chile's food labelling and marketing law, along with public awareness campaigns, should be integrated to leverage synergies, reinforce healthy norms, and maximise impact.

5.2. Meat reduction policies

Meaningful reductions of animal product consumption required to meet environmental goals of the EU are not to be expected without significant policy intervention (see Chapter 4, p. 83). Following the key messages in Chapter 2, p. 52, this section focuses on red and processed meat (but not dairy) consumption. However, there are so far very few real-world policy examples of specifically targeting the reduction of meat consumption (WBAE, 2020). This section first collects the reasons why significant policy interventions to reduce the consumption of animal products are absent in industrialised countries. Second, it highlights a number of examples of implemented or discussed

Selected examples of policies

policies targeting the reduction of meat consumption from relevant jurisdictions. Third, it summarises findings on options for raising public support for effective meat reduction policies. Reducing meat consumption to achieve more sustainable and healthy diets relates more to lack of motivation and lack of opportunity, but less to lack of capability (see section 3.3, p. 80).

As much meat consumption is in fact habitual and due to preference learning (Hawkes et al., 2015), this section looks especially into price interventions (economic instruments), public procurement (organisation) and public discourse (informative instruments) to shift behaviour. It looks at meat consumption of median consumers in EU nations; the specific role of the share of low-meat consumers is beyond scope. Unintended side-effects especially of price interventions, such as a reduction of red and processed meat leading to more consumption of poultry and its effect on animal welfare, are discussed in section 4.1, p. 85.

Barriers for policies reducing meat consumption

There are several political barriers to reducing meat consumption. Governance challenges to decrease red and processed meat consumption include industry lobbying, shaping the public discourse, market power, government-industry dependence and trade policy conflicts (Sievert et al., 2021). As one example, in 2019, only four out of the 35 largest meat and dairy companies stated that they have an explicit commitment to net zero greenhouse gas emissions by 2050 (Lazarus et al., 2021). In accordance with the focus of the report, this section focuses on the role of the consumers and the public in establishing political feasibility despite these barriers.

Notably, there is evidence that meat industry representative bodies have obstructed public discourse on the negative impacts from meat consumption in rich countries (Clare et al., 2022; Sievert, Lawrence, Parker, Russell, et al., 2022). Parallels have been observed between tactics used by major players in the meat sector (evidenced by documents from meat industry representative bodies) and the strategies of other politically powerful industries on which regulation was imposed due to desirable social transitions, such as the tobacco industry (Brownell & Warner, 2009) and the fossil fuel industry (Clare et al., 2022).

Clare et al. (2022) find three main strategies used by meat industry interest groups to shape public debates in the UK:

They emphasise the consumer's free choice, in particular with respect to health effects. This can also come with framing the reduction of meat consumption as an elitist "vegan agenda" (Sievert, Lawrence, Parker, Russell, et al., 2022) in UK, USA, Australia and New Zealand.

- They portray the industry as part of the solution, rather than the problem. This emphasises voluntary self-regulation, the (limited) possibilities for better practices within the production processes, and (alleged) positive side effects regarding carbon sequestration and biodiversity preservation of some of the sector's activities.
- They influence public discussion by deflecting attention from the need to reduce animal products by consistently using the following four main framings: "Iharmfulness of meat consumption is] still open for debate", "most people need not worry" (only a small subgroup of consumers are threatened by health risks), "keep eating meat to be healthy", "no need to cut down to be green" (Clare et al., 2022).

Discourse analysis of reactions to proposals to reduce meat consumption reveals that populist and anti-elitist sentiments are also barriers to the adoption of low-meat diets (Michielsen & van der Horst, 2022), and that discourse on meat reduction among the public is more polarised than among food system stakeholders (Sievert, Lawrence, Parker, & Baker, 2022). Even though there are more 'flexitarians' in EU populations than in the past, the (erroneous) lay beliefs that eating lots of meat is natural and plant-based meals are unenjoyable still constitute a significant barrier (Perez-Cueto et al., 2022). Moreover, sociological and cultural studies have identified a connection between norms of masculinity and meat consumption (Nath, 2010; Rozin et al., 2012). This may contribute to vegetarianism being more widespread amongst females (Räty & Carlsson-Kanyama, 2010; Rosenfeld & Tomiyama, 2021). However, there may be potential in EU countries for the younger generation to be especially receptive to publicly support meat reduction. Schulze et al. (2022) find that, among young Germans (15–29 years), opinion leaders have high approval for policy interventions promoting sustainable diets.

Examples of meat consumption regulation

Despite the efforts of industry interest groups to obstruct animal product regulation, there is nevertheless a number of prominent discussions about policies for reducing meat consumption in an EU context. In Germany and the Netherlands, forms of taxes on meat are current government proposals, notably in Germany as an 'animal welfare levy' (redistributing the tax proceeds to farmers to increase rearing standards in Germany). New Zealand is introducing some form of emissions pricing on agricultural products from 2025, as the sector is responsible for 50% of the country's greenhouse gas emissions.⁵³ The price will apply to manufacturers rather than individual farmers and is opposed by the farming lobby. At the time of writing, the level of the tax has not yet been decided.

Between 2011 and 2013, Denmark had a tax on saturated fat tax which reduced consumption of processed beef and cream (Jensen et al., 2016). The tax was repealed due to political pressure from stakeholders in Danish society. While public health

⁵³ https://environment.govt.nz/publications/aotearoa-new-zealands-first-emissions-reduction-plan/

research was considered in the introduction of the tax (as part of a broader tax reform), it played no role in the discussion leading to its repeal (Jensen & Smed, 2018).

Changes to relative prices reducing meat consumption to some extent can also be brought about via differentiating the rate of value-added taxes (VAT). In Italy and Spain, basic foods such as fruits, vegetables and dairy, but not meat, are taxed at the very low VAT rate of 4%. To counteract inflation, Spain has currently reduced the VAT on basic products to 0%, but this does not include meat. Early results shows high pass-through of this tax reduction to consumers (De Amores Hernandez et al., 2023). In the EU, it has been possible for member states to reduce VAT on fruits and vegetables to zero since 2022. This option is currently under discussion in a number of EU member states including Belgium, Germany and the Netherlands, and very low VAT rates on vegetables also apply in Cyprus, Latvia and Luxembourg and Poland.

There are many legislative efforts across EU countries to curtail meat consumption at the local level. Predominantly vegetarian-only or vegetarian default meals in public canteens (sometimes on specific days) have been legislated by municipal governments, with examples including Gent, Freiburg, Grenoble and Helsinki (De Keyzer et al., 2012; Lombardini & Lankoski, 2013). There is also a meat advertisement ban in Haarlem (Netherlands).⁵⁴ France enacted a mandatory vegetarian option.⁵⁵ Conversely, livestock-related targets in public procurement schemes are so far omitted at national or EU levels (over other goals such as switching to organic produce; see section 5.3, p. 127), although decreasing meat consumption would actually be a primary objective for improved public health and environmental goals.

Green public procurement schemes in the EU have so far predominantly focused on the production of food packaging, and only to a lesser extent on goals for meal composition (Neto & Gama Caldas, 2018). However, the emissions-reduction potential of switching towards more plant-based diets is far greater than that of a switch to organic foods (see Chapter 2 and section 4.1, p. 85). For example, in a modelling study for Italian school meals (based on life-cycle analysis), the emissions reductions potential of procuring 100% organic foods is in the order of 11%–15%, while a switch to vegetarian meals results in a 20%–32% decrease (Cerutti et al., 2018). Selected case studies across the EU (for example, for school canteens) also show that reduction in meat can further balance out the higher cost of procuring a larger share of organic products (European Commission, 2014; European Committee of the Regions & Soldi, 2018). Reisch et al. (2013) therefore suggest that the lever of sustainable public procurement is underutilised, due to the frequent omission of concrete goals for reducing meat consumption.

Finally, a limited nascent number of policy initiatives seek to make meat replacement products more competitive by subsidising learning technology (Treich, 2021). In Israel and the Netherlands

⁵⁴ https://haarlem.nl/haarlem-wil-geen-reclame-meer-voor-vleesproducten

⁵⁵ https://agriculture.gouv.fr/egalim-depuis-le-1er-novembre-2019-un-menu-vegetarien-par-semaine-danstoutes-les-cantines

as of 2022, government funding sponsors companies, research institutes and research foundations for the scaling up and education on cultured meat.

Options for raising public support for effective meat reduction policies

On overcoming some of the political barriers to regulation for reduction of meat consumption, we focus on the consumer side.

An important and especially policy-relevant barrier for tools to reduce meat consumption is their support from the public. A recent body of literature studies how policies can be designed to reduce meat consumption while at the same time increasing public support (Fesenfeld et al., 2021; Pechey et al., 2022; Perino & Schwickert, 2023). Fesenfeld et al. (2021) conclude that strategic framing alone is not sufficient to increase support for ambitious environmental policies which come at a cost to citizens, such as price increases on meat. Nevertheless, Perino and Schwickert (2023) show that, in Germany, animal welfare is a more popular reason for raising prices on meat than climate change. Fesenfeld et al. (2020) show that policy packaging (i.e. combining various food policies) can increase public support, while revenue recycling is not as effective in the food sector as it may be for carbon taxation in other sectors (Klenert et al., 2018). Instead, combining environmental taxes with other policy instruments perceived as beneficial has greater effects on support, including raising animal welfare standards and reducing agricultural subsidies (Fesenfeld et al., 2020). Furthermore, naming policies can sometimes matter: on meat replacement products, consumers prefer to name it 'cultured' over 'lab-grown' or 'artificial' (Asioli et al., 2022).

5.3. Promoting organic food consumption

Under Pillar II of the EU Common Agricultural Policy, support for organic agriculture is one of the activities which member states are responsible for designing. The way in which EU member states have shaped national organic agriculture policies varies, and policy impact has been very different. Some member states have converted a considerable share of their farmland to organic production, while others have been less successful in this regard.

The standard organic policy model across most European countries is to provide farm subsidies to encourage farmers to convert to and maintain organic farming. The assumption on which this policy model is based is that increased supply will create its own demand and this will lead to a growing organic food market. A more advanced policy model, the active market-development policy model, combines a variety of policy instruments to stimulate supply and demand for organic food products simultaneously (Daugbjerg & Sønderskov, 2012; Halpin et al., 2011). Policy instruments to increase

Selected examples of policies

organic production already exist in the form of conversion subsidies to compensate farmers for the costs during the two-year transition period, as well as various subsidies to maintain organic farming. These can be combined with informative instruments aimed at encouraging farmers to farm organically.

When stimulating demand for organic food, governments can use all four types of instruments (Daugbjerg & Sønderskov, 2012; see also p. 116). Though governments can concentrate on the use of one instrument, a significant policy impact is more likely if several instruments are applied in combination:

- Informative instruments can be used to create awareness among consumers of the availability of organic food and persuade them to buy organic.
- Regulatory instruments can also be used. The EU has adopted baseline standards that producers are required to meet in order to market their food as organic. Furthermore, though there are no examples in practice, governments could potentially require that food services in the public sector must, as a minimum, ensure that a certain share of the food served is organic.
- Economic instruments can be used to subsidise organic marketing campaigns or other activities related to these, subsidise various training activities for kitchen staff or lower taxes on organic food.
- Organisation can also be used as an instrument to stimulate demand for organic food indirectly. For instance, public procurement services for advising kitchen managers and procurement bodies in the public sector on how to procure organic food can be established.

Ideally, the instruments should be combined in a way that will mutually reinforce each other.

Several countries have local, regional or national experiences with using sustainability criteria, including organic food products, in their public food procurement programmes (Molin et al., 2021; Neto & Gama Caldas, 2018). However, while overview articles identify organic food as an environmental sustainability criteria in procurement programmes, they do not describe best practices. What is missing are studies of national policy programmes to promote consumption of organic food.

The policy studies literature has called for more studies on policy success to learn how governments may succeed in addressing various types of policy challenges. Considering that studies of policy failure are disproportionately represented in the policy studies literature, the argument of the recent policy success literature is that there are lessons to be learned from studying policy successes that have hitherto been missed (Compton & 't Hart, 2019; de la Porte et al., 2022; Douglas et al., 2021; McConnell, 2010). In particular, four countries in the EU have been successful in promoting organic food consumption:

Denmark, Austria, Sweden and Luxembourg. In relation to the share of food retail sales that are organic, Denmark has consistently topped the list of countries since the mid-1990s when comparable international sales data became available (Willer & Yussefi, 2000). By 2020, Denmark was still leading, with 13% of the food sold by retailers and online outlets being organic. With 26.5% of agricultural land being farmed organically (Schlatter et al., 2022), Austria is often highlighted as the organic forerunner in the EU in terms of land conversion. On the consumption side, Austria is also doing well, reaching 11.3% in 2020. Luxembourg reached 9.1%. Sweden was a slow starter but has caught up and now ranks fourth in the EU with 8.7% (Schlatter et al., 2022). It should be noted, however, that much of the organic production in Sweden is livestock production: approximately 90% of the organic land area is devoted to feed production (Cederberg et al., 2011). Considering the need to switch to more plant-based diets, for organic production to be a part of the solution these systems must also deliver legumes, cereals, fruits and vegetables.

Austria initially relied on civil society and private retailers to provide outlets for organic produce. Producer-consumer collaboration was initiated to help organic farmers market their produce and to connect to consumer segments. Later, food retailers started marketing organic food products and are now the dominant outlet. The Austrian government's role was to support farm conversion and reframing the role of Austrian agriculture from a provider of cheap food to a provider of quality food (Schermer, 2015).

Denmark and Sweden have intervened more directly to promote organic consumption. Governments in Denmark and Sweden have launched policy initiatives directly to promote organic food consumption. In addition to encouraging the supply of organic produce, it was realised in both countries that to expand the organic agricultural sector, it was necessary to stimulate demand for organic food. However, the timing of the demand side initiatives in the two countries was quite different. Further, while both countries have achieved comparatively high levels of organic consumption, they arrived there through different policy strategies. Policy initiatives directed at increasing household consumption has been the main policy strategy in Denmark. It was later supplemented with a public procurement programme. Increasing organic food procurement in public sector institutions has been the key driver in Sweden.

Policy instruments to increase household demand for organic food in Denmark

Starting with the initial phases in the late 1980s, the Danish organic food and farming policy took shape as an active market-development policy (Schvartzman, 2012). The then Ministry of Agriculture argued that it was necessary to develop organic farming to cater for consumers with preferences for sustainable food (Daugbjerg & Møller, 2010; Schvartzman, 2012). Therefore, consumer demand was seen as a driver for the

Selected examples of policies

development of the organic food sector. As was stated in 1999 by the Organic Food Council, established under the Ministry of Agriculture and including a number of stakeholders: "The underlying logic is that the organic farming sector can best be developed in accordance with the market which is created by the demand for organic produce. Thus, conversion is based on voluntary action and positive motivation" (Strukturdirektoratet, 1999, p. 16, author's translation). This resulted in a policy design in which demand-side instruments played an important role. The government used an economic instrument, by providing funding to promote market development such as subsidising product development, developing sales concepts and implementing marketing campaigns (Schvartzman, 2012).

Implementing the instrument of subsidised marketing required that capacities in organic marketing were developed. While government has administrative capacity to implement organic support schemes directed at farmers, or can relatively easily develop it, it is likely to lack expertise and experience in working with retailers on marketing. Furthermore, competition law may prevent government agencies from engaging directly with retailers in marketing efforts. Therefore, implementing demand-side policy instruments to increase private organic consumption, in particular those requiring collaboration with retailers, may require partnership with private bodies. Recognising this early on, the Swedish government encouraged the organic association to develop capacity in organic food marketing.

Most demand-side measures were initially implemented by a private body, the Organic Service Centre, which later merged with the association for organic farming and became Organic Denmark. The implementation of marketing measures required coordination between processors and distributors and direct engagement with retailers in sales campaigns focused on specific organic products. These strong coordination activities were a significant factor in promoting sales of organic products on a large scale. Importantly, they brought together and created trust amongst market actors who were essential for expanding the organic market (Daugbjerg & Schvartzman, 2022; Schvartzman, 2012). While the end target group at which policy was aimed were consumers, the policy worked through private intermediaries with Organic Denmark in a coordinating role and the food retail sector in an indirect role as facilitator of promoting organic food.

Using public procurement to promote consumption of organic food in Denmark and Sweden

In government efforts to promote organic food consumption through public procurement, the target group is public sector institutions rather than individual citizens. While increased public sector procurement has a direct positive impact on organic food consumption, there can be a multiplier effect if people having positive experiences with eating organic meals in the public sector feel inspired to consume more organic food in their households. A key challenge in promoting public procurement is to develop policy mixes that motivate kitchens in the public sector to introduce and increase their use of organic food products in the meals that they serve. There are few examples of national organic food procurement programmes and very few studies analysing best policy practices in this field (Lindström et al., 2022).

An important barrier to public procurement of organic food is the financial cost increase caused by higher prices for organic food. The higher the share of organic food procured, the higher the potential cost increase (Lindström et al., 2022; Nuutila et al., 2019). However, menu planning can address this issue by substituting some meat with vegetables without compromising the nutritional quality of the food (Risku-Norja & Løes, 2017). Nonetheless, a high organic procurement share may result in some cost increase (Lindström et al., 2022). Availability of organic food products for institutional kitchens is another practical barrier (Risku-Norja & Løes, 2017). As will be shown below, this barrier has to a large extent been overcome in Denmark and Sweden by engaging food wholesalers and including requirements to offer a range of organic food products in procurement contracts. Further, there may be a lack of legal knowledge at local or kitchen level on how to create a procurement contract for organic food and how much flexibility they involve for both parties (Krogh et al., 2013). The policy effort to promote organic food sales in the private sector in Denmark was supplemented by a public procurement programme for organic food in 2012 that was in force until 2014 (Ministeriet for Fødevarer, 2012). The programme was revived in 2022 and will run until 2025. This time it is administered by the semi-public Foundation for Organic Agriculture. Apart from broadening the programme to additionally cover canteens in private companies, it is a copy of the previous programme. The original programme relied on economic instruments backed by informative instruments, whereas the current programme is based purely on economic instruments. Based on assessment of applications for specific kitchen conversion projects, the previous and the current programme offer economic subsidies for training of and advisory services for kitchen staff. Furthermore, both programmes encourage the kitchens to become certified under the government's organic cuisine label, though this is not a mandatory requirement (Daugbjerg, 2022).

The original programme produced a significant impact. The kitchens participating in the original procurement programme increased their procurement of organic food products by 24% over a two-and-a-half year period and certification under the organic cuisine level increased significantly. Prior to participating in the procurement programme, 57% of kitchens qualified for certification under the organic cuisine label (but not all of them were certified). By 2015, the share of kitchens qualifying for certification under the label had increased to 90% (Sørensen et al., 2016). Though public procurement has remained a secondary measure in the Danish demand-side policy strategy, it did boost the public sector's demand for organic food products and increased the organic certification of

kitchens in public-sector institutions. In 2021, 29% of the food procured by Danish public sector institutions was organic (data from Statistics Denmark, provided by the Danish Veterinary and Food Administration, 2022).

Similarly to the programme aimed at household consumption of organic food, the public procurement programme relied on partnering up with private actors. It was led by the Danish Veterinary and Food Administration, with the Agricultural Agency administering the funds. It was centrally governed, with national government agencies collaborating closely with mainly national private associations, most notably Organic Denmark, in designing and implementing the policy instruments. Organic Denmark represents the organic industry, including farmers, consumers, retailers and food processors. Further, actors with specific expertise in organic food procurement and kitchen management, most notably, the Copenhagen House of Food, ⁵⁶ also took part in relevant parts of the process (Daugbjerg, 2022).

Engaging food wholesalers was identified as an important key to the success of the procurement programme. It was essential that they offered a range of organic products (Fødevarestyrelsen, 2012). Organic Denmark played an important role in activating wholesalers. It organised short courses aimed at sales agents and managers to motivate and assist them in developing strategies for offering organic products. The effort to engage food wholesalers proved successful, as it was possible for kitchens to source all necessary organic food products from wholesalers already in early 2013 (Daugbjerg, 2022).

The timing and the way in which organic food consumption has been promoted by government policy was quite different in Sweden. Whilst Danish policy has mainly concentrated on promoting household consumption from the initial phases of the organic food policy in the late 1980s, it was not until 2006 that the Swedish government directed attention to the demand-side, by adopting an organic consumption goal for the public sector. Up until then, the organic farming policy was focused on reaching the goal to convert 20% of farmland to organic farming (Jordbruksdepartementet, 2006), which it has achieved (Schlatter et al., 2022).

At national level, the Swedish procurement programme for organic food is essentially limited to a statement of a goal. This has remained the main national measure to increase consumption of organic food. Policy instruments supporting the goal are few and very modest in terms of funding (Burman et al., 2020; Miljö — och jordbruksutskottet, 2010). The original goal stated that organic food procurement in the public sector should increase from around 6% in 2006 to 25% by 2010 (Jordbruksdepartementet, 2006). Though procurement increased, the 25% goal was not reached until 2013 (Koch et al., 2018). In 2017, it was decided that by 2030, 60% of the food procured in the public sector

⁵⁶ https://kbh-madhus.webflow.io/english/aboutus

should be certified organic (Näringsdepartementet, 2017). The Swedish programme for procurement of organic food has been successful despite few and very modestly funded national policy instruments to back the goal. In 2021, 38% of the food procured by Swedish public sector institutions was organic, which was slightly lower than the 39% in the pre-corona year 2019 (Ekomatcentrum, 2022b).

Reaching the goal relied on local and regional governments introducing their own consumption goals and measures to increase organic food procurement within their kitchens and canteens. It was voluntary to set local/regional goals. Local and regional government found the national goals helpful in increasing organic food procurement (Miljö — och jordbruksutskottet, 2010) 85% of the Swedish municipalities had adopted goals for organic consumption by 2013 (Riksrevisjonen, 2016). The share peaked with 88% in 2017 but decreased to 67% in 2020. The Organic Food Centre (Ekomatcentrum) argues that the main reason for this decline was an increased procurement focus on local - and Swedish-produced food (Ekomatcentrum, 2022a). Lindström and associates have analysed municipal consumption data for the period 2003 - 2016 and showed that the adoption of local consumption goals has had a positive impact on the share of organic food purchases (Lindström et al., 2022). While the national consumption goals were voluntary for local and regional government, it had what has been described as a catapult effect in setting local and regional ambitions and creating encouragement amongst local procurement officers and kitchen managers (Daugbjerg, 2023). A report produced by the Organic Food Centre highlighted the adoption of local and regional procurement goals as a key factor: 'the national goals have resulted in municipalities and regions formulating and adopting their own local organic goals. It is these local goals that in turn have driven the increased purchases of organic food in the country's municipalities and regions' (Ekomatcentrum, 2022a, pp. 6-7, author's translation).

As in Denmark, food wholesalers were important to the success of the procurement programme. But unlike in Denmark, the effort to motivate them was driven by requirements in local procurement contracts that resulted in a wide range of organic food products being offered (Daugbjerg, 2023).

Denmark and Sweden have introduced organic cuisine labels under which private and public kitchens can be certified. Both labelling schemes distinguish between three levels of organic kitchen certification. The Danish label was introduced in 2009. The bronze label can be obtained when 30-60% of the food products used are organic, the silver label when the organic share is between 60 and 90%, and the gold label when it is between 90 and 100%. Sørensen et al. (2020) found that kitchen certification under the label was likely to maintain or even increase the share of organic food procurement. The organic cuisine label in Sweden was introduced in 1997 by the private organic certifier KRAV. Under the KRAV label, kitchens can achieve the one-star label if at least 25% of the food purchased

is organic (alternatively at least 15 organic products), the two-star label is awarded if the share is at least 50%, and three stars can be achieved if the share is 90% or above.

The Danish and Swedish cases demonstrate that demand-side instruments can be effective in creating demand for sustainable food and that such instruments can indeed be integrated in a coherent organic food policy that also promotes organic food production. Policy innovation requires an outlook which goes beyond directing policy at farmers. Retailers, food services and consumers are key actors in the European food system and therefore policy should also aim at changing their behaviour to demand more sustainable food. In turn, this will be an encouragement for farmers to adopt sustainable farming practices and produce sustainable food for which there will be a growing demand. A concern could be that the implementation of public procurement programmes may abruptly increase demand for organic food and cause price increases. However, this is unlikely to happen. Public procurement is likely to increase gradually as there will be a process of trial-and-error in individual kitchens. This enables supply to catch up.

While the Danish strategy to promote household consumption of organic food requires partnership with private organisations and development of new expertise in organic marketing, a green public procurement strategy is less demanding in terms of policy capacity. The EU's action plan for organic farming recognises the need to stimulate demand for organic food but has few suggestions for applying demands-side policy instruments. It does, however, highlight public procurement as an available tool to stimulate demand (European Commission, 2021), but remains rather unspecific in terms of outlining policy initiatives. An opinion from the European Committee of the Regions (2022) highlights the need to develop policy capacities for public procurement of organic food. Denmark and Sweden have exhibited a high degree of policy innovation in organic food procurement in public sector institutions that can serve as inspiration for other countries. Though it is unlikely that the two countries' procurement programmes for organic food are transferable to other countries in their entirety, components may be relevant to introduce elsewhere.

5.4. Key messages

- The selected policy examples demonstrate that governments use several policy instruments to promote sustainable and healthy food consumption and are considering new instruments. While there are examples of governments using a combination of policy instruments, current policy debates indicate that there is potential for expanded use of such packages of instruments.
- In order to effectively reduce sugar consumption, a comprehensive policy package is needed covering fiscal policies targeting high sugar foods and drinks, as well as restrictions on advertising and health-related food labelling policies, among others. One of the biggest barriers to designing and implementing such policies are the sugar and beverage industries' attempts to influence policies they consider harmful to their interests.
- Despite barriers to implementation such as industry lobbying, shaping of the public discourse, market power and government-industry dependence, notable policy initiatives exist to reduce meat consumption. For example, an 'animal welfare levy' is a government proposal in Germany, and New Zealand has scheduled implementation of a price for greenhouse gas emissions from agriculture. Local initiatives across European cities indicate that public procurement of more plantbased and less animal products is viable and effective and could be rolled out at the national or EU level.
- Policies to reduce meat consumption can be designed so that they garner sufficient public support. Strategic framing alone and strategically spending tax proceeds for political buy-in are insufficient. However, combining various food policies that include forms of meat pricing can increase public support to win majority approval, especially when motivated by animal welfare concerns.
- Policy innovation in promoting the consumption of organic food requires an outlook that goes beyond directing policy at farmers. Government should actively attempt to stimulate demand for organic food. Experiences from Denmark and Sweden demonstrate that such policies can be successfully implemented through public procurement and campaigns directed at households. In both instances, it is important to collaborate with private partners.

Chapter 6. Evidence-based policy elements

Policies steering nutritional and environmental outcomes are fragmented across multiple policy levels ranging from municipalities to regions, countries and ultimately the EU. Historically, EU-level policies have been stronger in the environmental domain than the nutritional domain, while some policies affecting food environments operate mainly at local level (for example, urban development and zoning laws). In addition, member states differ in how they organise their welfare systems, which affects factors like school procurement.

However, the challenge to shift towards healthy and sustainable diets requires integrated and coherent approaches at a systems level, both horizontally (that is, across different domains at a given policy level) and vertically (across different policy levels). This is necessary to achieve synergetic effects, deal with trade-offs in outcomes, and address possible reactions that counteract the intended effects. Several examples discussed in Chapter 5 show that such an approach is possible and desirable. This is why it is important to have a legislative framework for sustainable food systems that lays down the guiding principles to enable effective and efficient actions at multiple policy levels in a coherent and systemic way.⁵⁷

Both the legislative framework and the actions taken should not only consider the various barriers that consumers face in overcoming a shift towards healthy and sustainable diets; they should also acknowledge and take advantage of the food system as a whole, including non-market, informal and increasingly digital elements of the food system.

Rather than listing a set of policy options, we present here a varied set of policy elements that can be combined into policy mixes. Sections 6.1, p. 137, and 6.2, p. 138, list a set of elements that are important in enabling the right policy mixes: 6.2 calls attention for policy elements that capture the systemic nature of food environments and food systems in which consumer behaviour is embedded, while 6.3 summarises the main elements that form the policy objectives, namely healthy and sustainable diets. Section 6.3, p. 139, then provides elements that can be combined into policy instruments that constitute the right policy mixes.

⁵⁷ https://food.ec.europa.eu/horizontal-topics/farm-fork-strategy/legislative-framework_en

6.1. The systemic nature of food systems and environments in which consumption is embedded

Changing food consumption is a key lever to achieve the objectives of the EU's Green Deal and Farm-to-Fork Strategy. Shifting to more sustainable and healthy food choices is a complex process, as it entails changing diets, choosing foods produced using more sustainable production and distribution methods, and reducing food waste. Changing consumption practices requires consumers to reconsider trade-offs between various aspects, such as cost, health, taste, impact and time.

Current policy practices at different levels primarily focus on providing information and education to stimulate behavioural change. In the same vein, the actions proposed by the Farm-to-Fork Strategy mainly build on the premise that a well-informed consumer will make 'rational' choices, i.e. choices that are better for their own health and the planet. Such an approach relies on consumers' intentionality and knowledge about the effects of their choices. It also assumes consumers have agency to make these better choices, once armed with knowledge. This is problematic, as it assumes "that people need to be shown the consequences of their actions in order to be motivated to change behaviour, to take responsibility, to become more caring for the world around them" (Barnett & Land, 2007, p. 1070). The implication of this understanding would be that people do not care for the world and do not act responsibly unless they are exposed to educational campaigns (Sovová et al., 2021). This is patently not the case, as there is increasing recognition of different types of barriers hindering consumers to make such choices.

Therefore, the main question that this report addresses is:

What tools could be used at EU level, in addition to those mentioned in the 2020 Farm-to-Fork Strategy, to overcome the barriers preventing consumers to adopt sustainable and healthy diets, fostering the necessary change towards sustainability in the food environment? The Group's advice should be based on an analysis that identifies the elements refraining consumers from making healthy and sustainable choices.

This report acknowledges the role of personal factors (such as the cognitive processes underlying choices) that influence consumer behaviour, and the barriers to healthy and sustainable diets that consumers need to overcome. There is increasing evidence that consumer behaviour in everyday life is less deliberative and reflective and more automatic, emotion-driven and the outcome of habituation. Consumers tend to eat in a partially-distracted way (or semi-automatic way) and these features should be increasingly taken into account in sustainable food consumption policy.

This report has also established the crucial role of the food environment in shaping consumer choices, concerns and priorities, and how companies in turn are influencing food environments in ways that are not conducive to fostering healthy and sustainable

Evidence-based policy elements

diets. Consumer behaviour models should draw on systemic, integrative and interdisciplinary knowledge and include the relations between social, material and digital contexts; cultural conventions, social norms and values; meanings, beliefs and motivations; mental and physical/body features; emotions and feelings; know-how, skills and competences.

Finally, this report situates food environments in the larger context of food systems that are affected by multiple drivers, such as demographic changes, globalisation and geopolitical developments.

We have called attention to two important features of food systems dynamics. First, it is important to acknowledge the diversity of food systems in Europe and to account for both formal and informal food provisioning systems, not only market but crucially non-market based relations of exchange (for example, gift economy, donations and food redistribution). They not only continue to have importance in eastern and southern European countries as legacies of their recent rural histories, but are also gaining track everywhere in Europe as places of social innovation and experimentation to mitigate the effects of economic, climate, energy, food and sanitary crises. Many Europeans in their daily lives navigate and move seamlessly across various combinations of these systems. Second, digitalisation has become a key driver of change of food environments. Increasingly, digital infrastructures and technologies mediate how people seek, share and interpret food and eating-related information and practices. The effects of the digital food environment on (un)sustainable and (un)healthy food consumption are mixed and interrelated, and there is a risk of 'technological determinism' in the debate.

Finally, it should be mentioned that shifting consumer diets also creates business opportunities, by creating positive feedback loops to the sector. Innovations addressing consumer concerns such as unhealthy ingredients in meat substitutes are needed to accelerate the shift to healthy and sustainable diets (Aschemann-Witzel et al., 2021).

6.2. Direction of change: high-impact behaviours reducing negative environmental and health outcomes related to food systems

Dietary patterns and food consumption vary considerably from region to region in Europe, but also over time. There is a broad consensus in the recommended dietary pattern across countries: to predominantly eat a plant-based diet, rich in vegetables, fruits, whole grains, pulses and fish and seafood sourced from sustainably managed stocks, with moderate amounts of low-fat dairy products, and limited amounts of red and processed meat, salt, added sugar and high-fat animal products. These recommendations are primarily based on personal health considerations, reducing the incidence of overweight, obesity and chronic diseases.

Overall, these recommendations also support environmental outcomes, as food systems are major drivers of environmental impacts, especially in terms of biodiversity loss, eutrophication, water stress, land degradation and climate change. Animal-sourced foods in general have substantially higher environmental impacts compared to plantbased foods, especially in terms of climate change, both per kilogram of food and in total. Hence, there is a broad consensus consensus that limiting the consumption of meat and dairy, especially in affluent settings where consumption is high, is a crucial strategy to mitigate climate change, stop biodiversity loss, halt obesity and fight chronic non-transmissible diseases. In addition, ruminant animals such as cattle and sheep cause higher greenhouse gas emissions per kilogram of meat than pigs and poultry, but at the same time, ruminants are not dependent on human edible crops like cereals and grain legumes as pigs and poultry are. Meanwhile, organic production has benefits such as the avoidance of synthetic pesticides with positive impact on biodiversity, but has similar climate impacts per kilogram of produce as conventional production. This means that organic production without demand-side changes (reduced consumption of animal products and reduced waste) is not a climate mitigation strategy. Savings of greenhouse gas emissions from choosing local foods are minor due to the relatively small contribution from transport to overall emissions from the food system, but local food systems can have other (social) benefits depending on the indicators used to compare local and global supply chains.

Supporting this direction of change requires the development of metrics and other indicators describing the nutritional and environmental impacts of food consumption. Therefore, an important precondition is the development of monitoring schemes using harmonised methodologies calculating nutritional and environmental outcomes.

6.3. Overcoming barriers to more sustainable and healthy food consumption

Behavioural change can be conceptualised as the result of the interaction of motivation, capability and opportunity. In other words, to change their behaviour, consumers need to be motivated, dispose of the necessary resources and have the opportunity to make healthy and sustainable choices. As a result, consumers may face a large number of barriers when trying to change their behaviour, many of which are beyond their control.

The large magnitude and variety of barriers to sustainable and healthy food consumption calls for a broad set of policy interventions and tools to be implemented. Such a policy mix needs to take into account the type of barrier, the nature of the target behaviour,

Evidence-based policy elements

the target group and potential unintended side effects. Most existing public policy interventions use cognitive barriers and lack of personal capabilities as the entry point for behaviour change and try to build capacity through information provision and — to a lesser extent — aim to provide financial (dis)incentives. Target group-specific public policy interventions and the use of personalised feedback tools are rare. The current approach neglects the fact that food-related behaviour is often dominated by (semi-)automatic decision-making and affective processes.

Given that habits, routines, semi-autonomous processes and affective processes are important determinants of food choice, policy measures need to address these. In this regard, interdisciplinary approaches are important, including insights from sociology, anthropology, human geography and other interpretive social sciences. Disruptive measures that alter the context of food-related behaviour, such as taxes, bans, and mandatory reformulations, can alter routines and semi-autonomous processes, especially when they influence the physical food environment. Such approaches are important in creating a food environment that favours healthy and sustainable choices without the need for high agency at consumer level, and a powerful way of reducing dietary inequalities given the strong evidence that people from disadvantaged backgrounds have diets of poorer quality. In summary, policies should aim to make healthy and sustainable meals the easiest, most convenient and reasonable choice for all EU citizens.

Policy interventions should also keep an eye on possible unintended consequences. Regulations that intervene in the close food environment are often less effective, because the action of firms and the outcomes of markets may complement or frustrate policy outcomes (Hobbs & Roosen, 2022). This includes reactions to labelling policies and strategic use of alternative labelling options (Villas-Boas et al., 2020), opposing pricing strategies (Dubois et al., 2017) and manipulation of ingredients and nutrients to enable front-of-pack claims. In addition, policymakers should be aware of the multiple ways industry may operationalise its power to influence and slow down policy implementation through lobbying, finance, marketing, science, supply chains and waste, labour and employment, and reputational management. Therefore, rules on conflicts of interest, competition policy and lobbying should be scrutinised in order to mitigate industry action.

This report summarises the available empirical evidence on the impact of interventions in the food environment on behavioural change. Based on this evidence, we put forward the following policy elements, emphasising that these options should be taken in combination with each other to increase efficiency and effectiveness:

The economic environment and fiscal food policies

Taxes on less healthy and less sustainable foods, and subsidies or reduced taxes (like VAT) for more healthy and sustainable foods, tend to be effective, efficient and

have limited distributional effects. Relatively high prices need to be levied for change to account for the negative environmental outcomes related to meat consumption. Nevertheless, food subsidies supporting healthy diets targeted at low-income households are shown to be effective.

Theoretically, taxing inputs into agricultural production (such as feedstuffs, fertilisers or pesticides) leads to even higher efficiency. However, reaching higher efficiency depends on the abatement potential of production (i.e. the potential of greenhouse gas reduction), which may be limited. It may require a carbon tariff on imported carbon-intensive products, through border adjustment mechanisms.

Physical availability

Modest effectiveness for increasing health outcomes has been shown for improving the availability and placement of fruit and vegetables, while decreasing that of foods and beverages rich in added sugar, salt and fat, in all food environment settings ranging from specific points of purchase (retail, schools, restaurants & canteens) to neighbourhood environments. Similar results are to be expected for environmental outcomes and for the shift from meat to more plant-based foods, but evidence is limited, as too few studies have investigated these relationships. Informal and non-market food systems like urban gardens are shown to have positive health outcomes.

Governments may also directly influence physical availability through public procurement. The Farm-to-Fork Strategy proposes mandatory minimum sustainability criteria in public procurement, which may be expanded to also include mandatory minimum criteria for healthy diets. More attention should be paid to best procurement practices for healthy and sustainable food and their potential for transfer to other settings.

Food composition

The reformulation of product recipes can help improve dietary quality, but this positive effect depends on the degree to which the consumer substitutes reformulated products with other, less healthy products, and whether the reformulation is mandatory or voluntary for industry. Effective policies would require the EU to set mandatory reformulation targets for salt, added sugar and saturated fat for foods and meals sold in retail outlets, catering and restaurants. This can also be part of public procurement criteria.

The information environment

As mentioned in the introduction to this report, the Farm-to-Fork Strategy contains a lot of proposals related to the information environment. Evidence generally supports a moderate impact of nutrition labelling on (un)healthy consumption in different contexts

Evidence-based policy elements

(retail, out-of-home). Sustainability oriented labels tend to reach those who are already motivated and interested, and they strongly depend on the trustworthiness of labels, given that sustainability cannot be directly observed by consumers. However, there is much less research devoted to sustainability labelling in comparison to nutritional labelling. To conclude, shaping the information environment through labelling is necessary but not sufficient to advance healthy and sustainable diets.

Advertisements form an important part of consumers' information environment. Most voluntary agreements on advertisement restrictions tend to focus on advertisement restrictions to children. They tend to have only moderate effects due to their narrow focus (for example, only children's programmes on TV) and due to counteracting measures (for example, lowering prices).

A recent but promising set of instruments are food apps supporting healthy and sustainable food choices in various ways. Evidence shows positive but moderate effects on fruit and vegetable consumption and diet quality. Food apps tend to work better for certain groups of consumers seeking convenience. Personalised feedback also tends to be effective in the desired direction both in relation to health and environmental outcomes. Attention can be given to non-commercial digital platforms and apps run by volunteers or non-profit organisations aimed at non-monetised sharing of food (that is not framed as potential food waste or a source of food for food banks).

The social environment

The social environment exerts a powerful influence on consumer choices and past policy examples, for example the regulation of the tobacco industry, has shown that social norms can be shifted by using a multitude of policy instruments, from taxes and advertisement restrictions to smoking bans in public spaces. In the food environment, the effect of peer influence has been shown to be successful in improving fruit and vegetable intake and limiting fast food consumption. There is evidence that influence and social norms lead to reduced meat consumption.

To conclude, given the complexity of the food system, available scientific evidence confirms that a policy mix of hard and soft measures will be needed to overcome the barriers preventing consumers from adopting sustainable and healthy diets. The current policy focus on the provision of information and education is not effective enough, and must be mixed with a range of other policy elements. Harder measures such as taxes or bans stand out as a very powerful tool to change behaviours across the food system. However, they must be combined with the policy elements presented above. Taken separately, these may seem to have moderate effects, but they offer promising outcomes if they are deployed jointly and in a coordinated manner, and monitored.

UNPUBLISHED DRAFT

References

- Aertsens, J., Verbeke, W., Mondelaers, K., & Van Huylenbroeck, G. (2009). Personal determinants of organic food consumption: a review. *British Food Journal*, 111(10), 1140-1167. <u>https://doi. org/10.1108/00070700910992961</u>
- Agarwal, D., Ravi, P., Purohit, B., & Priya, H. (2021). The effect of energy and fat content labeling on food consumption pattern: a systematic review and meta-analysis. *Nutrition Reviews, 80*(3), 453-466. <u>https://</u> doi.org/10.1093/nutrit/nuab035
- Ahn, S., & Lusk, J. L. (2021). Non-Pecuniary Effects of Sugar-Sweetened Beverage Policies. *American Journal of Agricultural Economics*, *103*(1), 53-69. <u>https://doi.</u> <u>org/10.1111/ajae.12134</u>
- Akesson, A., Andersen, L. F., Kristjánsdóttir, A. G., Roos, E., Trolle, E., Voutilainen, E., & Wirfält, E. (2013, Oct 9). Health effects associated with foods characteristic of the Nordic diet: a systematic literature review. *Food & Nutrition Research, 57*. <u>https://doi.org/10.3402/fnrv57i0.22790</u>
- Alamri, E. S. (2021). Effects of COVID-19 home confinement on eating behavior: A review. *Journal of Public Health Research*, *10*(3). <u>https://doi.org/10.4081/jphr.2021.2088</u>
- Albrecht, C., & Smithers, J. (2018). Reconnecting through local food initiatives? Purpose, practice and conceptions of 'value'. *Agriculture and Human Values, 35*(1), 67-81. <u>https://doi. org/10.1007/s10460-017-9797-5</u>
- Allais, O., Bonnet, C., Réquillart, V., & Spiteri, M. (2020). Reformulation and taxes for healthier consumption: Empirical evidence in the French Dessert market. *TSE Working Paper, 20*, 1082. <u>https://www. tse-fr.eu/publications/reformulation-andtaxes-healthier-consumption-empiricalevidence-french-dessert-market</u>
- Allcott, H., Lockwood, B. B., & Taubinsky, D. (2019). Regressive sin taxes, with an application to the optimal soda tax. *The Quarterly Journal of Economics*, *134*(3), 1557-1626. <u>https://doi.org/10.3386/w25841</u>
- Alston, J. M., MacEwan, J. P., & Okrent, A. M. (2016). The Economics of Obesity and

Related Policy. *Annual Review of Resource Economics*, *8*, 443-465. <u>https://www.jstor.org/stable/26773374</u>

- Álvarez-Sánchez, C., Contento, I., Jiménez-Aguilar, A., Koch, P., Gray, H. L., Guerra, L. A., Rivera-Dommarco, J., Uribe-Carvajal, R., & Shamah-Levy, T. (2018). Does the Mexican sugar-sweetened beverage tax have a signaling effect? ENSANUT 2016. *PLOS ONE*, *13*(8), e0199337. <u>https://doi. org/10.1371/journal.pone.0199337</u>
- An, R. (2013). Effectiveness of subsidies in promoting healthy food purchases and consumption: a review of field experiments. *Public Health Nutrition*, *16*(7), 1215-1228. <u>https://doi.org/10.1017/</u> <u>S1368980012004715</u>
- An, R., He, L., & Shen, M. J. (2020). Impact of neighbourhood food environment on diet and obesity in China: a systematic review. *Public Health Nutrition, 23*(3), 457-473. <u>https://doi.org/10.1017/</u> <u>\$1368980019002167</u>
- An, R., Liu, J., Liu, R., Barker, A. R., Figueroa, R. B., & McBride, T. D. (2020). Impact of Sugar-Sweetened Beverage Warning Labels on Consumer Behaviors: A Systematic Review and Meta-Analysis. *American Journal of Preventive Medicine*, 60(1), 115-126. <u>https:// doi.org/10.1016/j.amepre.2020.07.003</u>
- An, R., Shi, Y., Shen, J., Bullard, T., Liu, G., Yang, Q., Chen, N., & Cao, L. (2021). Effect of front-of-package nutrition labeling on food purchases: a systematic review. *Public Health*, *191*, 59-67. <u>https://doi.org/10.1016/j.</u> <u>puhe.2020.06.035</u>
- Andreyeva, T., Long, M. W., & Brownell, K. D. (2010). The impact of food prices on consumption: a systematic review of research on the price elasticity of demand for food. *American journal of public health*, 100(2), 216-222. https://doi.org/10.2105/ AJPH.2008.151415
- Andreyeva, T., Marple, K., Marinello, S., Moore, T. E., & Powell, L. M. (2022). Outcomes Following Taxation of Sugar-Sweetened Beverages: A Systematic Review and Meta-analysis. *JAMA Network Open*,

144
5(6), e2215276-e2215276. <u>https://doi.</u> org/10.1001/jamanetworkopen.2022.15276

- Andreyeva, T., Marple, K., Moore, T. E., & Powell, L. M. (2022). Evaluation of Economic and Health Outcomes Associated With Food Taxes and Subsidies: A Systematic Review and Meta-analysis. *JAMA Network Open, 5*(6), e2214371-e2214371. <u>https://doi. org/10.1001/jamanetworkopen.2022.14371</u>
- Appleton, K. M., Hemingway, A., Saulais, L., Dinnella, C., Monteleone, E., Depezay, L., Morizet, D., Armando Perez-Cueto, F. J., Bevan, A., & Hartwell, H. (2016). Increasing vegetable intakes: rationale and systematic review of published interventions. *European Journal of Nutrition*, *55*(3), 869-896. <u>https://</u> doi.org/10.1007/s00394-015-1130-8
- Armitage, C. J., & Conner, M. (2001). Efficacy of a minimal intervention to reduce fat intake. *Social Science & Medicine*, *52*(10), 1517-1524. <u>https://doi.org/10.1016/</u> <u>S0277-9536(00)00265-3</u>
- Aschemann-Witzel, J., De Hooge, I., Amani, P., Bech-Larsen, T., & Oostindjer, M. (2015). Consumer-related food waste: Causes and potential for action. *Sustainability*, 7(6), 6457-6477. <u>https://doi.org/10.3390/</u> <u>su7066457</u>
- Aschemann-Witzel, J., Gantriis, R. F., Fraga, P., & Perez-Cueto, F. J. A. (2021). Plant-based food and protein trend from a business perspective: markets, consumers, and the challenges and opportunities in the future. *Critical Reviews in Food Science and Nutrition*, *61*(18), 3119-3128. <u>https://doi.org/1</u> 0.1080/10408398.2020.1793730
- Asioli, D., Bazzani, C., & Nayga Jr, R. M. (2022). Are consumers willing to pay for invitro meat? An investigation of naming effects. *Journal of Agricultural Economics*, *73*(2), 356-375. <u>https://doi.</u> org/10.1111/1477-9552.12467
- Atanasova, P., Kusuma, D., Pineda, E., Frost, G., Sassi, F., & Miraldo, M. (2022, Apr). The impact of the consumer and neighbourhood food environment on dietary intake and obesity-related outcomes: A systematic review of causal impact studies. *Social Science & Medicine*, *299*, 114879. <u>https://doi.org/10.1016/j.</u> <u>socscimed.2022.114879</u>
- Ayres, I., Raseman, S., & Shih, A. (2012). Evidence from Two Large Field Experiments that Peer Comparison

Feedback Can Reduce Residential Energy Usage. *The Journal of Law, Economics, and Organization, 29*(5), 992-1022. <u>https://doi.</u> org/10.1093/jleo/ews020

- Azadi, H., & Ho, P. (2010, Jan-Feb). Genetically modified and organic crops in developing countries: a review of options for food security. *Biotechnology Advances, 28*(1), 160-168. <u>https://doi.org/10.1016/j. biotechadv.2009.11.003</u>
- Azaïs-Braesco, V., Sluik, D., Maillot, M., Kok, F., & Moreno, L. A. (2017). A review of total & added sugar intakes and dietary sources in Europe. *Nutrition Journal*, *16*(1), 6. <u>https://</u> doi.org/10.1186/s12937-016-0225-2
- Baker, P., Machado, P., Santos, T., Sievert, K., Backholer, K., Hadjikakou, M., Russell, C., Huse, O., Bell, C., Scrinis, G., Worsley, A., Friel, S., & Lawrence, M. (2020). Ultraprocessed foods and the nutrition transition: Global, regional and national trends, food systems transformations and political economy drivers. *Obesity reviews*, *21*(12), e13126. <u>https://doi.org/10.1111/</u> <u>obr.13126</u>
- Balázs, B., Pataki, G., & Lazányi, O. (2016). Prospects for the future: Community supported agriculture in Hungary. *Futures*, *83*, 100-111. <u>https://doi.org/10.1016/j.</u> <u>futures.2016.03.005</u>
- Bandy, L., Scarborough, P., Harrington, R., Rayner, M., & Jebb, S. (2020). Reductions in sugar sales from soft drinks in the UK from 2015 to 2018. *BMC medicine*, *18*(1), 1-10. <u>https://doi.org/10.1186/s12916-019-1477-4</u>
- Barham, E. (2003). Translating terroir: the global challenge of French AOC labeling. *Journal of Rural Studies, 19*(1), 127-138. <u>https://doi.org/10.1016/</u> <u>S0743-0167(02)00052-9</u>
- Barkemeyer, R., Young, C. W., Chintakayala, P. K., & Owen, A. (2023). Eco-labels, conspicuous conservation and moral licensing: An indirect behavioural rebound effect. *Ecological Economics*, *204*, 107649. <u>https://doi.org/10.1016/j.</u> <u>ecolecon.2022.107649</u>
- Barnett, C., & Land, D. (2007). Geographies of generosity: Beyond the 'moral turn'. *Geoforum, 38*(6), 1065-1075. <u>https://doi. org/10.1016/j.geoforum.2007.02.006</u>
- Barretto, A. G., Berndes, G., Sparovek, G., & Wirsenius, S. (2013). Agricultural intensification in Brazil and its effects

on land-use patterns: an analysis of the 1975-2006 period. *Global Change Biology*, *19*(6), 1804-1815. <u>https://doi.org/10.1111/</u> <u>gcb.12174</u>

- Barrientos-Gutierrez, T., Zepeda-Tello, R., Rodrigues, E. R., Colchero-Aragonés, A., Rojas-Martínez, R., Lazcano-Ponce, E., Hernández-Ávila, M., Rivera-Dommarco, J., & Meza, R. (2017). Expected population weight and diabetes impact of the 1-peso-per-litre tax to sugar sweetened beverages in Mexico. *PLOS ONE*, *12*(5), e0176336. <u>https://doi.org/10.1371/journal.</u> <u>pone.0176336</u>
- Bartelmeß, T., & Godemann, J. (2022). Exploring the Linkages of Digital Food Communication and Analog Food Behavior: A Scoping Review. International Journal of Environmental Research and Public Health, 19(15). https://doi.org/10.3390/ ijerph19158990
- Basto-Abreu, A., Braverman-Bronstein, A., Camacho-García-Formentí, D., Zepeda-Tello, R., Popkin, B. M., Rivera-Dommarco, J., Hernández-Ávila, M., & Barrientos-Gutiérrez, T. (2018). Expected changes in obesity after reformulation to reduce added sugars in beverages: A modeling study. *PLOS Medicine*, *15*(10), e1002664. <u>https://doi. org/10.1371/journal.pmed.1002664</u>
- Bastounis, A., Buckell, J., Hartmann-Boyce, J., Cook, B., King, S., Potter, C., Bianchi, F., Rayner, M., & Jebb, S. A. (2021, Jul 31). The Impact of Environmental Sustainability Labels on Willingness-to-Pay for Foods: A Systematic Review and Meta-Analysis of Discrete Choice Experiments. *Nutrients*, 13(8). https://doi.org/10.3390/nu13082677
- Batis, C., Rivera, J. A., Popkin, B. M., & Taillie, L. S. (2016). First-Year Evaluation of Mexico's Tax on Nonessential Energy-Dense Foods: An Observational Study. *PLOS Medicine*, 13(7), e1002057. <u>https://doi.org/10.1371/journal.pmed.1002057</u>
- Bauer, J. M., & Reisch, L. A. (2019). Behavioural Insights and (Un)healthy Dietary Choices: a Review of Current Evidence. *Journal of Consumer Policy*, *42*(1), 3-45. <u>https://doi.</u> <u>org/10.1007/s10603-018-9387-y</u>
- Bazerghi, C., McKay, F. H., & Dunn, M. (2016). The Role of Food Banks in Addressing Food Insecurity: A Systematic Review. *Journal of Community Health*, 41(4), 732-740. https:// doi.org/10.1007/s10900-015-0147-5

- Bedard, K., & Kuhn, P. (2015). Micro-marketing healthier choices: Effects of personalized ordering suggestions on restaurant purchases. *Journal of health economics*, *39*, 106-122. <u>https://doi.org/10.1016/j.</u> jhealeco.2014.10.006
- Béhar, M. (1976). European diets vs traditional foods. *Food policy*, 1(5), 432-435. <u>https://doi.org/10.1016/0306-9192(76)90088-9</u>
- Béné, C., Fanzo, J., Prager, S. D., Achicanoy, H.
 A., Mapes, B. R., Alvarez Toro, P., & Bonilla Cedrez, C. (2020). Global drivers of food system (un)sustainability: A multi-country correlation analysis. *PLOS ONE*, *15*(4), e0231071. <u>https://doi.org/10.1371/journal.</u> <u>pone.0231071</u>
- Béné, C., Prager, S. D., Achicanoy, H. A. E., Toro, P. A., Lamotte, L., Cedrez, C. B., & Mapes,
 B. R. (2019). Understanding food systems drivers: A critical review of the literature. *Global Food Security*, 23, 149-159. <u>https://</u> doi.org/10.1016/j.gfs.2019.04.009
- Benelam, B. (2009). Satiation, satiety and their effects on eating behaviour. *Nutrition Bulletin, 34*(2), 126-173. <u>https://doi.</u> org/10.1111/j.1467-3010.2009.01753.x
- Beylot, A., Secchi, M., Cerutti, A., Merciai, S., Schmidt, J., & Sala, S. (2019). Assessing the environmental impacts of EU consumption at macro-scale. *Journal of Cleaner Production, 216*, 382-393. <u>https://doi.</u> org/10.1016/j.jclepro.2019.01.134
- Biasini, B., Rosi, A., Giopp, F., Turgut, R., Scazzina, F., & Menozzi, D. (2021). Understanding, promoting and predicting sustainable diets: A systematic review. *Trends in Food Science & Technology,* 111, 191-207. <u>https://doi.org/10.1016/j.</u> <u>tifs.2021.02.062</u>
- Birch, L. L. (1999). Development of Food References. *Annual Review of Nutrition, 19*(1), 41-62. <u>https://doi.org/10.1146/annurev.</u> <u>nutr.19.1.41</u>
- Black, A. P., Brimblecombe, J., Eyles, H., Morris, P., Vally, H., & ODea, K. (2012). Food subsidy programs and the health and nutritional status of disadvantaged families in high income countries: a systematic review. *BMC Public Health*, *12*(1), 1099. <u>https://doi.org/10.1186/1471-2458-12-1099</u>
- Black, C., Moon, G., & Baird, J. (2014). Dietary inequalities: what is the evidence for the effect of the neighbourhood food environment? *Health Place*, *27*,

229-242. <u>https://doi.org/10.1016/j.</u> healthplace.2013.09.015

- Blay-Palmer, A., Sonnino, R., & Custot, J. (2016). A food politics of the possible? Growing sustainable food systems through networks of knowledge. *Agriculture and Human Values*, *33*(1), 27-43. <u>https://doi.</u> org/10.1007/s10460-015-9592-0
- Bloom, J. D., & Hinrichs, C. C. (2011). Informal and Formal Mechanisms of Coordination in Hybrid Food Value Chains. *Journal of Agriculture, Food Systems, and Community Development, 1*(4), 143-156. <u>https://doi.</u> org/10.5304/jafscd.2011.014.016
- Bonnet, C., Bouamra-Mechemache, Z., & Corre, T. (2018). An environmental tax towards more sustainable food: empirical evidence of the consumption of animal products in France. *Ecological Economics*, *147*, 48-61. <u>https://doi.org/10.1016/j.</u> <u>ecolecon.2017.12.032</u>
- Borman, G. D., de Boef, W. S., Dirks, F., Gonzalez, Y. S., Subedi, A., Thijssen, M. H., Jacobs, J., Schrader, T., Boyd, S., ten Hove, H. J., van der Maden, E., Koomen, I., Assibey-Yeboah, S., Moussa, C., Uzamukunda, A., Daburon, A., Ndambi, A., van Vugt, S., Guijt, J., Kessler, J. J., Molenaar, J. W., & van Berkum, S. (2022). Putting food systems thinking into practice: Integrating agricultural sectors into a multi-level analytical framework. *Global Food Security*, *32*, 100591. <u>https://doi.org/10.1016/j.</u> gfs.2021.100591
- Bos, E., & Owen, L. (2016). Virtual reconnection: The online spaces of alternative food networks in England. *Journal of Rural Studies, 45,* 1-14. <u>https://doi. org/10.1016/j.jrurstud.2016.02.016</u>
- Bowen, S., & Mutersbaugh, T. (2014). Local or localized? Exploring the contributions of Franco-Mediterranean agrifood theory to alternative food research. *Agriculture and Human Values, 31*(2), 201-213. <u>https://doi. org/10.1007/s10460-013-9461-7</u>
- Bowles, S. (2016). *The moral economy: why good incentives are no substitute for good citizens*. Yale University Press.
- Boyland, E., McGale, L., Maden, M., Hounsome, J., Boland, A., & Jones, A. (2022, Aug). Systematic review of the effect of policies to restrict the marketing of foods and non-alcoholic beverages to which children

are exposed. *Obesity reviews, 23*(8), e13447. <u>https://doi.org/10.1111/obr.13447</u>

- Briers, B., Huh, Y. E., Chan, E., & Mukhopadhyay, A. (2020). The unhealthy = tasty belief is associated with BMI through reduced consumption of vegetables: A cross-national and mediational analysis. *Appetite*, *150*, 104639. <u>https://doi.</u> org/10.1016/j.appet.2020.104639
- Briggs, A. D., Kehlbacher, A., Tiffin, R., Garnett, T., Rayner, M., & Scarborough, P. (2013).
 Assessing the impact on chronic disease of incorporating the societal cost of greenhouse gases into the price of food: an econometric and comparative risk assessment modelling study. *BMJ open*, 3(10), e003543. <u>https://doi.org/doi</u>: 10.1136/ bmjopen-2013-003543
- Briggs, A. D., Mytton, O. T., Kehlbacher, A., Tiffin, R., Elhussein, A., Rayner, M., Jebb, S. A., Blakely, T., & Scarborough, P. (2017). Health impact assessment of the UK soft drinks industry levy: a comparative risk assessment modelling study. *The Lancet Public Health, 2*(1), e15-e22. <u>https://doi. org/10.1016/S2468-2667(16)30037-8</u>
- Broeks, M. J., Biesbroek, S., Over, E. A., van Gils, P. F., Toxopeus, I., Beukers, M. H., & Temme, E. H. (2020). A social cost-benefit analysis of meat taxation and a fruit and vegetables subsidy for a healthy and sustainable food consumption in the Netherlands. *BMC Public Health, 20*(1), 1-12. <u>https://doi. org/10.1186/s12889-020-08590-z</u>
- Brouwer, I. D., van Liere, M. J., de Brauw, A., Dominguez-Salas, P., Herforth, A., Kennedy, G., Lachat, C., Omosa, E. B., Talsma, E. F., Vandevijvere, S., Fanzo, J., & Ruel, M. (2021). Reverse thinking: taking a healthy diet perspective towards food systems transformations. *Food Security*, *13*(6), 1497-1523. <u>https://doi.org/10.1007/</u> <u>\$12571-021-01204-5</u>
- Brownell, K. D., & Warner, K. E. (2009). The perils of ignoring history: Big Tobacco played dirty and millions died. How similar is Big Food? *The Milbank Quarterly*, *87*(1), 259-294. <u>https://doi.org/doi</u>: 10.1111/j.1468-0009.2009.00555.x.
- Brug, J. (2008). Determinants of healthy eating: motivation, abilities and environmental opportunities. *Family Practice, 25*(suppl_1), i50-i55. <u>https://doi.org/10.1093/fampra/cmn063</u>

Brunner, T. A., Van der Horst, K., & Siegrist, M. (2010). Convenience food products. Drivers for consumption. *Appetite*, *55*(3), 498-506. <u>https://doi.org/10.1016/j.appet.2010.08.017</u>

Brunori, G., & Galli, F. (2016). Sustainability of Local and Global Food Chains: Introduction to the Special Issue. *Sustainability, 8*(8), 765. <u>https://www.mdpi.</u> <u>com/2071-1050/8/8/765</u>

Brunori, G., Galli, F., Barjolle, D., Van Broekhuizen, R., Colombo, L., Giampietro, M., Kirwan, J., Lang, T., Mathijs, E., Maye, D., De Roest, K., Rougoor, C., Schwarz, J., Schmitt, E., Smith, J., Stojanovic, Z., Tisenkopfs, T., & Touzard, J.-M. (2016). Are Local Food Chains More Sustainable than Global Food Chains? Considerations for Assessment. *Sustainability, 8*(5), 449. <u>https://</u> www.mdpi.com/2071-1050/8/5/449

Bryngelsson, D., Wirsenius, S., Hedenus, F., & Sonesson, U. (2016). How can the EU climate targets be met? A combined analysis of technological and demandside changes in food and agriculture. *Food policy*, *59*, 152-164. <u>https://doi. org/10.1016/j.foodpol.2015.12.012</u>

Brzezina, N., Kopainsky, B., & Mathijs, E. (2016). Can Organic Farming Reduce Vulnerabilities and Enhance the Resilience of the European Food System? A Critical Assessment Using System Dynamics Structural Thinking Tools. *Sustainability, 8*(10), 971. <u>https://www.mdpi. com/2071-1050/8/10/971</u>

Bucher, T., Collins, C., Rollo, M. E., McCaffrey, T. A., De Vlieger, N., Van der Bend, D., Truby, H., & Perez-Cueto, F. J. A. (2016). Nudging consumers towards healthier choices: a systematic review of positional influences on food choice. *British Journal of Nutrition*, *115*(12), 2252-2263. <u>https://doi.org/10.1017/</u> <u>S0007114516001653</u>

Buja, A., Grotto, G., Montecchio, L., De Battisti, E., Sperotto, M., Bertoncello, C., Cocchio, S., Baldovin, T., & Baldo, V. (2021).
Association between health literacy and dietary intake of sugar, fat and salt: a systematic review. *Public Health Nutrition*, 24(8), 2085-2097. <u>https://doi.org/10.1017/ S1368980020002311</u>

Burman, C., Johansson, S., & Davelid, A. (2020). Utvärdering och uppföljning av livsmedelsstrategin: årsrapport år 2020. Rapport 2020:3. <u>https://webbutiken.</u> jordbruksverket.se/sv/artiklar/ra203.html Buttlar, B., & Walther, E. (2018). Measuring the meat paradox: How ambivalence towards meat influences moral disengagement. *Appetite, 128,* 152-158. <u>https://doi. org/10.1016/j.appet.2018.06.011</u>

Byerly, H., Balmford, A., Ferraro, P. J., Hammond Wagner, C., Palchak, E., Polasky, S., Ricketts, T. H., Schwartz, A. J., & Fisher, B. (2018). Nudging pro-environmental behavior: evidence and opportunities. *Frontiers in Ecology and the Environment*, *16*(3), 159-168. <u>https://doi.org/10.1002/</u> <u>fee.1777</u>

Cabrera Escobar, M. A., Veerman, J. L., Tollman, S. M., Bertram, M. Y., & Hofman, K. J. (2013). Evidence that a tax on sugar sweetened beverages reduces the obesity rate: a meta-analysis. *BMC Public Health*, *13*(1), 1-10. <u>https://doi. org/10.1186/1471-2458-13-1072</u>

Cadario, R., & Chandon, P. (2020). Which Healthy Eating Nudges Work Best? A Meta-Analysis of Field Experiments. *Marketing Science, 39*(3), 465-486. <u>https://doi.</u> org/10.1287/mksc.2018.1128

Caillavet, F., Fadhuile, A., & Nichèle, V. (2016). Taxing animal-based foods for sustainability: environmental, nutritional and social perspectives in France. *European Review of Agricultural Economics, 43*(4), 537-560. <u>https://doi.org/10.1093/</u> <u>erae/jbv041</u>

Caillavet, F., Fadhuile, A., & Nichèle, V. (2019). Assessing the distributional effects of carbon taxes on food: Inequalities and nutritional insights in France. *Ecological Economics*, *163*, 20-31. <u>https://doi.</u> org/10.1016/j.ecolecon.2019.04.020

Calabro, R., Kemps, E., & Prichard, I. (2023). Socio-cognitive determinants of sugarsweetened beverage consumption among young people: A systematic review and meta-analysis. *Appetite*, *180*, 106334. <u>https://doi.org/10.1016/j.</u> <u>appet.2022.106334</u>

Campbell, N., Mialon, M., Reilly, K., Browne, S., & Finucane, F. M. (2020). How are frames generated? Insights from the industry lobby against the sugar tax in Ireland. *Social Science & Medicine*, *264*, 113215. <u>https://doi. org/10.1016/j.socscimed.2020.113215</u>

Cane, J., O'Connor, D., & Michie, S. (2012). Validation of the theoretical domains framework for use in behaviour change and implementation research. Implementation Science, 7(1), 37. <u>https://doi.org/10.1186/1748-5908-7-37</u>

- Capacci, S., & Mazzocchi, M. (2011). Fivea-day, a price to pay: an evaluation of the UK program impact accounting for market forces. *Journal of health economics*, *30*(1), 87-98. <u>https://doi.org/10.1016/j.</u> jhealeco.2010.10.006
- Caraher, M., & Cavicchi, A. (2014). Old crises on new plates or old plates for a new crises? Food banks and food insecurity. *British Food Journal*, *116*(9). <u>https://doi.</u> org/10.1108/BFJ-08-2014-0285
- Carey, R. N., Connell, L. E., Johnston, M., Rothman, A. J., de Bruin, M., Kelly, M. P., & Michie, S. (2018). Behavior Change Techniques and Their Mechanisms of Action: A Synthesis of Links Described in Published Intervention Literature. *Annals of Behavioral Medicine*, *53*(8), 693-707. <u>https://</u> doi.org/10.1093/abm/kay078
- Carolan, M. (2020). Automated agrifood futures: robotics, labor and the distributive politics of digital agriculture. *The Journal of Peasant Studies, 47*(1), 184-207. <u>https://doi. org/10.1080/03066150.2019.1584189</u>
- Casado, J., Brigden, K., Santillo, D., & Johnston, P. (2019). Screening of pesticides and veterinary drugs in small streams in the European Union by liquid chromatography high resolution mass spectrometry. *Science of The Total Environment, 670*, 1204-1225. <u>https://doi.org/10.1016/j.</u> <u>scitotenv.2019.03.207</u>
- Caspi, C. E., Sorensen, G., Subramanian, S. V., & Kawachi, I. (2012, Sep). The local food environment and diet: a systematic review. *Health & Place*, *18*(5), 1172-1187. <u>https://doi. org/10.1016/j.healthplace.2012.05.006</u>
- Cederberg, C., Wallman, M., Berglund, M., & Gustavsson, J. (2011). *Klimatavtryck av ekologiska jordbruksprodukter*. <u>https://</u> www.diva-portal.org/smash/get/ diva2:943924/FULLTEXT01.pdf
- Centola, D., Becker, J., Brackbill, D., & Baronchelli, A. (2018). Experimental evidence for tipping points in social convention. *Science, 360*(6393), 1116-1119. <u>https://doi.org/doi</u>:10.1126/science. aas8827
- Cerutti, A. K., Ardente, F., Contu, S., Donno, D., & Beccaro, G. L. (2018, 2018/01/01). Modelling, assessing, and ranking public

procurement options for a climate-friendly catering service. *The International Journal of Life Cycle Assessment, 23*(1), 95-115. <u>https://</u> doi.org/10.1007/s11367-017-1306-y

- Chaboud, G., & Daviron, B. (2017). Food losses and waste: Navigating the inconsistencies. *Global Food Security, 12*, 1-7. <u>https://doi. org/10.1016/j.gfs.2016.11.004</u>
- Chambers, S. A., Freeman, R., Anderson, A. S., & MacGillivray, S. (2015). Reducing the volume, exposure and negative impacts of advertising for foods high in fat, sugar and salt to children: A systematic review of the evidence from statutory and self-regulatory actions and educational measures. *Preventive Medicine*, *75*, 32-43. <u>https://doi.org/10.1016/j.</u> ypmed.2015.02.011
- Chaudhary, A., & Brooks, T. M. (2019). National Consumption and Global Trade Impacts on Biodiversity. *World Development, 121*, 178-187. <u>https://doi.org/10.1016/j.</u> worlddev.2017.10.012
- Chen, P.-J., & Antonelli, M. (2020). Conceptual Models of Food Choice: Influential Factors Related to Foods, Individual Differences, and Society. *Foods*, *g*(12), 1898. <u>https://</u> www.mdpi.com/2304-8158/9/12/1898
- Christoph, M. J., & An, R. (2018). Effect of nutrition labels on dietary quality among college students: a systematic review and meta-analysis. *Nutrition Reviews*, *76*(3), 187-203. <u>https://doi.org/10.1093/nutrit/ nux069</u>
- Chung, A., Vieira, D., Donley, T., Tan, N., Jean-Louis, G., Kiely Gouley, K., & Seixas, A. (2021). Adolescent Peer Influence on Eating Behaviors via Social Media: Scoping Review [Review]. *Journal of Medical Internet Research*, *23*(6), e19697. <u>https://doi.org/10.2196/19697</u>
- Clapp, J. (2023). Concentration and crises: exploring the deep roots of vulnerability in the global industrial food system. *The Journal of Peasant Studies*, *50*(1), 1-25. https://doi.org/10.1080/03066150.2022.21 29013
- Clapp, J., Moseley, W. G., Burlingame, B., & Termine, P. (2022). Viewpoint: The case for a six-dimensional food security framework. *Food policy, 106*, 102164. <u>https://doi.</u> org/10.1016/j.foodpol.2021.102164
- Clare, K., Maani, N., & Milner, J. (2022). Meat, money and messaging: How the

environmental and health harms of red and processed meat consumption are framed by the meat industry. *Food policy*, *109*, 102234. <u>https://doi.org/10.1016/j.</u> <u>foodpol.2022.102234</u>

- Clark, M. A., Domingo, N. G., Colgan, K., Thakrar, S. K., Tilman, D., Lynch, J., Azevedo, I. L., & Hill, J. D. (2020). Global food system emissions could preclude achieving the 1.5 and 2 C climate change targets. *Science*, *370*(6517), 705-708. <u>https://doi.org/10.1126/</u> <u>science.aba7357</u>
- Clark, M., & Tilman, D. (2017). Comparative analysis of environmental impacts of agricultural production systems, agricultural input efficiency, and food choice. *Environmental Research Letters*, *12*(6), 064016. <u>https://doi. org/10.1088/1748-9326/aa6cd5</u>
- Clarke, N., Pechey, E., Kosīte, D., König, L. M., Mantzari, E., Blackwell, A. K., Marteau, T. M., & Hollands, G. J. (2021). Impact of health warning labels on selection and consumption of food and alcohol products: systematic review with meta-analysis. *Health Psychology Review*, *15*(3), 430-453. <u>https://doi.org/10.1080/17437199.2020.17</u> <u>80147</u>
- Cobb, L. K., Appel, L. J., Franco, M., Jones-Smith, J. C., Nur, A., & Anderson, C. A. (2015). The relationship of the local food environment with obesity: A systematic review of methods, study quality, and results. *Obesity (Silver Spring), 23*(7), 1331-1344. https://doi.org/10.1002/oby.21118
- Colchero, M. A., Popkin, B. M., Rivera, J. A., & Ng, S. W. (2016). Beverage purchases from stores in Mexico under the excise tax on sugar sweetened beverages: observational study. *BMJ*, *352*, h6704. <u>https://doi.org/10.1136/bmj.h6704</u>
- Colchero, M. A., Rivera-Dommarco, J., Popkin, B. M., & Ng, S. W. (2017). In Mexico, evidence of sustained consumer response two years after implementing a sugarsweetened beverage tax. *Health Affairs*, *36*(3), 564-571. <u>https://doi.org/10.1377/</u> <u>hlthaff.2016.1231</u>
- Collin, J., & Hill, S. (2019). Structure and tactic of the tobacco, alcohol, and sugary beverage industries - Background Paper for the Task Force on Fiscal Policy for Health. <u>https://data.bloomberglp.</u> <u>com/dotorg/sites/2/2019/04/</u>

<u>Structure-and-Tactics-of-the-Tobacco-</u> <u>Alcohol-and-Sugary-Beverage-Industries.</u> <u>pdf</u>

- Compton, M. E., & 't Hart, P. (2019). *Great policy successes* (First edition ed.). Oxford University Press.
- Contento, I. R., & Koch, P. (2021). Chapter 2: Determinants of Food Choice and Dietary Change: Implications for Nutrition Education. In *Nutrition Education: Linking Research, Theory, and Practice* (4 ed.). Jones & Bartlett.
- Contois, E. J. H., & Kish, Z. (2022). Food Instagram - Identity, Influence, and Negotiation. University of Illinois Press. <u>http://www.jstor.org/stable/10.5406/j.</u> <u>ctv2d6jrkj</u>
- Correa, T., Reyes, M., Taillie, L. S., Corvalán, C., & Dillman Carpentier, F. R. (2020). Food Advertising on Television Before and After a National Unhealthy Food Marketing Regulation in Chile, 2016–2017. *American journal of public health*, *110*(7), 1054-1059. <u>https://doi.org/10.2105/ajph.2020.305658</u>
- Corvalán, C., Reyes, M., Garmendia, M. L., & Uauy, R. (2019). Structural responses to the obesity and non-communicable diseases epidemic: Update on the Chilean law of food labelling and advertising. *Obesity reviews, 20*(3), 367-374. <u>https://doi. org/10.1111/obr.12802</u>
- Coucke, N., Vermeir, I., Slabbinck, H., & Van Kerckhove, A. (2019). Show Me More! The Influence of Visibility on Sustainable Food Choices. *Foods*, 8(6). <u>https://doi.org/10.3390/foods8060186</u>
- Coucke, N., Vermeir, I., Slabbinck, H., Geuens, M., & Choueiki, Z. (2022). How to reduce agri-environmental impacts on ecosystem services: the role of nudging techniques to increase purchase of plant-based meat substitutes. *Ecosystem Services*, *56*, 101444. <u>https://doi.org/10.1016/j.</u> <u>ecoser.2022.101444</u>
- Crippa, M., Solazzo, E., Guizzardi, D., Monforti-Ferrario, F., Tubiello, F. N., & Leip, A. (2021). Food systems are responsible for a third of global anthropogenic GHG emissions. *Nature Food*, 2(3), 198-209. <u>https://doi. org/10.1038/s43016-021-00225-9</u>
- Crockett, R. A., King, S. E., Marteau, T. M., Prevost, A. T., Bignardi, G., Roberts, N. W., Stubbs, B., Hollands, G. J., & Jebb, S. A. (2018). Nutritional labelling for healthier

food or non-alcoholic drink purchasing and consumption. *Cochrane Database Syst Rev, 2*(2), Cd009315. <u>https://doi.</u> org/10.1002/14651858.CD009315.pub2

- Croker, H., Packer, J., Russell, S. J., Stansfield, C., & Viner, R. (2020). Front of pack nutritional labelling schemes: a systematic review and metaanalysis of recent evidence relating to objectively measured consumption and purchasing. *Journal of Human Nutrition and Dietetics*, *33*(4), 518-537. <u>https://doi.org/10.1111/jhn.12758</u>
- Crosetto, P., Lacroix, A., Muller, L., & Ruffieux, B. (2019). Nutritional and economic impact of five alternative front-of-pack nutritional labels: experimental evidence. *European Review of Agricultural Economics*, *47*(2), 785-818. <u>https://doi.org/10.1093/erae/</u> jbz037
- Cuevas García-Dorado, S., Cornselsen, L., Smith, R., & Walls, H. (2019). Economic globalization, nutrition and health: a review of quantitative evidence. *Globalization and Health*, *15*(1), 15. <u>https://doi.org/10.1186/</u> <u>\$12992-019-0456-z</u>
- Currie, J., DellaVigna, S., Moretti, E., & Pathania, V. (2010). The Effect of Fast Food Restaurants on Obesity and Weight Gain. *American Economic Journal: Economic Policy, 2*(3), 32-63. <u>https://doi.org/10.1257/</u> <u>pol.2.3.32</u>
- Daley, A. J., McGee, E., Bayliss, S., Coombe, A., & Parretti, H. M. (2020). Effects of physical activity calorie equivalent food labelling to reduce food selection and consumption: systematic review and meta-analysis of randomised controlled studies. *Journal of epidemiology and community health*, 74(3), 269-275. <u>https://doi.org/10.1136/</u> jech-2019-213216
- Daugbjerg, C. (2022). Against the odds: How policy capacity can compensate for weak instruments in promoting sustainable food. *Policy Sciences*, *55*(3), 451-467. <u>https://doi. org/10.1007/s11077-022-09466-2</u>
- Daugbjerg, C. (2023). Using public procurement of organic food to promote pesticide-free farming: A comparison of governance modes in Denmark and Sweden. *Environmental Science & Policy*, 140, 271-278. <u>https://doi.org/10.1016/j.</u> <u>envsci.2022.12.012</u>

Daugbjerg, C., & Møller, D. K. (2010). Økologipolitikken og de økologiske interesseorganisationers kapaciteter i Danmark, Sverige og Tyskland. *Politica*, 42(1), 69-89. <u>https://doi.org/10.7146/</u> <u>politica.v42i1.69795</u>

- Daugbjerg, C., & Schvartzman, Y. (2022). Organic Food and Farming Policy in Denmark. *Successful Public Policy in the Nordic Countries: Cases, Lessons, Challenges,* 2. <u>https://doi.org/10.1093/</u> <u>oso/9780192856296.003.0002</u>
- Daugbjerg, C., & Sønderskov, K. M. (2012). Environmental policy performance revisited: Designing effective policies for green markets. *Political Studies*, 60(2), 399-418. <u>https://doi. org/10.1111/j.1467-9248.2011.00910.x</u>
- Davis, T., Hennes, E. P., & Raymond, L. (2018). Cultural evolution of normative motivations for sustainable behaviour. *Nature Sustainability*, 1(5), 218-224. <u>https://doi.</u> org/10.1038/s41893-018-0061-9
- De Amores Hernandez, A., Barrios, S., Speitmann, R., & Stoehlker, D. (2023). Price Effects of Temporary VAT Rate Cuts: Evidence from Spanish Supermarkets. https://publications.jrc.ec.europa.eu/ repository/handle/JRC132542
- De Bauw, M., Franssens, S., & Vranken, L. (2022). Trading off environmental attributes in food consumption choices. *Food policy*, *112*, 102338. <u>https://doi.org/10.1016/j.</u> <u>foodpol.2022.102338</u>
- De Jong, I. C., & Guémené, D. (2011). Major welfare issues in broiler breeders. *World's Poultry Science Journal*, 67(1), 73-82. <u>https://</u> doi.org/10.1017/S0043933911000067
- De Keyzer, W., Van Caneghem, S., Heath, A.-L. M., Vanaelst, B., Verschraegen, M., De Henauw, S., & Huybrechts, I. (2012). Nutritional quality and acceptability of a weekly vegetarian lunch in primary-school canteens in Ghent, Belgium: 'Thursday Veggie Day'. *Public Health Nutrition*, *15*(12), 2326-2330. <u>https://doi.org/10.1017/</u> <u>S1368980012000870</u>
- de la Porte, C., Eydal, G. B., Kauko, J., Nohrstedt, D., t Hart, P., & Tranøy, B. S. (2022). *Successful Public Policy in the Nordic Countries: Cases, Lessons, Challenges.* Oxford University Press. <u>https://doi.</u> <u>org/10.1093/oso/9780192856296.001.0001</u>
- De Schutter, O., Jacobs, N., & Clément, C. (2020). A 'Common Food Policy' for Europe: How governance reforms can spark a

shift to healthy diets and sustainable food systems. *Food policy, 96*, 101849. <u>https://</u> doi.org/10.1016/j.foodpol.2020.101849

- DeVerteuil, G., & Golubchikov, O. (2016). Can resilience be redeemed? *City, 20*(1), 143-151. <u>https://doi.org/10.1080/13604813.2015.112</u> 5714
- DiFilippo, K. N., Huang, W.-H., Andrade, J. E., & Chapman-Novakofski, K. M. (2015). The use of mobile apps to improve nutrition outcomes: A systematic literature review. *Journal of Telemedicine and Telecare, 21*(5), 243-253. <u>https://doi. org/10.1177/1357633x15572203</u>
- Diószegi, J., Llanaj, E., & Ádány, R. (2019). Genetic background of taste perception, taste preferences, and its nutritional implications: a systematic review. *Frontiers in genetics*, *10*, 1272. <u>https://doi. org/10.3389/fgene.2019.01272</u>
- Djojosoeparto, S. K., Kamphuis, C. B. M., Vandevijvere, S., Murrin, C., Stanley, I., Romaniuk, P., Harrington, J. M., Poelman, M. P., & Consortium, t. P. (2022). Strength of EU-level food environment policies and priority recommendations to create healthy food environments. *European Journal of Public Health*, *32*(3), 504-511. <u>https://doi. org/10.1093/eurpub/ckac010</u>
- Douglas, S., Schillemans, T., 't Hart, P., Ansell, C., Bøgh Andersen, L., Flinders, M., Head, B., Moynihan, D., Nabatchi, T., O'Flynn, J., Peters, B. G., Raadschelders, J., Sancino, A., Sørensen, E., & Torfing, J. (2021). Rising to Ostrom's challenge: an invitation to walk on the bright side of public governance and public service. *Policy Design and Practice*, 4(4), 441-451. <u>https://doi.org/10.1080/25741</u> 292.2021.1972517
- Dowler, E. A., & O'Connor, D. (2012). Rightsbased approaches to addressing food poverty and food insecurity in Ireland and UK. *Social Science & Medicine*, 74(1), 44-51. <u>https://doi.org/10.1016/j.</u> <u>socscimed.2011.08.036</u>
- Downs, S. M., Ahmed, S., Fanzo, J., & Herforth, A. (2020). Food Environment Typology: Advancing an Expanded Definition, Framework, and Methodological Approach for Improved Characterization of Wild, Cultivated, and Built Food Environments toward Sustainable Diets. *Foods*, *9*(4), 532. <u>https://www.mdpi.</u> <u>com/2304-8158/9/4/532</u>

- Dubois, P., Albuquerque, P., Allais, O., Bonnet, C., Bertail, P., Combris, P., Lahlou, S., Rigal, N., Ruffieux, B., & Chandon, P. (2021). Effects of front-of-pack labels on the nutritional quality of supermarket food purchases: evidence from a large-scale randomized controlled trial. *Journal of the Academy of Marketing Science*, 49(1), 119-138. <u>https://</u> doi.org/10.1007/s11747-020-00723-5
- Dubois, P., Griffith, R., & O'Connell, M. (2020). How well targeted are soda taxes? *American Economic Review, 110*(11), 3661-3704. <u>https://doi.org/10.1257/aer.20171898</u>
- Dubois, P., Griffith, R., & O'Connell, M. (2017). The Effects of Banning Advertising in Junk Food Markets. *The Review of Economic Studies, 85*(1), 396-436. <u>https://doi.</u> org/10.1093/restud/rdx025
- Eakin, H., Connors, J. P., Wharton, C., Bertmann, F., Xiong, A., & Stoltzfus, J. (2017). Identifying attributes of food system sustainability: emerging themes and consensus. *Agriculture and Human Values*, 34(3), 757-773. <u>https://doi.org/10.1007/</u> <u>\$10460-016-9754-8</u>
- Edwards, P., & Roberts, I. (2009). Population adiposity and climate change. *International Journal of Epidemiology, 38*(4), 1137-1140. <u>https://doi.org/10.1093/ije/dyp172</u>
- Ekomatcentrum. (2022a). Hållbara måltider genom målsamverkan. <u>http://</u> <u>ekomatcentrum.se/wp-content/</u> <u>uploads/2022/06/Rapport-hallbara-</u> <u>maltider-genom-malsamverkan-2022juni.</u> <u>pdf</u>
- Ekomatcentrum. (2022b). *Marknadsrapport: Ekologiskt i offentlig sektor 2021*. <u>http://</u> <u>ekomatcentrum.se/wp-content/</u> <u>uploads/2022/05/Rapport-</u> <u>Marknadsrapport-EMC-2022-17-maj-1.pdf</u>
- Eli, K., Dolan, C., Schneider, T., & Ulijaszek, S. (2016). Mobile activism, material imaginings, and the ethics of the edible: Framing political engagement through the Buycott app. *Geoforum*, 74, 63-73. <u>https://doi. org/10.1016/j.geoforum.2016.04.002</u>
- Eli, K., McLennan, A. K., & Schneider, T. (2015). Configuring relations of care in an online consumer protection organization. In E.-J. Abbots, A. Lavis, & L. Attala (Eds.), *Careful Eating: Bodies, Food and Care* (pp. 173-193). Ashgate.
- Elofsson, K., Bengtsson, N., Matsdotter, E., & Arntyr, J. (2016). The impact of climate

information on milk demand: Evidence from a field experiment. *Food policy*, *58*, 14-23. <u>https://doi.org/10.1016/j.</u> foodpol.2015.11.002

- Ericksen, P. J. (2008). Conceptualizing food systems for global environmental change research. *Global Environmental Change*, *18*(1), 234-245. <u>https://doi.org/10.1016/j.</u> <u>gloenvcha.2007.09.002</u>
- Ericksen, P., Bohle, H.-G., & Stewart, B. (2012). Vulnerability and resilience of food systems. In *Food security and global environmental change* (pp. 87-97). Routledge.
- Ericksen, P., Stewart, B., Dixon, J., Barling, D., Loring, P., Anderson, M., & Ingram, J. (2010). The Value of a Food System Approach. In *Food Security and Global Environmental Change* (pp. 21). Routledge.
- Essman, M., Taillie, L. S., Frank, T., Ng, S. W., Popkin, B. M., & Swart, E. C. (2021). Taxed and untaxed beverage intake by South African young adults after a national sugar-sweetened beverage tax: A beforeand-after study. *PLOS Medicine*, *18*(5), e1003574. <u>https://doi.org/10.1371/journal.</u> <u>pmed.1003574</u>
- European Commission, Joint Research Centre, Bertoldi, P., Notarnicola, B., Monforti-Ferrario, F., Renzulli, P., Sala, S., Saouter, E., Medarac, H., Pennington, D., Motola, V., Tassielli, G., Labanca, N., Pinedo Pascua, I., Castellazzi, L., Dallemand, J., Banja, M., Goralczyk, M., Scarlat, N., & Schau, E. (2015). *Energy use in the EU food sector : state of play and opportunities for improvement.* Publications Office. <u>https://doi.org/</u> <u>doi/10.2790/158316</u>
- European Commission. (2007). White Paper on A Strategy for Europe on Nutrition, Overweight and Obesity related health issues. . <u>https://health.ec.europa.eu/</u> <u>nutrition-and-physical-activity/overview/</u> <u>strategy-nutrition-overweight-and-obesityrelated-health-issues_en</u>
- European Commission. (2014). Procurement of 100% organic, seasonal food -Municipality of Copenhagen, Denmark. *GPP In practice series*(47). <u>https://ec.europa.</u> <u>eu/environment/gpp/pdf/news_alert/</u> <u>Issue47_Case_Study97_Copenhagen.pdf</u>
- European Commission. (2020). *Farm* to Fork strategy. <u>https://food.</u>

ec.europa.eu/horizontal-topics/ farm-fork-strategy_en#info

- European Commission. (2021). Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions on an Action Plan for the Development of Organic Production. https://eur-lex.europa.eu/legal-content/ EN/TXT/?uri=CELEX% 3A52021DC0141R%2801%29
- European Committee of the Regions, & Soldi, R. (2018). *Sustainable public procurement of food*. <u>https://op.europa.</u> <u>eu/en/publication-detail/-/publication/</u> <u>b1b7d65b-5334-11e8-be1d-01aa75ed71a1/</u> <u>language-en</u>
- European Committee of the Regions. (2022). *Opinion of the European Committee of the Regions — EU action plan for organic farming*. <u>https://eur-lex.europa.eu/legal-</u> <u>content/EN/TXT/?uri=CELEX</u>% 3A52021IR1968&qid=1652655655807
- Evans, D. (2011). Thrifty, green or frugal: Reflections on sustainable consumption in a changing economic climate. *Geoforum*, 42(5), 550-557. <u>https://doi.org/10.1016/j.</u> geoforum.2011.03.008
- Evans, D. (2014). *Food waste: Home consumption, material culture and everyday life*. Bloomsbury publishing.
- Evans, D. M. (2019). What is consumption, where has it been going, and does it still matter? *The Sociological Review*, *67*(3), 499-517. <u>https://doi.</u> org/10.1177/0038026118764028
- Evers, C., Dingemans, A., Junghans, A. F., & Boevé, A. (2018). Feeling bad or feeling good, does emotion affect your consumption of food? A meta-analysis of the experimental evidence. *Neuroscience & Biobehavioral Reviews, 92*, 195-208. <u>https://</u> doi.org/10.1016/j.neubiorev.2018.05.028
- Fakhri, M. (2022). Conflict and the right to food - Report of the Special Rapporteur on the right to food. https://www.ohchr.org/en/ documents/thematic-reports/ahrc5240conflict-and-right-food-report-specialrapporteur-right-food
- Falk, A., & Heckman, J. J. (2009). Lab experiments are a major source of knowledge in the social sciences. *Science*, *326*(5952), 535-538. <u>https://doi.org/10.1126/</u> <u>science.1168244</u>

References

- Fallaize, R., Newlove, J., White, A., & Lovegrove, J. A. (2020). Nutritional adequacy and content of food bank parcels in Oxfordshire, UK: a comparative analysis of independent and organisational provision. *Journal of Human Nutrition and Dietetics*, *33*(4), 477-486. <u>https://doi. org/10.1111/jhn.12740</u>
- Fanzo, J. (2019). Healthy and Sustainable Diets and Food Systems: the Key to Achieving Sustainable Development Goal 2? *Food Ethics*, 4(2), 159-174. <u>https://doi.</u> org/10.1007/s41055-019-00052-6
- Fanzo, J. C., Downs, S., Marshall, Q. E., de Pee, S., & Bloem, M. W. (2017). Value Chain Focus on Food and Nutrition Security. In S. de Pee, D. Taren, & M. W. Bloem (Eds.), *Nutrition and Health in a Developing World* (pp. 753-770). Springer International Publishing. <u>https:// doi.org/10.1007/978-3-319-43739-2_34</u>
- Fanzo, J., Bellows, A. L., Spiker, M. L., Thorne-Lyman, A. L., & Bloem, M. W. (2021). The importance of food systems and the environment for nutrition. *The American Journal of Clinical Nutrition*, *113*(1), 7-16. <u>https://doi.org/10.1093/ajcn/nqaa313</u>
- Fanzo, J., Davis, C., McLaren, R., & Choufani, J. (2018). The effect of climate change across food systems: Implications for nutrition outcomes. *Global Food Security*, *18*, 12-19. <u>https://doi.org/10.1016/j.gfs.2018.06.001</u>
- Fanzo, J., Haddad, L., Schneider, K. R., Béné, C., Covic, N. M., Guarin, A., Herforth, A. W., Herrero, M., Sumaila, U. R., Aburto, N. J., Amuyunzu-Nyamongo, M., Barguera, S., Battersby, J., Beal, T., Bizzotto Molina, P., Brusset, E., Cafiero, C., Campeau, C., Caron, P., Cattaneo, A., Conforti, P., Davis, C., DeClerck, F. A. J., Elouafi, I., Fabi, C., Gephart, J. A., Golden, C. D., Hendriks, S. L., Huang, J., Laar, A., Lal, R., Lidder, P., Loken, B., Marshall, Q., Masuda, Y. J., McLaren, R., Neufeld, L. M., Nordhagen, S., Remans, R., Resnick, D., Silverberg, M., Torero Cullen, M., Tubiello, F. N., Vivero-Pol, J.-L., Wei, S., & Rosero Moncayo, J. (2021, 2021/10/01/). Viewpoint: Rigorous monitoring is necessary to guide food system transformation in the countdown to the 2030 global goals. Food policy, 104, 102163. https://doi.org/10.1016/j. foodpol.2021.102163
- FAO, & PAHO. (2017). Approval of a New Food Act in Chile: Process Summary. <u>https://www.paho.org/chi/dmdocuments/Approval%20</u>

of%20a%20new%20food%20act%20in%20 Chile.pdf

- FAO, & WHO. (2019). Sustainable healthy diets: guiding principles. <u>https://www.fao.org/3/</u> <u>ca6640en/ca6640en.pdf</u>
- FAO. (2022a). The future of food and agriculture – Drivers and triggers for transformation. <u>https://www.fao.org/</u> documents/card/en/c/cc0959en
- FAO. (2022b). *Statistical Database*. <u>http://www.fao.org/faostat/en/#data</u>
- Feldman, Z. (2021). 'Good food' in an Instagram age: Rethinking hierarchies of culture, criticism and taste. *European Journal of Cultural Studies, 24*(6), 1340-1359. https://doi.org/10.1177/13675494211055733
- Fendrychová, L., & Jehlička, P. (2018). Revealing the hidden geography of alternative food networks: The travelling concept of farmers' markets. *Geoforum*, *95*, 1-10. <u>https://doi.org/10.1016/j.</u> <u>geoforum.2018.06.012</u>
- Fernandez, M. A., & Raine, K. D. (2019). Insights on the influence of sugar taxes on obesity prevention efforts. *Current Nutrition Reports*, *8*, 333-339. <u>https://doi.org/10.1007/</u> <u>\$13668-019-00282-4</u>
- Fernandez, M. A., & Raine, K. D. (2021). Digital Food Retail: Public Health Opportunities. *Nutrients*, *13*(11), 3789. <u>https://www.mdpi.</u> <u>com/2072-6643/13/11/3789</u>
- Fesenfeld, L. P., Sun, Y., Wicki, M., & Bernauer, T. (2021). The role and limits of strategic framing for promoting sustainable consumption and policy. *Global Environmental Change*, *68*, 102266. <u>https://</u> doi.org/10.1016/j.gloenvcha.2021.102266
- Fesenfeld, L. P., Wicki, M., Sun, Y., & Bernauer, T. (2020). Policy packaging can make food system transformation feasible. *Nature Food*, 1(3), 173-182. <u>https://doi.org/10.1038/</u> <u>\$43016-020-0047-4</u>
- Fødevarestyrelsen. (2012). Kommunikationsplan: Det offentlige køkken som nationalt økologisk fyrtårn (Køkkenprojektet) 2012–2020. Unpublished archival material.
- Forssell, S., & Lankoski, L. (2015). The sustainability promise of alternative food networks: an examination through "alternative" characteristics. *Agriculture and Human Values*, *32*(1), 63-75. <u>https://doi.</u> org/10.1007/s10460-014-9516-4

Fox, A., Feng, W., & Asal, V. (2019). What is driving global obesity trends? Globalization or "modernization"? *Globalization and Health*, *15*(1), 32. <u>https://doi.org/10.1186/</u> <u>\$12992-019-0457-y</u>

Fragapane, S., & Mortara, A. (2022). The Value of Networks Against Food Waste: The Case of "Too Good To Go". *Italian Sociological Review*, *12*(3), 1111-1137. <u>https://</u> doi.org/10.13136/isr.v12i3.605

Fuentes, C., & Samsioe, E. (2021). Devising food consumption: complex households and the socio-material work of meal box schemes. *Consumption Markets & Culture,* 24(5), 492-511. <u>https://doi.org/10.1080/1025</u> <u>3866.2020.1810027</u>

Fuentes, M., & Fuentes, C. (2022). Reconfiguring food materialities: plantbased food consumption practices in antagonistic landscapes. *Food, Culture & Society, 25*(3), 520-539. <u>https://doi.org/10.10</u> <u>80/15528014.2021.1903716</u>

Fulponi, L. (2007). *The globalization of private standards and the agri-food system*. CABI International. <u>https://doi.org/10.1079/9781845931858.0005</u>

Funke, F., Mattauch, L., Bijgaart, I. v. d., Godfray, H. C. J., Hepburn, C., Klenert, D., Springmann, M., & Treich, N. (2022). Toward optimal meat pricing: is it time to tax meat consumption? *Review of Environmental Economics and Policy*, 16(2), 219-240. <u>https://www.journals.uchicago.edu/</u> <u>doi/10.1086/721078</u>

FUSIONS project. (2016). Recommendations and guidelines for a common European food waste policy framework. https://www.eu-fusions. org/phocadownload/Publications/ D3.5%20recommendations%20and%20 guidelines%20food%20waste%20policy%20 FINAL.pdf

Gaigné, C., & Le Mener, L. (2014). Agricultural Prices, Selection, and the Evolution of the Food Industry. *American Journal of Agricultural Economics*, *96*(3), 884-902. <u>https://doi.org/10.1093/ajae/aat080</u>

Galbraith-Emami, S., & Lobstein, T. (2013). The impact of initiatives to limit the advertising of food and beverage products to children: a systematic review. *Obesity reviews*, 14(12), 960-974. https://doi.org/10.1111/obr.12060

Gallet, C. A. (2010). Meat meets meta: a quantitative review of the price elasticity

of meat. American Journal of Agricultural Economics, 92(1), 258-272. <u>http://www.jstor.</u> org/stable/40647981

Galli, F., Cavicchi, A., & Brunori, G. (2019). Food waste reduction and food poverty alleviation: a system dynamics conceptual model. *Agriculture and Human Values*, 36(2), 289-300. <u>https://doi.org/10.1007/</u> <u>\$10460-019-09919-0</u>

Galli, F., Hebinck, A., & Carroll, B. (2018). Addressing food poverty in systems: governance of food assistance in three European countries. *Food Security*, *10*(6), 1353-1370. <u>https://doi.org/10.1007/</u> <u>\$12571-018-0850-z</u>

Galli, F., Prosperi, P., Favilli, E., D'Amico, S., Bartolini, F., & Brunori, G. (2020). How can policy processes remove barriers to sustainable food systems in Europe? Contributing to a policy framework for agri-food transitions. *Food policy*, *g6*, 101871. <u>https://doi.org/10.1016/j.</u> <u>foodpol.2020.101871</u>

García-Muros, X., Markandya, A., Romero-Jordán, D., & González-Eguino, M. (2017). The distributional effects of carbon-based food taxes. *Journal of Cleaner Production*, *140*, 996-1006. <u>https://doi.org/10.1016/j.</u> jclepro.2016.05.171

Garcia, M. T., Ribeiro, S. M., Germani, A., & Bógus, C. M. (2018). The impact of urban gardens on adequate and healthy food: a systematic review. *Public Health Nutr, 21*(2), 416-425. <u>https://doi.org/10.1017/</u> <u>\$1368980017002944</u>

Gasparatos, A., Stromberg, P., & Takeuchi, K. (2011). Biofuels, ecosystem services and human wellbeing: Putting biofuels in the ecosystem services narrative. *Agriculture, Ecosystems & Environment, 142*(3), 111-128. https://doi.org/10.1016/j.agee.2011.04.020

GCSA. (2020). Towards a Sustainable Food System. <u>https://op.europa.eu/</u> <u>en/web/eu-law-and-publications/</u> <u>publication-detail/-/publication/</u> <u>ca8ffeda-99bb-11ea-aac4-01aa75ed71a1</u>

Gephart, J. A., Henriksson, P. J. G., Parker, R.
W. R., Shepon, A., Gorospe, K. D., Bergman, K., Eshel, G., Golden, C. D., Halpern, B.
S., Hornborg, S., Jonell, M., Metian, M., Mifflin, K., Newton, R., Tyedmers, P., Zhang, W., Ziegler, F., & Troell, M. (2021). Environmental performance of blue foods.

References

Nature, 597(7876), 360-365. <u>https://doi.</u> org/10.1038/s41586-021-03889-2

- Gereffi, G., Humphrey, J., & Sturgeon, T. (2005). The governance of global value chains. *Review of International Political Economy*, *12*(1), 78-104. <u>https://doi.</u> org/10.1080/09692290500049805
- Gibson-Graham, J. K. (2006). *A postcapitalist politics*. University of Minnesota Press.
- Gibson-Graham, J.-K. (2002). A diverse economy: rethinking economy and economic representation. <u>http://avery.</u> wellesley.edu/Economics/jmatthaei/ <u>transformationcentral/solidarity/</u> <u>solidaritydocuments/diverseeconomies.</u> <u>pdf</u>
- Gilmore, A. B., Fabbri, A., Baum, F., Bertscher, A., Bondy, K., Chang, H.-J., Demaio, S., Erzse, A., Freudenberg, N., Friel, S., Hofman, K. J., Johns, P., Abdool Karim, S., Lacy-Nichols, J., de Carvalho, C. M. P., Marten, R., McKee, M., Petticrew, M., Robertson, L., Tangcharoensathien, V., & Thow, A. M. (2023). Defining and conceptualising the commercial determinants of health. *The Lancet*. <u>https://doi.org/10.1016/</u> <u>S0140-6736(23)00013-2</u>
- Godfray, H. C. J., Aveyard, P., Garnett, T., Hall, J. W., Key, T. J., Lorimer, J., Pierrehumbert, R. T., Scarborough, P., Springmann, M., & Jebb, S. A. (2018). Meat consumption, health, and the environment. *Science*, *361*(6399), eaam5324. <u>https://doi.org/doi</u>:10.1126/ science.aam5324
- Godfray, H. C. J., Beddington, J. R., Crute, I. R., Haddad, L., Lawrence, D., Muir, J. F., Pretty, J., Robinson, S., Thomas, S. M., & Toulmin, C. (2010). Food Security: The Challenge of Feeding 9 Billion People. *Science*, *327*(5967), 812-818. <u>https://doi.org/ doi</u>:10.1126/science.1185383
- González-Monroy, C., Gómez-Gómez, I., Olarte-Sánchez, C. M., & Motrico, E. (2021). Eating Behaviour Changes during the COVID-19 Pandemic: A Systematic Review of Longitudinal Studies. *Int J Environ Res Public Health*, *18*(21). <u>https://doi. org/10.3390/ijerph182111130</u>
- Goodman, D., Melanie, D. E., & Goodman, M. K. (2013). *Alternative food networks: knowledge, practice and politics*. Routledge.
- Goodman, M. K., & Jaworska, S. (2020). Mapping digital foodscapes: Digital food influencers and the grammars of good

food. *Geoforum*, 117, 183-193. <u>https://doi.</u> org/10.1016/j.geoforum.2020.09.020

- Gordon, L. J., Bignet, V., Crona, B., Henriksson, P. J., Van Holt, T., Jonell, M., Lindahl, T., Troell, M., Barthel, S., & Deutsch, L. (2017). Rewiring food systems to enhance human health and biosphere stewardship. *Environmental Research Letters*, *12*(10), 100201. <u>https://doi. org/10.1088/1748-9326/aa81dc</u>
- Goryakin, Y., & Suhrcke, M. (2014). Economic development, urbanization, technological change and overweight: What do we learn from 244 Demographic and Health Surveys? *Economics & Human Biology*, 14, 109-127. <u>https://doi.org/10.1016/j.</u> <u>ehb.2013.11.003</u>
- Graça, J., Godinho, C. A., & Truninger, M. (2019). Reducing meat consumption and following plant-based diets: Current evidence and future directions to inform integrated transitions. *Trends in Food Science & Technology*, *91*, 380-390. <u>https://</u> doi.org/10.1016/j.tifs.2019.07.046
- Gradidge, S., Zawisza, M., Harvey, A. J., & McDermott, D. T. (2021). A Structured Literature Review of the Meat Paradox. *Social Psychological Bulletin*, *16*(3), 1-26. <u>https://doi.org/10.32872/spb.5953</u>
- Graffe, M. I. M., Pala, V., De Henauw, S., Eiben, G., Hadjigeorgiou, C., Iacoviello, L., Intemann, T., Jilani, H., Molnar, D., Russo, P., Veidebaum, T., & Moreno, L. A. (2020). Dietary sources of free sugars in the diet of European children: the IDEFICS Study. *European Journal of Nutrition*, 59(3), 979-989. <u>https://doi.org/10.1007/</u> <u>s00394-019-01957-y</u>
- Graham-Rowe, E., Jessop, D. C., & Sparks, P. (2014). Identifying motivations and barriers to minimising household food waste. *Resources, Conservation and Recycling, 84*, 15-23. <u>https://doi.org/10.1016/j.</u> <u>resconrec.2013.12.005</u>
- Granheim, S. I. (2019). The digital food environment. UNSCN Nutr., 44, 115-121. <u>https://www.unscn.org/uploads/web/</u> <u>news/UNSCN-Nutrition44-WEB-21aug.pdf</u>
- Granheim, S. I., Løvhaug, A. L., Terragni, L., Torheim, L. E., & Thurston, M. (2022). Mapping the digital food environment: A systematic scoping review. *Obesity reviews*, 23(1), e13356. <u>https://doi.org/10.1111/</u> obr.13356

Gravert, C., & Shreedhar, G. (2022). Effective carbon taxes need green nudges. *Nature Climate Change, 12*(12), 1073-1074. <u>https://</u> <u>doi.org/10.1038/s41558-022-01515-1</u>

Gressier, M., Swinburn, B., Frost, G., Segal, A. B., & Sassi, F. (2021). What is the impact of food reformulation on individuals' behaviour, nutrient intakes and health status? A systematic review of empirical evidence. *Obesity reviews, 22*(2), e13139. <u>https://doi. org/10.1111/obr.13139</u>

Griffith, R. (2022). Obesity, poverty and public policy. *The Economic Journal, 132*(644), 1235-1258. <u>https://doi.org/10.1093/ej/ ueac013</u>

Griffith, R., & O'Connell, M. (2010). Public policy towards food consumption. *Fiscal Studies, 31*(4), 481-507. <u>http://www.jstor.org/</u> <u>stable/24440089</u>

Griffith, R., Jenneson, V., James, J., & Taylor, A. (2021). The impact of a tax on added sugar and salt. <u>https://ifs.org.uk/publications/</u> impact-tax-added-sugar-and-salt

Griffith, R., O'Connell, M., & Smith, K. (2017). The Importance of Product Reformulation Versus Consumer Choice in Improving Diet Quality. *Economica*, *84*(333), 34-53. <u>https://doi.org/10.1111/ecca.12192</u>

Griffith, R., O'Connell, M., Smith, K., & Stroud, R. (2020). What's on the menu? Policies to reduce young people's sugar consumption. *Fiscal Studies, 41*(1), 165-197. <u>https://doi. org/10.1111/1475-5890.12194</u>

Grosjean, G., Acworth, W., Flachsland, C., & Marschinski, R. (2016). After monetary policy, climate policy: is delegation the key to EU ETS reform? *Climate Policy*, *16*(1), 1-25. <u>https://doi.org/10.1080/14693062.2014.96</u> 5657

Grummon, A. H., & Hall, M. G. (2020). Sugary drink warnings: A meta-analysis of experimental studies. *PLOS Medicine*, *17*(5), e1003120. <u>https://doi.org/10.1371/journal.</u> <u>pmed.1003120</u>

Grummon, A. H., Smith, N. R., Golden, S. D., Frerichs, L., Taillie, L. S., & Brewer, N. T. (2019). Health Warnings on Sugar-Sweetened Beverages: Simulation of Impacts on Diet and Obesity Among U.S. Adults. American Journal of Preventive Medicine, 57(6), 765-774. <u>https://doi. org/10.1016/j.amepre.2019.06.022</u> Grundy, E. A. C., Slattery, P., Saeri, A. K., Watkins, K., Houlden, T., Farr, N., Askin, H., Lee, J., Mintoft-Jones, A., Cyna, S., Dziegielewski, A., Gelber, R., Rowe, A., Mathur, M. B., Timmons, S., Zhao, K., Wilks, M., Peacock, J. R., Harris, J., Rosenfeld, D. L., Bryant, C., Moss, D., & Zorker, M. (2022, 2022/06/01/). Interventions that influence animal-product consumption: A metareview. *Future Foods*, *5*, 100111. <u>https://doi. org/10.1016/j.fufo.2021.100111</u>

Grunert, K. G. (2011). Sustainability in the Food Sector: A Consumer Behaviour Perspective. *International Journal on Food System Dynamics, 2*(3), 207-218. <u>https://doi.org/10.18461/ijfsd.vzi3.232</u>

Grunert, K. G., & Wills, J. M. (2007). A review of European research on consumer response to nutrition information on food labels. *Journal of Public Health*, *15*(5), 385-399. <u>https://doi.org/10.1007/</u> <u>\$10389-007-0101-9</u>

Gupta, A., Smithers, L. G., Harford, J., Merlin, T., & Braunack-Mayer, A. (2018). Determinants of knowledge and attitudes about sugar and the association of knowledge and attitudes with sugar intake among adults: A systematic review. *Appetite*, *126*, 185-194. <u>https://doi.org/10.1016/j.appet.2018.03.019</u>

Hahn, S. L., Hazzard, V. M., Loth, K. A., Larson, N., Klein, L., & Neumark-Sztainer, D. (2022). Using apps to self-monitor diet and physical activity is linked to greater use of disordered eating behaviors among emerging adults. *Preventive Medicine*, *155*, 106967. <u>https://doi.org/10.1016/j.</u> <u>ypmed.2022.106967</u>

Halkier, B., Keller, M., Truninger, M., & T.-A., W. (2017). Consumption research revisited: Charting of the territory and introducing the handbook. In M. H. In Keller, B.; Wilska; T.-A.; Truninger, M. (Ed.), *Routledge Handbook on Consumption* (pp. 1-11). Routledge.

Hallström, E., Carlsson-Kanyama, A., & Börjesson, P. (2015). Environmental impact of dietary change: a systematic review. *Journal of Cleaner Production*, *91*, 1-11. <u>https://doi.org/10.1016/j.</u> jclepro.2014.12.008

Halpin, D., Daugbjerg, C., & Schvartzman, Y. (2011). Interest-group capacities and infant industry development: State-sponsored growth in organic farming. *International Political Science Review*, *32*(2), 147-166. https://doi.org/10.1177/019251211037243

- Händel, M. N., Cardoso, I., Rasmussen, K. M., Rohde, J. F., Jacobsen, R., Nielsen, S. M., Christensen, R., & Heitmann, B. L. (2019).
 Processed meat intake and chronic disease morbidity and mortality: An overview of systematic reviews and meta-analyses. *PLOS ONE*, *14*(10), e0223883. <u>https://doi. org/10.1371/journal.pone.0223883</u>
- Hansen, K. L., Golubovic, S., Eriksen, C.
 U., Jørgensen, T., & Toft, U. (2022).
 Effectiveness of food environment policies in improving population diets: a review of systematic reviews. *European Journal of Clinical Nutrition*, 76(5), 637-646. <u>https://doi. org/10.1038/s41430-021-01008-y</u>
- Hansen, K., Misra, K., & Sanders, R. E. (2021). Consumer (In)attention to Expiration Dates: A Field Study. *SRPN: Consumer Behavior Issues (Topic)*. <u>https://doi.org/http</u>://dx.doi. org/10.2139/ssrn.3898800
- Hansmann, R., Baur, I., & Binder, C. R. (2020). Increasing organic food consumption: An integrating model of drivers and barriers. *Journal of Cleaner Production*, *275*, 123058. <u>https://doi.org/10.1016/j.</u> jclepro.2020.123058
- Hartcher, K. M., & Lum, H. K. (2020). Genetic selection of broilers and welfare consequences: a review. *World's Poultry Science Journal*, *76*(1), 154-167. <u>https://doi.</u> org/10.1080/00439339.2019.1680025
- Hasselbach, J. L., & Roosen, J. (2015). Consumer Heterogeneity in the Willingness to Pay for Local and Organic Food. *Journal of Food Products Marketing*, *21*(6), 608-625. <u>https://doi.org/10.1080/10454446.2014.88</u> <u>5866</u>
- Hawkes, C., Smith, T. G., Jewell, J., Wardle, J., Hammond, R. A., Friel, S., Thow, A. M., & Kain, J. (2015). Smart food policies for obesity prevention. *The Lancet, 385*(9985), 2410-2421. <u>https://doi.org/10.1016/</u> <u>S0140-6736(14)61745-1</u>
- Hawkins, R., & Horst, N. (2020). Ethical consumption? There's an app for that. Digital technologies and everyday consumption practices. *Canadian Geographies / Les géographies canadiennes, 64*(4), 590-601. <u>https://doi.</u> org/10.1111/cag.12616
- Hebinck, A., Galli, F., Arcuri, S., Carroll, B., O'connor, D., & Oostindie, H. (2018). Capturing change in European food assistance practices: A transformative

social innovation perspective. *Local Environment, 23*(4), 398-413. <u>https://doi.org/</u> 10.1080/13549839.2017.1423046

- Heidenstrøm, N., & Hebrok, M. (2022). Towards realizing the sustainability potential within digital food provisioning platforms: The case of meal box schemes and online grocery shopping in Norway. *Sustainable Production and Consumption, 29*, 831-850. https://doi.org/10.1016/j.spc.2021.06.030
- Hempel, C., & Hamm, U. (2016). How important is local food to organic-minded consumers? *Appetite*, *96*, 309-318. <u>https://</u><u>doi.org/10.1016/j.appet.2015.09.036</u>
- Hendrie, G. A., Lease, H. J., Bowen, J., Baird, D. L., & Cox, D. N. (2017). Strategies to increase children's vegetable intake in home and community settings: a systematic review of literature. *Matern Child Nutr, 13*(1). <u>https://</u> doi.org/10.1111/mcn.12276
- Henry, R. C., Alexander, P., Rabin, S., Anthoni, P., Rounsevell, M. D. A., & Arneth, A. (2019). The role of global dietary transitions for safeguarding biodiversity. *Global Environmental Change*, *58*, 101956. <u>https://</u> <u>doi.org/10.1016/j.gloenvcha.2019.101956</u>
- Hercberg, S., Touvier, M., & Salas-Salvado, J. (2022). The Nutri-Score nutrition label. *Int J Vitam Nutr Res, 92*(3-4), 147-157. <u>https://doi. org/10.1024/0300-9831/a000722</u>
- Herforth, A., & Ahmed, S. (2015). The food environment, its effects on dietary consumption, and potential for measurement within agriculture-nutrition interventions. *Food Security*, 7(3), 505-520. https://doi.org/10.1007/s12571-015-0455-8
- Herforth, A., Arimond, M., Álvarez-Sánchez,
 C., Coates, J., Christianson, K., & Muehlhoff,
 E. (2019). A Global Review of Food-Based
 Dietary Guidelines. Adv Nutr, 10(4), 590-605. https://doi.org/10.1093/advances/nmy130
- Hernández-F, M., Batis, C., Rivera, J. A., & Colchero, M. A. (2019). Reduction in purchases of energy-dense nutrientpoor foods in Mexico associated with the introduction of a tax in 2014. *Preventive Medicine*, *118*, 16-22. <u>https://doi. org/10.1016/j.ypmed.2018.09.019</u>
- Hertz, F. D., & Halkier, B. (2017). Meal box schemes a convenient way to avoid convenience food? Uses and understandings of meal box schemes among Danish consumers. *Appetite*,

114, 232-239. <u>https://doi.org/10.1016/j.</u> appet.2017.03.016

- Hestermann, N., Le Yaouanq, Y., & Treich, N. (2020). An economic model of the meat paradox. *European Economic Review*, *129*, 103569. <u>https://doi.org/10.1016/j.</u> <u>euroecorev.2020.103569</u>
- Hinrichs, C. C. (2000). Embeddedness and local food systems: notes on two types of direct agricultural market. *Journal of Rural Studies*, *16*(3), 295-303. <u>https://doi. org/10.1016/S0743-0167(99)00063-7</u>
- Hirvonen, K., Bai, Y., Headey, D., & Masters, W. A. (2020). Affordability of the EAT–Lancet reference diet: a global analysis. *The Lancet Global Health*, 8(1), e59–e66. <u>https://doi. org/10.1016/S2214-109X(19)30447-4</u>
- Hitchings, R., Collins, R., & Day, R. (2013). Inadvertent environmentalism and the action-value opportunity: reflections from studies at both ends of the generational spectrum. *Local Environment, 20*(3), 369-385. <u>https://doi.org/10.1080/13549839</u> .2013.852524
- HLPE. (2017). Nutrition and food systems. https://www.fao.org/3/i7846e/i7846e.pdf
- HLPE. (2020). Food security and nutrition: building a global narrative towards 2030. https://www.fao.org/3/ca9731en/ ca9731en.pdf
- Hobbs, J. E., & Roosen, J. (2022). Healthoriented nutrition policies. In J. E. Hobbs & J. Roosen (Eds.), *A Modern Guide to Food Economics* (pp. 113–143). Elgar Modern Guides. <u>https://doi.org/10.4337/978180037</u> 2054.00013
- Hoelscher, V., & Chatzidakis, A. (2021). Ethical Consumption Communities Across Physical and Digital Spaces: An Exploration of Their Complementary and Synergistic Affordances. *Journal of Business Ethics*, *172*(2), 291-306. <u>https://doi.org/10.1007/</u> <u>\$10551-020-04477-6</u>
- Hoff, K., & Stiglitz, J. E. (2016). Striving for balance in economics: Towards a theory of the social determination of behavior. *Journal of Economic Behavior* & Organization, 126, 25-57. https://doi. org/10.1016/j.jebo.2016.01.005
- Holloway, L., Kneafsey, M., Venn, L., Cox, R., Dowler, E., & Tuomainen, H. (2007). Possible Food Economies: a Methodological Framework for Exploring Food

Production–Consumption Relationships. *Sociologia Ruralis, 47*(1), 1-19. <u>https://doi.org/10.1111/j.1467-9523.2007.00427.x</u>

- Hood, C. (1983). *Tools of government*. Macmillan International Higher Education.
- Hood, C., & Margetts, H. (2007). *The tools of government in the digital age*. Bloomsbury Publishing.
- Howlett, M., & Rayner, J. (2007). Design Principles for Policy Mixes: Cohesion and Coherence in 'New Governance Arrangements'. *Policy and Society*, *26*(4), 1-18. <u>https://doi.org/10.1016/</u> <u>S1449-4035(07)70118-2</u>
- Howlett, M., Ramesh, M., & Capano, G. (2020). Policy-makers, policy-takers and policy tools: Dealing with behaviourial issues in policy design. *22*(6), 487-497. <u>https://doi. org/10.1080/13876988.2020.1774367</u>
- Hughner, R. S., McDonagh, P., Prothero, A., Shultz II, C. J., & Stanton, J. (2007). Who are organic food consumers? A compilation and review of why people purchase organic food. *Journal of Consumer Behaviour, 6*(2-3), 94-110. <u>https://doi.</u> org/10.1002/cb.210
- Hume, C., Grieger, J. A., Kalamkarian, A., D'Onise, K., & Smithers, L. G. (2022).
 Community gardens and their effects on diet, health, psychosocial and community outcomes: a systematic review. *BMC Public Health*, *22*(1), 1247. <u>https://doi.org/10.1186/ s12889-022-13591-1</u>
- Hunter, L., Gerritsen, S., & Egli, V. (2023). Changes in eating behaviours due to crises, disasters and pandemics: a scoping review. *Nutrition & Food Science*, *53*(2), 358-390. <u>https://doi.org/10.1108/NFS-12-2021-0385</u>
- Huyghe, E., Verstraeten, J., Geuens, M., & Van Kerckhove, A. (2017). Clicks as a Healthy Alternative to Bricks: How Online Grocery Shopping Reduces Vice Purchases. *Journal* of Marketing Research, 54(1), 61-74. <u>https://</u> doi.org/10.1509/jmr.14.0490
- Ikonen, I., Sotgiu, F., Aydinli, A., & Verlegh, P. W. (2020). Consumer effects of front-of-package nutrition labeling: An interdisciplinary meta-analysis. *Journal of the Academy of Marketing Science*, 48(3), 360-383.
- Ilbery, B., Morris, C., Buller, H., Maye, D., & Kneafsey, M. (2005). Product, Process and Place:An Examination of Food Marketing

and Labelling Schemes in Europe and North America. *European Urban and Regional Studies, 12*(2), 116-132. <u>https://doi.</u> org/10.1177/0969776405048499

- Inauen, J., Bolger, N., Shrout, P. E., Stadler, G., Amrein, M., Rackow, P., & Scholz, U. (2017).
 Using Smartphone-Based Support Groups to Promote Healthy Eating in Daily Life: A Randomised Trial. *Applied Psychology: Health and Well-Being*, 9(3), 303-323.
 https://doi.org/10.1111/aphw.12093
- Ingram, J. (2011). A food systems approach to researching food security and its interactions with global environmental change. *Food Security*, *3*(4), 417-431. <u>https://</u><u>doi.org/10.1007/s12571-011-0149-9</u>
- IPBES. (2019). Global assessment report on biodiversity and ecosystem services of the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services. https://ipbes.net/global-assessment
- IPCC. (2022). Climate Change 2022: Impacts, Adaptation, and Vulnerability. Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. <u>https://www.ipcc.ch/</u> <u>report/ar6/wg2/</u>
- Ismail, M. R., Seabrook, J. A., & Gilliland, J. A. (2021). Process evaluation of fruit and vegetables distribution interventions in school-based settings: a systematic review. *Preventive Medicine Reports*, *21*, 101281. <u>https://doi.org/10.1016/j.</u> pmedr.2020.101281
- Jachimowicz, J. M., Hauser, O. P., O'Brien, J. D., Sherman, E., & Galinsky, A. D. (2018). The critical role of second-order normative beliefs in predicting energy conservation. *Nature Human Behaviour,* 2(10), 757-764. <u>https://doi.org/10.1038/</u> <u>\$41562-018-0434-0</u>
- Jahn, G., Schramm, M., & Spiller, A. (2005). The Reliability of Certification: Quality Labels as a Consumer Policy Tool. *Journal of Consumer Policy*, *28*(1), 53-73. <u>https://doi.</u> <u>org/10.1007/s10603-004-7298-6</u>
- Janssen, M., & Hamm, U. (2012). Product labelling in the market for organic food: Consumer preferences and willingnessto-pay for different organic certification logos. *Food Quality and Preference*, *25*(1), 9-22. <u>https://doi.org/10.1016/j.</u> <u>foodqual.2011.12.004</u>

- Jehlička, P., & Daněk, P. (2017). Rendering the Actually Existing Sharing Economy Visible: Home-Grown Food and the Pleasure of Sharing. *Sociologia Ruralis*, *57*(3), 274-296. <u>https://doi.org/10.1111/soru.12160</u>
- Jehlička, P., Ančić, B., Daněk, P., & Domazet, M. (2021). Beyond hardship and joy: Framing home gardening on insights from the European semi-periphery. *Geoforum*, *126*, 150-158. <u>https://doi.org/10.1016/j.</u> <u>geoforum.2021.05.018</u>
- Jehlička, P., Daněk, P., & Vávra, J. (2019). Rethinking resilience: home gardening, food sharing and everyday resistance. *Canadian Journal of Development Studies / Revue canadienne d'études du développement, 40*(4), 511-527. <u>https://doi. org/10.1080/02255189.2018.1498325</u>
- Jehlička, P., Grīviņš, M., Visser, O., & Balázs, B. (2020). Thinking food like an East European: A critical reflection on the framing of food systems. *Journal of Rural Studies*, 76, 286-295. <u>https://doi.org/10.1016/j.</u> jrurstud.2020.04.015
- Jensen, J. D., & Sommer, I. (2017). Reducing calorie sales from supermarkets – 'silent' reformulation of retailer-brand food products. *International Journal of Behavioral Nutrition and Physical Activity*, 14(1), 104. <u>https://doi.org/10.1186/s12966-017-0559-y</u>
- Jensen, J. D., Smed, S., Aarup, L., & Nielsen, E. (2016). Effects of the Danish saturated fat tax on the demand for meat and dairy products. *Public Health Nutrition*, *19*(17), 3085-3094. <u>https://doi.org/10.1017/</u> <u>S1368980015002360</u>
- Jensen, J., & Smed, S. (2018). State-ofthe-art for food taxes to promote public health. *Proceedings of the Nutrition Society*, 77(2), 100-105. <u>https://doi.org/10.1017/</u> <u>S0029665117004050</u>
- Jordbruksdepartementet. (2006). Ekologisk produktion och konsumtion - Mål och inriktning till 2010, Skr. 2005/06:88. <u>https://</u> www.regeringen.se/rattsliga-dokument/ skrivelse/2006/03/skr.-20050688
- Juhl, H. J., Fenger, M. H. J., & Thøgersen, J. (2017). Will the Consistent Organic Food Consumer Step Forward? An Empirical Analysis. *Journal of Consumer Research*, 44(3), 519-535. <u>https://doi.org/10.1093/jcr/ ucx052</u>
- Kalamov, Z. Y., & Runkel, M. (2022). Taxation of unhealthy food consumption and the

intensive versus extensive margin of obesity. *International Tax and Public Finance*, *29*(5), 1294-1320. <u>https://doi.org/10.1007/</u> <u>s10797-021-09704-y</u>

Kamiyama, C., Hashimoto, S., Kohsaka, R., & Saito, O. (2016). Non-market food provisioning services via homegardens and communal sharing in satoyama socioecological production landscapes on Japan's Noto peninsula. *Ecosystem Services*, *17*, 185-196. <u>https://doi.org/10.1016/j.</u> <u>ecoser.2016.01.002</u>

Kankanhalli, A., Saxena, M., & Wadhwa, B. (2019). Combined interventions for physical activity, sleep, and diet using smartphone apps: A scoping literature review. *International Journal of Medical Informatics*, 123, 54-67. <u>https://doi.org/10.1016/j.</u> ijmedinf.2018.12.005

- Karlsson, J. O., Parodi, A., van Zanten, H. H. E., Hansson, P.-A., & Röös, E. (2021). Halting European Union soybean feed imports favours ruminants over pigs and poultry. *Nature Food, 2*(1), 38-46. <u>https://doi. org/10.1038/s43016-020-00203-7</u>
- Kehlbacher, A., Tiffin, R., Briggs, A., Berners-Lee, M., & Scarborough, P. (2016). The distributional and nutritional impacts and mitigation potential of emission-based food taxes in the UK. *Climatic Change*, 137(1), 121-141. <u>https://doi.org/10.1007/</u> <u>\$10584-016-1673-6</u>
- Kenkel, D. S. (1991). Health Behavior, Health Knowledge, and Schooling. *Journal of Political Economy*, 99(2), 287-305. <u>http://</u> www.jstor.org/stable/2937682
- Kent, R. (2020). Self-tracking and digital food cultures - Surveillance and selfrepresentation of the moral 'healthy' body. In D. Lupton & Z. Feldman (Eds.), *Digital Food Cultures* (pp. 16). Routledge.
- Kent, R. (2021). Digital food tracking: Combining traditional and digital ethnographic methods to identify the influence of social media sharing of health and foods upon users' everyday lives. In J. Leer & S. G. S. Krogager (Eds.), *Research Methods in Digital Food Studies* (pp. 143-163). Routledge.
- Kern, F., & Howlett, M. (2009). Implementing transition management as policy reforms: a case study of the Dutch energy sector. *Policy Sciences*, *42*(4), 391-408. <u>https://doi. org/10.1007/s11077-009-9099-x</u>

- Kickbusch, I., Allen, L., & Franz, C. (2016). The commercial determinants of health. *The Lancet Global Health*, 4(12), e895-e896. <u>https://doi.org/10.1016/</u> <u>S2214-109X(16)30217-0</u>
- Kleis, L. D., Schulte, E. A., & Buyken, A. E. (2020). Reformulation across Europe -Results from strategies implemented in European countries other than Germany and identification of factors contributing to their success – part 2. *Ernaehrungs Umschau International*, *11*(67), 200-205. <u>https://doi.org/10.4455/eu.2020.053</u>

Klenert, D., Funke, F., & Cai, M. (2022). Would a Meat Tax in Europe Inevitably Burden the Poor? *Social Science Research Network SSRN*. <u>https://doi.org/http://dx.doi.</u> <u>org/10.2139/ssrn.4298405</u>

Klenert, D., Mattauch, L., Combet, E., Edenhofer, O., Hepburn, C., Rafaty, R., & Stern, N. (2018). Making carbon pricing work for citizens. *Nature Climate Change*, *8*(8), 669-677. <u>https://doi.org/10.1038/</u> <u>\$41558-018-0201-2</u>

Klerkx, L., Jakku, E., & Labarthe, P. (2019). A review of social science on digital agriculture, smart farming and agriculture 4.0: New contributions and a future research agenda. *NJAS: Wageningen Journal of Life Sciences, 90-91*(1), 1-16. https://doi.org/10.1016/j.njas.2019.100315

- Knutson, A., & de Soysa, I. (2019). Does social globalisation through access to information communication technologies drive obesity among youth? An empirical analysis, 1990– 2013. *Global Public Health*, *14*(12), 1911-1926. <u>https://doi.org/10.1080/17441692.2019.16</u> 52335
- Kobez, M. (2018). 'Restaurant reviews aren't what they used to be': digital disruption and the transformation of the role of the food critic. *Communication Research and Practice*, 4(3), 261-276. <u>https://doi.org/10.10</u> <u>80/22041451.2018.1476797</u>
- Koch, C., Ascard, J., Falkenek, A., Jönsson, A., André, C. B., Fröman, E., Gotting. M., & Andresen, N. (2018). Åtgärdsplan för att öka produktion, konsumtion och export av ekologiska livsmedel: Rapport från ett regeringsuppdrag om ekologiska livsmedel. https://webbutiken.jordbruksverket.se/sv/ artiklar/ra1816.html
- Kopainsky, B., Tribaldos, T., & Ledermann, S. T. (2018). A Food Systems Perspective for

Food and Nutrition Security beyond the Post-2015 Development Agenda. *Systems Research and Behavioral Science*, *35*(2), 178-190. <u>https://doi.org/10.1002/sres.2458</u>

- Koretskaya, O., & Feola, G. (2020). A framework for recognizing diversity beyond capitalism in agri-food systems. *Journal of Rural Studies, 80*, 302-313. <u>https://</u> <u>doi.org/10.1016/j.jrurstud.2020.10.002</u>
- Kormaňáková, M., Remešová, M., & Vančová, T. (2021). Food waste in municipal mixed waste produced at household level: empirical evidence from Czechia. *Journal* of Material Cycles and Waste Management, 23(4), 1348-1364. <u>https://doi.org/10.1007/</u> <u>\$10163-021-01214-7</u>
- Krieger, J., Bleich, S. N., Scarmo, S., & Ng, S. W. (2021). Sugar-Sweetened Beverage Reduction Policies: Progress and Promise. *Annual Review of Public Health*, 42(1), 439-461. <u>https://doi.org/10.1146/</u> <u>annurev-publhealth-090419-103005</u>
- Krogh, L. H., Hansen, M. W., & Dal, J. K. (2013). Kvalitativ undersøgelse af økologi i offentlige køkkener: Fra beslutning til praksis. <u>https://vbn.aau.dk/ws/portalfiles/</u> portal/177750482/Kvalitativ_unders_ gelse_af_kologi_i_offentlige_k_kkener_Fra_ beslutning_til_praksis.pdf
- Kubíčková, L., Veselá, L., & Kormaňáková, M. (2021). Food Waste Behaviour at the Consumer Level: Pilot Study on Czech Private Households. *Sustainability*, *13*(20), 11311. <u>https://www.mdpi.</u> <u>com/2071-1050/13/20/11311</u>
- Kushwah, S., Dhir, A., Sagar, M., & Gupta, B. (2019). Determinants of organic food consumption. A systematic literature review on motives and barriers. *Appetite*, 143, 104402. <u>https://doi.org/10.1016/j.</u> appet.2019.104402
- Lacy-Nichols, J., & Marten, R. (2021). Power and the commercial determinants of health: ideas for a research agenda. *BMJ Global Health*, 6(2), e003850. <u>https://doi. org/10.1136/bmjgh-2020-003850</u>

Lambie-Mumford, H., & Silvasti, T. (2020). *The rise of food charity in Europe*. Policy Press Bristol.

Landwehr, S. C., & Hartmann, M. (2020). Industry self-regulation of food advertisement to children: Compliance versus effectiveness of the EU Pledge. Food policy, 91, 101833. <u>https://doi.</u> org/10.1016/j.foodpol.2020.101833

- Lanz, B., Wurlod, J.-D., Panzone, L., & Swanson, T. (2018). The behavioral effect of Pigovian regulation: Evidence from a field experiment. *Journal of Environmental Economics and Management*, *87*, 190-205. https://doi.org/10.1016/j.jeem.2017.06.005
- Latka, C., Kuiper, M., Frank, S., Heckelei, T., Havlík, P., Witzke, H.-P., Leip, A., Cui, H. D., Kuijsten, A., & Geleijnse, J. M. (2021). Paying the price for environmentally sustainable and healthy EU diets. *Global Food Security*, *28*, 100437. <u>https://doi.org/10.1016/j.</u> <u>gfs.2020.100437</u>
- Lauber, K., Hunt, D., Gilmore, A. B., & Rutter, H. (2021). Corporate political activity in the context of unhealthy food advertising restrictions across Transport for London: A qualitative case study. *PLOS Medicine*, *18*(9), e1003695. <u>https://doi.org/10.1371/journal.</u> <u>pmed.1003695</u>
- Lauber, K., Rippin, H., Wickramasinghe, K., & Gilmore, A. B. (2022). Corporate political activity in the context of sugar-sweetened beverage tax policy in the WHO European Region. *European Journal of Public Health*, 32(5), 786-793. <u>https://doi.org/10.1093/</u> <u>eurpub/ckac117</u>
- Lazarus, O., McDermid, S., & Jacquet, J. (2021). The climate responsibilities of industrial meat and dairy producers. *Climatic Change, 165*(1), 30. <u>https://doi.org/10.1007/</u> <u>\$10584-021-03047-7</u>
- Leer, J., & Krogager, S. G. S. (2021). *Research methods in digital food studies*. Routledge.
- Leip, A., Billen, G., Garnier, J., Grizzetti, B., Lassaletta, L., Reis, S., Simpson, D., Sutton, M. A., de Vries, W., Weiss, F., & Westhoek, H. (2015). Impacts of European livestock production: nitrogen, sulphur, phosphorus and greenhouse gas emissions, land-use, water eutrophication and biodiversity. *Environmental Research Letters*, 10(11), 115004. <u>https://doi.</u> org/10.1088/1748-9326/10/11/115004
- Leroy, P., Réquillart, V., Soler, L. G., & Enderli, G. (2016). An assessment of the potential health impacts of food reformulation. *European Journal of Clinical Nutrition*, 70(6), 694-699. <u>https://doi.org/10.1038/</u> <u>ejcn.2015.201</u>
- Lewis, T. (2020). *Digital food: from paddock to platform* (1 ed.). Bloomsbury Academic.

- Libery, B., & Kneafsey, M. (1998). Product and Place:Promoting Quality Products and Services in the Lagging Rural Regions of the European Union. *European Urban and Regional Studies*, *5*(4), 329-341. <u>https://doi. org/10.1177/096977649800500404</u>
- Lim, S. Y., Lee, K. W., Seow, W.-L., Mohamed, N. A., Devaraj, N. K., & Amin-Nordin, S. (2021). Effectiveness of Integrated Technology Apps for Supporting Healthy Food Purchasing and Consumption: A Systematic Review. *Foods*, *10*(8), 1861. <u>https://www. mdpi.com/2304-8158/10/8/1861</u>
- Lindström, H., Lundberg, S., & Marklund, P.-O. (2022). Green public procurement: An empirical analysis of the uptake of organic food policy. *Journal of Purchasing and Supply Management, 28*(3), 100752. <u>https://</u> doi.org/10.1016/j.pursup.2022.100752
- Lindstrom, K. N., Tucker, J. A., & McVay, M. (2023). Nudges and choice architecture to promote healthy food purchases in adults: A systematized review. *Psychol Addict Behav*, *37*(1), 87-103. <u>https://doi.</u> org/10.1037/adb000892
- Linseisen, J., Kesse, E., Slimani, N., Bueno-de-Mesquita, H. B., Ocké, M. C., Skeie, G., Kumle, M., Iraeta, M. D., Gómez, P. M., Janzon, L., Stattin, P., Welch, A. A., Spencer, E. A., Overvad, K., Tjønneland, A., Clavel-Chapelon, F., Miller, A. B., Klipstein-Grobusch, K., Lagiou, P., Kalapothaki, V., Masala, G., Giurdanella, M. C., Norat, T., & Riboli, E. (2002). Meat consumption in the European Prospective Investigation into Cancer and Nutrition (EPIC) cohorts: results from 24-hour dietary recalls. *Public Health Nutrition*, 5(6b), 1243-1258. <u>https://doi.org/10.1079/</u> PHN2002402
- Livingstone, K. M., Celis-Morales, C., Navas-Carretero, S., San-Cristobal, R., Forster, H., Woolhead, C., O'Donovan, C. B., Moschonis, G., Manios, Y., Traczyk, I., Gundersen, T. E., Drevon, C. A., Marsaux, C. F. M., Fallaize, R., Macready, A. L., Daniel, H., Saris, W. H. M., Lovegrove, J. A., Gibney, M., Gibney, E. R., Walsh, M., Brennan, L., Martinez, J. A., & Mathers, J. C. (2021). Personalised nutrition advice reduces intake of discretionary foods and beverages: findings from the Food4Me randomised controlled trial. *International Journal of Behavioral Nutrition and Physical Activity*, *18*(1), 70. <u>https://doi.</u> org/10.1186/s12966-021-01136-5
- Lombardini, C., & Lankoski, L. (2013). Forced Choice Restriction in Promoting

Sustainable Food Consumption: Intended and Unintended Effects of the Mandatory Vegetarian Day in Helsinki Schools. *Journal of Consumer Policy*, *36*(2), 159-178. <u>https://</u> <u>doi.org/10.1007/s10603-013-9221-5</u>

- Long, M. W., Gortmaker, S. L., Ward, Z. J., Resch, S. C., Moodie, M. L., Sacks, G., Swinburn, B. A., Carter, R. C., & Claire Wang, Y. (2015). Cost Effectiveness of a Sugar-Sweetened Beverage Excise Tax in the U.S. *American Journal of Preventive Medicine*, 49(1), 112-123. <u>https://doi.org/10.1016/j.</u> amepre.2015.03.004
- Long, M. W., Tobias, D. K., Cradock, A. L., Batchelder, H., & Gortmaker, S. L. (2015). Systematic Review and Meta-analysis of the Impact of Restaurant Menu Calorie Labeling. *American journal of public health*, *105*(5), e11-e24. <u>https://doi.org/10.2105/</u> <u>ajph.2015.302570</u>
- Loopstra, R. (2018). Interventions to address household food insecurity in high-income countries. *Proceedings of the Nutrition Society*, 77(3), 270-281. <u>https://doi.</u> org/10.1017/S002966511800006X
- Love, P., Whelan, J., Bell, C., & McCracken, J. (2019). Measuring Rural Food Environments for Local Action in Australia: A Systematic Critical Synthesis Review. *Int J Environ Res Public Health*, *16*(13). <u>https://doi.</u> org/10.3390/ijerph16132416
- Luchs, M. G., Naylor, R. W., Irwin, J. R., & Raghunathan, R. (2010). The sustainability liability: Potential negative effects of ethicality on product preference. *Journal* of Marketing, 74(5), 18-31. <u>https://doi.</u> org/10.1509/jmkg.74.5.018
- Lupton, D. (2021). 'All at the tap of a button': Mapping the food app landscape. *European Journal of Cultural Studies, 24*(6), 1360-1381. <u>https://doi.</u> org/10.1177/13675494211055732
- Lupton, D., & Feldman, Z. (2020). *Digital food cultures*. Routledge/Taylor & Francis Group.
- Maani, N., van Schalkwyk, M. C., Petticrew, M., & Buse, K. (2022). The pollution of health discourse and the need for effective counter-framing. *British Medical Journal*(377:01128). <u>https://doi.org/10.1136/</u> <u>bmj.01128</u>
- MacDonald, G. K. (2013). Eating on an interconnected planet. *Environmental Research Letters, 8*(2), 021002. <u>https://doi.</u> org/10.1088/1748-9326/8/2/021002

- Mah, C. L., Luongo, G., Hasdell, R., Taylor, N. G., & Lo, B. K. (2019). A systematic review of the effect of retail food environment interventions on diet and health with a focus on the enabling role of public policies. *Current Nutrition Reports, 8*, 411-428. <u>https://doi.org/10.1007/</u> <u>\$13668-019-00295-z</u>
- Mancino, L., Kuchler, F., & Leibtag, E. (2008). Getting consumers to eat more wholegrains: The role of policy, information, and food manufacturers. *Food policy*, *33*(6), 489-496. <u>https://doi.org/10.1016/j. foodpol.2008.05.005</u>
- Mandracchia, F., Llauradó, E., Tarro, L., del Bas, J. M., Valls, R. M., Pedret, A., Radeva, P., Arola, L., Solà, R., & Boqué, N. (2019). Potential Use of Mobile Phone Applications for Self-Monitoring and Increasing Daily Fruit and Vegetable Consumption: A Systematized Review. *Nutrients*, *11*(3), 686. <u>https://www.</u> mdpi.com/2072-6643/11/3/686
- Marsden, T., Banks, J., & Bristow, G. (2002). The Social Management of Rural Nature: Understanding Agrarian-Based Rural Development. *Environment and Planning A: Economy and Space, 34*(5), 809-825. <u>https://</u> doi.org/10.1068/a3427
- Mattauch, L., Hepburn, C., Spuler, F., & Stern, N. (2022). The economics of climate change with endogenous preferences. *Resource and Energy Economics, 69,* 101312. <u>https://doi.org/10.1016/j.</u> <u>reseneeco.2022.101312</u>
- May, P. J. (2003). Policy design and implementation. In B. G. Peters & J. Pierre (Eds.), *Handbook of public administration* (Vol. 223, pp. 223-233). SAGE Publications Ltd.
- Mazarello Paes, V., Hesketh, K., O'Malley, C., Moore, H., Summerbell, C., Griffin, S., van Sluijs, E. M. F., Ong, K. K., & Lakshman, R. (2015). Determinants of sugar-sweetened beverage consumption in young children: a systematic review. *Obesity reviews*, 16(11), 903-913. <u>https://doi.org/10.1111/obr.12310</u>
- McConnell, A. (2010). *Understanding policy success: rethinking public policy*. Palgrave Macmillan.
- McMichael, P. (2015). Value Theory and Food Regime Temporalities. *Review (Fernand Braudel Center), 38*(4), 307-328. <u>https:// www.jstor.org/stable/26563030</u>

- McSpadden, K. E., Patrick, H., Oh, A. Y., Yaroch, A. L., Dwyer, L. A., & Nebeling, L. C. (2016). The association between motivation and fruit and vegetable intake: The moderating role of social support. *Appetite*, *96*, 87-94. https://doi.org/10.1016/j.appet.2015.08.031
- Meier, M. S., Stoessel, F., Jungbluth, N., Juraske, R., Schader, C., & Stolze, M. (2015). Environmental impacts of organic and conventional agricultural products – Are the differences captured by life cycle assessment? *Journal of Environmental Management, 149,* 193-208. <u>https://doi. org/10.1016/j.jenvman.2014.10.006</u>
- Meiksin, R., Er, V., Thompson, C., Adams, J., Boyland, E., Burgoine, T., Cornelsen, L., de Vocht, F., Egan, M., Lake, A. A., Lock, K., Mytton, O., White, M., Yau, A., & Cummins, S. (2022). Restricting the advertising of high fat, salt and sugar foods on the Transport for London estate: Process and implementation study. *Social Science & Medicine*, 292, 114548. <u>https://doi. org/10.1016/j.socscimed.2021.114548</u>
- Messer, M., McClure, Z., Norton, B., Smart, M., & Linardon, J. (2021). Using an app to count calories: Motives, perceptions, and connections to thinness- and muscularity-oriented disordered eating. *Eating Behaviors, 43*, 101568. <u>https://doi. org/10.1016/j.eatbeh.2021.101568</u>
- Messner, R., Richards, C., & Johnson, H. (2020). The "Prevention Paradox": food waste prevention and the quandary of systemic surplus production. *Agriculture and Human Values, 37*, 805-817. <u>https://doi.</u> org/10.1007/s10460-019-10014-7
- Micha, R., Karageorgou, D., Bakogianni, I., Trichia, E., Whitsel, L. P., Story, M., Peñalvo, J. L., & Mozaffarian, D. (2018). Effectiveness of school food environment policies on children's dietary behaviors: A systematic review and meta-analysis. *PLOS ONE*, *13*(3), e0194555. <u>https://doi.org/10.1371/journal.</u> <u>pone.0194555</u>
- Michel-Villarreal, R., Hingley, M., Canavari, M., & Bregoli, I. (2019). Sustainability in Alternative Food Networks: A Systematic Literature Review. *Sustainability*, 11(3), 859. <u>https://www.mdpi.</u> <u>com/2071-1050/11/3/859</u>
- Michie, S., Van Stralen, M. M., & West, R. (2011). The behaviour change wheel: A new method for characterising and designing behaviour change interventions.

Implementation Science, 6(1), 42. <u>https://doi.org/10.1186/1748-5908-6-42</u>

- Michielsen, Y. J., & van der Horst, H. M. (2022). Backlash against meat curtailment policies in online discourse: Populism as a missing link. *Appetite*, *171*, 105931. <u>https://doi. org/10.1016/j.appet.2022.105931</u>
- Milestad, R., Bartel-Kratochvil, R., Leitner, H., & Axmann, P. (2010). Being close: The quality of social relationships in a local organic cereal and bread network in Lower Austria. *Journal of Rural Studies*, *26*(3), 228-240. <u>https://doi.org/10.1016/j.</u> jrurstud.2010.01.004
- Miljö- och jordbruksutskottet. (2010). Uppföljning av ekologisk produktion och offentlig konsumtion. <u>https://www.</u> <u>riksdagen.se/sv/dokument-lagar/</u> <u>dokument/rapport-fran-riksdagen/</u> <u>uppfoljning-av-ekologisk-produktion-och-</u> <u>offentlig_GYOWRFR1</u>
- Minadeo, M., & Pope, L. (2022). Weightnormative messaging predominates on TikTok—A qualitative content analysis. *PLOS ONE, 17*(11), e0267997. <u>https://doi. org/10.1371/journal.pone.0267997</u>
- Ministeriet for Fødevarer, L. o. F. (2012). Økologisk handlingsplan 2020. https:// fvm.dk/fileadmin/user_upload/FVM. dk/Dokumenter/Landbrug/Indsatser/ Oekologi/Oekologisk_Handlingsplan_2020. pdf
- Moberg, E., Säll, S., Hansson, P.-A., & Röös, E. (2021). Taxing food consumption to reduce environmental impacts–Identification of synergies and goal conflicts. *Food policy*, *101*, 102090. <u>https://doi.org/10.1016/j.</u> <u>foodpol.2021.102090</u>
- Mohr, B., Dolgopolova, I., & Roosen, J. (2019). The influence of sex and self-control on the efficacy of nudges in lowering the energy content of food during a fast food order. *Appetite*, *141*, 104314. <u>https://doi. org/10.1016/j.appet.2019.06.006</u>
- Molin, E., Martin, M., & Björklund, A. (2021). Addressing Sustainability within Public Procurement of Food: A Systematic Literature Review. *Sustainability*, *13*(23), 13395. <u>https://www.mdpi.</u> <u>com/2071-1050/13/23/13395</u>
- Mottet, A., de Haan, C., Falcucci, A., Tempio, G., Opio, C., & Gerber, P. (2017). Livestock: On our plates or eating at our table? A new analysis of the feed/food debate.

Global Food Security, 14, 1-8. <u>https://doi.org/10.1016/j.gfs.2017.01.001</u>

- Mount, P. (2012). Growing local food: scale and local food systems governance. *Agriculture and Human Values, 29*(1), 107-121. <u>https://</u> <u>doi.org/10.1007/s10460-011-9331-0</u>
- Mousa, T. Y., & Freeland-Graves, J. H. (2019). Impact of food pantry donations on diet of a low-income population. *International Journal of Food Sciences and Nutrition*, 70(1), 78-87. <u>https://doi.org/10.1080/09637486.2</u> 018.1466271
- Mozaffarian, D. (2016). Dietary and Policy Priorities for Cardiovascular Disease, Diabetes, and Obesity: A Comprehensive Review. *Circulation*, *133*(2), 187-225. <u>https://</u> doi.org/10.1161/circulationaha.115.018585
- Muir, S., Dhuria, P., & Vogel, C. (2022). Government must proceed with landmark anti-obesity regulations in England. *BMJ*, *378*, 02358. <u>https://doi.org/10.1136/bmj</u>. <u>02358</u>
- Munt, A. E., Partridge, S. R., & Allman-Farinelli, M. (2017). The barriers and enablers of healthy eating among young adults: a missing piece of the obesity puzzle: A scoping review. *Obesity reviews*, *18*(1), 1-17. https://doi.org/10.1111/obr.12472
- Mytton, O., Gray, A., Rayner, M., & Rutter, H. (2007). Could targeted food taxes improve health? *Journal of Epidemiology & Community Health*, *61*(8), 689-694. <u>https://</u> doi.org/10.1136/jech.2006.047746
- Nadricka, K., Millet, K., & Verlegh, P. W. (2020). When organic products are tasty: Taste inferences from an Organic= Healthy Association. *Food Quality and Preference*, *83*, 103896. <u>https://doi.org/10.1016/j.</u> <u>foodqual.2020.103896</u>
- Nagy, L. B., Lakner, Z., & Temesi, Á. (2022). Is it really organic? Credibility factors of organic food-A systematic review and bibliometric analysis. *PLOS ONE*, *17*(4), e0266855. <u>https://doi.org/10.1371/journal.</u> <u>pone.0266855</u>
- Näringsdepartementet. (2017). En livsmedelsstrategi för Sverige – fler jobb och hållbar tillväxt i hela landet: Regeringens handlingsplan. <u>https://www.regeringen.se/</u> rattsliga-dokument/proposition/2017/01/ prop.-201617104
- Nath, J. (2010). 'God is a vegetarian': The food, health and bio-spirituality of Hare Krishna,

Buddhist and Seventh-Day Adventist devotees. *Health Sociology Review*, 19(3), 356-368. <u>https://doi.org/10.5172/</u> <u>hesr.2010.19.3.356</u>

- Nathan, N., Janssen, L., Sutherland, R., Hodder, R. K., Evans, C. E. L., Booth, D., Yoong, S. L., Reilly, K., Finch, M., & Wolfenden, L. (2019). The effectiveness of lunchbox interventions on improving the foods and beverages packed and consumed by children at centre-based care or school: a systematic review and meta-analysis. *International Journal of Behavioral Nutrition and Physical Activity*, *16*(1), 38. <u>https://doi.org/10.1186/ s12966-019-0798-1</u>
- Nayga, R. M. (2000). Nutrition Knowledge, Gender, and Food Label Use. *The Journal of Consumer Affairs*, *34*(1), 97-112. <u>http://www.</u> jstor.org/stable/23859983
- Neto, B., & Gama Caldas, M. (2018). The use of green criteria in the public procurement of food products and catering services: a review of EU schemes. *Environment*, *Development and Sustainability*, 20(5), 1905-1933. https://doi.org/10.1007/ s10668-017-9992-y
- Newman, G. E., Gorlin, M., & Dhar, R. (2014). When going green backfires: How firm intentions shape the evaluation of socially beneficial product enhancements. *Journal of Consumer Research, 41*, 823-839. <u>https://</u> doi.org/10.1086/677841
- Ng, S. W., Rivera, J. A., Popkin, B. M., & Colchero, M. A. (2019). Did high sugarsweetened beverage purchasers respond differently to the excise tax on sugarsweetened beverages in Mexico? *Public Health Nutrition, 22*(4), 750-756. <u>https://doi. org/10.1017/S136898001800321X</u>
- Nguyen, J., Ferraro, C., Sands, S., & Luxton, S. (2022). Alternative protein consumption: A systematic review and future research directions. *International Journal of Consumer Studies*, 46(5), 1691-1717. <u>https://</u> doi.org/10.1111/jjcs.12797
- Ni Mhurchu, C., Vandevijvere, S., Waterlander, W., Thornton, L. E., Kelly, B., Cameron, A. J., Snowdon, W., & Swinburn, B. (2013). Monitoring the availability of healthy and unhealthy foods and non-alcoholic beverages in community and consumer retail food environments globally. *Obesity reviews, 14 Suppl 1*, 108-119. <u>https://doi.</u> <u>org/10.1111/obr.12080</u>

- Nikolaou, C. K., Hankey, C. R., & Lean, M. E. J. (2015, 2015/03/01). Calorie-labelling: does it impact on calorie purchase in catering outlets and the views of young adults? *International Journal of Obesity*, 39(3), 542-545. <u>https://doi.org/10.1038/</u> ijo.2014.162
- Nordborg, M., Davis, J., Cederberg, C., & Woodhouse, A. (2017). Freshwater ecotoxicity impacts from pesticide use in animal and vegetable foods produced in Sweden. *Science of The Total Environment*, *581-582*, 448-459. <u>https://doi.org/10.1016/j.</u> <u>scitotenv.2016.12.153</u>
- Nordic Nutrition Recommendations 2012 : Integrating nutrition and physical activity. (2014). (5 ed.). Nordisk Ministerråd. <u>https://</u> norden.diva-portal.org/smash/get/ diva2:704251/FULLTEXT01.pdf
- Nuutila, J., Risku-Norja, H., & Arolaakso, A. (2019). Public kitchen menu substitutions increase organic share and school meal sustainability at equal cost. *Organic Agriculture*, *9*(1), 117-126. <u>https://doi.</u> <u>org/10.1007/s13165-018-0215-0</u>
- Nyborg, K., Anderies, J. M., Dannenberg, A., Lindahl, T., Schill, C., Schlüter, M., Adger, W. N., Arrow, K. J., Barrett, S., Carpenter, S., Chapin, F. S., Crépin, A.-S., Daily, G., Ehrlich, P., Folke, C., Jager, W., Kautsky, N., Levin, S. A., Madsen, O. J., Polasky, S., Scheffer, M., Walker, B., Weber, E. U., Wilen, J., Xepapadeas, A., & de Zeeuw, A. (2016). Social norms as solutions. *Science, 354*(6308), 42-43. <u>https://doi.org/ doi</u>:10.1126/science.aaf8317
- O'Connor, N., Farag, K., & Baines, R. (2016). What is food poverty? A conceptual framework. *British Food Journal*, *118*(2), 429-449. <u>https://doi.org/10.1108/</u> <u>BFJ-06-2015-0222</u>
- Obeid, C. A., Gubbels, J. S., Jaalouk, D., Kremers, S. P. J., & Oenema, A. (2022). Adherence to the Mediterranean diet among adults in Mediterranean countries: a systematic literature review. *Eur J Nutr*, *61*(7), 3327-3344. <u>https://doi.org/10.1007/ s00394-022-02885-0</u>
- Oke, S., & Tan, M. (2022). Techniques for Advertising Healthy Food in School Settings to Increase Fruit and Vegetable Consumption. *INQUIRY: The Journal of Health Care Organization, Provision, and Financing, 59*, 00469580221100165. <u>https://</u> doi.org/10.1177/00469580221100165

- Onorati, M. G., & Giardullo, P. (2020). Social media as taste re-mediators: emerging patterns of food taste on TripAdvisor. *Food*, *Culture & Society*, *23*(3), 347-365. <u>https://doi. org/10.1080/15528014.2020.1715074</u>
- Orfanos, P., Naska, A., Trichopoulos, D., Slimani, N., Ferrari, P., van Bakel, M., Deharveng, G., Overvad, K., Tjønneland, A., Halkjær, J., Santucci de Magistris, M., Tumino, R., Pala, V., Sacerdote, C., Masala, G., Skeie, G., Engeset, D., Lund, E., Jakszyn, P., Barricarte, A., Chirlague, M.-D., Martinez-Garcia, C., Amiano, P., Quirós, J. R., Bingham, S., Welch, A., Spencer, E. A., Key, T. J., Rohrmann, S., Linseisen, J., Ray, J., Boeing, H., Peeters, P. H., Bueno-de-Mesquita, H. B., Ocke, M., Johansson, I., Johansson, G., Berglund, G., Manjer, J., Boutron-Ruault, M.-C., Touvier, M., Clavel-Chapelon, F., & Trichopoulou, A. (2007). Eating out of home and its correlates in 10 European countries. The European Prospective Investigation into Cancer and Nutrition (EPIC) study. Public Health Nutrition, 10(12), 1515-1525. https:// doi.org/10.1017/S1368980007000171
- Osmani, S. R. (1993). The Entitlement Approach to Famine: An Assessment - Working Paper No. 107. <u>https://www.wider.unu.edu/sites/ default/files/WP107.pdf</u>
- Ostrom, E. (2009). A General Framework for Analyzing Sustainability of Social-Ecological Systems. *Science*, *325*(5939), 419-422. <u>https://doi.org/doi</u>:10.1126/ science.1172133
- Pancrazi, R., van Rens, T., & Vukotić, M. (2022). How distorted food prices discourage a healthy diet. *Science Advances, 8*(13), eabi8807. <u>https://doi.org/10.1126/sciadv.</u> <u>abi8807</u>
- Pechey, R., Reynolds, J. P., Cook, B., Marteau, T. M., & Jebb, S. A. (2022). Acceptability of policies to reduce consumption of red and processed meat: A populationbased survey experiment. *Journal of Environmental Psychology*, *81*, 101817. <u>https://doi.org/10.1016/j.jenvp.2022.101817</u>
- Peñalvo, J. L., Cudhea, F., Micha, R., Rehm, C. D., Afshin, A., Whitsel, L., Wilde, P., Gaziano, T., Pearson-Stuttard, J., O'Flaherty, M., Capewell, S., & Mozaffarian, D. (2017). The potential impact of food taxes and subsidies on cardiovascular disease and diabetes burden and disparities in the United States. *BMC medicine*, *15*(1), 208. <u>https://doi.org/10.1186/s12916-017-0971-9</u>

- Penchansky, R., & Thomas, J. W. (1981). The concept of access: definition and relationship to consumer satisfaction. *Medical Care, 19*(2), 127-140. <u>https://doi. org/10.1097/00005650-198102000-00001</u>
- Pendrill, F., Persson, U. M., Godar, J., Kastner, T., Moran, D., Schmidt, S., & Wood, R. (2019). Agricultural and forestry trade drives large share of tropical deforestation emissions. *Global Environmental Change*, 56, 1-10. <u>https://doi.org/10.1016/j.</u> <u>gloenvcha.2019.03.002</u>
- Penne, T., & Goedemé, T. (2021). Can low-income households afford a healthy diet? Insufficient income as a driver of food insecurity in Europe. *Food policy*, *99*, 101978. <u>https://doi.org/10.1016/j.</u> <u>foodpol.2020.101978</u>
- Penney, T. L., Almiron-Roig, E., Shearer, C., McIsaac, J. L., & Kirk, S. F. (2014). Modifying the food environment for childhood obesity prevention: challenges and opportunities. *Proceedings of the Nutrition Society*, 73(2), 226-236. <u>https://doi.org/10.1017/</u> <u>s0029665113003819</u>
- Pérez Roda, M. A., Gilman, E., Huntington, T., Kennelly, S. J., & Suuronen, P. (2019). *A third assessment of global marine fisheries discards*. Food and Agriculture Organization of the United Nations. <u>https://</u> <u>www.fao.org/documents/card/en/c/</u> <u>CA0300EN/</u>
- Perez-Cueto, F. J. A., Rini, L., Faber, I., Rasmussen, M. A., Bechtold, K.-B., Schouteten, J. J., & De Steur, H. (2022). How barriers towards plant-based food consumption differ according to dietary lifestyle: Findings from a consumer survey in 10 EU countries. *International Journal of Gastronomy and Food Science, 29*, 100587. https://doi.org/10.1016/j.ijgfs.2022.100587
- Perino, G., & Schwickert, H. (2023). Animal welfare is a stronger determinant of public support for meat taxation than climate change mitigation in Germany. Nature Food, 4(2), 160-169. <u>https://doi. org/10.1038/s43016-023-00696-y</u>
- Perino, G., & Schwirplies, C. (2022). Meaty arguments and fishy effects: Field experimental evidence on the impact of reasons to reduce meat consumption. *Journal of Environmental Economics and Management, 114,* 102667. <u>https://doi. org/10.1016/j.jeem.2022.102667</u>

- Pieper, M., Michalke, A., & Gaugler, T. (2020). Calculation of external climate costs for food highlights inadequate pricing of animal products. *Nature Communications*, *11*(1), 1-13. <u>https://doi.org/10.1038/</u> <u>\$41467-020-19474-6</u>
- Pinho, M., Mackenbach, J. D., Oppert, J.
 M., Charreire, H., Bárdos, H., Rutter, H., Compernolle, S., Beulens, J., Brug, J., & Lakerveld, J. (2019). Exploring absolute and relative measures of exposure to food environments in relation to dietary patterns among European adults. *Public Health Nutrition, 22*(6), 1037-1047. <u>https://doi. org/10.1017/s1368980018003063</u>
- Pinto-Correia, T., Rivera, M., Guarín, A., Grivins, M., Tisenkopfs, T., & Hernández, P. A. (2021). Unseen food: The importance of extramarket small farm's production for rural households in Europe. *Global Food Security*, *30*, 100563. <u>https://doi.org/10.1016/j.</u> <u>gfs.2021.100563</u>
- Pitt, A., & Bendavid, E. (2017). Effect of meat price on race and gender disparities in obesity, mortality and quality of life in the US: a model-based analysis. *PLOS ONE*, *12*(1), e0168710. <u>https://doi.org/10.1371/</u> journal.pone.0168710
- Ponte, S. (2012). The Marine Stewardship Council (MSC) and the Making of a Market for 'Sustainable Fish'. Journal of Agrarian Change, 12(2-3), 300-315. <u>https://doi. org/10.1111/j.1471-0366.2011.00345.x</u>
- Poore, J., & Nemecek, T. (2018). Reducing food's environmental impacts through producers and consumers. *Science*, *360*(6392), 987-992. <u>https://doi.org/10.1126/</u> <u>science.aaq0216</u>
- Popkin, B. M., & Ng, S. W. (2022). The nutrition transition to a stage of high obesity and noncommunicable disease prevalence dominated by ultra-processed foods is not inevitable. *Obesity reviews, 23*(1), e13366. <u>https://doi.org/10.1111/obr.13366</u>
- Portingale, J., Eddy, S., Fuller-Tyszkiewicz, M., Liu, S., Giles, S., & Krug, I. (2023). Tonight, I'm disordered eating: The effects of food delivery app use, loneliness, and mood on daily body dissatisfaction and disordered eating urges. *Appetite, 180*, 106310. <u>https://</u> doi.org/10.1016/j.appet.2022.106310
- Potter, C., Bastounis, A., Hartmann-Boyce, J., Stewart, C., Frie, K., Tudor, K., Bianchi, F., Cartwright, E., Cook, B., Rayner, M., & Jebb,

S. A. (2021). The Effects of Environmental Sustainability Labels on Selection, Purchase, and Consumption of Food and Drink Products: A Systematic Review. *Environment and Behavior, 53*(8), 891-925. https://doi.org/10.1177/0013916521995473

Powell, L. M., Chriqui, J. F., Khan, T., Wada, R., & Chaloupka, F. J. (2013). Assessing the potential effectiveness of food and beverage taxes and subsidies for improving public health: a systematic review of prices, demand and body weight outcomes. *Obesity reviews*, *14*(2), 110-128. <u>https://doi.org/10.1111/obr.12002</u>

Public Health England. (2020). Sugar reduction - Report on progress between 2015 and 2019. https://assets.publishing.service.gov.uk/ government/uploads/system/uploads/ attachment_data/file/984282/Sugar_ reduction_progress_report_2015_to_2019-1. pdf

- Qi, D., Li, R., Penn, J., Houghtaling, B., Prinyawiwatkul, W., & Roe, B. E. (2022). Nudging greater vegetable intake and less food waste: A field experiment. *Food policy*, *112*, 102369. <u>https://doi.org/10.1016/j.</u> <u>foodpol.2022.102369</u>
- Quested, T. E., Marsh, E., Stunell, D., & Parry, A. D. (2013). Spaghetti soup: The complex world of food waste behaviours. *Resources, Conservation and Recycling,* 79, 43-51. <u>https://doi.org/10.1016/j.</u> <u>resconrec.2013.04.011</u>
- Ranta, M. (2014). Sources for Comparative Analysis of European Dietary Habits in Relation to Health and Risk Issues: A short note. *Anthropology of food*(S10). <u>https://doi. org/10.4000/aof.7676</u>
- Räty, R., & Carlsson-Kanyama, A. (2010). Energy consumption by gender in some European countries. *Energy Policy, 38*(1), 646-649. <u>https://doi.org/10.1016/j.</u> <u>enpol.2009.08.010</u>
- Read, Q. D., Brown, S., Cuéllar, A. D., Finn, S. M., Gephart, J. A., Marston, L. T., Meyer, E., Weitz, K. A., & Muth, M. K. (2020). Assessing the environmental impacts of halving food loss and waste along the food supply chain. *Science of The Total Environment*, *712*, 136255. <u>https://doi.org/10.1016/j.</u> <u>scitotenv.2019.136255</u>
- Reganold, J. P., & Wachter, J. M. (2016). Organic agriculture in the twenty-first

century. *Nature Plants, 2*(2), 15221. <u>https://</u> doi.org/10.1038/nplants.2015.221

- Reisch, L. A., Sunstein, C. R., Andor, M. A., Doebbe, F. C., Meier, J., & Haddaway, N. R. (2021). Mitigating climate change via food consumption and food waste: A systematic map of behavioral interventions. *Journal of Cleaner Production, 279*, 123717. <u>https://doi. org/10.1016/j.jclepro.2020.123717</u>
- Reisch, L., Eberle, U., & Lorek, S. (2013). Sustainable food consumption: an overview of contemporary issues and policies. *Sustainability: Science, Practice and Policy, g*(2), 7-25. <u>https://doi.org/10.108</u> 0/15487733.2013.11908111
- Renting, H., Marsden, T. K., & Banks, J. (2003). Understanding Alternative Food Networks: Exploring the Role of Short Food Supply Chains in Rural Development. *Environment and Planning A: Economy and Space, 35*(3), 393-411. <u>https://doi.org/10.1068/a3510</u>
- Reyes, M., Garmendia, M. L., Olivares, S., Aqueveque, C., Zacarías, I., & Corvalán, C. (2019). Development of the Chilean frontof-package food warning label. *BMC Public Health*, *19*(1), 906. <u>https://doi.org/10.1186/</u> <u>\$12889-019-7118-1</u>
- Reyes, M., Smith Taillie, L., Popkin, B., Kanter, R., Vandevijvere, S., & Corvalán, C. (2020). Changes in the amount of nutrient of packaged foods and beverages after the initial implementation of the Chilean Law of Food Labelling and Advertising: A nonexperimental prospective study. *PLOS Medicine*, *17*(7), e1003220. <u>https://doi.</u> org/10.1371/journal.pmed.1003220
- Riches, G. (2018). Food bank nations: Poverty, corporate charity and the right to food. Routledge.
- Riches, G., & Silvasti, T. (2014). *First world hunger revisited: Food charity or the right to food?* Springer.
- Riksrevisjonen. (2016). *Riksrevisjonens* undersøking av arbeidet til styresmaktene for å nå måla om økologisk landbruk. Dokument 3:7 (2015–2016). <u>https://www.</u> riksrevisjonen.no/globalassets/rapporter/ no-2015-2016/okologisklandbruk.pdf
- Rippin, H. L., Hutchinson, J., Jewell, J., Breda, J. J., & Cade, J. E. (2017). Adult Nutrient Intakes from Current National Dietary Surveys of European Populations. *Nutrients, g*(12). https://doi.org/10.3390/nug121288

- Rippin, H. L., Hutchinson, J., Jewell, J., Breda, J. J., & Cade, J. E. (2019). Child and adolescent nutrient intakes from current national dietary surveys of European populations. *Nutrition Research Reviews*, 32(1), 38-69. <u>https://doi.org/10.1017/</u> <u>\$0954422418000161</u>
- Risku-Norja, H., & Løes, A.-K. (2017, 2017/06/01). Organic food in food policy and in public catering: lessons learned from Finland. *Organic Agriculture*, 7(2), 111-124. <u>https://doi.org/10.1007/</u> <u>\$13165-016-0148-4</u>
- Rivera-Ferre, M. G., Ortega-Cerdà, M., & Baumgärtner, J. (2013). Rethinking Study and Management of Agricultural Systems for Policy Design. *Sustainability*, 5(9), 3858-3875. <u>https://www.mdpi.</u> <u>com/2071-1050/5/9/3858</u>
- Roberto, C. A., Ng, S. W., Ganderats-Fuentes, M., Hammond, D., Barquera, S., Jauregui, A., & Taillie, L. S. (2021). The influence of frontof-package nutrition labeling on consumer behavior and product reformulation. *Annual Review of Nutrition*, 41, 529-550. <u>https://doi. org/10.1146/annurev-nutr-111120-094932</u>
- Rodríguez-González, P., Iglesias, D., Fernandez-Rio, J., & Gao, Z. (2023). Effectiveness of interventions using apps to improve physical activity, sedentary behavior and diet: An umbrella review. *Complementary Therapies in Clinical Practice*, *50*, 101711. <u>https://doi.</u> <u>org/10.1016/j.ctcp.2022.101711</u>
- Rodríguez-Ortega, T., Oteros-Rozas, E., Ripoll-Bosch, R., Tichit, M., Martín-López, B., & Bernués, A. (2014). Applying the ecosystem services framework to pasturebased livestock farming systems in Europe. *Animal*, 8(8), 1361-1372. <u>https://doi.</u> org/10.1017/S1751731114000421
- Rogers, N. T., Cummins, S., Forde, H., Jones, C. P., Mytton, O., Rutter, H., Sharp, S. J., Theis, D., White, M., & Adams, J. (2023). Associations between trajectories of obesity prevalence in English primary school children and the UK soft drinks industry levy: An interrupted time series analysis of surveillance data. *PLOS Medicine, 20*(1), e1004160. <u>https://doi. org/10.1371/journal.pmed.1004160</u>
- Ronis, M. J. J., Pedersen, K. B., & Watt, J. (2018). Adverse Effects of Nutraceuticals and Dietary Supplements. *Annual Review of Pharmacology and Toxicology,*

58(1), 583-601. <u>https://doi.org/10.1146/</u> <u>annurev-pharmtox-010617-052844</u>

Röös, E., Bajželj, B., Smith, P., Patel, M., Little, D., & Garnett, T. (2017). Greedy or needy? Land use and climate impacts of food in 2050 under different livestock futures. *Global Environmental Change*, *47*, 1-12. <u>https://doi. org/10.1016/j.gloenvcha.2017.09.001</u>

Röös, E., Mayer, A., Muller, A., Kalt, G.,
Ferguson, S., Erb, K.-H., Hart, R., Matej, S.,
Kaufmann, L., Pfeifer, C., Frehner, A., Smith,
P., & Schwarz, G. (2022). Agroecological
practices in combination with healthy diets
can help meet EU food system policy
targets. Science of The Total Environment,
847, 157612. https://doi.org/10.1016/j.
scitotenv.2022.157612

- Röös, E., Mie, A., Wivstad, M., Salomon, E., Johansson, B., Gunnarsson, S., Wallenbeck, A., Hoffmann, R., Nilsson, U., & Sundberg, C. (2018). Risks and opportunities of increasing yields in organic farming. A review. Agronomy for sustainable development, 38, 1-21. <u>https://doi. org/10.1007/s13593-018-0489-3</u>
- Roosen, J., & Marette, S. (2011). Making the 'right' choice based on experiments: regulatory decisions for food and health. *European Review of Agricultural Economics,* 38(3), 361-381. <u>https://doi.org/10.1093/</u> <u>erae/jbr026</u>
- Roosen, J., Staudigel, M., & Rahbauer, S. (2022). Demand elasticities for fresh meat and welfare effects of meat taxes in Germany. *Food policy*, *106*, 102194. <u>https://</u> doi.org/10.1016/j.foodpol.2021.102194
- Rosenfeld, D. L., & Tomiyama, A. J. (2021). Gender differences in meat consumption and openness to vegetarianism. *Appetite*, *166*, 105475. <u>https://doi.org/10.1016/j.</u> <u>appet.2021.105475</u>

Rotz, S., Gravely, E., Mosby, I., Duncan, E., Finnis, E., Horgan, M., LeBlanc, J., Martin, R., Neufeld, H. T., Nixon, A., Pant, L., Shalla, V., & Fraser, E. (2019). Automated pastures and the digital divide: How agricultural technologies are shaping labour and rural communities. *Journal of Rural Studies*, *68*, 112-122. <u>https://doi.org/10.1016/j.</u> jrurstud.2019.01.023

Roux, N., Kastner, T., Erb, K.-H., & Haberl, H. (2021). Does agricultural trade reduce pressure on land ecosystems? Decomposing drivers of the embodied human appropriation of net primary production. *Ecological Economics*, *181*, 106915. <u>https://doi.org/10.1016/j</u>. <u>ecolecon.2020.106915</u>

Rozin, P., Hormes, J. M., Faith, M. S., & Wansink, B. (2012). Is meat male? A quantitative multimethod framework to establish metaphoric relationships. *Journal of Consumer Research*, *39*(3), 629-643. <u>https://</u> doi.org/10.1086/664970

- Säll, S. (2018). Environmental food taxes and inequalities: Simulation of a meat tax in Sweden. *Food policy, 74,* 147-153. <u>https://</u> <u>doi.org/10.1016/j.foodpol.2017.12.007</u>
- Samsioe, E., & Fuentes, C. (2022). Digitalizing shopping routines: Re-organizing household practices to enable sustainable food provisioning. *Sustainable Production and Consumption, 29*, 807-819. <u>https://doi. org/10.1016/j.spc.2021.07.019</u>

Sánchez-Bayo, F., & Wyckhuys, K. A. G. (2019). Worldwide decline of the entomofauna: A review of its drivers. *Biological Conservation*, 232, 8-27. <u>https://doi.org/10.1016/j.</u> <u>biocon.2019.01.020</u>

Sánchez-Romero, L. M., Penko, J., Coxson, P. G., Fernández, A., Mason, A., Moran, A. E., Ávila-Burgos, L., Odden, M., Barquera, S., & Bibbins-Domingo, K. (2016). Projected Impact of Mexico's Sugar-Sweetened Beverage Tax Policy on Diabetes and Cardiovascular Disease: A Modeling Study. *PLOS Medicine, 13*(11), e1002158. <u>https://doi. org/10.1371/journal.pmed.1002158</u>

Sandström, V., Valin, H., Krisztin, T., Havlík, P., Herrero, M., & Kastner, T. (2018, 2018/12/01/). The role of trade in the greenhouse gas footprints of EU diets. *Global Food Security*, 19, 48-55. <u>https://doi. org/10.1016/j.gfs.2018.08.007</u>

SAPEA. (2020). A Sustainable Food System for the European Union. <u>https://sapea.info/ topic/food/</u>

Saulais, L., Massey, C., Perez-Cueto, F. J. A., Appleton, K. M., Dinnella, C., Monteleone, E., Depezay, L., Hartwell, H., & Giboreau, A. (2019). When are "Dish of the Day" nudges most effective to increase vegetable selection? *Food policy*, *85*, 15-27. <u>https://doi. org/10.1016/j.foodpol.2019.04.003</u>

Scapin, T., Fernandes, A. C., Curioni, C.
C., Pettigrew, S., Neal, B., Coyle, D. H.,
Rodrigues, V. M., Bernardo, G. L., Uggioni, P.
L., & Proença, R. P. (2021). Influence of sugar

label formats on consumer understanding and amount of sugar in food choices: a systematic review and meta-analyses. *Nutrition Reviews, 79*(7), 788-801. <u>https://</u> doi.org/10.1093/nutrit/nuaa108

- Scarborough, P., Adhikari, V., Harrington, R. A., Elhussein, A., Briggs, A., Rayner, M., Adams, J., Cummins, S., Penney, T., & White, M. (2020). Impact of the announcement and implementation of the UK Soft Drinks Industry Levy on sugar content, price, product size and number of available soft drinks in the UK, 2015-19: A controlled interrupted time series analysis. *PLOS Medicine*, *17*(2), e1003025. <u>https://doi.</u> org/10.1371/journal.pmed.1003025
- Scarry, A., Rice, J., O'Connor, E. M., & Tierney, A. C. (2022). Usage of Mobile Applications or Mobile Health Technology to Improve Diet Quality in Adults. *Nutrients*, *14*(12). <u>https://</u> <u>doi.org/10.3390/nu14122437</u>
- Schanes, K., Dobernig, K., & Gözet, B. (2018). Food waste matters-A systematic review of household food waste practices and their policy implications. *Journal of Cleaner Production, 182*, 978-991. <u>https://doi. org/10.1016/j.jclepro.2018.02.030</u>
- Schermer, M. (2015). From "Food from Nowhere" to "Food from Here:" changing producer–consumer relations in Austria. *Agriculture and Human Values*, *32*(1), 121-132. <u>https://doi.org/10.1007/</u> <u>\$10460-014-9529-z</u>
- Schlatter, B., Trávníček, J., Meier, C., & Willer, H. (2022). Current statistics on organic agriculture worldwide: Area, operators and market. In H. Willer, J. Trávníček, C. Meier, & B. Schlatter (Eds.), *The world of organic agriculture: statistics & emerging trends* 2022 (pp. 32-137). FiBL IFOMA-Organics international.
- Schmitt, E., Galli, F., Menozzi, D., Maye, D., Touzard, J.-M., Marescotti, A., Six, J., & Brunori, G. (2017). Comparing the sustainability of local and global food products in Europe. *Journal of Cleaner Production*, *165*, 346-359. <u>https://doi.org/10.1016/jjclepro.2017.07.039</u>
- Schmutzler, A., & Goulder, L. H. (1997). The choice between emission taxes and output taxes under imperfect monitoring. *Journal of Environmental Economics and Management, 32*(1), 51-64. <u>https://doi. org/10.1006/jeem.1996.0953</u>

- Schneider, S., Schilling, L., & Osenbrügge, N. (2021). Determinants of soft drink consumption among children and adolescents in developed countries - a systematic review. *Central European Journal of Public Health, 29*(4), 290-300. <u>https://doi.org/http://dx.doi.org/10.21101/</u> <u>ceiph.a6755</u>
- Schneider, T., & Eli, K. (2021). Fieldwork in online foodscapes: How to bring an ethnographic approach to studies of digital food and digital eating. In J. K. Leer, S. G.
 S. (Ed.), *Research Methods in Digital Food Studies* (pp. 71-85). Routledge.
- Schneider, T., & Eli, K. (2022). The digital labor of ethical food consumption: a new research agenda for studying everyday food digitalization. *Agriculture and Human Values*. <u>https://doi.org/10.1007/</u> <u>\$10460-022-10390-7</u>
- Schneider, T., Eli, K., Dolan, C. S., & Ulijaszek, S. J. (2018). *Digital food activism* (First issued in paperback ed.). Routledge Taylor & Francis Group.
- Schulz, C.-A., Oluwagbemigun, K., & Nöthlings, U. (2021). Advances in dietary pattern analysis in nutritional epidemiology. *European Journal of Nutrition, 60*(8), 4115-4130. <u>https://doi.org/10.1007/</u> <u>\$00394-021-02545-9</u>
- Schulze, M., Spiller, A., & Jürkenbeck, K. (2022). Politicised opinion leaders in the younger generation: to meat or not to meat? *British Food Journal*, *124*(11), 3907-3921. <u>https://doi.org/10.1108/BFJ-07-2021-0817</u>
- Schupp, J. L., & Sharp, J. S. (2012, 2012/03/01). Exploring the social bases of home gardening. *Agriculture and Human Values*, *29*(1), 93-105. <u>https://doi.org/10.1007/</u> <u>\$10460-011-9321-2</u>
- Schvartzman, Y. (2012). Metastyring af markedsudvikling: policystrategier og netværkskoordinering i udvikling af nye markeder : et komparativt studie af udviklingen af det danske og det svenske økologimarked. Forlaget Politica. https://politica.dk/fileadmin/politica/ Dokumenter/Afhandlinger/yonatan_ schvartzman.pdf
- Scott, W. R. (1995). Institutions and organizations: ideas, interests, and identities. SAGE.
- Searchinger, T. D., Wirsenius, S., Beringer, T., & Dumas, P. (2018). Assessing the efficiency of

changes in land use for mitigating climate change. *Nature*, *564*(7735), 249-253. <u>https://</u> <u>doi.org/10.1038/s41586-018-0757-z</u>

- Sen, A. (1981). Poverty and Famines: An Essay on Entitlement and Deprivation. Oxford University Press. <u>https://doi.org/10.1093/0</u> 198284632.001.0001
- Sen, A. (1984). *Resources, Values and Development.* Basil Blackwell.
- Seufert, V., & Ramankutty, N. (2017). Many shades of gray—The context-dependent performance of organic agriculture. *Science Advances, 3*(3), e1602638. <u>https://</u> <u>doi.org/10.1126/sciadv.1602638</u>
- Shangguan, S., Afshin, A., Shulkin, M., Ma, W., Marsden, D., Smith, J., Saheb-Kashaf, M., Shi, P., Micha, R., & Imamura, F. (2019). A meta-analysis of food labeling effects on consumer diet behaviors and industry practices. *American Journal of Preventive Medicine*, *56*(2), 300-314. <u>https://doi. org/10.1016/j.amepre.2018.09.024</u>
- Shaw, S. C., Ntani, G., Baird, J., & Vogel, C. A. (2020). A systematic review of the influences of food store product placement on dietary-related outcomes. *Nutrition Reviews*, 78(12), 1030-1045. <u>https://</u> doi.org/10.1093/nutrit/nuaa024
- Shove, E. (2010). Beyond the ABC: Climate Change Policy and Theories of Social Change. *Environment and Planning A: Economy and Space, 42*(6), 1273-1285. <u>https://doi.org/10.1068/a42282</u>
- Sievert, K., Lawrence, M., Parker, C., & Baker, P. (2021). Understanding the political challenge of red and processed meat reduction for healthy and sustainable food systems: a narrative review of the literature. *International journal of health policy and management, 10*(Special Issue on Political Economy of Food Systems), 793-808. https://doi.org/10.34172/IJHPM.2020.238
- Sievert, K., Lawrence, M., Parker, C., & Baker, P. (2022). What's really at 'steak'? Understanding the global politics of red and processed meat reduction: A framing analysis of stakeholder interviews. *Environmental Science & Policy, 137*, 12-21. <u>https://doi.org/10.1016/j.</u> <u>envsci.2022.08.007</u>
- Sievert, K., Lawrence, M., Parker, C., Russell, C. A., & Baker, P. (2022). Who has a beef with reducing red and processed meat consumption? A media framing

analysis. *Public Health Nutrition, 25*(3), 578-590. <u>https://doi.org/10.1017/</u> <u>s1368980021004092</u>

- Silva, A., Higgins, L. M., & Hussein, M. (2015). An Evaluation of the Effect of Child-Directed Television Food Advertising Regulation in the United Kingdom. *Canadian Journal of Agricultural Economics/Revue canadienne d'agroeconomie*, *63*(4), 583-600. <u>https://doi. org/10.1111/cjag.12078</u>
- Sinclair, S. E., Cooper, M., & Mansfield, E. D. (2014). The Influence of Menu Labeling on Calories Selected or Consumed: A Systematic Review and Meta-Analysis. *Journal of the Academy of Nutrition and Dietetics*, 114(9), 1375-1388.e1315. <u>https://doi.org/10.1016/j.jand.2014.05.014</u>
- Sing, F., & Backholer, K. (2023). Strengthening Global Legislative Actions to Protect Children from the Harmful Impacts of Unhealthy Food and Non-alcoholic Beverage Marketing. *Current Obesity Reports*. <u>https://doi.org/10.1007/</u> <u>\$13679-023-00492-6</u>
- Slimani, N., Fahey, M., Welch, A., Wirfält,
 E., Stripp, C., Bergström, E., Linseisen, J.,
 Schulze, M. B., Bamia, C., Chloptsios, Y.,
 Veglia, F., Panico, S., Bueno-de-Mesquita,
 H. B., Ocké, M. C., Brustad, M., Lund, E.,
 González, C. A., Barcos, A., Berglund,
 G., Winkvist, A., Mulligan, A., Appleby,
 P., Overvad, K., Tjønneland, A., ClavelChapelon, F., Kesse, E., Ferrari, P., Van
 Staveren, W. A., & Riboli, E. (2002). Diversity
 of dietary patterns observed in the
 European Prospective Investigation into
 Cancer and Nutrition (EPIC) project. *Public Health Nutrition*, *5*(6b), 1311-1328. https://doi.org/10.1079/PHN2002407
- Smith, J., & Jehlička, P. (2013). Quiet sustainability: Fertile lessons from Europe's productive gardeners. *Journal* of Rural Studies, 32, 148-157. <u>https://doi.</u> org/10.1016/j.jrurstud.2013.05.002
- Smith, J., Kostelecký, T., & Jehlička, P. (2015). Quietly does it: Questioning assumptions about class, sustainability and consumption. *Geoforum, 67*, 223-232. <u>https://doi. org/10.1016/j.geoforum.2015.03.017</u>
- Smith, L. G., Kirk, G. J. D., Jones, P. J., & Williams, A. G. (2019). The greenhouse gas impacts of converting food production in England and Wales to organic methods. *Nature Communications*, *10*(1), 4641. <u>https://</u> doi.org/10.1038/s41467-019-12622-7

- Smith, P., Haberl, H., Popp, A., Erb, K.-h., Lauk, C., Harper, R., Tubiello, F. N., de Siqueira Pinto, A., Jafari, M., Sohi, S., Masera, O., Böttcher, H., Berndes, G., Bustamante, M., Ahammad, H., Clark, H., Dong, H., Elsiddig, E. A., Mbow, C., Ravindranath, N. H., Rice, C. W., Robledo Abad, C., Romanovskaya, A., Sperling, F., Herrero, M., House, J. I., & Rose, S. (2013). How much land-based greenhouse gas mitigation can be achieved without compromising food security and environmental goals? *Global Change Biology*, *19*(8), 2285-2302. <u>https:// doi.org/10.1111/gcb.12160</u>
- Song, J., Brown, M. K., Tan, M., MacGregor, G. A., Webster, J., Campbell, N. R., Trieu, K., Ni Mhurchu, C., Cobb, L. K., & He, F. J. (2021). Impact of color-coded and warning nutrition labelling schemes: A systematic review and network meta-analysis. *PLOS Medicine*, *18*(10), e1003765. <u>https://doi. org/10.1371/journal.pmed.1003765</u>
- Sonnino, R., & Griggs-Trevarthen, C. (2013). A resilient social economy? Insights from the community food sector in the UK. *Entrepreneurship & Regional Development,* 25(3-4), 272-292. <u>https://doi.org/10.1080/0</u> 8985626.2012.710268
- Sonnino, R., & Marsden, T. (2005). Beyond the divide: rethinking relationships between alternative and conventional food networks in Europe. *Journal of Economic Geography*, 6(2), 181-199. <u>https://doi.org/10.1093/jeg/ lbi006</u>
- Sørensen, N. N., Sørensen, M.-L. K., Trolle, E., & Lassen, A. D. (2020). Organic Food in Public Catering: How the Danish Organic Cuisine Label May Maintain Organic Food Production in the Longer Term. *Journal of Culinary Science & Technology*, *18*(4), 255-269. <u>https://doi.org/10.1080/15428052</u> .2019.1582122
- Sørensen, N. N., Tetens, I., Løje, H., & Lassen, A. D. (2016). The effectiveness of the Danish Organic Action Plan 2020 to increase the level of organic public procurement in Danish public kitchens. *Public Health Nutrition*, *19*(18), 3428-3435. <u>https://doi.</u> org/10.1017/S1368980016001737
- Sosna, D., Brunclíková, L., & Galeta, P. (2019). Rescuing things: Food waste in the rural environment in Czechia. *Journal of Cleaner Production, 214,* 319-330. <u>https://doi. org/10.1016/j.jclepro.2018.12.214</u>

- Soutjis, B. (2020). The new digital face of the consumerist mediator: the case of the 'Yuka' mobile app. *Journal of Cultural Economy*, *13*(1), 114-131. <u>https://doi.org/10.1</u> 080/17530350.2019.1603116
- Sovová, L., Jehlička, P., & Daněk, P. (2021). Growing the Beautiful Anthropocene: Ethics of Care in East European Food Gardens. *Sustainability*, *13*(9), 5193. <u>https://www.mdpi.</u> <u>com/2071-1050/13/9/5193</u>
- Sparkman, G., Howe, L., & Walton, G. (2021). How social norms are often a barrier to addressing climate change but can be part of the solution. *Behavioural public policy*, 5(4), 528-555. <u>https://doi.org/10.1017/</u> <u>bpp.2020.42</u>
- Springmann, M., Godfray, H. C. J., Rayner, M., & Scarborough, P. (2016). Analysis and valuation of the health and climate change cobenefits of dietary change. *Proceedings of the National Academy of Sciences, 113*(15), 4146-4151. <u>https://doi.org/doi</u>:10.1073/ pnas.1523119113
- Springmann, M., Mason-D'Croz, D., Robinson, S., Wiebe, K., Godfray, H. C. J., Rayner, M., & Scarborough, P. (2018). Health-motivated taxes on red and processed meat: A modelling study on optimal tax levels and associated health impacts. *PLOS ONE*, *13*(11), e0204139. <u>https://doi.org/10.1371/journal.</u> <u>pone.0204139</u>
- Springmann, M., Spajic, L., Clark, M. A., Poore, J., Herforth, A., Webb, P., Rayner, M., & Scarborough, P. (2020). The healthiness and sustainability of national and global food based dietary guidelines: modelling study. *BMJ*, 370, m2322. <u>https://doi.org/10.1136/ bmj.m2322</u>
- Springmann, M., Wiebe, K., Mason-D'Croz, D., Sulser, T. B., Rayner, M., & Scarborough, P. (2018). Health and nutritional aspects of sustainable diet strategies and their association with environmental impacts: a global modelling analysis with countrylevel detail. *The Lancet Planetary Health*, 2(10), e451-e461. <u>https://doi.org/10.1016/ S2542-5196(18)30206-7</u>
- Stacey, N., Edoka, I., Hofman, K., Swart, E. C., Popkin, B., & Ng, S. W. (2021). Changes in beverage purchases following the announcement and implementation of South Africa's Health Promotion Levy: an observational study. *The Lancet Planetary Health*, *5*(4), e200-e208. <u>https://doi.</u> org/10.1016/S2542-5196(20)30304-1

- Staudigel, M., & Anders, S. (2020). Effects of the FDA's sodium reduction strategy in the U.S. market for chip products. *Journal of Economic Behavior & Organization, 173*(C), 216-238. <u>https://EconPapers.repec.org/</u> <u>RePEc</u>:eee:jeborg:v:173:y:2020:i:c:p:216-238
- Stautz, K., Zupan, Z., Field, M., & Marteau, T. M. (2018). Does self-control modify the impact of interventions to change alcohol, tobacco, and food consumption? A systematic review. *Health Psychol Rev*, *12*(2), 157-178. <u>https://doi.org/10.1080/17437199.2017.142</u> 1477
- Stoll-Kleemann, S., & Schmidt, U. J. (2017). Reducing meat consumption in developed and transition countries to counter climate change and biodiversity loss: a review of influence factors. *Regional Environmental Change*, *17*(5), 1261-1277. <u>https://doi.</u> org/10.1007/s10113-016-1057-5
- Strukturdirektoratet. (1999). Aktionsplan II: Økologi i udvikling. <u>https://www.yumpu.</u> <u>com/da/document/view/17631980/</u> <u>aktionsplan-ii-kologi-i-udvikling-</u> <u>strukturdirektoratet-januar-1999</u>
- Stuber, J. M., Lakerveld, J., Kievitsbosch, L. W., Mackenbach, J. D., & Beulens, J. W. J. (2022). Nudging customers towards healthier food and beverage purchases in a real-life online supermarket: a multi-arm randomized controlled trial. *BMC medicine*, *20*(1), 10. <u>https://doi.org/10.1186/ s12916-021-02205-z</u>
- Stuckler, D., McKee, M., Ebrahim, S., & Basu, S. (2012). Manufacturing Epidemics: The Role of Global Producers in Increased Consumption of Unhealthy Commodities Including Processed Foods, Alcohol, and Tobacco. *PLOS Medicine*, *9*(6), e1001235. <u>https://doi.org/10.1371/journal.</u> <u>pmed.1001235</u>
- Sundin, N., Rosell, M., Eriksson, M., Jensen, C., & Bianchi, M. (2021). The climate impact of excess food intake - An avoidable environmental burden. *Resources, Conservation and Recycling,* 174, 105777. https://doi.org/10.1016/j. resconrec.2021.105777
- Suri, S., Kumar, V., Kumar, S., Goyal, A., Tanwar, B., Kaur, J., & Kaur, J. (2020). DASH Dietary Pattern: A Treatment for Non-communicable Diseases. *Current Hypertension Reviews*, *16*(2), 108-114. <u>https://</u> doi.org/10.2174/1573402115666191007144 <u>608</u>

- Svedäng, H., & Hornborg, S. (2014). Selective fishing induces density-dependent growth. *Nature Communications*, *5*(1), 4152. <u>https://</u><u>doi.org/10.1038/ncomms5152</u>
- Swinburn, B. A., Sacks, G., Hall, K. D., McPherson, K., Finegood, D. T., Moodie, M. L., & Gortmaker, S. L. (2011). The global obesity pandemic: shaped by global drivers and local environments. *Lancet*, *378*(9793), 804-814. <u>https://doi.org/10.1016/</u> <u>s0140-6736(11)60813-1</u>
- Swinburn, B., Kraak, V., Rutter, H., Vandevijvere, S., Lobstein, T., Sacks, G., Gomes, F., Marsh, T., & Magnusson, R. (2015). Strengthening of accountability systems to create healthy food environments and reduce global obesity. *Lancet, 385*(9986), 2534-2545. <u>https://doi.org/10.1016/</u> <u>s0140-6736(14)61747-5</u>
- Swinburn, B., Sacks, G., Vandevijvere, S., Kumanyika, S., Lobstein, T., Neal, B., Barquera, S., Friel, S., Hawkes, C., Kelly, B., L'Abbé, M., Lee, A., Ma, J., Macmullan, J., Mohan, S., Monteiro, C., Rayner, M., Sanders, D., Snowdon, W., & Walker, C. (2013). INFORMAS (International Network for Food and Obesity/non-communicable diseases Research, Monitoring and Action Support): overview and key principles. *Obesity reviews, 14 Suppl 1*, 1-12. <u>https://doi. org/10.1111/obr.12087</u>
- Taillie, L. S., Bercholz, M., Popkin, B., Reyes, M., Colchero, M. A., & Corvalán, C. (2021). Changes in food purchases after the Chilean policies on food labelling, marketing, and sales in schools: a before and after study. *Lancet Planet Health*, 5(8), e526-e533. <u>https://doi.org/10.1016/ s2542-5196(21)00172-8</u>
- Taillie, L. S., Reyes, M., Colchero, M. A., Popkin, B., & Corvalán, C. (2020). An evaluation of Chile's Law of Food Labeling and Advertising on sugar-sweetened beverage purchases from 2015 to 2017: A beforeand-after study. *PLOS Medicine*, 17(2), e1003015. <u>https://doi.org/10.1371/journal.</u> <u>pmed.1003015</u>
- Taufik, D., Verain, M. C. D., Bouwman, E. P., & Reinders, M. J. (2019). Determinants of reallife behavioural interventions to stimulate more plant-based and less animal-based diets: A systematic review. *Trends in Food Science & Technology*, *93*, 281-303. <u>https://</u> doi.org/10.1016/j.tifs.2019.09.019

Teitelbaum, S., & Beckley, T. M. (2006). Harvested , Hunted and Home Grown : The Prevalence of Self-Provisioning in Rural Canada. *Journal of Rural and Community Development, 1*(2), 114-130. <u>https://journals.</u> <u>brandonu.ca/jrcd/article/view/9/4</u>

- Teng, A. M., Jones, A. C., Mizdrak, A., Signal, L., Genç, M., & Wilson, N. (2019). Impact of sugarsweetened beverage taxes on purchases and dietary intake: Systematic review and metaanalysis. *Obesity reviews*, 20(9), 1187-1204. <u>https://doi.org/10.1111/ obr.12868</u>
- Thomas, C., Breeze, P., Cummins, S., Cornelsen, L., Yau, A., & Brennan, A. (2022). The health, cost and equity impacts of restrictions on the advertisement of high fat, salt and sugar products across the transport for London network: a health economic modelling study. *International Journal of Behavioral Nutrition and Physical Activity*, 19(1), 93. https://doi.org/10.1186/ s12966-022-01331-y
- Thomé, K. M., Cappellesso, G., Ramos, E. L. A., & Duarte, S. C. d. L. (2021). Food Supply Chains and Short Food Supply Chains: Coexistence conceptual framework. *Journal of Cleaner Production*, *278*, 123207. <u>https://doi.org/10.1016/j.</u> jclepro.2020.123207
- Tilman, D., & Clark, M. (2014, 2014/11/01). Global diets link environmental sustainability and human health. *Nature*, *515*(7528), 518-522. <u>https://doi.org/10.1038/</u> <u>nature13959</u>
- Titis, E., Procter, R., & Walasek, L. (2022). Assessing physical access to healthy food across United Kingdom: A systematic review of measures and findings. *Obesity Science & Practice*, 8(2), 233-246. <u>https://doi.</u> <u>org/10.1002/osp4.563</u>
- Tobi, R. C., Harris, F., Rana, R., Brown, K. A., Quaife, M., & Green, R. (2019). Sustainable diet dimensions. Comparing consumer preference for nutrition, environmental and social responsibility food labelling: a systematic review. *Sustainability*, *11*(23), 6575.
- Torres-Álvarez, R., Barrán-Zubaran, R., Canto-Osorio, F., Sánchez-Romero, L. M., Camacho-García-Formentí, D., Popkin, B. M., Rivera, J. A., Meza, R., & Barrientos-Gutiérrez, T. (2020). Body weight impact of the sugar-sweetened beverages tax in Mexican children: A modeling study.

Pediatric Obesity, 15(8), e12636. <u>https://doi.org/10.1111/ijp0.12636</u>

- Toti, E., Di Mattia, C., & Serafini, M. (2019). Metabolic Food Waste and Ecological Impact of Obesity in FAO World's Region. *Frontiers in Nutrition, 6.* <u>https://doi.</u> <u>org/10.3389/fnut.2019.00126</u>
- Treich, N. (2021). Cultured Meat: Promises and Challenges. *Environmental and Resource Economics*, 79(1), 33-61. <u>https://doi.</u> <u>org/10.1007/s10640-021-00551-3</u>
- Trewern, J., Chenoweth, J., Christie, I., & Halevy, S. (2022). Does promoting plantbased products in Veganuary lead to increased sales, and a reduction in meat sales? A natural experiment in a supermarket setting. *Public Health Nutrition*, *25*(11), 3204-3214. <u>https://doi.org/10.1017/</u> <u>S1368980022001914</u>
- Trichopoulou, A., Martínez-González, M. A., Tong, T. Y. N., Forouhi, N. G., Khandelwal, S., Prabhakaran, D., Mozaffarian, D., & de Lorgeril, M. (2014). Definitions and potential health benefits of the Mediterranean diet: views from experts around the world. *BMC medicine*, *12*(1), 112. <u>https://doi. org/10.1186/1741-7015-12-112</u>
- Trichopoulou, A., Soukara, S., & Vasilopoulou, E. (2007). Traditional foods: a science and society perspective. *Trends in Food Science* & *Technology*, *18*(8), 420-427. <u>https://doi. org/10.1016/j.tifs.2007.03.007</u>
- Tselengidis, A., & Östergren, P.-O. (2019). Lobbying against sugar taxation in the European Union: Analysing the lobbying arguments and tactics of stakeholders in the food and drink industries. *Scandinavian Journal of Public Health*, *47*(5), 565-575. <u>https://doi.org/10.1177/1403494818787102</u>
- Turner, C., Aggarwal, A., Walls, H., Herforth, A., Drewnowski, A., Coates, J., Kalamatianou, S., & Kadiyala, S. (2018). Concepts and critical perspectives for food environment research: A global framework with implications for action in low- and middleincome countries. *Global Food Security*, 18, 93-101. <u>https://doi.org/10.1016/j.</u> <u>qfs.2018.08.003</u>
- UNEP. (2022). Emissions Gap Report 2022: The Closing Window — Climate crisis calls for rapid transformation of societies. <u>https://</u> www.unep.org/emissions-gap-report-2022
- van Berkum, S., Dengerink, J., & Ruben, R. (2018). The food systems approach:

sustainable solutions for a sufficient supply of healthy food. *Wageningen Economic Research Memorandum*(64), 29. <u>https://doi.</u> org/10.18174/451505

- van Bers, C., Delaney, A., Eakin, H., Cramer, L., Purdon, M., Oberlack, C., Evans, T., Pahl-Wostl, C., Eriksen, S., Jones, L., Korhonen-Kurki, K., & Vasileiou, I. (2019). Advancing the research agenda on food systems governance and transformation. *Current Opinion in Environmental Sustainability*, 39, 94-102. <u>https://doi.org/10.1016/j.</u> <u>cosust.2019.08.003</u>
- Van Dam, I., Allais, O., & Vandevijvere, S. (2022). Market concentration and the healthiness of packaged food and non-alcoholic beverage sales across the European single market. *Public Health Nutr*, *25*(11), 3131-3136. <u>https://doi.org/10.1017/</u> <u>\$1368980022001926</u>
- van der Werf, H. M. G., Knudsen, M. T., & Cederberg, C. (2020). Towards better representation of organic agriculture in life cycle assessment. *Nature Sustainability*, 3(6), 419-425. <u>https://doi.org/10.1038/</u> <u>\$41893-020-0489-6</u>
- van Doorn, J., & Verhoef, P. C. (2011). Willingness to pay for organic products: Differences between virtue and vice foods. *International Journal of Research in Marketing, 28*(3), 167-180. <u>https://doi. org/10.1016/j.ijresmar.2011.02.005</u>
- Van Doorn, J., Risselada, H., & Verhoef, P. C. (2021). Does sustainability sell? The impact of sustainability claims on the success of national brands' new product introductions. *Journal of Business Research*, *137*, 182-193. <u>https://doi.org/10.1016/j.</u> jbusres.2021.08.032
- van Selm, B., Frehner, A., de Boer, I. J. M., van Hal, O., Hijbeek, R., van Ittersum, M. K., Talsma, E. F., Lesschen, J. P., Hendriks, C. M. J., Herrero, M., & van Zanten, H. H. E. (2022). Circularity in animal production requires a change in the EAT-Lancet diet in Europe. *Nature Food*, *3*(1), 66-73. <u>https://doi. org/10.1038/s43016-021-00425-3</u>
- Vandenbroele, J., Slabbinck, H., Van Kerckhove, A., & Vermeir, I. (2021). Mock meat in the butchery: Nudging consumers toward meat substitutes. *Organizational Behavior and Human Decision Processes*, *163*, 105-116. <u>https://doi.org/10.1016/j.</u> <u>obhdp.2019.09.004</u>

- Vandevijvere, S., & Swinburn, B. (2014). Towards global benchmarking of food environments and policies to reduce obesity and diet-related non-communicable diseases: design and methods for nation-wide surveys. *BMJ open, 4*(5), e005339. <u>https://doi. org/10.1136/bmjopen-2014-005339</u>
- Vandevijvere, S., & Vanderlee, L. (2019). Effect of Formulation, Labelling, and Taxation Policies on the Nutritional Quality of the Food Supply. *Curr Nutr Rep*, *8*(3), 240-249. <u>https://doi.org/10.1007/</u> <u>\$13668-019-00289-x</u>
- VanEpps, E. M., Molnar, A., Downs, J. S., & Loewenstein, G. (2021). Choosing the Light Meal: Real-Time Aggregation of Calorie Information Reduces Meal Calories. *Journal of Marketing Research, 58*(5), 948-967. <u>https://doi. org/10.1177/00222437211022367</u>
- Vedung, E. (1998). Policy instruments: Typologies and theories. In E. Vedung, R. C. Rist, & M.-L. Bemelmans-Videc (Eds.), *Carrots, sticks & sermons: policy instruments and their evaluation* (pp. 21-58). Transaction Books.
- Verdonschot, A., Follong, B. M., Collins, C. E., de Vet, E., Haveman-Nies, A., & Bucher, T. (2022). Effectiveness of school-based nutrition intervention components on fruit and vegetable intake and nutrition knowledge in children aged 4-12 years old: an umbrella review. *Nutr Rev.* <u>https://doi.</u> <u>org/10.1093/nutrit/nuac057</u>
- Vermeir, I., Weijters, B., De Houwer, J., Geuens, M., Slabbinck, H., Spruyt, A., Van Kerckhove, A., Van Lippevelde, W., De Steur, H., & Verbeke, W. (2020). Environmentally Sustainable Food Consumption: A Review and Research Agenda From a Goal-Directed Perspective [Review]. *Frontiers in Psychology*, *11*. <u>https://doi.org/10.3389/</u> <u>fpsyg.2020.01603</u>
- Villas-Boas, S. B., Kiesel, K., Berning, J. P., Chouinard, H. H., & McCluskey, J. J. (2020). Consumer and Strategic Firm Response to Nutrition Shelf Labels. *American Journal of Agricultural Economics*, *102*(2), 458-479. <u>https://doi.org/10.1002/ajae.12002</u>
- VKM. (2022). Benefit and risk assessment of fish in the Norwegian diet - Scientific Opinion of the Scientific Committee of the Norwegian Scientific Committee for Food and Environment. https://vkm.no/

english/riskassessments/allpublications/ benefitandriskassessmentoffish inthenorwegiandiet.4. 7b65040716afa427d7ec5d3a.html

- Vogel, C., Crozier, S., Penn-Newman, D., Ball, K., Moon, G., Lord, J., Cooper, C., & Baird, J. (2021). Altering product placement to create a healthier layout in supermarkets: Outcomes on store sales, customer purchasing, and diet in a prospective matched controlled cluster study. *PLOS Medicine*, *18*(9), e1003729. <u>https://doi. org/10.1371/journal.pmed.1003729</u>
- von Philipsborn, P., Stratil, J. M., Burns, J., Busert, L. K., Pfadenhauer, L. M., Polus, S., Holzapfel, C., Hauner, H., & Rehfuess, E. (2019). Environmental interventions to reduce the consumption of sugarsweetened beverages and their effects on health. *Cochrane Database Syst Rev, 6*(6), Cd012292. <u>https://doi.</u> org/10.1002/14651858.CD012292.pub2
- Wägeli, S., Janssen, M., & Hamm, U. (2016). Organic consumers' preferences and willingness-to-pay for locally produced animal products. *International Journal of Consumer Studies*, 40(3), 357-367. <u>https://</u> doi.org/10.1111/ijcs.12262
- Walker, R. E., Keane, C. R., & Burke, J. G. (2010). Disparities and access to healthy food in the United States: A review of food deserts literature. *Health Place*, *16*(5), 876-884. <u>https://doi.org/10.1016/j.</u> <u>healthplace.2010.04.013</u>
- Warde, A. (2005). Consumption and Theories of Practice. *Journal of Consumer Culture, 5*(2), 137. <u>https://doi.</u> <u>org/10.1177/1469540505053090</u>
- Warde, A. (2010). Introduction. In A. Warde (Ed.), *Consumption* (pp. xxi-l 30 p.). Sage Publications Ltd.
- Warde, A. (2016). *The practice of eating*. John Wiley & Sons.

Warde, A. (2022). Society and consumption. *Consumption and Society, 1*(1), 11-30. <u>https://</u> <u>doi.org/10.1332/gtye7193</u>

- Warde, A., Paddock, J., & Whillans, J. (2020). *The social significance of dining out - A study of continuity and change* (1 ed.). Manchester University Press. <u>http://www.jstor.org/</u> <u>stable/j.ctv11g95k0</u>
- Waterlander, W. E., Steenhuis, I. H. M., de Boer, M. R., Schuit, A. J., & Seidell, J. C. (2012).

Introducing taxes, subsidies or both: The effects of various food pricing strategies in a web-based supermarket randomized trial. *Preventive Medicine, 54*(5), 323-330. <u>https://doi.org/10.1016/j.ypmed.2012.02.009</u>

- Waters, E., Hall, B., Armstrong, R., Doyle, J., Pettman, T., & de Silva-Sanigorski, A. (2011). Essential components of public health evidence reviews: capturing intervention complexity, implementation, economics and equity. *Journal of Public Health*, *33*(3), 462-465. <u>https://doi.org/10.1093/pubmed/ fdr064</u>
- Watts, D. C. H., Ilbery, B., & Maye, D. (2005). Making reconnections in agrofood geography: alternative systems of food provision. *Progress in Human Geography*, *29*(1), 22-40. <u>https://doi.</u> <u>org/10.1191/0309132505ph5260a</u>
- WBAE. (2020). Politik für eine nachhaltigere Ernährung - Eine integrierte Ernährungspolitik entwickeln und faire Ernährungsumgebungen gestalten. <u>https://</u> www.bmel.de/SharedDocs/Downloads/ DE/_Ministerium/Beiraete/agrarpolitik/ wbae-gutachten-nachhaltige-ernaehrung. pdf?__blob=publicationFile&v=3
- WCRF. (2019). Building momentum: lessons on implementing a robust front-of-pack food label. <u>https://www.</u> wcrf.org/policy/our-publications/ building-momentum-series/
- Weber, A. (2021). Mobile apps as a sustainable shopping guide: The effect of eco-score rankings on sustainable food choice. *Appetite, 167*, 105616. <u>https://doi. org/10.1016/j.appet.2021.105616</u>
- Welch, D., Swaffield, J., & Evans, D. (2021). Who's responsible for food waste? Consumers, retailers and the food waste discourse coalition in the United Kingdom. *Journal of Consumer Culture, 21*(2), 236-256. https://doi.org/10.1177/1469540518773801
- WHO Regional Office for Europe. (2021). Slide to order: a food systems approach to meals delivery apps: WHO European Office for the Prevention and Control of Noncommunicable diseases. <u>https://apps.</u> who.int/iris/handle/10665/350121
- WHO Regional Office for Europe. (2022a). The out-of-home food environment: report of a WHO Regional Office for Europe and Public Health England expert meeting,

10 June 2021. https://apps.who.int/iris/ handle/10665/356899

- WHO Regional Office for Europe. (2022b). *Policy brief: taxing sugar-sweetened beverages*. <u>https://www.who.int/europe/publications/i/item/</u>WHO-EURO-2022-5721-45486-65112
- WHO Regional Office for Europe. (2022c). Sugar-sweetened beverage taxes in the WHO European Region: success through lessons learned and challenges faced CC BY-NC-SA 3.0 IGO). https://apps.who.int/ iris/handle/10665/351781
- WHO Regional Office for Europe. (2022d). *WHO European Regional Obesity Report* 2022. World Health Organization. Regional Office for Europe. <u>https://apps.who.int/iris/</u> <u>handle/10665/353747</u>
- WHO. (1990). Diet, nutrition, and the prevention of chronic diseases: report of a WHO study group on Diet, Nutrition and Prevention of Noncommunicable Diseases Imeeting held in Geneva from 6-13 March 1989J. <u>https://</u> apps.who.int/iris/handle/10665/39426
- WHO. (2003). Diet, nutrition, and the prevention of chronic diseases: report of a joint WHO/ FAO expert consultation (Vol. 916). World Health Organization. <u>http://apps.who.int/</u> iris/bitstream/handle/10665/42665/ WHO_TRS_916.pdf?sequence=1
- WHO. (2017). Tackling NCDs: 'best buys' and other recommended interventions for the prevention and control of noncommunicable diseases. <u>https://apps.who.int/iris/</u> <u>handle/10665/259232</u>
- WHO. (2022). WHO manual on sugarsweetened beverage taxation policies to promote healthy diets. <u>https://www.who.int/</u> publications/i/item/9789240056299
- Wijen, F., & Chiroleu-Assouline, M. (2019). Controversy Over Voluntary Environmental Standards: A Socioeconomic Analysis of the Marine Stewardship Council. Organization & Environment, 32(2), 98-124. <u>https://doi. org/10.1177/1086026619831449</u>
- Willer, H., & Yussefi, M. (2000). Organic agriculture worldwide. *Statistics and Future Prospects. Bad Dürkheim, Germany.* <u>https://www.cabdirect.org/cabdirect/</u> <u>abstract/20003032403</u>
- Willett, W., Rockström, J., Loken, B., Springmann, M., Lang, T., Vermeulen, S., Garnett, T., Tilman, D., DeClerck, F., &

Wood, A. (2019). Food in the Anthropocene: the EAT-Lancet Commission on healthy diets from sustainable food systems. *The Lancet, 393*(10170), 447-492. <u>https://doi.</u> org/10.1016/S0140-6736(18)31788-4

- Winter, M. (2003). Embeddedness, the new food economy and defensive localism. *Journal of Rural Studies, 19*(1), 23-32. <u>https://</u> <u>doi.org/10.1016/S0743-0167(02)00053-0</u>
- Wirsenius, S., Hedenus, F., & Mohlin, K. (2011). Greenhouse gas taxes on animal food products: rationale, tax scheme and climate mitigation effects. *Climatic Change*, *108*(1), 159-184. <u>https://doi.org/10.1007/</u> <u>\$10584-010-9971-x</u>
- Wolfenden, L., Barnes, C., Lane, C., McCrabb, S., Brown, H. M., Gerritsen, S., Barquera, S., Véjar, L. S., Munguía, A., & Yoong, S. L. (2021). Consolidating evidence on the effectiveness of interventions promoting fruit and vegetable consumption: an umbrella review. *International Journal of Behavioral Nutrition and Physical Activity*, 18, 1-21. <u>https://doi.org/10.1186/</u> <u>\$12966-020-01046-y</u>
- Wood, A., Moberg, E., Curi-Quinto, K., van Rysselberge, P., & Röös, E. (2023). From "good for people" to "good for people and planet" – Placing health and environment on equal footing when developing food-based dietary guidelines. *Food policy*. <u>https://doi.org/10.1016/j.</u> <u>foodpol.2023.102444</u>
- Wood, B., McCoy, D., Baker, P., Williams, O., & Sacks, G. (2021). The double burden of maldistribution: a descriptive analysis of corporate wealth and income distribution in four unhealthy commodity industries. *Critical public health*, 33(2), 135-147. <u>https://</u> doi.org/10.1080/09581596.2021.2019681
- Woolhead, C., Walsh, M. C., Gibney, M. J., Daniel, H., Drevon, C. A., Lovegrove, J. A., Manios, Y., Martinez, J. A., Mathers, J. C., Traczyk, I., Saris, W. H. M., Gibney, E. R., & Brennan, L. (2015). Dietary patterns in Europe: the Food4Me proof of principle study. *Proceedings of the Nutrition Society*, 74(OCE4), E233, Article E233. https://doi. org/10.1017/S002966511500275X
- Wouters, S., Thewissen, V., Duif, M., van Bree, R. J., Lechner, L., & Jacobs, N. (2018). Habit strength and between-meal snacking in daily life: the moderating role of level of education. *Public Health Nutrition*, *21*(14),

2595-2605. https://doi.org/10.1017/ S1368980018001283

- Wright, B., & Bragge, P. (2018). Interventions to promote healthy eating choices when dining out: A systematic review of reviews. *British Journal of Health Psychology*, *23*(2), 278-295. <u>https://doi.org/10.1111/bjhp.12285</u>
- Wright, L. T., Nancarrow, C., & Kwok, P. M. (2001). Food taste preferences and cultural influences on consumption. *British Food Journal*. <u>https://doi.</u> org/10.1108/00070700110396321
- Xu, X., Sharma, P., Shu, S., Lin, T.-S., Ciais, P., Tubiello, F. N., Smith, P., Campbell, N., & Jain, A. K. (2021). Global greenhouse gas emissions from animal-based foods are twice those of plant-based foods. *Nature Food*, 2(9), 724-732. <u>https://doi.org/10.1038/</u> <u>\$43016-021-00358-x</u>
- Yach, D., Stuckler, D., & Brownell, K. D. (2006). Epidemiologic and economic consequences of the global epidemics of obesity and diabetes. *Nature Medicine*, *12*(1), 62-66. <u>https://doi.org/10.1038/nm0106-62</u>
- Yau, A., Berger, N., Law, C., Cornelsen, L., Greener, R., Adams, J., Boyland, E. J.,
 Burgoine, T., de Vocht, F., Egan, M., Er, V., Lake, A. A., Lock, K., Mytton, O., Petticrew,
 M., Thompson, C., White, M., & Cummins,
 S. (2022). Changes in household food and drink purchases following restrictions on the advertisement of high fat, salt, and sugar products across the Transport for London network: A controlled interrupted time series analysis. *PLOS Medicine*, 19(2),

e1003915. <u>https://doi.org/10.1371/journal.</u> pmed.1003915

- Yee, A. Z. H., Lwin, M. O., & Ho, S. S. (2017). The influence of parental practices on child promotive and preventive food consumption behaviors: a systematic review and meta-analysis. *International Journal of Behavioral Nutrition and Physical Activity, 14*(1), 47. <u>https://doi.org/10.1186/</u> <u>\$12966-017-0501-3</u>
- Zhao, J., Li, Z., Gao, Q., Zhao, H., Chen, S., Huang, L., Wang, W., & Wang, T. (2021). A review of statistical methods for dietary pattern analysis. *Nutr J, 20*(1), 37. <u>https://doi. org/10.1186/s12937-021-00692-7</u>
- Zimmermann, A., & Rapsomanikis, G. (2021). Trade and Sustainable Food Systems: Food Systems Summit Brief Prepared by Research Partners of the Scientific Group for the Food Systems Summit June 8, 2021. 25-pages. <u>https://doi.org/10.48565/</u> <u>SCFSS2021-ZQ03</u>
- Zurek, M., Hebinck, A., Leip, A., Vervoort, J., Kuiper, M., Garrone, M., Havlík, P., Heckelei, T., Hornborg, S., Ingram, J., Kuijsten, A., Shutes, L., Geleijnse, J. M., Terluin, I., Van 't Veer, P., Wijnands, J., Zimmermann, A., & Achterbosch, T. (2018). Assessing Sustainable Food and Nutrition Security of the EU Food System—An Integrated Approach. *Sustainability*, *10*(11), 4271. <u>https://</u> www.mdpi.com/2071-1050/10/11/4271

Annex 1. Responsibilities and working structure within the Scientific Advice Mechanism

- The Group of Chief Scientific Advisors is responsible for developing the Scientific Opinion, which contains evidence-based policy recommendations. Four members of the Group were involved with the project (Eric Lambin, Nicole Grobert, Nebojsa Nakicenovic, and Eva Zažímalová), as well as former members Janusz Bujnicki and Carina Keskitalo. Eric Lambin was appointed as lead scientific advisor for the topic.
- The Science Policy, Advice and Ethics Unit at DG RTD assists the GCSA in the development of the Scientific Opinion. Lucia Selfa Aspiroz, Nicola Magnani and Leonard Engels coordinated the project.
- SAPEA is responsible for independently producing the evidence review report that informs the Scientific Opinion. Within SAPEA, ALLEA served as lead academy network for the topic. Céline Tschirhart, ALLEA Scientific Policy Officer, coordinated the report's development, with the support of the SAPEA team of scientific policy officers: Marie Franquin (Euro-CASE), Louise Edwards (Academia Europaea), Rafael Carrascosa Marzo (Academia Europaea), and Rúben Castro (FEAM).

To jointly coordinate the project between the three parts of the SAM, regular coordination meetings took place in different configurations. From SAPEA, the chair of the SAPEA working group, the president of the network leading on the topic, and members of staff supporting the project participated.
Annex 2. Selection of experts

Following our Quality Assurance Guidelines, we set up an interdisciplinary working group with 17 members from 11 European countries, chaired by Erik Mathijs.

The chair of the working group was proposed by the lead academy network, ALLEA, and approved by the SAPEA board after his Declaration of Interest form was assessed.

We issued a call for nominations describing the scope, timeline and expertise required. The areas of expertise were previously discussed with the working group chair, in coordination with the GCSA and the Unit. The call for nominations was sent via the academy networks to their member academies, who were invited to nominate experts. Experts were also identified through desk research by the academy networks.

The selection committee for the working group met on 8 September 2022. Following our Quality Assurance Guidelines, the Selection Committee comprised:

- the working group chair (Erik Mathijs, KU Leuven)
- the president of the lead academy network, ALLEA (Antonio Loprieno)
- the president of another academy network, FEAM (Stefan Constantinescu)
- a second subject expert (Peter Jackson, University of Sheffield, former SAPEA working group chair on sustainable food systems)

We received a total of 70 nominations for the working group. The experts were selected on the basis of scientific excellence and disciplinary requirements as a priority, taking into account commitment and time availability, the criteria set out in our Strategy of Diversity and Inclusiveness, and other requirements communicated to the committee in advance:

- inter- and multidisciplinarity
- involvement in the wider scientific community, i.e. not Fellows of academies;
- inclusion of early- and mid-career researchers
- gender balance
- wide geographical coverage, including from Widening countries

In the final working group, 70% of selected experts were female and 53% were early- or mid-career researchers. 11 European countries are represented in the group, with 1 member from central/eastern Europe, 3 from southern Europe, 4 from northern Europe and 9 from Western Europe.

The composition of the working group was approved by the SAPEA board. All working group members were required to fill out the Standard Declaration of Interest Form of the European Commission, in accordance with SAPEA's Quality Guidelines. In the assessment, no conflicts of interests were detected.

Annex 3. Evidence review process

We compiled this evidence review report based on input from the experts and their indepth knowledge of the field, together with literature reviews conducted systematically on specific topics for chapter writing teams (see Annex 5, p. 189). A mapping of the EU policy landscape was also carried out to inform the work (see Annex 4, p. 188). In terms of data management, we commit to Open Science and FAIR principles.

The evidence necessary to respond to the question in the Scoping Paper was discussed, debated and assessed by the Working Group members at Working Group meetings, and was written up in iterative drafts of the Report. Chapter contributors also reviewed the relevant literature based on explicit methodological criteria, detailed in the relevant chapters. The literature reviewed for this report was not systematically checked for sponsorship or authors' conflict of interest statements. The authors acknowledge that systematic reviews may yield ambiguous results or more favourable outcomes in studies that were sponsored by industry or where authors reported conflict of interest than in studies where this was not the case (Bes-Rastrollo et al., 2013; Mandrioli et al., 2016).

The first consolidated draft was reviewed by external experts during an Expert Workshop, and the final draft underwent a double-blind peer review.

Timeline

- September 2022 Final formation of working group
- October 2022
 Working group meeting
- November 2022 Working group meeting
- December 2022 Working group meeting
- January 2023 Production of first draft
- February 2023 Expert workshop
- March 2023 Production of second draft
 - Report sent to peer reviewers
- April 2023 Working group addresses peer reviewers' comments
 Production of final draft
- May 2023 SAPEA endorsement
- June 2023 Publication of scientific opinion and evidence review report

182

Requested literature reviews

A Literature Review Team was formed, comprising the director of the Specialist Unit for Review Evidence, the manager of the European Information Service at Cardiff University, and SAPEA staff. The European Information Service was also responsible for developing an EU policy mapping, especially related to the Farm-to-Fork Strategy, to support the work (Annex 4, p. 188).

To complement their knowledge, the working group made individual requests for literature searches on:

- the Farm-to-Fork Strategy
- the role of the food environment for sustainable food choices
- drivers and interventions on household and consumer food waste
- labelling and consumer behaviour
- field experiments in supermarkets and canteens/restaurants regarding interventions to foster sustainable food choices
- political dimensions of meat consumption reduction
- organic food procurement

The search strategies can be found in Annex 5, p. 189.

The reviews were conducted systematically, based on a template completed by the working group member requesting the search. Protocols were recorded and submitted alongside the screened results, and EndNote files were retained with all the extracted results. Scopus and Web of Science were used in the literature searches, alongside further screening of grey literature and using EUR-LEX, the EU Publications Office catalogue and other databases, such as Overton and European Sources Online. The inclusion/exclusion criteria were discussed with appropriate members of the Working Group (when necessary), as well as other members of the Literature Review Team.

Expert workshop

In line with our Quality Assurance Guidelines, we organised an expert workshop on 3 February 2023 in Brussels and online (hybrid format) to discuss and review the evidence, especially:

- to tackle potential blind spots or biases of the working group
- to ensure that the scope and the scale of the evidence and the way it is provided covers the actual discussions in the stakeholder scene
- to discuss the practical applicability of the options proposed in the ERR

to perform a critical appraisal of the evidence.

This workshop took place before the peer review process.

In order to select experts to participate, SAPEA Scientific Policy Officers compiled a list of experts that was based on previous academy and network nominations; on new suggestions by the Group of Chief Scientific Advisors. the SAM Unit and the working group; and further desk research. Experts were prioritised based on the selection criteria below and with the help of the working group chair. The list of potential experts was discussed and approved by the selection committee for the working group, and finally by the SAPEA board.

The criteria for the selection of experts for the expert workshop were:

- scientific background with applied or policy context knowledge in the field of sustainable food consumption
- complementarity of backgrounds, expertise and interests to cover topics covered in the ERR
- inclusion of early- and mid-career experts
- gender balance
- wide geographical coverage, including from Widening countries
- commitment and availability

In the final selection, 13 scientists or experts with applied knowledge in the field were invited to comment on the draft report. 62% of selected experts were female, and 23% were early- or mid-career researchers. 9 European countries were represented in the group, with 1 member from eastern Europe, 1 from southern Europe, 2 from northern Europe and 9 from western Europe.

Other participants included members of the working group, as well as a group of observers composed of SAPEA representatives, members of the Group of Chief Scientific Advisors, and staff of the European Commission.

The expert workshop followed an established format with the Chatham House rule applied to the entire workshop. Participants had received a draft confidential copy of the report in advance of the workshop. After a general introduction to the report, a keynote speaker presented an overall assessment of the report, with initial observations on strengths, possible limitations and gaps. Each of the main chapters was then introduced, followed by feedback from an invited discussant and then an opportunity for open discussion. The discussions helped to refine the draft evidence-based policy options in the report, to ensure they have practical implications for real-world scenarios on timescales that are relevant for EU policy development. After the workshop, members of the working group considered the feedback and agreed on the actions that should be taken to address it. The draft evidence review report was then revised prior to undergoing formal peer review.

The expert workshop report is published separately, as a companion document to this report, and is available on the SAPEA website.¹

Peer review

In line with our Quality Assurance Guidelines, we followed a double-blind peer review process. ALLEA, the lead network for this report, established the areas of expertise needed for peer reviewers based on the draft report, namely trade and economy, health, social sciences, and consumer behaviour.

The partner network Euro-CASE compiled a list of experts based on previous academy and network nominations, on new suggestions by the Group of Chief Scientific Advisors and the SAM Unit, and further desk research. Euro-CASE suggested a list of experts to the SAPEA noard based on the areas of expertise defined by ALLEA, complementarity of expertise, expertise that included a broad overview of the field rather than indepth knowledge in a narrow field, taking into account gender and geographical balance, and inclusion of early and mid-career experts. Euro-CASE performed an internet screening for major conflicts of interests from the potential peer-reviewers. The SAPEA board, excluding ALLEA, gave the final approval for the list of peer reviewers to be invited.

Following these directions, three reviewers accepted the invitation. Of these reviewers, 2 are female, 1 is an early- or mid-career researcher. 2 are from northern Europe and 1 from north America.

Responses were received in March 2023, anonymised by Euro-CASE and then shared with ALLEA and the working group. Members of the working group reviewed the responses and agreed on the actions that should be taken to address them. The draft evidence review report was then revised.

Revisions following peer review

Peer reviewer comments were overall positive. They found that the report satisfactorily addressed the questions posed in the scoping paper, that the literature cited was up-to-date (and several additional literature sources suggested by peer reviewer were incorporated into the text by the working group), that arguments advanced in the report showed the requisite degree of analytical rigour, that conclusions and policy options

^{1 &}lt;a href="https://sapea.info/topic/food-consumption/">https://sapea.info/topic/food-consumption/

were well supported by the scientific evidence, and that there were no signs of biases or undue influence from individuals or interest groups.

Overall, peer reviewers highlighted two points that may be strengthened:

- the socio-cultural dimensions of food consumption
- the power dynamics and interests within the food system that influence food consumption

In response, the working group strengthened these aspects in the text where relevant, to better reflect the relation between policies, norms and conventions, and adding scientific evidence on the effects of power dynamics on food consumption. The working group also acknowledged that given the timeframe for the report, some aspects cannot be dealt with in too much depth.

Additional adaptations following peer review comments included:

- emphasising the reasons for the choice of the COM-B model
- adding information on the roles of industry, both in the potential positive feedback loop with innovation resulting from changing consumer diet, and in the capacity of industries to slow down policy implementation
- adding a note that the reviews carried out for the report were not systematically checked for sponsorship or authors' conflict of interest statements

Further comments from reviewers suggested minor changes. In response, information was provided by the working group to the following areas:

- systemic violence as a driver of food insecurity
- digitalisation (apps and potential impact on sustainability)
- commercial determinants of health
- out-of-home consumption
- the case of Chile
- organic meat and meat policy interventions
- policy instruments for organic production
- the food and nutrition security dimensions in relation to food system outcomes

After the reviewers' comments were addressed by the working group, the peer reviewers' responses, the working groups' rebuttals and actions were sent to the SAPEA board which approved the outcome of the peer review process.

Plagiarism check

In accordance with the Quality Assurance Guidelines, a plagiarism check on the main report was run by Cardiff University using Turnitin software.

Publication

This evidence review report will be handed over to the Group of Chief Scientific Advisors on ~date. At the time of writing, it is planned to publish in June 2023, along with the Advisors' scientific opinion.

The main report will be accompanied by two parallel documents: one expert workshop report, and one policy landscape mapping. All documents can be accessed on the SAPEA website.²

^{2 &}lt;u>https://sapea.info/topic/food-consumption/</u>

Annex 4. Policy landscape summary

A separate policy landscape document was developed for the working group and the Group of Chief Scientific Advisors. It provides an overview of some of the main areas of action at EU level regarding sustainable food consumption. The main aim is to identify legal acts and preparatory documents relevant to understanding policy development. It also seeks to highlight some of the trends and challenges found in the policy documents over the years.

This document shows how legislation on foodstuffs emerged from the need to harmonise national rules to establish an internal market and free movement of products. Concerns over product safety, public health and consumer protection were apparent from an early stage. In this context, consumer information became particularly relevant in legislative initiatives adopted over the years. Wider trends of EU integration also influenced relevant legislation. At the turn of the 21st century, legislative consolidation and simplification was a clear focus of the European Commission, as were new topics such as sustainable development. The intersection of foodstuffs legislation with environmental considerations became more frequent and explicit over the last few years. The European Green Deal, adopted by the European Commission in December 2019, sought to embed environmental and climate action across all policy fields, in an attempt to create synergies.

The backbone of this policy landscape is therefore the Farm-to-Fork Strategy and associated actions. This approach was informed by the scoping paper and additional guidance provided by the European Commission's SAM Unit. The drafting of this document also took into account the work carried out by the working group and some of the main aspects of the evidence review report. While the core of this narrative developed around the actions put forward by the Farm-to-Form Strategy, other action plans and policies are still mentioned in passing whenever mentioned explicitly by the main policy documents or otherwise deemed relevant.

The narrative developed from text analysis of legal acts and preparatory documents published by the co-legislators of the European Union. Particular focus was provided to preparatory documents made available by the European Commission, as the sole institution with the power of legislative initiative. Research was carried out using the EUR-LEX database, the EU's main official resource of legislative information. Policy texts were screened for context and hints to other relevant policy documents. Search terms used for other literature reviewing work in the framework of the evidence review report were used to search in the text. EuroVoc keywords were also used to search the database.

The full document can be found on the SAPEA website.³

³ https://sapea.info/topic/food-consumption/

Annex 5. Literature search strategies

SAPEA supports open and transparent science practices. The following search strategies were designed in response to requests for literature reviews made by members of the working group. The strategies show the date of the search, sources searched, keywords and date limits (if applicable). 'N' shows the number of potentially relevant results that were scanned. Where multiple sources have been searched, a deduplication process has taken place.

Farm-to-Fork Strategy

Nature of search: An initial search on the policy landscape relating to the Farm-to-Fork Strategy, subsequently updated (see Annex 4, p. 188) Date: 2020 – October 2022 Sources: Scopus and Web of Science TITLE-ABS-KEY ("farm to fork" AND strateg* AND europe*) Number relevant = 85

The role of the food environment for sustainable food choices

Nature of search: The role of food environment for sustainable food choices, in the context of neighbourhoods, retail outlets, food deserts Date: 1999 - November 2022 Source: Scopus TITLE (food AND (environment OR desert OR equal* OR outlet*) AND (neighbo* OR loca* OR rural OR urban* OR swamp*)) AND PUBYEAR > 1999 AND (LIMIT-TO (DOCTYPE , "re")

Number retrieved = Not reported

Number relevant = 35

TITLE (food AND (environment OR desert OR swamp* OR equal* OR outlet*) AND (neighbo* OR loca* OR rural OR urban*) AND review*) AND PUBYEAR > 1999 AND PUBYEAR > 1999

Number relevant = 21

Source: Web of Science

food AND (environment OR desert OR equal* OR outlet*) AND (neighbo* OR loca* OR rural OR urban* OR swamp*) Title Search + Reviews

Number relevant = 26

Source: healthevidence.org

food AND (environment OR neighbourhood OR outlet* OR desert* OR swamp*) AND (diet* OR choice OR consum*)

Date: 2018 - November 2022

Number relevant = 6

189

Literature search strategies

Date: 2000-2017 Number relevant = 3

Source: Epistemonikos.org/Scopus/Web of Science

food AND (environment OR neighbourhood OR outlet* OR desert* OR swamp*) AND (diet* OR choice OR consum*) AND ("meta-analysis" OR "meta analysis" OR "systematic review") Date: 2018 - November 2022 Number relevant = Epistemoikos 40; Scopus 14; Web of Science 12

Date: 2000-2017

Number relevant = Epistemoikos 13; Scopus 5; Web of Science 2

Source: Medline/PubMed

Food AND (environment OR neighbourhood OR outlet^{*} OR desert^{*} OR swamp^{*}) AND (diet^{*} OR choice OR consum^{*}) AND (meta-analysis OR meta analysis OR systematic review)

Limited to title [also AND meta-analysis.pt in Medline]

Date: 2018 - November 2022

Number relevant = 11

Date: 2000-2017

Number relevant = 4

Nature of search: Drivers of consumer and household food waste, and interventions and other levers to mitigate it

Date: [Start date not reported] - 18 November 2022

Source: Scopus

TITLE (("food waste" OR "food loss" AND (house" OR home" OR domestic OR consum" AND (LIMIT-TO (DOCTYPE , "re"))

Number relevant = 30

Source: Epistemonikos Number relevant: 1

Labelling and consumer behaviour

Nature of search: To what extent food labelling and certification impacts consumption behaviour.

Date of searches: 2018 - November 2022

Source: healthevidence.org

((food OR nutrition) AND (label* OR certificat* OR "health warning"))

Number relevant = 22

Source: Epistemonikos.org

((food OR nutrition) AND (label* OR certificat* OR "health warning")) AND ("meta-analysis" OR "meta analysis" OR "systematic review") Number relevant = 6

Source: Scopus

((food OR nutrition) AND (label* OR certificat* OR "health warning")) AND ("meta-analysis" OR "meta analysis" OR "systematic review"). Limited to title only. Number relevant = 15



Source: Medline

((food OR nutrition) AND (label* OR certificat* OR "health warning")) AND (("meta-analysis" OR "meta analysis" or "systematic review").tw OR meta-analysis.pt)) Limited to title other than meta-analysis.pt. Number relevant = 15

Source: Scopus

TITLE-ABS-KEY (consum* AND behavio* AND (label* OR certifi*) AND food AND (organi* OR nutri* OR "animal welfare" OR "fair trade" OR "traffic lights")) Reviews only Number relevant = 38

Merged and deduplicated results set from the searches above = 54 potentially relevant reviews

Field experiments in supermarkets and canteens/restaurants regarding interventions to foster sustainable food choices

Nature of search: Evidence from field experiments in supermarkets and canteens/restaurants regarding interventions to foster sustainable food choices

Date: 2002 - December 2022

Sources: Scopus & Web of Science

Title + Abstract

food AND (field OR experimen* OR observ* OR test* OR investiga*) AND (supermarket* OR canteen* OR restaurant*) AND (interven* OR instrument* OR nudg* OR pric* OR label*) AND (sustainab* OR climate OR organic* OR biodivers* OR water OR "animal welfare" OR climate) Number relevant = 67

Political dimensions of meat consumption reduction

Nature of search: What worked or could work politically to reduce meat consumption

Date: 2013 - January 2023

Source: Scopus & Web of Science

Title only

meat AND (consum* OR reduc* OR curtail*) AND (stakeholder* OR government* OR polic* OR politic* OR regulat*).

Number relevant = 14

Organic food procurement

Nature of search: Organic food and public procurement Date: 2010 - February 2023

Source: Scopus & Web of Science

Title only

(Organic [within two words of*] (food OR procurement OR kitchen* OR catering OR agriculture) AND (policy OR "action plan" OR strateg* OR guideline* OR legislat* OR "public kitchen" OR "public sector")) Number relevant = 13

[Search syntax for 'within two words of' for Scopus is W/2 and for Web of Science is NEAR/2]

Annex 6. Acknowledgements

SAPEA wishes to thank the following people for their valued contributions and support in the production of this report.

Working group

The names of the working group members who wrote this report are on p. 8.

Peer reviewers

- Jennifer Clapp, University of Waterloo, Canada
- Lotte Holm, University of Copenhagen, Denmark
- Armando Perez-Cueto, Umeå University, Sweden

Expert workshop participants

- Koen Boone, The Sustainability Consortium, Netherlands
- Christine Cherbut, National Research Institute for Agriculture, Food and the Environment (INRAE), France
- Dessislava Dimitrova, Bulgarian Academy of Sciences, Bulgaria
- Edel O. Elvevoll, The Arctic University of Norway, Norway
- David Evans, University of Bristol, United Kingdom
- Klaus Grunert, Aarhus University, Denmark
- Preetmoninder Lidder, Food and Agriculture Organization, Italy
- Theresa Marteau, University of Cambridge, United Kingdom
- Aifric O'Sullivan, University College Dublin, Ireland
- Krijn J. Poppe, Council for the Environment and Infrastructure, Netherlands
- Kai Purnhagen, University of Bayreuth, Germany
- Lucia Reisch, University of Cambridge, United Kingdom
- Monika Zurek, Oxford University, United Kingdom

Members of the selection committee

- Antonio Loprieno, ALLEA president and chair of the SAPEA board
- Stefan Constantinescu, FEAM president
- Peter Jackson, University of Sheffield, United Kingdom
- Erik Mathijs, working group chair, KU Leuven, Belgium

SAPEA staff members

- Céline Tschirhart, Scientific Policy Officer, ALLEA
- Marie Franquin, Scientific Policy Officer, Euro-CASE
- Louise Edwards, Scientific Policy Officer, AE
- Rafael Carrascosa Marzo, Scientific Policy Officer, AE
- Rúben Castro, Scientific Policy Officer, FEAM
- Hannah Whittle, Scientific Policy Officer, FEAM
- Rudolf Hielscher, Coordinator, acatech
- Toby Wardman, Head of Communications

Literature reviews and policy mapping

- Frederico Rocha, Manager of European Information Service, Cardiff University
- Alison Weightman, Director of the Specialist Unit for Review Evidence, Cardiff University

Science writers

- Gavin Wren, Wren&Co
- Cheron Constance, Wageningen University

Group of Chief Scientific Advisors to the European Commision

Eric Lambin (lead for the topic)

Science Policy, Advice and Ethics Unit at DG RTD, European Commission

- Lucia Selfa Aspiroz
- Nicola Magnani
- Leonard Engels



SAPEA is part of the European Commission's Scientific Advice Mechanism, which provides independent, interdisciplinary, and evidence-based scientific advice on policy issues to the European Commission.

This Evidence Review Report informs the Group of Chief Scientific Advisors' Scientific Opinion on the topic.



www.sapea.info @SAPEAnews